Measuring What Works: A Mixed-Methods Evaluation of Women's Groups on Maternal Health Uptake in Rural Nepal.

Sheetal Sharma

In partial fulfilment of the requirements of Bournemouth University for the degree of Doctor of Philosophy

Bournemouth University, U.K.

June 2017

Copyright Statement

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and due acknowledgement must always be made of the use of any materials contained in, or derived from, this thesis. This thesis has no competing interests.

Abstract: Measuring what Works: A Mixed-Methods Evaluation of Women's Groups on Maternal Health Uptake in Rural Nepal

Sheetal Sharma

Background: There is a need for more studies that analyse evaluation methods in the context of maternal health promotion. These should assess the effectiveness of health promotion interventions on health outcomes, factors contributing to impact, and transferability. This thesis reports on an evaluation of one such intervention in Nepal targeting rural women to promote maternal health.

Methods: A mixed-methods approach was used where, first, a Difference-in-Difference (DiD) estimation assessed the effects of the intervention on selected outcome variables while controlling for: 1) a constructed wealth index; and 2) women's socio-economic characteristics in a five-year controlled, non-randomised, repeated cross-sectional study of a community-based health promotion intervention targeting maternal health in Nepal. Second, the qualitative data were analysed to explore the knowledge, attitudes, and beliefs of women post-intervention. Finally, the financial data were analysed to identify resources needed and estimate the cost of the health promotion intervention.

Results: After five years, women in the intervention area were more likely to seek antenatal care at least once, to take iron/folic acid, and to attend postnatal care. The intervention did not influence women's place of birth or likelihood of receiving care from a skilled birth attendant. However, it did improve attendance for the recommended four antenatal visits for the first two and a half years. The qualitative findings helped explain some of the changes or lack thereof, where in the intervention area women were perceived, by the researcher, as empowered, confident, and the family as supportive. The cost of providing the health promotion intervention per group/woman and the evaluation process consisted of only 10% of the total programme cost.

Conclusion: This is the first community-based health promotion intervention that has demonstrated a greater impact during pregnancy (i.e., uptake of antenatal care) than around birth (i.e., changes in delivery care). Other factors, not easily resolved through health promotion interventions, may influence birth outcomes, such as financial liquidity or geographical constraints. The evaluation showed that using mixed methods provided valuable information that would not have been extracted through one method alone. While DiD is a precise tool for measurement, the qualitative research provided insight into why the intervention had an impact in pregnancy but not at birth.

Table of contents

Copyrig	Copyright Statement2		
Abstract:			
Table of	Table of contents4		
List of Figures and Tables8			
Acknow	ledgments	9	
Declarat	lion	10	
List of A	bbreviations	11	
Chapter	'0' Introduction to Thesis	13	
Chapter	1 Introduction	15	
1.1	Setting the scene	15	
1.2	Evaluation of community-based interventions		
1.3	Maternal health policy & community interventions		
1.3.1	Maternal health access and outcome indicators for evaluation		
1.3.2	Three Delays impact on maternal health		
1.4	Background on Nepal		
1.4.1	Population & nealth policies and programmes in Nepal		
1.4.2	Negali Maternal health and delivery practices		
1.4.3	Demographic drivers of dealing in MMP	37 20	
1.4.4	Pational for this thosis	30 12	
1.5	Summary	42 //3	
1.0	Summary	40	
Chapter	2 Evaluation health promotion & GTN		
2.1	GTN intervention aims (or approach) and health promotion	44 44	
2.1 2.2	GTN intervention aims (or approach) and health promotion	44 44 45	
2.1 2.2 2.3	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion	44 44 45 51	
2.1 2.2 2.3 2.4	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches	44 44 51 53	
2.1 2.2 2.3 2.4 2.5 2.5	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion	44 45 51 53 55	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle	44 45 51 53 55 56	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.2	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle		
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.3	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation	44 45 51 53 55 56 58 60 61	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation	44 44 51 53 55 56 58 60 61	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4 2.5.4.1 2.5.4.2	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation	44 45 51 53 55 56 58 60 61 61	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4 2.5.4.1 2.5.4.2 2.6	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs	44 44 51 53 56 56 60 61 61 66 68	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6 2.6	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion	44 44 51 55 56 56 61 61 66 68 68 68	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps	44 45 51 55 56 56 60 61 61 61 68 68 70 74	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention site and control community. The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps	44 45 51 53 55 56 58 60 61 61 61 66 68 70 74	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps 3 Evaluation .	44 44 51 53 55 56 58 60 61 61 66 68 70 74 75	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1 3.2	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps 3 Evaluation: its nature in health promotion Evaluation Evaluation Evaluation	44 45 53 56 56 56 60 61 61 61 61 64 68 70 74 75 75 75	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1 3.2 3.2.1	GTN intervention aims (or approach) and health promotion Health promotion theories. Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle. Programme planning cycle. Planning: intervention site and control community. The GTN intervention implementation & the evaluation. Implementation cycle. Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps 3 Evaluation . Evaluation. Evaluation. Evaluation, philosophical underpinnings. Evidence of effectiveness	44 45 51 53 55 56 58 60 61 61 61 61 61 61 62 70 74 75 75 77 79	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1 3.2 3.2.1 3.2.2	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps 3 Evaluation: its nature in health promotion Evaluation. Evaluation, philosophical underpinnings Evidence of effectiveness Realist evaluation.	44 45 51 53 55 56 58 60 61 61 61 61 66 68 70 74 75 75 77 79 81	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1 3.2 3.2.1 3.2.2 3.2.3	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community. The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps 3 Evaluation: its nature in health promotion Evaluation Evaluation, philosophical underpinnings Evidence of effectiveness Realist evaluation Evaluation in health promotion	44 45 53 56 56 56 60 61 61 61 61 61 68 70 74 75 75 75 79 79 81 83	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1 3.2 3.2.1 3.2.2 3.2.3 3.2.3.1	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps 3 Evaluation: its nature in health promotion Evaluation Evaluation, philosophical underpinnings Evidence of effectiveness Realist evaluation Process evaluation	44 45 51 53 55 56 58 60 61 61 61 61 61 61 61 68 70 74 75 75 77 79 81 83 88	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1 3.2 3.2.1 3.2.2 3.2.3 3.2.3.1 3.2.3.2	GTN intervention aims (or approach) and health promotion Health promotion theories Community-based health promotion Health promotion approaches GTN and health promotion Needs assessment cycle Programme planning cycle Planning: intervention site and control community The GTN intervention implementation & the evaluation Implementation cycle Evaluation Maternal health promotion interventions in LMICs Literature review effectiveness of maternity health promotion Summary of literature review and next steps 3 Evaluation: its nature in health promotion Evaluation, philosophical underpinnings Evidence of effectiveness Realist evaluation Process evaluation Impact evaluation	44 45 51 53 55 56 58 60 61 61 61 61 66 68 70 71 75 77 75 77 79 81 83 88 88	
2.1 2.2 2.3 2.4 2.5 2.5.1 2.5.2 2.5.3 2.5.4 2.5.4.1 2.5.4.2 2.6 2.6.1 2.7 Chapter 3.1 3.2 3.2.1 3.2.2 3.2.3.1 3.2.3.2 3.2.3.1 3.2.3.2 3.2.3.3	GTN intervention aims (or approach) and health promotion	44 44 45 53 56 56 56 56 60 61 61 61 61 61 63 70 74 75 75 75 79 79 83 83 83 83 89 93	

3.4	Costing of evaluations	94
3.5	Time & indicators	95
3.6	Evaluations strengths & weaknesses	95
3.7	Aims & objectives	98
3.8	Summary	99
Chanter 4	Methodology & Methods	100
	Introduction	100
4.1 12	Rational of mixed methods	101
т .2 Л 2 1	Mixing the data & data collection using different designs	101
4.2.1	Advantages & challenges in mixed methods	105
4.2.1	The critique of quantitative and qualitative methods	107
4.2.1.1	Quantitative research & its suidenes	107
4.2.1.2	Qualitative research a fixed methods study	. 107
4.2.1.3	Qualitative component of mixed-methods study	110
4.2.	Summary of methodology	۲۱۱
4.3	Mixed methods in this thesis	113
4.3.1	Mixed-methods evaluation	113
4.3.2	Maternal health services uptake conceptual framework	116
4.3.3.	GIN survey & survey population: sampling	118
4.3.4.1	Qualitative Interview sampling	118
4.3.4.2	Quantitative sampling	120
4.3.4.3	GTN survey	120
4.3.4.3.1	Training of field staff for sampling	121
4.3.4.3.2	Data management	122
4.3.5.	Study area & population	123
4.3.6	Baseline data	125
4.3.7	Study population	125
4.3.8	Qualitative methods	126
4.3.8.1	Interviews data collection	128
4.3.8.1.1	Focus groups	128
4.3.8.1.2	Semi-structured interviews	129
4.3.8.2	Qualitative data analysis	129
4.3.9	Quantitative study of this mixed-methods study	133
4.3.9.1	Data analysis: secondary analysis, GTN	134
4.3.9.1.2	Dependent & independent variables	134
4.3.9.1.3	Chi-square test	135
4.3.9.1.4	Difference-in-Differences analysis	135
4.3.9.1.5	Wealth index construction	141
4.3.9.1.6	Caste	. 142
4.3.9.1.7	Cost analysis extrapolation	. 143
4.4	Validity, reliability and goodness of fit of study methods	144
4.5	Ethical Considerations in mixed-methods	146
4.6	Summary	148
1.0		
Chapter 5	Quantitative study findings	149
5.1	Introduction	149
5.2	Findings	149
5.3	Socio-demographic & socio-economic characteristics of samples	150
5.4	Household decision on the utilisation of maternal care	153
5.5	Confounding factors	157
5.6	Utilisation of maternal care	158
5.7	Impact of the intervention	160
5.8	Impact of covariates on outcomes	163
5.8.1	Wealth and caste	163
5.8.2	Age, education & time	167
5.9	Continuity of care	168

5.11		168
E 40	The cost of the intervention	170
5.12	Summary	174
Chanter (S Qualitative study and findings	175
6.1	Introduction to the qualitative findings	175
6.2	Participants' characteristics	
6.2.1	Health promoters' evaluation, key informants	179
6.3	Focus group findings	185
6.4	Thematic analysis of the study population	186
6.4.1.1	Changes in the last five years in the area/community	
6.4.1.2	Changes in attitudes with respect to seeking care	
6.4.1.3	Perceived progress of maternal health	191
6.4.1.3.1	Changes in attitudes towards motherhood and female children	191
6.4.1.4	Groups working in the area	
6.3.1.4.1	Women and workload	196
6.4.1.5	Shyness	197
6416	Men's involvement in maternal health	199
6.4.1.6.1	Effects of the aroups	200
64162	Health workers working with GTN	201
6417	Empowerment	202
6418	Changes in decision-making patterns	203
64181	Decision-making in PNC	205
6419	Barriers and limitations to attending groups and health services	207
64191	Socio-economic barriers and limitations	207
64102	Cultural harriers	200
64110	Reflexive section	203
0.4.1.10		
Chapter 7	7 Discussion	213
7.1	Introduction	213
7.2	Difference-in-Difference	213
7.2.1	Mixed-measurement evaluation	218
	DiD & confounding factors	220
7.2.1.1		
7.2.1.1 7.2.2	Qualitative	220
7.2.1.1 7.2.2 7.2.3	Qualitative Groups	220 222 226
7.2.1.1 7.2.2 7.2.3 7.2.4	Qualitative Groups Cost	220 222 226 226
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5	Qualitative Groups Cost Empowerment & decision-making	220 222 226 226 227
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths	220 222 226 226 227 229
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations	220 222 226 226 227 229 229 229
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths	220 222 226 226 227 229 229 229 231
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths	220 222 226 226 227 229 229 231
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8 1	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths B Thesis conclusion Evaluation context	220 222 226 226 227 229 229 231 231 235 236
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context	220 222 226 226 227 229 229 231 235 236
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis	220 222 226 226 227 229 229 231 235 236 236
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview	220 220 222 226 226 227 229 229 231 231 236 236 240
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN	220 222 226 226 227 229 231 231 236 236 240 240 240
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for researchers	220 220 222 226 227 229 231 231 236 236 240 240 240 243
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners	220 222 226 226 227 229 229 231 235 236 236 240 240 240 243 243
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for practitioners Recommendations for policy-makers	220 220 222 226 226 227 229 231 231 236 240 240 240 240 244 244 246
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5 Appendix	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for policy-makers	220 220 222 226 226 227 229 231 236 236 240 240 240 240 243 244 246 246
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5 Appendix	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for policy-makers	220 220 222 226 227 229 231 236 236 236 240 240 240 240 244 244 244 244 246
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5 Appendix Appendix	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for policy-makers 1 - Ethics Letter II - Table 2. Definitions & Sources	220 220 222 226 227 229 229 231 235 236 240 240 240 243 244 244 244 247 247 248
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5 Appendix Appendix Appendix	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for policy-makers 1 - Ethics Letter II - Table 2, Definitions & Sources III - Questionnaire for Women in English and Nepali	220 222 226 226 227 229 229 231 235 236 240 240 240 243 244 244 244 244 245 247 247 248 251
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5 Appendix Appendix Appendix Appendix	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for practitioners Recommendations for policy-makers I - Ethics Letter II - Table 2, Definitions & Sources III - Questionnaire for Women in English and Nepali IV - Variables Description	220 220 222 226 226 227 229 231 236 236 240 240 240 240 240 246 247 247 247 248 251 279
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5 Appendix Appendix Appendix Appendix Appendix	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for policy-makers I - Ethics Letter II - Table 2, Definitions & Sources III - Questionnaire for Women in English and Nepali IV - Variables Description V - Annex-1: Qualitative topic quide Intervention Area	220 222 226 226 227 229 229 231 236 236 236 240 240 240 240 240 244 246 247 247 248 247 248 251 279 281
7.2.1.1 7.2.2 7.2.3 7.2.4 7.2.5 7.3 7.3.1 7.3.2 Chapter 8 8.1 Chapter 9 9.1 9.2 9.3 9.4 9.5 Appendix Appendix Appendix Appendix Appendix Appendix	Qualitative Groups Cost Empowerment & decision-making Research limitations & strengths Limitations Strengths 3 Thesis conclusion Evaluation context 9 Recommendations from the thesis Overview Recommendations for GTN Recommendations for practitioners Recommendations for practitioners Recommendations for policy-makers 1 - Ethics Letter II - Table 2, Definitions & Sources III - Questionnaire for Women in English and Nepali IV - Variables Description V - Annex-1: Qualitative topic guide, Intervention Area V - Annex-2; gualitative topic guide. Control area	220 220 222 226 227 229 229 231 235 236 240 240 240 240 240 243 244 244 246 247 247 248 251 279 281 285

Appendix VI – Definitions	
Appendix VII – Published papers	
Bibliography	

List of Figures and Tables

Figure 1 Rural Nepal, Pharping area	31
Figure 2 Organisational structure of the Nepal Health System	36
Figure 3 Health promotion intervention cycle	56
Figure 4 Objectives of the GTN intervention	59
Figure 5 Rural Nepal, health promoters	61
Figure 6 Visual cards	62
Figure 7 GTN Visual cards	63
Figure 8 Women's group meeting in rural Nepal	65
Figure 9 Timeline of the programme and PhD	67
Figure 10 Flow chart of literature search	69
Figure 11 Donabedian model	85
Figure 12 Overview of mixed methods1	14
Figure 13 Merging the data in a single study1	15
Figure 14 Conceptual framework used in the evaluation of GTN1	17
Figure 15 Map of area as part of Kathmandu district1	23
Figure 16 Wealth distribution by caste1	65
Figure 17 Wealth distribution in high castes1	66
Figure 18 Wealth distribution in middle castes1	63
Figure 19 Wealth index distribution among lower castes1	67
Table 1 Demographic and health indicators	41
Table 2 Characteristics of intervention and control community1	25
Table 3 Focus group target population 1	29
Table 4 DiD Estimation1	37
Table 5 Number of surveyed women with last child <2 years by area and year1	50
Table 6 Characteristics of the Control and Intervention area1	51
Table 7 Decision-maker for healthcare in the household: percentage changes over	۶r
time & area1	54
Table 8 Household Assets of Respondents1	56
Table 9 Correlation of intervention outcomes and living in the intervention area at	
baseline1	57
Table 10 Maternal Health Uptake (%) of health services in the intervention and	
control area1	59
Table 11 DiD analysis of maternal health uptake at midline evaluation (OR & CI) 1	61
Table 12 DiD analysis of maternal health uptake at final evaluation (OR & CI)1	62
Table 13 Caste distribution1	63
Table 14 Wealth distribution1	64
Table 15 SBA uptake post-ANC1	68
Table 16 Predicted probabilities of DiD analysis of maternal health uptake change	s
in intervention & control area, final evaluation (%)	69
Table 17 Cost of the implementation of the intervention1	71
Table 18 Cost of the evaluation of the intervention1	72
Table 19 Cost of the likelihood of ANC uptake for the population in the intervention	n
area	73
Table 20 Focus Groups1	77
Table 21 Interviews 1	78

Acknowledgments

Thank you to my remarkable and challenging supervisors; they have acted above and beyond the call of duty: Professor Edwin van Teijlingen, Professor Vanora Hundley, Dr. Catherine Angell, Professor Padam Simkhada and Dr. Elisa Sicuri. I am grateful for their unending support, their friendship, their extensive knowledge in the interpretation of data, and their patience when commenting on multiple drafts and approving the final submission.

I also wish to thank the institutions who supported me these past years, the Faculty of Health and Social Sciences, Bournemouth University (BU), UK; Institute for Clinical Effectiveness and Health Policy (IECS), Buenos Aires, Argentina and Barcelona Institute for Global Health (ISGlobal), Hospital Clínic - Universitat de Barcelona, Barcelona, Spain. These research mobility activities were possible with Bournemouth University and Santander Universities' funding. I am indebted to Dr. José M. Belizán from IECS for his mentorship.

I am grateful to Sofia Castro Lopes, Nagawa Lule and Gita DeVaney for their friendship and technical advises. I wish to say thank you to the Eva Papadopoulou, Sara Glithro, and Louise Bryant and the BU Graduate School.

I wish to thank all research participants as well as the staff of Green Tara Nepal and the health posts and the Female Community Health Volunteers. This study had financial support from Green Tara Trust, UK, BU, and Santander Universities and I thank them.

I am grateful to the rural communities in Nepal who participated in the research studies, community interventions and gave advice and to Green Tara Nepal for initiating this challenging approach to maternity care and health promotion aimed at childbearing women.

A last but not least, thank you to my dear, dear family for their unconditional support and patience. Merci.

This work is dedicated to my mother, Naznin, brave and inspiring until the very end.

Declaration

I declare that I conducted the work represented in this thesis, entitled *Measuring What Works: A Mixed-Methods Evaluation of Women's Groups on Maternal Health Uptake in Rural Nepal.* It is in accordance with the requirements for the degree of Ph.D. at Bournemouth University. The work that is presented to the best of my knowledge is original except as acknowledged in the script. All quotations have been distinguished by quotation marks and sources of information have been specifically acknowledged. This thesis has not been submitted previously, either as a whole or in part, for a degree at this or any other university.

Sheetal Sharma 25 June 2017

List of Abbreviations

AIDS	Acquired Immunodeficiency Syndrome
AHW	Auxiliary Health worker
ANC	Antenatal Care
ANM	Auxiliary Nurse-Midwives
С	Control
СВА	Cost-benefit analysis
CBS	Central Bureau of Statistics
CEA	Cost-effectiveness analysis
CHW	Community Health Worker
СМА	Community Medical Assistant
CMACE	Centre for Maternal and Child Enguiries
CI	Confidence Interval
CUA	Cost-utility analysis
DALY	Disability-Adjusted Life Year
DC	Delivery Care
DHS	Demographic Health Surveys
DiD	Difference-in-Difference
EmONC	Emergency obstetric and newborn care
FCHV	Female Community Health Volunteer
FP	Family Planning
FP/MCH	Nepal Family Planning and Maternal and Child Health Plan
FP/MCH HDI	Nepal Family Planning and Maternal and Child Health Plan Human Development Index
FP/MCH HDI I	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention
FP/MCH HDI I GDP	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion)
FP/MCH HDI I GDP GoN	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal
FP/MCH HDI I GDP GoN GNI	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income
FP/MCH HDI I GDP GoN GNI GTN	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal
FP/MCH HDI I GDP GoN GNI GTN GTT	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Trust
FP/MCH HDI I GDP GoN GNI GTN GTT HP	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Trust Health Post
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV ICD-10	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus International statistical classification of diseases and related health problems, 10 th edition
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV ICD-10	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus International statistical classification of diseases and related health problems, 10 th edition The WHO application of ICD-10 to deaths
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV ICD-10 ICD-MM	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus International statistical classification of diseases and related health problems, 10 th edition The WHO application of ICD-10 to deaths during pregnancy, childbirth and the puerperium: ICD Maternal Mortality
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV ICD-10 ICD-MM	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus International statistical classification of diseases and related health problems, 10 th edition The WHO application of ICD-10 to deaths during pregnancy, childbirth and the puerperium: ICD Maternal Mortality Institutional Delivery
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV ICD-10 ICD-MM ID ITS	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus International statistical classification of diseases and related health problems, 10 th edition The WHO application of ICD-10 to deaths during pregnancy, childbirth and the puerperium: ICD Maternal Mortality Institutional Delivery Interrupted Time Series
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV ICD-10 ICD-MM ID ITS IMR	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus International statistical classification of diseases and related health problems, 10 th edition The WHO application of ICD-10 to deaths during pregnancy, childbirth and the puerperium: ICD Maternal Mortality Institutional Delivery Interrupted Time Series Infant Mortality Rate
FP/MCH HDI I GDP GoN GNI GTN GTT HP HIV ICD-10 ICD-10 ICD-MM ID ITS IMR M&E	Nepal Family Planning and Maternal and Child Health Plan Human Development Index Intervention Gross Domestic product per capita (based on Purchasing Power Parity conversion) Government of Nepal Gross National Income Green Tara Nepal Green Tara Nepal Green Tara Trust Health Post Human Immunodeficiency virus International statistical classification of diseases and related health problems, 10 th edition The WHO application of ICD-10 to deaths during pregnancy, childbirth and the puerperium: ICD Maternal Mortality Institutional Delivery Interrupted Time Series Infant Mortality Rate Monitoring and Evaluation

MCHW	Maternal and Child Health Worker
MDGs	Millennium Development Goals
MDSR	Maternal Death Surveillances and Response
MNH	Maternal and Newborn Health
MMR	Maternal Mortality Rate
MH	Maternal Health
MOHP	Ministry of Health and Population
NFHP	Nepal Family Health Programme
PCA	Principal Component Analysis
PHCCs	Primary Healthcare Centres
PHN	Primary Health Nurse
PPP	Purchasing Power Parity
QALY	Quality-Adjusted Life Year
RCT	Randomised Controlled Trial
RR	Relative Risk
SBA	Skilled birth attendant
SDGs	Sustainable Development Goals
SES	Socio-economic Status
SHP	Sub-health Post
SRH	Sexual and Reproductive Health
STI	Sexually Transmitted Infections
ТВА	Traditional Birth Attendant
TFR	Total Fertility Rate
OR	Odds ratio
UN	United Nations
UNICEF	United Nations Children's Fund
UNFPA	United Nations Population Fund
USA	United States of America
USD	United States dollar
VDC	Village Development Committee
VHW	Volunteer Health worker
WHO	World Health Organization

Chapter '0' Introduction to Thesis

This introductory chapter provides a road map to the thesis, which the reader may feel is laid out in an unconventional way. This doctoral work is concerned with identifying and applying a pragmatic and achievable method to evaluate healthcare interventions in low- and middle-income countries. In this thesis the substantive research in Nepal centres on the intervention implemented by Green Tara Nepal, this health promotion intervention acts as a case study to test the methods. The health promotion intervention was run by the Non-Governmental Organisation (NGO) *Green Tara Nepal* (GTN) and financially supported by a Buddhist charity based in London called *Green Tara Trust (GTT)*. GTN and GTT are interlinked, and henceforth the intervention was designed by GTT in close collaboration with researchers based at the University of Aberdeen. It aimed to increase the uptake and knowledge of maternal health and thereby improve maternal healthcare and decision-making by individual mothers in two districts in Kathmandu Valley, Nepal.

While a more conventional thesis may have started with an outline of the intervention under research, one key focus of this thesis is the pragmatic approach to evaluation research and the most appropriate methods of evaluation. In particular, this research was concerned with how community interventions in maternal health promotion can be evaluated in low-income settings. Thus, this PhD study is substantively based on an evaluation designed to determine the mechanisms at work in the GTN intervention. This evaluation is concerned not only with assessing the effectiveness (here, impact) of the programme, but also with extracting the findings from the intervention so that they can be plausibly applied to other interventions. Additionally, this mixed-methods study explored various sociodemographic characteristics of study participants using primary and secondary data. The study approach is based on a controlled-before-and-after design; hence data were collected in both the intervention and control areas to identify whether these factors acted as enablers or barriers to the uptake of maternal healthcare, knowledge, and decision-making. Finally, the researcher assessed the costs of the intervention and the evaluation. Following this outline, the overview of chapters satisfactorily explains the internal logic of the thesis.

The main body of the thesis is organised into nine chapters. *Chapter 1* provides an outline of evaluations, the global situation for maternal health, its indicators of

progress, the global burden of maternal mortality, and the global initiatives that target the latter. Brief aspects focusing on health promotion in the field of maternity care and the uptake indicators in low and middle-income countries follow it. These indicators are sometimes referred to as "proxy outcomes" or simply "outcomes". As the GTN intervention took place in rural Nepal, the chapter ends with providing context for the evaluation by presenting Nepal and the local situation for maternal health. Also discussed is the justification for this study.

Chapter 2 expands upon the notions of health promotion. Health promotion is explored as a discipline concerned with maintaining health rather than preventing disease in order to improve community-based health. Also detailed in this chapter is the GTN intervention and the findings from the literature review on evaluations of community health promotion in maternal health interventions. *Chapter Three* explores the underlying philosophies of evaluation and evaluation approaches to the practice of health promotion. *Chapter Four* focuses on methodology and the methods or the tools of evaluation employed in carrying out the mixed-methods evaluation of the GTN intervention.

Chapters Five and Six present, in detail, the findings of this thesis with respect to this particular project in Nepal, which is important in its own right. These findings and interpretations will of course be of interest to GTN and similar organisations concerned with social improvement.

In *Chapter Seven*, the results from the different methods in this mixed-methods study are synthesised into a comprehensive discussion, while discussing the strengths and limitations of this work. Finally, *Chapter Eight* concludes on both the substantive Green Tara Intervention as well as on methodological issues raised. *Chapter Nine* provides recommendations that arose from the analyses in this PhD research. Separate recommendations are provided for practitioners, educators, researcher and policy-makers.

Chapter 1 Introduction

1.1 Setting the scene

Evaluations of community-based interventions can be complex, especially those on a larger scale. It is important to measure the outcomes of health and health promotion interventions everywhere, but perhaps more so in low-income countries. The knowledge such evaluations generate is important as an exercise in accountability for researchers, funders, policy makers, practitioners in the field, and to most importantly (potential) recipients. Moreover, the aim of evaluation research is to address enablers and obstacles of the study in question for determining "what works" in order to continue programme activities or upscale. An evaluation study can be undertaken using various research approaches or within the context of a philosophy, such as realism, positivism, or pragmatism. In addition, evaluation is one particular type of research that gives a distinctive account of the nature of programmes and how they work, of what is involved in explaining and understanding programmes, of the research methods needed to understand how the programme works, and of the products/outcomes of evaluation research (Clancy 2002; Pawson and Tilley 2004).

This PhD study is based on evaluation research designed using mixed methods to determine the mechanisms at work in a maternal health promotion and community-based intervention in a low-resource setting. Health promotion is a 'salutogenic' (preventative) approach to improving community-based health (Judd et al. 2001). The health promotion programme was implemented by Green Tara Nepal (GTN) and is discussed in detail in Chapter 2. This chapter sets the scene for evaluations and the particular Green Tara Intervention is placed in context of the wider country (Nepal). The chapter has been divided into three parts: the first part discusses evaluations of community-based interventions, the second part discusses the chosen outcomes indicators/outcomes to evaluate maternal health, and the third examines the context behind this evaluation.

1.2 Evaluation of community-based interventions

In 1998, the World Health Organization (WHO) called for more robust evaluations of implementations from the field (WHO 1998) in order to determine impact on lives, scaleability, replicability (whether programmes were generalisable), validity, to provide accountability to stakeholders and funders alike, and to set certain criteria/standards of evaluation (Judd et al. 2001; Duflo 2004; Godin et al. 2006).

As discussed in Sections 1.1 and 2.5, however, nearly two decades later there have been few evaluations of health promotion interventions in Low- and Middle-Income Countries (LMICs). This section explores the challenges of evaluating these types of health promotion interventions due to the nature of such interventions and the cost of properly designed effectiveness evaluations of interventions. In LMICs, there exists a risk of providing simple or no comparisons to a given intervention – thus many evaluations do not answer the question "did it work?". Further impacting this risk is the existence of publication bias, which hypothesises that only "successful" studies are published (Duflo 2004). Several examples of evaluations and their outcomes are given below.

Consider the example of a school allocation lottery programme where the evaluation only compared attendance rates, choice of school, and performance (Duflo 2004). It appeared that students performed better when they were able to choose their school. However, a further regression analysis of lottery winners and losers showed that students who chose among the neighbourhood schools had low performance outcomes. This shows that a simple analysis of choice versus no choice did not take into consideration the kind of choices people made (the factors involved in choosing).

Obstetrics also has its fair share of poorly evaluated interventions that have been introduced. Three separate Cochrane Collaboration reviews have shown how some simple maternity-care interventions were harmful or ineffective. These included shaving women's perineal area, which does not decrease infection risk (Basevi & Lavender 2014); using enemas during labour, which led to an increase in intrapartum infection (Reveiz et al. 2007); and the use of fundal pressure, which was associated with an increase in anal sphincter tears (Verheijen et al. 2009).

Evaluations can also yield measures of effectiveness. Effectiveness is used to describe evaluations or understand the findings of the studies conducted in real-world settings by individuals who are not part of a research staff (Glasgow et al. 1999). An example of effectiveness is provided by Banerjee and Duflo (2011), who evaluated a development project addressing food aid. The intervention did not yield a positive effect; despite the provision of food aid, trial participants did not eat healthier. The unintended consequence (Sections 3.2 and 3.2.3.2) was that the money received was spent on alcohol, tobacco, and festivals. As culture, politics, history, laws, infrastructure, and individuals shape society, introducing an intervention into a given community requires adaptation. When something is improved, it is changed in an expected way but also in an unexpected way; "unintended consequences" may occur. For instance, an increase in uptake, although desired in principle, may lead to overwhelmed staff, uptake of poor quality care, or the reduction of health provision by local governments (Thrall 2011). Hence, appropriate evaluation methods ought to be chosen for low-resource settings to account for these, such as mixed methods (Alderman et al. 2009). Mixed methods address both guantitative and gualitative questions and as such may help explain unexpected outcomes and unintended consequences of an intervention. For instance, a mixed-method review saw that women groups had positive effects on various dimensions of women's empowerment, including economic, social, and political. Yet, there were unintended consequences of the interventions for these women empowerment groups and were detailed as: intimate partner violence, stigma, disappointment, and reduced subjective wellbeing. In essence, the mixed-method approach enables the research to capture a broader range of evidence than a review of quantitative studies alone to answer relevant policy questions more comprehensively (Brody et al. 2016).

Evaluations can also provide evidence on transferability. In Kenya, a school deworming programme improved attendance, pupils' performance, and graduation rates. Based on this evidence the funders decided to replicate this deworming programme in India. In India, the programme was not as successful as the intervention was rolled out rapidly (with little planning for the local context) and without a comparison of control and treatment groups. This made it difficult to assess the gaps in implementation in the India setting (Hawkes 2013).

Furthermore, when evaluating the effectiveness of an intervention, it is important to have a comparison group. A good example of a large-scale community intervention without comparison (i.e. no control villages) is the Millennium Villages Project (2004).

This programme built houses, schools, roads, health clinics, and provided education, nutrition, health, training as well as financial resources in nine millennium villages. To evaluate the impact of the project on these villages, 18 indicators were measured, ranging from child mortality and maternal health to measles immunisation and the use of anti-malaria bed nets. The study concluded post-intervention there were significant reductions in child mortality (Pronyk et al. 2012). Although, the baseline data of the control villages had to be retrofitted by surveying participants three years later about indicators at baseline, which created recall bias. Comparisons were needed to test this intervention for impact. Examples include a control group of villages to avoid the above recall bias and a comparison to national trends during that time period (Anon. 2012; Malenga and Molyneux 2012).

Thus, research and evaluation enables us to determine what will work in a given community, where programme strategies may vary from better infrastructure (hospital buildings) to more health workers, lower user fees, better transport, or incentives (Glasgow et al. 1999). Although there are several methods available to an evaluator, some are not suitable. For instance, simple average changes in percentages of descriptive studies, Pearson's chi-squares (x²), or counts may not yield the precision required to understand whether the intervention has 'worked' (i.e., if it was effective). At the other end of the spectrum, sophisticated techniques such as the gold-standard randomised controlled trial (RCT) have gained popularity in social/non-clinical interventions. Many have said that RCT-type evaluations are appropriate, practical and ethical (Glasgow et al. 1999; Tollefson 2015). One of the first RCTs in a social intervention was *PROGRESA* incentivising school and health clinic attendance in low-income communities, and comparing the results to the control area showed effectiveness (Tollefson 2015).

However, RCTs may be inappropriate for community-based interventions as they are expensive or inadequate due to the large sample size required - "there is little research on interventions that address whole populations, (that) are long lasting, or (that) become 'institutionalised' (generalised)" (Glasgow et al. 1999, page 1322-1327). The authors continue to state: "*low-intensity interventions that are less efficacious but that can be delivered to large numbers of people may have a more pervasive impact*". RCTs have proved to be expensive and lacking in external validity and generalisability due to their selective eligibility criteria (e.g. participants are often excluded if they have concomitant medication use, medical comorbidities or are women). RCTs are explanatory but they are also artificial as the evidence is generated under the most

favourable circumstances. In other words, the RCT exclusion criteria may mean the RCT does not reflect the 'real world'. However, experimental non-randomised controlled trials (community trials/controlled before and after studies) are studies that, if well designed, offer external validity but may not offer internal validity if covariates, such as age or education, are not taken into account (Clancy 2002; Van Spall et al. 2007).

Evidence-based community health promotion needs evaluation, but these health promotion interventions are often not evaluated for evidence of effectiveness (Rychetnik 2002). Resources for health promotion interventions are often limited, which means funding for evaluations can be limited. In addition, health promotion evaluations are often not designed and conducted by researchers. In order to better understand the NGO GTN's intervention and its evaluation, and why there is a need to conduct evaluations, the next section presents key issues around maternal health and its indicators for evaluation and health promotion. Those who deliver and plan health promotion services have a very broad range of questions for evaluation studies to answer, of which "does it work?" is one. Evaluation in health promotion is expected to be useful - it influences decisions about whether to stop, continue, or extend a project and how it should be changed or improved. Therefore, there should be an emphasis on the core health promotion values at the outset of the evaluation. Building on core values and adopting a participatory approach to evaluation by involving both stakeholders and the client group has been found to be crucial to the success of evaluation studies in a wide variety of settings and disciplines (Patton 1996; Rootman et al. 2001). Finally, in LMICs, NGOs and academics have collaborated to implement and evaluate projects for the benefit of stakeholders, funders, and science (Duflo 2004; Alderman et al. 2009; Morrison et al. 2014).

At a local level, evaluations usually have several purposes, according to the values and priorities of the people involved. When planning an evaluation to determine whether a programme has "worked", it is important to check what it is that the programme manager/provider and funders consider success, or what they consider to be important for good health promotion performance. Perkins et al. (1999), recommend a negotiated approach to evaluation, in other words, one that takes account of the stakeholder values and expectations which provides a practical framework for gathering evidence. Key reasons for evaluation in health promotion practice which guide the mixed-method approach of this evaluation are to:

- improve the design or performance of a health promotion project, policy, activity or service;
- 2. make choices between health promotion activities;
- Aid decisions about which activities should be funded and which initiatives have greatest impact;
- 4. learn how a particular health promotion project or activity might be repeated and sustained elsewhere;
- 5. find out whether an activity is conducted according to an agreed plan, objective and time frame;
- establish whether a project provides value for money (costeffectiveness) and;
- 7. test whether new ideas will work in practice (Nutbeam 1998; O'Connor-Fleming et al. 2006).

In addition, the planning of an evaluation needs practical research experience (Perkins et al. 1999). An experienced researcher will select the measures/type of evaluation and be able to plan the evaluation. He or she will be able to determine the indicators and outcomes to be measured given the time and resources available. Perkins et al. (1999) suggest that an "outside" researcher (one that is not part of the delivery of the intervention) should conduct the evaluation. The above-mentioned external (neutral) researcher should be involved as a partner with the local health promotion practitioners as the project staff may not have time and/or implementers or governments may not see the value of conducting an evaluation of their activities. As the latter may take time away from programme activities and they may already have the evidence or guidelines at hand (Judd et al. 2001). Lastly, the cost of an evaluation may be a further deterrent (Duflo 2004; Hobbes 2014). Therefore, the responsibility for the evaluation researcher is to understand the resource constraints, the requirements of the stakeholders and the health promotion principles and values that underpin the intervention (Perkins et al. 1999; Judd et al. 2001).

In the health promotion field, there has been considerable debate about the nature of evidence and how we can assess effectiveness (Glasgow et al. 1999). One of the issues that arise in evaluation is that a simple input–output model of evaluation cannot fully address the complexity of the health promotion programme, which is usually a multi-dimensional endeavour. Indicators of 'success' are not a single, one-size-fits-all measure; and meeting a set standard does not equal efficiency (Judd et al. 2001). For example, in an intervention, NGOs implement programmes, train staff, build links with

the community, and give away incentives (Godin et al. 2007). These process measures need to be accounted for in the evaluation, not only the global outcomes. Doing so yields qualitative information - themes and observations that sometimes are not comparable across interventions that permit implementers to spend time (collecting data) getting to know the community. This latter process helps design the data collection. Complementary qualitative research methods are applied in order to understand social phenomena in natural settings allowing researchers to draw meanings, experiences, and views of all participants, including those in the control area. Qualitative methods are concerned not with "how often", as quantitative methods are, but with "why" something happens, "how" it works, and "what" people think (Judd et al. 2001; Clancy 2002; Godin et al. 2007a).

It is well recognised that evaluations should be concerned with process, impact and outcome indicators (Green and Tones 1999). Process and outcome evaluation examine whether targets have been implemented and achieved over the long-term. While an impact evaluation is structured to answer the questions: how would outcomes, such as individual participant's wellbeing, have changed in the absence of the programme? Or how would those who did not receive the intervention (i.e., the control group) have benefitted if it had been available to them. This involves counterfactual analysis, that is, "a comparison between what actually happened and what would have happened in the absence of the intervention" (White 2006). In essence, impact evaluations seek to answer cause-and-effect questions: those changes in outcome that are directly attributable to the programme (Gertler et al. 2011). The pragmatic mixed-methods evaluation was chosen for this PhD study to assess the impact of both the intended and unintended consequences that can be attributed to the GTN intervention (Khandker et al. 2010a; Westhrop 2014; J-PAL 2015).

The evaluation aimed to address counterfactual questions and provide an average impact on the group in question, as there may have been unrelated changes concurrent to the programme implementation. Accounting for these changes is essential and often not solely possible through statistical methods. Concerns about the possible dominance of a quantitative methodological agenda in impact evaluations, such as an RCT (Green 2000a) and its limited applicability to health promotion evaluation, were discussed in Chapter 3.

These considerations informed the evaluation of the intervention in an area where health promotion is paramount, maternity care as the outcome of such an intervention may impact two lives - the mother and the baby. The next section discusses maternal health promotion in improving women's knowledge of and access to maternity services in LMICs, such as Nepal. Also explained are the maternal health indicators, set as standards, using evidence that measures progress to aid in the reduction of maternal morbidity and mortality.

1.3 Maternal health policy & community interventions

Maternal health is a matter of concern on the international health agenda and an important public health and health promotion issue in LMICs. It is estimated that 289,000 women die annually (approximately 800 per day) from pregnancy or childbirth-related complications with more than 99% of these maternal deaths taking place in the LMICs, and the highest rates of mortality in Sub-Saharan Africa and South-Asia (Simkhada et al. 2006; Lozano et al. 2011; WHO 2013). Maternal mortality is defined as:

"The death of a woman while pregnant or within 42 days of termination of pregnancy, irrespective of the duration and the site of the pregnancy, from any cause related to or aggravated by the pregnancy or its management, but not from accidental or incidental causes" (WHO 1992:1238).

Maternal mortality levels have been advocated as a marker for a country's development. The Maternal Mortality Ratio (MMR) is one of the key indicators of the status of reproductive healthcare service delivery and utilisation, and also of women's overall status in society (Defo 1997). Complications of pregnancy and childbirth are still the leading cause of death and disability among women of reproductive age in LMICs (WHO 2009b). In LMICs, 80% of all maternal morbidities or deaths are due to obstetric complications that occur during pregnancy, labour, or puerperium. Five direct causes are responsible for nearly three quarters of all maternal deaths: unsafe abortion (13%), sepsis (15%), haemorrhage (24%), eclampsia and hypertensive disorders (12%), and obstructed labour (8%), (Ronsmans and Graham 2006). The remaining deaths are indirectly caused by or associated with diseases such as malaria and Acquired Immuno-Deficiency Syndrome (AIDS) during pregnancy (WHO 2012a; WHO et al. 2014b).

Poor maternal outcomes, which are defined as morbidity and/or mortality, have a significant impact on families, communities and societies and various socio-economic groups (Furuta and Salway 2006). The cost of treatment for complications can lead to considerable debt for women and their families (Ronsmans 2009). In addition, maternal morbidities can affect women's ability to work, resulting in a loss of productivity and negative outcomes for infants and children (Powell-Jackson and Hoque 2012).

The role of primary healthcare and prevention/health promotion is to improve health, including maternal health. Community-based preventive interventions and health-sector interventions designed to increase women's access to professional and quality medical care, were emphasised at the Alma-Ata conference in 1978 (WHO 1978). In 1987, the Safe Motherhood Initiative launched by the WHO and other international agencies prioritised the following: women's status, education of communities and the strengthening and expansion of core indicators of maternal health (Starrs 2006). For instance, ANC was chosen as a specific indicator at the World Summit for Children (1990). The summit included a recommendation that all pregnant women should have access to ANC, a skilled birth attendant (SBA) at delivery, and access to referral facilities (institutions) for high-risk pregnancies. It was recommended that these facilities should have the capacity to provide emergency obstetric care so that both the mother and child would have improved health outcomes (AbouZahr 2003). The maternal health service indicators, used as global standards for evaluation, are ANC, delivery care (DC) and postnatal care (PNC), both at the community and referral level.

Global initiatives have further galvanised maternal health progress. For instance, the 1987 Safe Motherhood Initiative was a commitment among political leaders: it highlighted the idea that maternal death is an "issue". The term "Safe Motherhood", with its implied focus on women's childbearing role (Starrs 2006) was thus in the public-health realm and a core component of reproductive health policy. Both the initiative's 20th anniversary and the Millennium Declaration strengthened the Safe Motherhood advocacy in the last 20 years.

In September 2000, the United Nations adopted the Millennium Declaration and set eight Millennium Declaration Goals (MDGs), the fifth (MDG5) of which was "to improve maternal health" (UNGA 2000). This goal was translated into two targets: the first one was to reduce maternal mortality by three quarters between 1990 and 2015, and the second was to achieve universal access to reproductive health by 2015 (UN 2011a:30-31). The two key MDG5 indicators for monitoring the progress towards the first target are: (a) the MMR (i.e. the number of maternal deaths per 100,000 live births); and (b) the proportion of births attended by a skilled health attendant.

The MMR indicator represents the obstetric risk associated with each pregnancy. However, measuring MMR is problematic in many countries due to the challenges of obtaining accurate data on the number of pregnancies and determining whether maternal deaths are due to obstetric causes, especially in the community (home) versus hospitals/clinics. Most women in LMICs die in the community and in many countries, this is where they give birth. MMR is often measured in LMICs by surveys from respondents about the deaths of their sisters using the *sisterhood method* of estimation (Graham et al. 1989). This method is based on highly uncertain data. It is also subject to further variance due to poor health facility record keeping (Ameh et al. 2014; Graham et al. 2004). Thus, as an indicator MMR is controversial: if taken as a measure of maternal health progress, it is very variable. In the GTN intervention MMR was not available, as it is a rare event (WHO & UNICEF 2014b).

As a result of the introduction of MDG5, maternal health has received increased attention (Morrison et al. 2008). There has been a 43% decline in maternal mortality between 1990 and 2015 (Hogan et al. 2010; Lozano et al. 2011; WHO et al. 2012; EWEC 2015; WHO & UNICEF 2014b). The most recent progress on MDG5 indicated that nine countries had met the target (out of 75 countries) with a high burden of maternal mortality (WHO et al. 2015). Technical measurements and data are needed to determine progress or assess interventions and the accurate recording of deaths (Ameh et al. 2014; Graham et al. 2002). It should be noted these MMR data did spur the maternal health community, and some countries have made significant progress in maternal health.

Progress in maternal health and MMR is also credited to health and social factors. For example, the Safe Motherhood campaigns took inspiration from the 2006 initiative, which has increased the availability of skilled birth attendants and antiretroviral therapy as well as decreased pregnancy rates as a result of family planning and use of contraception (Hogan et al. 2010; WHO et al. 2012). Furthermore, the wide use of antibiotics led to a decrease in MMR (Costello et al. 2006). Progress was also due to data improvements in death registration, increased investigation into the causes of

deaths of women of reproductive age, vertical health programmes in the 1980s (promotion of breastfeeding, oral rehydration, and immunisations), and implementation of national programmes to improve maternal and child health and to promote women's health. Finally, socioeconomic and demographic changes (economic growth, improved education of women, and decreased fertility rates), and interventions outside the health sector (for example, conditional cash transfer programmes and improvements in water and sanitation) also played a role (Victora et al. 2011). Particularly in China, one-child policies arguably also contributed (Yanqiu et al. 2009).

Also, vital to progress was ensuring a continuum of care from ANC to PNC during the launch of Partnership for Maternal, Newborn and Child Health (2005). This global consortium took on the goal of reducing maternal mortality advocating and integrating it with newborn and child mortality in a continuum of care (Starrs 2006; WHO et al. 2011). In order to continue the progress in achieving MDG5, women's socio-economic status and the health system conditions should be addressed as these also present a risk (WHO 2004). Lozano et al. (2011) highlighted that interventions are still needed for disadvantaged/marginalised women.

Recently, the MDGs reached maturity in 2015 and were replaced by the Sustainable Development Goals (SDGs) towards continued development. The single health goal (Goal 3.7) aims to address the universal need for access to quality sexual and reproductive health services to meet the need of women and their families: "By 2030, ensure universal access to sexual and reproductive healthcare services, including for family planning, information and education, and the integration of reproductive health into national strategies and programmes". The specific targets for measurement are still under debate, yet SBA at birth and MMR are likely to still be used as indicators (UN 2016).

1.3.1 Maternal health access and outcome indicators for evaluation

In this section, the intricacies of provision of these maternal health indicators are discussed as they are used in this evaluation. One key strategy in achieving improvement in maternal health is to increase the uptake of maternal health services, particularly ANC. Uptake of ANC is relevant not only to LMICs. In high-income

countries, such as the UK, there is evidence that 26% of the women who died from direct or indirect causes related to pregnancy were poor ANC attendees (Lewis 2011). Access to antenatal health visits and medicines can prevent death from hypertensive disorders, while death due to sepsis can be averted by screening for prenatal maternal infection and sexually transmitted infections (STI) during antenatal visits and with hygienic infection control measures provided by SBA during birth (Ronsmans & Graham 2006). The antenatal period presents an important opportunity to identify danger signs, symptoms, and potential risks of labour and delivery. It is during the antenatal period that measuring women's blood pressure can identify women at risk of pre-eclampsia and treatment can prevent eclamptic convulsions (AbouZahr et al. 2003). Furthermore, if anaemia is targeted during ANC (especially during the first trimester) with a nutritional intervention, low birth-weight and mother and foetal outcomes can be improved (AbouZahr et al. 2003). Also, tetanus immunisation during pregnancy can be life-saving as it prevents both mother and child from contracting tetanus (Clostridium tetani) (AbouZahr et al. 2003). The antenatal period is also an opportunity for education and counselling: women can obtain information on birthspacing and on STIs including HIV prevention to improve maternal and infant survival (UN 2011b).

While levels of provision and attendance of ANC have increased in many parts of the world during the past decade, only 46% of women in LMICs attend any ANC at all (UN 2011b). It therefore remains a high priority to provide women with adequate ANC. In addition, just over a third of all pregnant women in LMICs have the recommended four ANC visits (Lincetto et al. 2006; WHO 2014c; UN 2011b). In comparison, 66% of women attend ANC in the first trimester in Latin American and the Caribbean and in the Middle East and North Africa, while in Asia this figure is nearly half of the above rate and in South Asia there is overall a low level of use. In Nepal, for example, 38% reported attending one visit and only 9% reported four of more visits (Tuladhar and Dhakal 2012). While, in Sub-Saharan Africa women tend to wait until the second trimester to attend ANC (Mushamiri et al. 2015).

One reason for low or delayed uptake is that in LMICs women have to travel long distances and wait long hours, which deters them from attending ANC, as seen in the *three-delay model* (Section 1.3.2). There are substantial financial and opportunity costs to women for such frequent attendance, i.e. one or more antenatal visits (MOH 2002). Furthermore, women who present for one ANC visit are likely to attend additional visits (AbouZahr 2003). The WHO model of ANC separates pregnant

women into two groups: routine ANC, 75% of the total population of pregnant women who have a minimum of four ANC visits, and the remaining 25% that necessitate special care (AbouZahr 2003).

The added value of ANC leading to better pregnancy outcomes is that it can increase the likelihood of a woman seeking delivery with a skilled healthcare provider (WHO 2004b; Fujita et al. 2005). Women who had four ANC visits were on average 3.3 times more likely to give birth in a health facility. There is a strong positive correlation between at least one visit and having a skilled birth attendant at delivery. ANC can potentially serve as a strategy to increase the uptake of SBA services and ensure access to emergency obstetric care.

All women need a midwife and some women need a doctor, the former to ensure a psychosocial birthing process, and the latter to help with complications (Sandall 2012; UNFPA et al. 2014; EWEC 2015). The WHO definition of a SBA is a health worker with midwifery skills "trained to proficiency in the skills needed to manage normal (uncomplicated) pregnancies, childbirth and the immediate postnatal period, and in the identification, management and referral of complications in women and newborns" (WHO 2004b). Traditional birth attendants (TBA), trained or untrained, were excluded from the definition of SBA because they lacked the clinical skills to manage complications, for example, haemorrhage, eclampsia, or severe infection (Starrs 2006). Thus, while ANC can provide an important opportunity to target maternal deaths, it requires the addition of a skilled attendant and PNC to effectively target MDG5 and provide women with the care they need (Fujita et al. 2005). The effectiveness of only providing ANC, however, is debated - the WHO consensus is that important elements of ANC are likely to improve maternal and/or perinatal health outcomes, but not necessarily maternal survival (AbouZahr et al. 2003; Zeitlin 2013). Also the risk approach (identifying the women who are most likely to go on to develop serious complications) has been shown to have limited effectiveness, as those identified as being at risk often have normal deliveries. As stated in Section 1.3, a multi-level continuum of care approach is needed to target maternal and newborn health (MNH) (WHO 2011). Thus, interventions in LMICs should target the full continuum of care, as timely management of pregnancy and labour, with intervention if needed, can make the difference particularly where morbidity and mortality levels among women of reproductive-age are high. Maternal morbidity and deaths can be prevented through simple cost-effective measures if these are available: blood transfusions, oxytocics to prevent bleeding, and/or manual removal of the placenta by

a SBA (Bayer 2001; UNFPA 2006). Most deaths occur during labour and delivery, hence the need for skilled care during delivery, which should be emphasised to women during ANC (UN 2011b). On the basis of such evidence, the WHO guidelines advise that women should have at least four antenatal visits in pregnancy, the first within the first trimester of pregnancy, and have a skilled attendant at birth with adequate resources and PNC immediately after birth and/or for at least 24 hours after birth (WHO et al. 2014a).

1.3.2 Three Delays impact on maternal health

Improving uptake of maternal health includes those interventions that target the uptake of ANC, SBA, ID, and PNC specifically in rural areas (Section 1.4). In general, maternal mortality is higher among women living in rural areas and poorer communities. In LMICs, poor rural women are the least likely to receive adequate healthcare, especially in regions with low numbers of skilled health workers, health services, or where there is a lack of transportation and the lack of personal finances to travel to and use the health services, such as sub-Saharan Africa and South Asia (Rogo et al. 2006). Therefore, to reduce the delays, there is a significant need for interventions that are delivered within the community setting. For example, most births and newborn deaths occur at home in Nepal (Sreeramareddy et al. 2011). Hence, to reduce mortality, behavioural change interventions are required to improve care at home and care-seeking behaviour (uptake).

Access to maternal health services is a key criterion for the health status of childbearing women across the globe. The *Three Delays* model (2004) proposed by Thaddeus and Maine postulate that access to healthcare could be delayed for a number of reasons. They proposed three points at which action should be taken: (1) when there is a delay in deciding to seek care (first delay), (2) when a woman fails to reach care in time (second delay) and (3) when a delay occurs in receiving adequate treatment at the facility (third delay) (Thaddeus and Maine 1994). Explicitly put, women are hindered from receiving or seeking care during pregnancy and childbirth due to factors such as empowerment (being the main the decision-maker for their health), poverty, distance, and lack of information, inadequate services and cultural practices (Section 1.4). Thaddeus and Maine also argue that women lack assertiveness and have low self-esteem and other social determinants for MNH service utilisation (such as literacy, education level, socio-economic status and parity). They are also unlikely to access services due to a lack of financial support (equity

funds), and staff attitudes at the health facility can be a barrier to access that is not addressed by community-based health interventions (UNFPA 2014).

In order to continue progress towards improving maternal health and decreasing MMR, there is a need for evidence-based health promotion and community programmes that improve access to services. For instance, programmes that are a part of solving the "delays" are applicable to non-attenders of ANC, during labour and birth and PNC where adequate treatment if provided, can mean averting pregnancyrelated mortality (Nour 2008). It is also necessary to increase the service delivery capacity of health providers and to address the four dimensions of access barriers (geographical access, availability, affordability and acceptability) (Jacobs et al. 2012; Khan & Bhardwaj 1994; UNFPA 2014). One of the main recommendations from the report on the MDGs on child and maternal mortality was the need to strengthen health systems, to improve access to maternal health services, and to introduce an evidence-based holistic approach where communities are empowered to demand high-quality services that include well-functioning referral and transport mechanisms (Lozano et al. 2011). Although emergencies cannot be eradicated through labour care, they can be reduced through skilled and motivated teams available at facilities and equipped with necessary medicines and commodities. The teams need to be able to work in enabling environments that promote evidence-based practices, and clientcentred and respectful maternity care services. The availability of good quality essential and emergency obstetric and neonatal care services is crucial for further improvements in maternal and neonatal outcomes (UNFPA 2014). Thus, to continue to improve maternal health the barriers that limit access to quality maternal health services must be identified and overcome at all levels of the health system (UN 2011; WHO 2012b). By strengthening existing health systems and access to these systems in countries and improving intrapartum and postnatal quality of care, treatment can be lifesaving for women (Sharma et al. 2016a; Sharma et al. 2016b).

However, as discussed in Section 1.3, maternal health interventions alone are not enough. Nor is strengthening access to health systems with comprehensive facilitybased midwifery and obstetrical care (Costello et al. 2006), as deaths occur outside the intrapartum state and the facility. Addressing equity, human rights and the economic and social benefits of saving women's lives will benefit infants and children as well (Starrs 2006). Equally important are partnerships across country and regional level between international, national, academic, governmental and non-governmental and community, donors, health professional associations, non-governmental organisations, and academic and research institutions. The need of collaboration among sectors within countries is essential for implementation at the level of the needed interventions (Judd et al. 2001; Duflo 2004). Additionally, more evidence is needed to first develop assessment methodologies and second, to develop interventions that are cost-effective (Campbell and Graham 2006).

For instance, community-based maternity health promotion has been conducted to empower communities in rural Nepal. On the basis that several components are needed to holistically improve maternal care, the Nepali NGO GTN designed an evidence-based health promotion strategy to improve uptake, knowledge and decision-making. Access to health services was a GTN priority; the programme took into account the fact that there was a lack of information and socio-cultural factors that influenced access to care in the community (Chapter 2). The next section discusses Nepal and its health and culture policies that have improved women's knowledge of and access to maternity services in the country.

1.4 Background on Nepal

Nepal is an impoverished low-income Asian country and has a population of close to 27 million (World Bank 2013). Approximately 49% of the population live in the southern Terai region, the most fertile area of Nepal, while 44% of the population live in the Central hill zone that includes the capital and only 7% live in the mountain region of northern Nepal (DFID 2010). Kathmandu is the capital city and the principal urban centre of Nepal. Like most low-income countries; it has a significant education and wealth disparity (UNU-WIDER 2005; World Bank 2006; MOHP 2012). Unemployment is at 45%, about 43% of adults are illiterate (Table 1), and foreign aid makes up 3.4% of Nepal's economy. Additionally, the country's human development index ranks 145th out of 188 countries and experiences large gender disparities (Index Mundi 2012; UNDP 2015). The average life expectancy is 68 years and the total fertility rate is 2.3 births per woman (Table 1). Approximately a third (34.9%) of the population lives in urban areas in 2011, and urban-rural differences are considerable, with rural poverty being nearly 1.8 times higher than urban poverty (UNDP 2014). Figure 1 shows the Pharping area in rural Nepal. Furthermore, Nepal is primarily a patriarchal society, with 77% of households headed by male members in 2013/4 (CBS 2015).



Figure 1 Rural Nepal, Pharping area

© Sheetal Sharma 2013

The country has been through a period of transition from an authoritarian Hindu kingdom to a constitutional monarchy and then to a democratic republic. The transition included events such as the massacre of members of the royal family in 2001, and the decade-long civil war between Maoist insurgents and the government that ended in a peace agreement in 2006. These events additionally stalled progress in the Federal Democratic Republic of Nepal. The removal of the monarchy altogether in 2008 enabled the creation of a democratic republic. Finally, 16,000 deaths were recorded at the end of the civil war (Brown and Felbab-Brown 2012; CIA 2013; Wasti et al. 2015; World Bank 2010).

The country is divided not only geographically but also by wealth, caste, ethnicity, religion, and a federal system (dividing the national government and the smaller local governments) (CIA 2013; van Teijlingen et al. 2015; World Bank 2010). Cultural diversity and complexity characterise the current social landscape of Nepal. The Census of 2011 recorded 125 caste/ethnic groups, and 123 documented languages (CBS 2012). With the introduction of the *Muluki Ain* (National Code) in 1854, the caste system in Nepal has been accepted as the primary organising principle and the major determinant of social identity. Although caste-based discrimination was outlawed in

Nepal in 1963 (in the National Civil Code), it is still present in society - dominant and subservient groups, and disparities in education and wealth are closely linked to the Hindu caste system. It divides the population into hereditary groups (circa 12). At the top of the social order are members of the Brahmin class (priests and scholars), followed by *Chhetris, Newar, Magur, Tharu, Tamang* and *Dalit* with a majority of the population being Hindu (80.6%) (MOHP 2012; Brown and Felbab-Brown 2012). Buddhists and Muslims are minorities who, along with lower-caste people (Dalits) and rural residents, have been historically marginalised (Brown and Felbab-Brown 2012).

The continued political instability contributed to stalled socio-economic development. Unemployment, poverty, socio-cultural and ethnic diversity, socio-economic exclusions, and slow economic growth remain major challenges to progress (WHO 2007). Another more recent challenge is the 2015 earthquake that has decimated selected areas of Nepal, particularly rural areas (Neupane 2015; Sharma 2015).

1.4.1 Population & health policies and programmes in Nepal

Nepal has a fairly comprehensive health policy framework. With the introduction of the Family Planning and Maternal and Child Health Plan (FP/MCH) in 1965, family planning has been a health priority. In the 1990s, health policy introduced preventive, promotive, curative, and rehabilitative care. From 1997, disparities in health status were addressed, assuring equitable access to quality healthcare services with full community participation and gender sensitivity. Initiatives were particularly aimed at vulnerable groups whose health needs are often not met, such as women and children, the poor, underprivileged, and marginalised groups. Policies also extended the primary healthcare system to the rural population, providing modern facilities and trained healthcare providers including female community health volunteers (FCHVs) and TBAs. The indicators Total Fertility Rate (TFR) and Infant Mortality Rate (IMR) were the focus of the MoHP. Policies also have had a role in liberating and empowering women in the village development committee (VDC) (Bishai et al. 2002). The Safe Motherhood Programme used evidence-based policies that stressed the importance of skilled birth attendants at every birth and embodied the government's commitment to improving women's health via skilled birth attendants. This led to the recognition of the importance of SBAs in reducing maternal and neonatal mortalities, further complemented by legislation of abortion (2002), and increased emphasis on equity issues in Safe Motherhood services (Wasti et al. 2015). Since the introduction of these policies, MMR was notably reduced (see also Section 1.4.4).

Programmes during 2001-2012 aimed to increase the quality and uptake of family planning (FP) services and MCH services by strengthening health service delivery systems, building local capacity and engaging stakeholders in the community. The Implementation Plan (2004-2009) emphasised the above with a major aim to achieve the health sector MDGs in Nepal, and a consequent reduction in poverty, through improved health outcomes for the poor and those living in remote areas. From 2010-2015, there was a further refinement of earlier polices and plans for the implementation of cost-effective, evidence-based interventions to make the best use of limited resources. Thus, maternal and child health indicators such as MMR, TFR, neonatal, infant and under-five mortality rates, contraceptive prevalence rates, and the percentage of underweight children were specifically targeted (MOHP 2010a; Dixit 2005). These initiatives were further complemented by the interim development plans drafted after the People's Movement (2006), where Nepal's Interim Constitution (2007) proposed the concept of "health for all". In practical terms this meant providing essential healthcare services free of charge and the right of every woman to a good standard of reproductive health. Part of the above concept was the Aama Surakchhya maternity incentive programme (AAMA), designed to improve indicators of maternal health, which distributed Nepali Rupees (NRs.) 400 (£ 2.4) to women for having four ANC visits, a hospital birth, and the 1st PNC visit. This programme was introduced nationally in 2010. The cash payments, given after birth at a facility, were increased in remote areas: NRs.1500 (£9.20) in mountain areas, NRs.1000 (£6.10) in hill areas, and NRs.500 (£3.10) in the Terai. Incentives also included payments to health facilities for the provision of free care covering normal delivery (NRs.1500 or £9.20 to health facilities with 25 or more beds and NRs.1000 or £6.10 to health facilities with less than 25 beds) and payments to health facilities for complicated deliveries (NRs. 3000 or £18.50) and caesarean-section deliveries (NRs.7000 or £ 43.20). Incentives to health workers for births at home had been reduced to discourage home-based delivery; if a woman gave birth at home, there was no incentive (Ensor et al. 2009; Witter et al. 2011).

Therefore, achieving equity in Nepal meant having concern for poor and marginalised populations. Providing health services, education, and employment in general and to the low-caste Dalits in particular was a priority of the government. To this end, the MoHP spends close to 51% of its funds in rural areas (which are typically poorer), 18% in semi-urban and 31% in urban areas. Despite policies aimed to reduce this marginalisation of groups from any economic, social and political participation and

representation in local and central state, the continuing caste/ethnic and regional disparities provided a medium for the growth of conflict and the decade-long Maoist insurgency, and these have not altered the status quo completely with a resurgence of ethnic and regional issues (van Teijlingen et al. 2015).

1.4.2 Health administration and health workforce

Nepal is administratively divided into 75 districts. Each district, managed by a chief district officer, is further divided into smaller units called VDCs and municipalities. The VDCs are rural areas whereas municipalities are urban. Currently there are 3,915 VDCs and 58 municipalities. Each VDC is composed of nine wards. In municipalities the number of wards varies from 9 to 35. The lowest level of formal healthcare starts from Sub-Health Posts at the VDC level to Health Posts, Primary Healthcare Centres and to specialised care at hospitals at the district, zonal, sub-regional, regional, and central levels (MoHP 2012; Wasti, 2015). In each VDC there is a health post or a subhealth post. A Primary Healthcare Centre is staffed with a Medical Officer, two staff nurses, two auxiliary nurse-midwives (ANMs), two auxiliary health workers (AHWs), and volunteers. Health posts are staffed with a Health Assistant or a Senior AHW, an ANM, two AHWs and a FCHV. Similarly, in a sub-health post the official positions are a CMA, a Maternal and Child Health Worker (MCHW), and a FCHV. Yet recruitment and retention of health workers is a problem, especially in remote rural areas (MOHP 2012). As health staffing has not increased, except for certain categories of doctors and the post of village health worker was created in response to the need for a trained health workforce. For instance, Village Health Workers were also introduced with sixweeks training in basic healthcare and ANM training courses to increase the manpower for the MCH programme. In addition, for maternity care, specialised training for ANM and SBA were also introduced in order to provide village-level delivery service care in a more efficient way. FCHVs provide basic primary health services and health education to promote community participation, to promote women's participation in the process of development, and to improve access to health services (Bishai et al. 2002). Therefore, community-based health workers FCHVs and TBAs act as links between the community, and the formal health system. Finally, volunteers are responsible for helping Village Health Workers to implement health promotion and preventive healthcare.

Although services for health, education and rural amenities have been well-planned they are often inadequately and inequitably delivered (Wasti et al. 2015). There are few health staff and government agencies in the many remote regions of Nepal from which it could easily be assumed that not all births or deaths are accurately recorded (Ameh and van den Broek 2014). Furthermore, there are an estimated 400,000 to 800,000 traditional healers in Nepal, but only 3,500 biomedical doctors. It has been argued that traditional healers can play a central role in scaling-up community healthcare (Poudyal et al. 2003). Yet it is more imperative to increase the nurse-midwife population (as per demand). The nursing ratio is only 5 nurses per 11,825 people (Index Mundi 2012).



Organizational Structure of the Department of Health Services



(Source: GoN, MoH and DHS, 2014).
1.4.3 Nepal: Maternal health and delivery practices

Also lagging in progress is maternal healthcare uptake - less than half of Nepalese pregnant women attend ANC, and over 80% of births occur at home. Despite efforts by the MoHP to encourage facility-based births (WHO 2009b), only 36% of births in 2011 were assisted by a SBA (WHO 2014c; Table 1). In the maternal health context, this places women at risk, as SBA, which in Nepal, contrary to the global definition (Section 1.3.1), includes TBA with only 12 days training (Falle et al. 2009). In LMICs, the risk of maternal death during childbirth is 2–4 times higher among adolescents (younger than 18) than among women aged 20 or older. In a country like Nepal this is of concern as the median age of women's first pregnancy is 20.2 years (MOHP, New ERA & ICF 2012).

Nepal, however, is one of the few nations in the world to have made great progress in decreasing maternal mortality (Shrestha et al. 2014) despite numerous challenges, including poverty and economic disparities, long distances, lack or cost of transport, shortage of staff, facility capacity to treat serious complications, and home birth preference without a SBA (MOHP, New Era & ICF 2012). Nepal achieved its target for MDG5 of reducing the MMR by three quarters between 1990 and 2015 (UNDP 2013), a target set by the Ministry of Health at 250/100,000 (MOHP 2011). From 1990 to 2015, the MMR in Nepal declined from 770 to 258 deaths per 100,000 live births (MOHP 2012; WHO 2015). A WHO report estimated the 2013 MMR at 190 (WHO & UNICEF 2014b; World Health Organization, United Nations Children's Fund, United Nations Population Fund, United Nations Population Division and World Bank 2014), compared to the UK where it was 10 per 100,000 live births between 2010 and 2012 (Manktelow et al. 2015). As discussed in Section 1.3, there exists a certain degree of in-country variability for MMR as it is based on modelled data (Pant et al. 2008). For instance, a study in eight districts reported an MMR of 229 deaths (Suvedi et al. 2009). An independent study in the Sarlahi district (which generally has more accurate statistics in health than the national average) found a MMR of 529 deaths per 100,000 live births, in other words, double the government target (Wee et al. 2010). It should be noted that regional differences in maternal health are not uncommon in South Asia and have been reported in many studies in Nepal (Khanal, Adhikari et al. 2013), India (Sreeramareddy, Harj, et al. 2011), and Pakistan (Badshah et al. 2008). MMR

reductions are credited to several demographic drivers, as discussed in the next section.

1.4.4 Demographic drivers of decline in MMR

It seems that the progress in crucial maternal health indicators (Section 1.3.1) alone is not enough to bring about such a dramatic decline in MMR, as the reduction in maternal mortality coincided with severe political upheavals. The reasons for decline in MMR are unclear since, as discussed in Section 1.3, MMR is based on highly uncertain data. There is some evidence that a reduction in fertility, changes in education and wealth, improvement in women's education, gender empowerment, and reduction of anaemia may each explain more than 10% of the district variation in maternal mortality (Shrestha et al. 2014; Hussein et al. 2011). Some of these factors are discussed below.

Education is considered to be one of the key factors in improving standards of living in modern life. Women's improved education status can contribute to a decline in MMR. In Nepal, where the patriarchal system has a strong influence in governance (Mullany et al. 2005), the education ratio between the respective male and female populations shows a considerable disparity. For example, there is a large disparity between literacy of males and females in Nepal. In 1994/95, 2003/4 and 2010/11 the difference between the male and female literacy rates were 34.1 per cent, 30.7 per cent and 27.1 per cent, respectively (World Bank 2011). Another factor in the decline in MMR is a rise in the age of first sexual intercourse. NDHS data (2011) showed differences in age at first sexual intercourse by sex. Nepali women aged 25-49 initiate sexual intercourse at a median age of 17.7, just after marriage (MOHP, New ERA & ICF International 2012). Women in Nepal marry at a young age - for women aged 25-49 years the median age of marriage was 17.5 years old (MOHP, New ERA & ICF International 2012). Likewise, there has been a rise in the age that women have their first birth in Nepal (median age of 20.2), which has also had an impact on MMR. Women with no education had their first birth four years earlier than women with higher education (median age of 19.7 compared to 23.7) (MOHP, New ERA & ICF International 2012). However, fertility in Nepal has declined over the past twenty years. Women have on average 2.6 children, a decrease from 4.6 in 1996, 4.1 in 2001, and 3.1 in 2006 (Pradhan et al. 1997; MOH, New ERA & ORC Macro 2002; MOHP, New ERA & Macro International 2007; MOHP, New ERA & ICF International 2012). This decline in fertility has impacted

MMR as maternal mortality risk depends on the number and timing of pregnancies in a woman's reproductive lifespan, by the presence of co-morbidities, and by obstetric care (Cleland et al. 2012). However, fertility varies by residence, region, women's education and economic status. Additionally, fertility increases as household wealth decreases. As of 2011, women who had higher education had an average of 1.7 children, while women with no schooling had an average of 3.7 children (MOHP, New ERA & ICF International 2012). Concomitantly, there has also been a marked increase in the use of contraceptives from 1996 to 2006, although usage remained the same from 2006 to 2011, probably due to male overseas economic migration (MOHP 2012; MOHP, New ERA & ICF International 2012). Contraception use was at 50% in Nepal in 2011 (Table 1). Use of modern family planning methods is fairly high in both urban and rural areas (50% and 42%) respectively). It is interesting to note that modern contraceptive use is lower among educated women; only 35% of women use a modern method compared to 49% of those without education. Thus, these trends in family planning led to birth spacing and reductions in the number of pregnancies (MOHP, New ERA & ICF International 2012).

FCHVS are thought to be key contributors of the reduction of maternal mortality in Nepal due the key factor of task shifting of maternity care education to the grassroots-level (i.e. FCHVS/VHW/MHCW) (Koirala 2012). This increase in health manpower was a vital point that has contributed to the reduction of maternal mortality in the country, especially in the rural part as it addressed a void in care (MOH, New Era and USAID 2014; WHO 2015).

Additional key factors in targeting the reduction of MMR are the uptake of maternal health services and indicators (Koblinsky 2003). The maternal health progress indicators used as standards in Nepal and worldwide are measured by the Demographic Health Surveys (DHS) (MOHP 2012; MOHP, New ERA & ICF International 2012). These surveys include maternal health indicators such as the number of ANC visits, the timing of the first ANC visit, components of ANC (tetanus toxoid vaccination and iron/folic tablets), place of birth (institutional delivery or ID), SBA at birth, characteristics of the birthing, birth complications, problems in accessing healthcare, items for delivery services, and essential supplies for delivery and attending PNC.

However, Nepal has areas of low uptake of maternal health services that are influenced by traditional healers and religion (Syed 2008; Sharma et al. 2016b). Home birth remains the preferred option for many (Section 1.4.3), and one key problem is the slow decision-making process at home when something goes wrong due to the lack of women's autonomy and poor recognition of complications of pregnancy (MOHP, New ERA & ICF International 2007; MOHP, New ERA & ICF International 2012).

Moreover, as detailed in Section 1.3, policy and the health system have also played a role - these legal changes have affected the provision of maternity care. For instance, until 1951, women in Nepal had no legal status and their legal rights were negligible. In 2002, the government outlawed child marriage and polygamy; legalised abortion of up to the first 12 weeks of pregnancy. The Interim Constitution of Nepal (2007) recognised reproductive health as a fundamental right, one of the first nations to do so (Nepal Law Commission 2007; Vijeyarasa 2009).

Table 1 puts Nepal's progress in terms of MMR into context, as India with double the Gross National Income 5,350\$ GNI PPP and a higher health worker expenditure has the same MMR (190). Afghanistan has a similar \$ GNI PPP (1,960\$ GNI PPP) but, predictably, a higher MMR (400) due to the remoteness of regions, high instability and insecurity in the country and weak reproductive rights for women (Arnold et al. 2015; World Health Organization, United Nations Children's Fund, United Nations Population Fund, United Nations Population Division & World Bank 2014), compared to the UK, where it was 8 per 100,000 live births between 2010 and 2012; (Manktelow et al. 2015). Nepal continues to do well in other progress indicators (Table 1) despite a civil war and a low GDP rank of 97/230 (2014). The 2,260\$ GNI PPP of Nepal (2013) does not include personal remittances, mainly from Nepalese men working abroad, which helps women to pay for care – this amounts to 22.2% of GDP. The data are in current U.S. dollars. The table definitions and sources are in Appendix II.

Demographic and health	Nepal	Bangladesh	India	Afghanistan	U.K.
indicators	(Year)	(Year)	(Year)	(Year)	(Year)
Population, millions ¹	27	153	1221	30 (2011)	63
Adult literacy rate, % of people aged 15 and above ²	57 (2011)	59 (2012)	-	32 (2011)	-
Life expectancy at birth, years ³	68 (2013)	71 (2013)	-	61 (2013)	81 (2013)
Maternal mortality ratio per 100,000 live births ⁴	190 (2013)	170 (2013)	190 (2013)	400 (2013)	8 (2013)
Neonatal mortality ratio per 1,000 live births (2013) ⁵	23 (2013)	24 (2013)	29 (2013)	36 (2013)	3 (2012)
Births attended by skilled health staff, % ⁶	36 (2011)	34 (2013)	52 (2008)	39 (2011)	100
Crude birth rate per 1,000 people ⁷	21 (2013)	20 (2013)	20 (2013)	34 (2013)	34 (2013)
Crude death rate per 1,000 people ⁸	7	6	8	8	9
Total fertility rate, total births per woman ⁹	2.3 (2012)	2.2 (2013)	2.5 (2013)	4.9 (2013)	1.9 (2013)
Contraceptive prevalence, % ¹⁰	50 (2011)	62 (2013)	55 (2008)	21 (2011)	84 (2009)
Pregnant women receiving prenatal care, % ¹¹	58 (2011)	53 (2013)	-	48 (2011)	-
Health expenditure per capita, per purchasing power parity ¹²	135	95	215	143 (2013)	3311
Out-of-pocket health expenditure, % ¹³	46.2	14.6	58.2	73.8	9.3
(\$) GNI per capita, purchasing power parity ¹⁴	2260	3190	5350	1960	37970
GDP per purchasing power parity rank ¹⁵	97 (2014)	36	4	101	11

Table 1 Demographic and health indicators

¹ rounded off to closest million

1.5 Rational for this thesis

This mixed-method evaluation aims to enable the development of a better understanding of potential causes of care-seeking behaviour in maternal health. This evaluation will assist in determining the best way to evaluate maternal health promotion interventions and whether there was an increase in the uptake of services, a change in knowledge, attitudes and beliefs and the cost of providing and evaluating health promotion. In order to evaluate the intervention from an "effectiveness perspective", experience was gained in conducting evaluations of health promotion interventions in LMICs, specifically evaluation techniques in the field of reproductive health while at the University of Barcelona, Spain and the University of Buenos Aires, Argentina. An "effectiveness" and economic analysis was also conducted, which is a new application in this field of health promotion evaluations, in order to assess what the cost of providing these interventions and evaluations are. These activities were supported by grants from both Bournemouth University and Santander Universities.

On a professional level, this PhD leads to an understanding of potential factors affecting care-seeking behaviour in maternal health, and developed the researcher's skills in mixed-methods research and evaluations. The latter are needed in order to understand how to measure "what works" given limited resources settings. This research will be relevant to those who wish to know how evaluations are conducted, and in evaluating whether health promotion was effective in improving maternal health. The findings may affect how interventions are evaluated in similar environments in the future. The findings may also assist relevant national bodies or NGOs in producing health promotion curricula for implementation, community mobilisation and evaluation.

Finally, this PhD focuses on issues that are important to the researcher at a personal level, as she encountered them growing up in Kenya, a LMICs country, and in the past few years spent considerable time in Nepal. She believes that this type of research can ensure accountability in programmes that aim to minimise inequalities and inequities in health, in particular in access to maternal and child health services. In research, a 'voice' is given through writing to those who do this important work of saving mothers' and children's lives and

additionally share "what works" for these women who are in great need of healthcare.

1.6 Summary

This introductory chapter has established the importance of evaluations in community-based health promotion interventions. It has also discussed an issue of importance in public health - maternal health and related causes of mortality, such as access to and uptake of services. It further described global initiatives and indicators, such as the role of MDG5, in addressing maternal health and the high MMR in LMICs, such as Nepal. The latter was also discussed to provide context to the GTN intervention. Most births and deaths occur at home in Nepal, hence to reduce mortality, behavioural change interventions are required to improve care at home and care-seeking behaviour. Evaluative research, as described here, can identify community-level maternal and newborn care practices and care seeking behaviour (or lack of), as well as inform the design and evaluation of health programmes, such as GTN, that target barriers to access to care, which will be discussed in Chapter 2.

Chapter 2 Evaluation, health promotion & GTN

This chapter describes the foundations and principles of health promotion and the Green Tara Nepal (GTN) maternal health promotion intervention (henceforth called "The GTN Intervention") that aims to improve the knowledge and access of women to maternal healthcare in Nepal. The intervention used health promotion to empower individuals with increased knowledge of maternal health and to encourage/empower expectant and new mothers to seek care. The chapter concludes with a literature review on similar interventions.

2.1 GTN intervention aims (or approach) and health promotion

Research is often passive: it is done to observe or measure change. However, "action research", as used by GTN, was done to facilitate change in the individual and the community. Action research is applicable to small-scale interventions (Glanz & Rimer 1997; Baker 1999). Action research aims to include service users and communities in both the delivery of health interventions and in research projects. User interactions are formed to develop more focused strategies and to cater to the needs of the members of society whose 'voices' are least heard, such as women and children (Osrin 2003; Akhund and Yousafzai 2011).

This type of intervention is part of solving the first delay of the "3 delays" model (Section 1.3.2), identified in the maternal morbidity and mortality conceptual framework, namely the delay in making the decision to seek care (Thaddeus and Maine 1994; Nour 2008). The key aims of the intervention were to (a) understand why pregnant women do not access existing services; (b) identify and address socio-cultural issues, psychosocial barriers and social organisation, i.e., women's status and influence of others in decision-making, and economic, geographic and financial access to maternal health services; and (c) meet the potential increase in demand by the concomitant strengthening of the existing service provision (Simkhada et al. 2006; van Teijlingen et al. 2012).

Essentially, the health promotion intervention aimed to improve access by addressing cultural and psychosocial barriers, and by achieving empowerment and community participation by working with the community to change both individual and group behaviour in order to improve awareness (van Teijlingen et al. 2012). The GTN intervention took the form of empowerment referred to by Hulton and colleagues (2007) - to improve knowledge, attitudes and beliefs, decision-making, and encourage uptake of health services (Section 2.1). The GTN was not a midwifery intervention. The programme took into account the fact that there was a lack of information and that socio-cultural factors in the community influenced access to maternal healthcare and could be addressed with health promotion. In order to have a better understanding of the intervention, the reader is introduced to key concepts of health promotion in the next section.

2.2 Health promotion theories

Health promotion and public health strategies are based on the understanding that health is a concept engaging social, mental, spiritual, and physical wellbeing. Public health is concerned with the assessment of the health of populations, formulating policies to prevent or respond to health problems, promoting healthy environments, and promoting societal efforts to invest in living conditions that create, maintain and protect health. This covers an extremely wide range of interventions aimed at improving health with various levels and types of intervention including health promotion (Waters et al. 2006).

While public health has tended to place more emphasis on the end results, health promotion has placed more value on the means used to achieving those ends (Green & South 2006). The most commonly applied health promotion definition is the one used in the Ottawa Charter (Page 1):

"Health promotion is the process of enabling people to increase control over, and to improve, their health. To reach a state of complete physical, mental and social wellbeing an individual or group must be able to identify and realise aspirations, to satisfy needs and to change or cope with the environment."

(WHO 1986)

A "holistic view on health" or holism was identified as one of the key principles of the Ottawa Charter for Health Promotion (WHO 1986) and is linked to social ecology with the determinants of health. First, social ecology is an approach which views one's socio-economic and policy environment as a key influencing factor of one's behaviour (McLeroy et al. 1988; Stokols 1996). Second, the determinants of health are interactions between social and economic conditions, the physical environment, individual lifestyles and health. Due to the influence of these approaches and the resultant complicated and intricate interactions, "we" are likely to need quite multi-dimensional complex interventions and therefore an evaluation needs to be cover this complexity to support improved health and learning outcomes (Booth and Samdal 1997).

At the policy level progressively, the Ottawa Charter (1986) and the Jakarta Declaration (1997), among others, enforced the notion of health promotion with goals of empowerment and a more long-term and fundamental shift in village, family, and gender power relations (WHO 1986; WHO 2005a). Over the years, health promotion moved beyond a focus on solely individual behaviour towards a wide range and depth of social and environmental interventions (WHO 2004a). For example, health promotion policy in the Bangkok Charter for "Health Promotion in a Globalised World" came to the consensus that health promotion should be central to the global development agenda, a core responsibility of all governments, a key focus of communities and civil societies, and a requirement of good global practice (WHO 2005). Bangkok re-enforced the global commitment to health promotion and effective interventions as a strategy for health promotion (WHO 2009a).

The Ottawa Charter, and its "successors", introduced the notion that health was a broad concept, and that the disease approaches were highly related to health education and promotion, i.e., active participation by people to directly affect their health and the broader determinants of it, or *holism* (McQueen and De Salazar 2011). Subsequently, health promotion is not a biomedical model of health. In a biomedical model, health and illness are distinct states. Medical science often looks at the human body as a complex physical system which is dysfunctional when affected by disease, i.e., a pathogen. In the biomedical model (pathogenic), the emphasis is to treat the pathogen and cure the

individual (van Teijlingen 2005). While health promotion has multiple determinants and is a multi-dimensional salutogenic concept, it emphasises the social and mental aspects of health in addition to the physical body (Antonovsky 1996). Therefore, health is something more than a state in which disease is absent (Green & South 2006). Also stated in the Charter, health promotion enables people to learn to make choices conducive to their health and throughout life to cope with illness and injuries. This learning was advocated for in school, home, work and community settings through educational, professional, commercial and voluntary bodies, and all supported with theoretical guidance (WHO 1986).

There are several theories involved in health promotion. Theories range from behaviour-change theory (at the individual, organisational and community levels) to social change theory which covers the community and policy development (Green and Tones 1999). These theories are used to help identify potential points of intervention. Regarding theory building in health promotion, evaluation is a useful contribution to fulfil the Ottawa Charter's action means for health-promotion strategies and programmes. They include the development of personal skills and motivation (e.g. to stand up to peer pressure), strengthening community action, create enabling environments and holistic policy, and reorient health services to a social model (Glanz & Rimmer 1997; Nutbeam 1998; WHO 1998; US Department of Health & Human services 2002).

To implement health promotion interventions and fulfilling the above-mentioned action means, Rootman et al. (2001) identified seven key principles concerning health promotion activities:

- empowering: enabling individuals and communities to assume more power over the factors that affect their health;
- 2. participatory: involving all concerned at all stages of the process;
- 3. holistic: fostering physical, mental, social and spiritual health;
- 4. intersectoral: involving the collaboration of agencies from the relevant sectors;
- 5. equitable: guided by a concern for equity and social justice;
- 6. sustainable: bringing about changes that can be maintained once the initial funding has ended;

7. multi-strategy: using a variety of approaches in implementation.

Therefore, the attributes of health promotion can be summarised as the need to implement community-driven health reform based on social participation. The latter refers to the willingness of communities to become empowered in determining long-term priorities. Attaining this effective and equitable health promotion therefore requires an understanding of the social ecology and determinants of health (Povlsen and Borup 2011). As the ecological models lend to that, the idea that an individual's behaviour is shaped by a dynamic interaction with the social environment, which includes influences at the interpersonal, organisational, community, and policy levels (McLeroy et al. 1988; Stokols 1996). The success or otherwise of interventions relate to the degree to which recipients value the intended change and internalise in the health promotion activity (Tones and Tilford 2001; Mittelmark 2002).

It can be argued that the central tenets of health promotion are holism and empowerment. According to Wallerstein (1992), health promotion empowerment is a social process that promotes participation towards the goals of increased individual and community control, political efficacy, improved quality of life and social justice. First, empowered individuals are needed to mobilise communities. The empowered communities can generate norms and support systems that enable individuals in greater numbers to acquire the competencies and characteristics of self-empowerment (Tones and Tilford 2001). In effect, this suggests that a better strategy for empowering more individuals lies not with individual empowerment programmes, but with the process that accompanies a whole community action empowerment strategy or a positive spillover effect (de Heer 2011; Vanderweele 2013). Socio-politically-oriented community development approaches are the most "ideal" form of health promotion practice (Green 2000; McQueen and De Salazar 2011; Tengland 2012). Moreover, empowerment is an interpretive concept, embraced differently in different contexts (Section 1.3.2): empowerment in LMICs combines several notions that arise from the Charter, including individual and collective capacity and participation. Over the years, health promotion empowerment evolved to a multilevel construct of a social process by which individuals gain control over their lives, their organisations, and their communities, in the context of changing

their social and political environment, to improve equity and quality of life (McQueen and De Salazar 2011). However, changing people's health and improving their quality of life is not easily done (Hawe et al. 1990a). For instance, success in health promotion is not a given; evidence from the past 20 years indicates that many community-based programmes have had only a modest impact. For instance, the model of community health workers (CHWs) providing the first line healthcare (i.e. primary healthcare with CHWs) was a health promotion intervention adopted by many governments and nongovernmental organisations after the Alma-Ata Declaration (Section 1.3). Yet by the 1990s, many government programmes for CHWs had disappeared because of the problems in integrating them into national programmes (Brauman et al. 1999). In terms of maternal survival, efforts of community approaches focusing on TBAs lacked clear evidence of effectiveness (Rosato et al. 2008). For example, the Projahnmo cRCT assessed the effectiveness of specially trained CHWs, who provided a home-care package showed a reduction in the neonatal mortality rate when compared with the comparison group (the two study arms). Yet the third community care arm, in which community mobilisers held community meetings with women in villages, showed no effect on neonatal mortality compared with the control arm (Bagui et al. 2008). The apparent conclusion is that the interpretation of the findings of any intervention must be considered carefully to guide policy makers (Section 1.2).

In short, the health promotion literature, over the last decade or so, has demonstrated a move from individual empowerment programmes to far more emphasis on policy-driven initiatives that work through research, particularly at the level of collective action (Whitehead 2004). In addition, the global health promotion strategy documents also recognised that active participation by people to directly affect their health and the broader determinants was imperative. They were reacting to many of the emerging ideas of the time that were outside the biomedical and public health sectors such as equity, salutogenesis, healthy cities, complexity, participation, context, and implementation. As a consequence, health promotion is implemented through various foci, such as individual, interpersonal, community, institutional and public policy and one of which is community-based interventions (Eriksson and Lindström 2008; McQueen and De Salazar 2011). The following section delves into the latter as was applied by GTN.

2.3 Community-based health promotion

In marginalised groups, the social environment affects health, and personal behaviours play a role. As discussed in Sections 1.3 and 1.4 maternal and child health are subject to risk factors, such as isolation, lack of social support, low self-esteem, and risk conditions, such as poverty, discrimination, and steep power hierarchies. Here, community engagement with a problem-solving process can collectively change marginalised groups' circumstances such as those in rural Nepal (Section 1.4) – perhaps only to mobilise their communities to initiate localised actions based on their immediate needs, such as health access, rather than broader social and political actions (Rosato et al. 2008). As community-based health promotion is concerned with a salutogenic orientation, it is important to start from a consideration of how health is created and maintained through community-based health promotion (Judd et al. 2001). Health promotion community participation in healthcare is attributable partly to the scarcity of resources committed to primary care, the perceived failure of conventional health education and primary healthcare to deliver health benefits by engaging users to adopt positive healthcare behaviours. For instance, a local health community may have little accountability to the community and the women. As the latter may be passive due to differing perception or low awareness of need (Bolam et al. 1998; Bryce et al. 2003; Manandhar et al. 2004; Victora et al. 2011). As discussed in the previous section, trends in the field of health promotion emphasise community-based programmes employing multiple interventions as the main strategy for achieving population-level change in risk behaviours and health and community mobilisation. Followed by active participation in achieving programme goals and implementing interventions in multiple community settings (Merzel and D'Afflitti 2003). Community mobilisation's key concepts include social planning, local development, and social action such as active community participation (Merzel and D'Afflitti 2003; WHO 2009a). Current trends in the field of health promotion emphasise that community-based programmes must influence multiple levels for achieving population-level change in risk behaviours and health in order to be effective. These multiple levels of influence are intrapersonal, interpersonal, institutional, community, and public policy (Godin et al. 2007). In essence, focusing on a community and population-based approach has steadily evolved from a shift in emphasis from individually focused explanations of health behaviour to ones

that also encompass social and environmental influences, as reflected in ecological models of health. Here, the community and its social and cultural processes have an important role in shaping maternal health promotion strategies (WHO 2015). Therefore there is not a single, objective and universal notion of what health promotion is. It seeks to permit multiple perspectives or approaches rather than focusing on a single goal or desirable outcome (Webb and Harinarayan 1999). Thus, health promotion through diverse means aims to provide positive maternal health. In maternal health, as seen in Section 1.3, childbirth is a "normal" psychosocial process for women (Sandall 2012). Whereas maternal health promotion "promotes" childbirth and not risk: i.e. a majority of women should have uncomplicated labours (Berg 2005). If a medical model is focused on in childbirth with counter-physiological practices, where pregnancy is treated as a risk, it may mean a relatively uncomplicated delivery with minimal intervention can "change" into a life-threatening emergency (Tracy & Tracy 2003; Hundley 2013, 2014). In LMICs, if women are empowered to seek supportive care throughout labour (Sections 2.4 and 2.5.1), behaviour change may mean they seek emergency care when needed (Hulton et al. 2007). Also, Rosato and colleagues (2008) and Wallerstein (1992) provided further insights applicable to health in marginalised groups (Sections 1.3 and 2.3) where many health problems are rooted in "powerlessness" (i.e. lack of decision-making), and could be addressed by social and political empowerment.

Therefore, health promotion is more holistic and empowering if it involves dialogue and problem solving rather than didactic messages where communities can develop a critical knowledge base to recognise and address the underlying social and political determinants of health (Rosato et al. 2008; Wallerstein, 1992). For instance, if there exist gender inequity constraints to improvements in maternal survival, such as in rural Nepal (Section 1.4.1), empowered groups could give women the understanding, confidence, and support to choose a healthy diet in pregnancy, and seek care or advice outside of their homes (Prost et al. 2013).

2.4 Health promotion approaches

There are several strategies to promote health in individuals and populations: medical or preventive, behaviour change, educational, empowerment, and social change. No one approach is or has been responsible for improvements in the health status of individuals or populations. Often a combination of some or all of these approaches is required (Naidoo and Wills 2000). The GTN intervention includes elements of education, behaviour change and empowerment approaches. Thus GTN used health promoters to help communities identify health and social problems, and to plan and implement strategies to address these problems.

First, the "education approach" is strongly linked to health education, a component of health promotion. Health education is an activity that seeks to inform the individual on the nature and causes of health/illness and that an individual's personal level of risk is associated with their lifestyle-related behaviours. Health education seeks to motivate the individual to accept a process of behavioural change through directly influencing their value, belief, and attitude systems, where it is deemed that the individual is particularly at risk or has already been affected by illness/disease or disability. The professional intention is that the "education" will culminate in behavioural change and lead to a positive health status outcome (Whitehead 2004). It seeks to provide knowledge and information, and to develop the necessary skills so that people can make informed decisions about their behaviour. The following assumptions exist that increasing knowledge will lead to change in attitudes that may result in behavioural change. Another is that education is intended to have a positive outcome. For example, in order to help an individual understand the effects of smoking on their health to then make a decision, whether or not to stop – the education approach here increases an individual's knowledge about healthy choices with the provision of medical or preventative information. The health promotion activity will be to help them to learn how to stop smoking. The approach can be described as a way of working which increases people's ability to change their social reality and that it is possible for them to change that reality (Macdonald et al. 1996). Here, in the education approach, the health promoters undertake community development work, often with others. For instance, the

community development professionals help communities to identify concerns and work with them to plan a programme of action to address these concerns, such as knowledge and access of maternal health (Godin et al. 2007). Moreover, health education assumes that the health professional has the necessary health-related information to impart and that the recipient is in need of and will benefit from this information. A further assumption is that if the recipient has correctly assimilated and disseminated this information - any further action on their part will involve change or modification in their behaviour. Recipients may be supported through this process, but they are ultimately personally responsible for any action that they may or may not undertake (Whitehead 2004).

Second, the "behaviour change" approach aims to encourage individuals to adopt "healthy" behaviours (exercise, good nutrition, and smoking cessation) or prevention behaviours. For example, giving persuasive education to prevent non-smokers from starting or to persuade smokers to cease (Ryan 2009; Dawson and Grill 2012). Behaviour is partly responses to conditions people live in, and the causes of these conditions may be outwith the individual control. Yet the behaviour change approach remains popular with health promotion agencies for it views health as the property of the individual (Ryan, 2012). Therefore there exist the assumption, or risk, that people can make real improvements to their health by changing their lifestyle. Furthermore, if people do not take responsibility for their actions, they are to blame for the consequences, i.e. a victim blaming approach. Over time, it is acknowledged that a complex relationship exists between individual behaviour, social, and environmental factors (Nutbeam and Harris 2004; Baum et al. 2006; Nutbeam 2006).

Lastly, under the empowerment approach comes the notion of helping people to identify their own concerns, gain skills, and make changes to their lives accordingly. In health promotion, the definition of empowerment has evolved; in the past, it was defined as a state (Green and Tones 1999b; Perkins et al. 1999). Recently it is taken as both a state of being empowered, and as a process to achieve this goal (Tengland 2012). For example, if anti-smoking is a concern and clients identify what, if anything they want to know and do about it (Section 2.1). It is also considered as the bottom-up approach where a health

promoter acts as a facilitator, rather than an expert, for change by supporting individuals and communities to make changes.

It is useful to combine these two approaches - behaviour and education - as they are complimentary. Yet to "only" change a person's behaviour may fail to address important issues, such as powerlessness or lack of control (empowerment). Therefore a population requires both the attainment goals envisaged in behaviour change projects, and empowering instrumental goals, such as increased real/tangible opportunities in life, for example, access to health services. Therefore, the behaviour-change model may not consider the right to autonomy (i.e. strengthening the 'whole' individual or group). The third approach empowerment, on the other hand, respects the participant's right to autonomy. It tends to increase the ability for autonomy as well as increasing other coping skills, and is likely to reduce (health) inequalities (Dawson and Grill 2012; Tengland 2012). When these three approaches are combined: healthrelated advice is provided in order that they can make sense of their actions and behaviours and consequently, will act on any tensions (stress) that may arise if they are empowered. With the assumption that an individual values and prioritises their health as important, and that it is reasonable for the health professional to act on the basis that the individual wants to avoid or reduce any negative health state by changing their behaviour. The delivery of these approaches by GTN is expanded upon in the next section.

2.5 GTN and health promotion

This section discusses the GTN intervention that was delivered in a repeated cycle in four stages: needs assessment, programme planning, implementation, and evaluation (Figure 3). The key aims of the intervention were to (a) understand why pregnant women do not access existing services; (b) identify and address socio-cultural issues and psychosocial barriers (social organisation, i.e., status and influence of others in decision-making, and economic, geographic, and financial access); and (c) meet the potential increase in demand by the concomitant strengthening of the mental health existing service provision (Simkhada et al. 2006; van Teijlingen et al. 2012). The objectives of the intervention are detailed in the implementation section.



Figure 3 Health promotion intervention cycle

(Source: van Teijlingen et al. 2012)

The intervention started with a needs assessment of the rural communities, based on research evidence that supports health promotion interventions (Section 2.1). The results of this phase showed that an increased demand for local maternity service provision was deemed viable, as uptake stayed within the health system capacity for maternal care. Thus, it was more likely to be sustainable compared to the introduction of an external intervention, which is new to the community and potentially expensive (Sharma et al. 2017).

2.5.1 Needs assessment cycle

Needs assessment is the first step in planning any health promotion initiative. It helps to identify and then to analyse a health problem and the nature of the target group. This activity is for the purpose of planning any health promotion action (Hawe et al. 1990). The second step is intervention planning, which includes mapping, group creation and curriculum development. Thus the GTN

intervention was designed based on the findings of the community needs assessment, including focus group discussions in the community, and complemented by a consultation process with stakeholders, including funders, academics, local policy makers and local healthcare providers. The importance of NGOs collaborations was mentioned in Section 1.2. From the outset all stakeholders were involved in the needs assessment, deciding which area of health promotion to focus on and planning the community household mapping and monitoring (Barnett 2012; CDC 2013). The needs assessment determined that improving maternal healthcare was a key priority.

The underlying philosophy of empowerment and community participation were essential in this low-cost health intervention to make it sustainable (Section 2.2 and 2.4). This was achieved by incorporating the diverse and changing needs of the local communities in the planning and to make best use of the existing resources whether government or NGOs to ensure the intervention was low-cost (e.g. government clinics/buildings etc.) (van Teijlingen et al. 2012).

In the design, implementation and evaluation of interventions, it is necessary to include formative research at the start (i.e. needs assessment) with the participation of community-based stakeholders (Morrison et al. 2008). Formative research in this context is the description of practices and beliefs and rapid rural assessment of local needs. Thus formative research can provide information about existing practices, and this is a way in which researchers and community members can head towards a solution to optimal health (Morrison et al. 2008). This improves the chances of empowerment, intervention ownership, participation, and sustainability once the intervention has ended, as the stakeholders are part of the decision-making (ADB 2001; Judd et al. 2001; IFAD & Tango International 2009).

Once the needs assessment was concluded, step 3 was implemented. The latter consisted of a continuous evaluation (a monitoring and evaluation or M&E) to determine response/uptake to health promotion activities (Section 2.4).

2.5.2 Programme planning cycle

The community, with the facilitation of GTN staff, influenced actions to target various socio-economic barriers that could limit the utilisation of maternal healthcare services by local rural women (Simkhada et al. 2010; van Teijlingen et al. 2012; Sharma et al. 2017). To target these barriers, GTN worked with women and with the people who influence their ability to access health services, medical facilities, and money for delivery (mothers-in-law and husbands). The socio-economic barriers were grouped into social organisation, status and influence of others in decision-making, economic, geographic and financial access (van Teijlingen et al. 2012; Sharma et al. 2017). It was not in the remit of the intervention to address economic barriers; as the introduction of economic incentives, although effective, are not sustainable over the long-term (Witter et al. 2011; Powell-Jackson & Hogue 2012). The health promotion intervention was designed to be low-cost, flexible, multi-disciplinary, potentially sustainable, and participatory (vanTeijlingen et al. 2012). Participatory approaches as used by GTN are advocated in health promotion (Rootman et al. 2001). They are detailed in the implementation cycle of the intervention (Section 2.5.4.1).

The reasoning GTN applied was that better maternal care results in fewer women suffering from complications during childbirth. Health promotion here improves knowledge, attitudes, and beliefs towards seeking care, especially during pregnancy complications. For example, bleeding or feeling weak during pregnancy and seeking a SBA, or exclusive breastfeeding from birth as opposed to the current practice in Nepal of supplementing breastfeeding with un-hygienic water or glucose water (Khanal et al. 2013a; GTN 2008; Sharma et al. 2017). The specific objectives of the GTN intervention were delivered in the cycle above:

- 1. to improve ANC, delivery care (DC) and postnatal care (PNC) practices in the community;
- 2. to strengthen the community capacity in identifying and solving the health problems related to maternal health and neonatal health;
- 3. to monitor the maternal health practice of each individual;
- 4. to support those women who have a problem by exploring appropriate solutions;
- 5. promotion of exclusive breastfeeding;
- promotion of adequate and timely complimentary feeding (at about 6 months of age);
- promotion of key hygiene behaviours (e.g. hand-washing with soap);
- to encourage family members to provide special care to expectant women;
- 9. to employ local staff;
- 10. to encourage local women and men to commit to group participation.

Figure 4 Objectives of the GTN intervention

(Source: van Teijlingen et al. 2012)

Overall through these objectives (Section 2.5.2, Figure 4), the GTN intervention aimed to increase the uptake of ANC, a skilled attendant at delivery, ID, and PNC in rural Nepal. These are recognised measures in targeting the reduction of maternal morbidity and mortality (Fujita et al. 2005). This also links to Rootman et al. (2001) as outlined in Section 2.2.

2.5.3 Planning: intervention site and control community

The selected intervention site was Pharping, a small rural town amenable to research due to its accessibility from Kathmandu. The intervention was rolled out in two VDCs with similar socio-economic status. In order to avoid "contamination" (or selection bias, see Section 3.2.3.2) between the intervention and control group, the sites were chosen approximately 40km from one another at opposite extremes from Kathmandu, in the northeast and southwest of the capital, with no direct transport line. The control community was selected on the basis of its location, population composition, facilities available, and its similarity to the intervention community (Section 4.3.5 and Table 2). Both communities include the villages and semi-urban communities of Kathmandu. The largest ethnic groups are Tamang and Brahmin-Chhetri (CBS 2001; Simkhada et al. 2009).

Health services characteristics were also similar between the intervention and control areas. In the two VDCs that formed part of the intervention, additionally, there was a private not-for-profit hospital (providing a mix of public/private services) with maternity services (i.e. Basic Emergency Obstetric Care Centre - Appendix VI) and two government health posts (providing primary care services). In the control area, there are two health posts and a primary care centre nearby (similar to the community hospital in the intervention area). The intervention area was chosen from a few pre-selected districts not far from Kathmandu that were (a) safe to work in at the time of Maoist rebellion (1996-2006), which was still on-going when the intervention was designed (2005/2006); (b) with the local maternal health needs identified by the community; (c) with local political commitment to making a change; and (d) staff recruitment (Sharma et al. 2016a).

As Nepal is geographically and culturally diverse (Section 1.4), it follows that any community intervention must be socially acceptable and culturally appropriate. As previously discussed, the majority (about 87%) of Nepal's population lives in rural areas (MOHP, New ERA & ICF International 2012). The topography of the study area is hilly (Figure 1) so service users walk up to three hours or more to the nearest health facility, which brings additional geographical barriers to accessing care.

2.5.4 The GTN intervention implementation & the evaluation

2.5.4.1 Implementation cycle

GTN launched the intervention on the basis of participatory learning and action research approaches (Hart 1996; Minkler 1997; Manandhar et al. 2004). The health promotion was designed by GTN and facilitated by the intervention's health promoter staff, one ANM and a CHW, with community input (Figure 5).



Figure 5 Rural Nepal, health promoters

© Sheetal Sharma 2013

The implementation of the intervention consisted of 24 group sessions of health promotion delivered in a repeated cycle (Figure 3). The health promoters carried out the health promotion activities with the intervention participants. The main

step of the intervention was to discuss issues regarding maternal care behaviours in the community each month (Sharma et al. 2017).



Figure 6 Visual cards

© Sheetal Sharma 2013

The content within the groups varied, as did strategies for delivery. This was due to the flexible nature of the GTN intervention. The enrolment began in 2006/7 until 2012. From 2007 to 2012, the project formed health promotion groups (Figure 8); organised different community-based training sessions focused on health promotion of maternal and neonatal health and in the supporting sub-health post outreach clinics for family planning, ANC and PNC check-ups. Mass health promotion events were also organised (for instance, during Tihar, the five-day festival of lights).



Figure 7 GTN Visual cards

© Sheetal Sharma 2013

The health promoters had training in participatory techniques and a health background (Figure 5). Their role was to activate, strengthen, and support groups through an action research cycle. Participatory activities used visual cards (Figure 6 and 7) that addressed prevention, treatment, and consultation for typical problems in mothers and babies. Role-playing activities were also conducted on the importance of contraception, ANC, iron/folic supplementation, danger signs of pregnancy, safe delivery, and postnatal care (see Section 1.3). The health promotion group strategies included:

- setting up a group;
- problem identification individually or in the groups;
- priority setting;
- introducing the aim of the group meeting;
- discussing why mothers and newborn infants die and how the intervention will work in the community;

- finding out about maternal and neonatal problems in the community and women's understanding of these issues;
- sharing health promotion information with regards to maternal and newborn health by:
 - role-playing as various family members and daily scenes they face;
 - encouraging participation by describing pictures of household chores, maternal care and danger signs recognition (bleeding, fever and feeling weak);
 - religious festivals drama enactment of maternal and child health activities;
 - identifying barriers to uptake: these were addressed adjusted and reapplied to address these barriers and meet the local needs of the population. For example, women who stopped attending the groups were identified, visited at home, and individually encouraged to re-join the activities;
- monitoring and evaluation (M&E) of GTN group attendance, household visits and costs (Sharma et al. 2017).



Figure 8 Women's group meeting in rural Nepal

© Sheetal Sharma 2013

As discussed in Section 2.1, health promotion is holistic and multi-layered. Hence, in addition to groups and mass events, GTN staff visited households to support women who were considered "most in need" and who were not able to leave the household (i.e., those who were physically weak, anaemic or did not have permission from their families to leave the house or attend the groups).

Moreover, the intervention supported the existing health services of sub-health posts by providing health communication training to MCHWs, FCHVs and Traditional Healers. Furthermore members of Mother-Child Health (MCH) hospital staff were given neonatal care training. The intervention also included typical strategies for maternal or infant care, including stretcher schemes to three health posts, mobile clinics health visits (including contraception and antenatal clinics) visits, distribution of clean delivery kits, and home visits by group members to newly pregnant mothers (Sharma et al. 2016a; Sharma et al. 2017). This activity entailed interactions outside the groups, which increased

awareness of the intervention, referred to in the literature and this thesis as a 'positive spillover effect' (Israel et al. 2001; Manandhar et al. 2004; de Heer 2011; Vanderweele et al. 2013).

As incentives can encourage women to attend intervention activities (Grant and Sugarman 2004; Cryder et al. 2010), and in accordance with the programme values/objectives (that the intervention is low-cost and participatory), the incentives were small gifts of less than 10 Nepali rupees (GBP £0.10). Similarly, maternal care gifts were given as incentives: a baby blanket on completion of four ANC visits, and safe delivery kits were made available at a subsidised price and sold through the women's groups. These incentives had the aim of encouraging women to attend groups and to incentivise their health behaviour towards seeking care. Finally, GTN monitored and evaluated their activities to report back to their funders (Sections 2.5.4.1 and 3.2.3.2).

2.5.4.2 Evaluation

The GTN survey data collection (prior to this PhD evaluation) consisted of a structured questionnaire in Nepali to collect "baseline, midline, and final" survey data on women's health status and knowledge of maternal services and socioeconomic status (Section 4.3.4 and Appendix III). First, GTN conducted their own M&E analysis of the data from the baseline, midline and final that looked at how many groups were formed (as were participants' ages, caste, the attendance data and number of children). At the time of the midline survey in 2010, there were 37 active groups reaching 1100 people. In addition, 134 household visits to support women most in need, for instance those that could not attend groups and/or needed to be at home and care for their families. At the time of the final survey of 2012, there were 40 active groups and more than 100 household visits reported. The total participants were 731 (Sharma et al. 2016a; Sharma et al. 2017).

The internal M&E activity acted as an exercise to help interim changes to the programme where the process was not working as well as planned. For example, the M&E showed that the intervention had not focused enough on the

knowledge of PNC and danger signs post-delivery (GTN 2008). Therefore, GTN refocused the intervention to improve the outcomes on the indicators of postnatal care.

The purpose of this thesis was to conduct a process, impact, and outcome evaluation using these quantitative data sets and the qualitative views of the participants even after the end of study period (Figure 9). The survey data of GTN were tested for association between the intervention and attendance, for instance how to identify "the gap": it is not known how a health promotion intervention based on community mobilisation/groups would work in Nepal. Therefore it was done all the while taking into account different factors are responsible for attendance (Section 1.3.1 and 1.4.4). The uptake of maternal health services was selected as a proxy for success of the intervention; based on the literature review on chosen indicators (Section 1.3.1 and 2.6.1) and health promotion evaluations of maternal health interventions discussed in Section 2.6.1.



Figure 9 Timeline of the programme and PhD

2.6 Maternal health promotion interventions in LMICs

Two decades of health promotion, since the Ottawa Charter, have been very much occupied with providing evidence for health promotion (McQueen and Jones 2007). The preoccupation with providing evidence in health promotion was due to the notion of evidence-based medicine in public health that began in the early 1990s (Section 2.2). As discussed in Section 1.1, many public institutions and NGOs rose to this challenge, including community-based maternal health promotion interventions. A literature review of maternity community-based health promotion interventions in LMICs is presented in this section.

A literature review guided this evaluation to find key elements of evaluation methods to assess the effect of community-based maternal health promotion interventions in low-income countries. The literature search was done on the PubMed database, for peer-reviewed literature complemented by a hand-search of library periodicals and a search of relevant reports by the WHO and other international agencies (1980-2015). The start date coincided with the introduction of the definition of health promotion as "*the science and art of helping people change their lifestyle to move toward a state of optimal health*" (Section 2.1). The keywords were: community participation, women's groups, developing countries, health promotion, and evaluation (Figure 10). Excluded were interventions that focused on individual women or non-health related outcomes, such as micro-credit or savings behaviour. The findings of the literature review are presented in Section 2.6.1.

Types of interventions:

Community health promotion for maternal and newborn care implemented and evaluated.

Settings:

Community-based (home, primary health facility, that providing primary level health care, dispensary, health post or Maternal and Child/Family planning MCH/FP) clinic which provides basic health services, health education and promotion, simple laboratory tests and treatment).

Types of participants:

All participants reside in developing countries and include: Women of child-bearing age/fertile; Pregnant women at any period of gestation; Mothers of <2 year-old children; Their spouses/partners; Other family members (i.e. mother-in-law).

Figure 10 Flow chart of literature search

2.6.1 Literature review effectiveness of maternity health promotion

Several studies, like GTN, have tried through participatory community trials to improve maternal health. These studies have looked at the participation in healthcare uptake as a primary intervention outcome. The studies ranged from descriptive studies of organising women's groups to sophisticated randomised controlled trials (RCTs) for improving mother and child health and maternal health proxy outcomes (uptake) in rural regions such as India, Nepal and Malawi (Hadi 2001; Boone et al. 2007).

Community-based local facilitators, such as auxiliary nurses, local women trained as counsellor/supervisors or community health workers convened the groups. The topics discussed ranged from promotion of family planning, prevention and management of malaria, breastfeeding, prevention and management of diarrhoea, and uptake and benefits of antenatal, natal, postnatal, and neonatal care. Also a number of activities were performed by women's groups: training community members in safe birthing techniques, generation of community funds for maternal and infant care, stretcher provision schemes, production and distribution of clean delivery kits, home visits by women's group members to newly pregnant mothers, awareness raising with the help of video films, social and psychological support, income generation, improvement of water supply and sanitary conditions, support for early initiation and maintenance of breastfeeding, literacy classes, and management of diarrhoeal diseases (Langer et al. 1998; Bhuiya and Chowdhury 2002; Dearden et al. 2002; Kouyate et al. 2008; Dennis et al. 2009).

In other community approaches to increase care-seeking and appropriate home prevention and care practices for mothers and newborns, interventions have used approaches such as making home visits to counsel mothers, providing newborn care, and facilitating referral (Bang et al. 2005; Baqui et al. 2008; Bhutta et al. 2011; Kumar et al. 2015). A third approach has involved women's groups in a four-phase participatory learning and action cycle (Manandhar et al. 2004; Tripathy et al. 2010; Azad et al. 2015). For instance, a systematic review of cluster-RCTs (Nepal, India, Bangladesh, and Malawi) assessed the effect of women's groups practising participatory learning and action (Prost et al. 2013).

These seven trials, similarly to the GTN programme, involved participatory women's groups as an intervention to reduce maternal and newborn deaths in poor communities. The role of the group facilitator was to activate, to strengthen, and to support groups through an action research cycle. The review findings were that women's groups practising participatory learning and action led to improvements in attendance and substantial reductions in neonatal and maternal mortalities in rural, low-resource settings. The key predictors of the intervention's impact were the proportion of pregnant women participating in groups and the population coverage of groups (Prost et al. 2013).

While in Malawi, groups of women, guided by a female Health Surveillance Assistant (i.e. CHW), met monthly to work through the action cycle to identify and prioritise key maternal and neonatal health problems to design and implement strategies to address their MNH problems and evaluate progress. A mixed-method evaluation was conducted with the quantitative study looking at percentage changes and the qualitative study using interviews and focus group discussions (Mseu et al. 2014). In Bangladesh, researchers have applied before-and-after quasi-experimental studies and RCTs of upskilling TBAs and CHWs, and their impact on maternal morbidity/mortality (Darmstadt et al. 2009). They found that CHWs might play a promising role in providing pregnancy and childbirth care, mobilising communities, and improving perinatal outcomes in low-income settings. Whilst, in Nepal, a health education RCT consisted of two 35-minutes maternal health education sessions, whereby the women were monitored until the postpartum period. Women who received education with their husbands were more likely to attend a postpartum visit, and were also nearly twice as likely as control group women to report making birth preparations. Moreover the study groups were similar with respect to attending the recommended number of antenatal care check-ups, delivering in a health institution or having a skilled provider at birth. The study reported an impact only on postpartum care (Mullany et al. 2007).

Additionally, Liu and colleagues (2010) aimed to evaluate the effects of the "Safe Motherhood" programme on maternal care utilisation. Pre-intervention and post-intervention cross-sectional surveys were conducted with questionnaires about the prenatal care utilisation in 2001 and 2005, respectively (Liu et al.

2010). The method of analysis difference-in-differences (DiD) was used to assess the effect of intervention on the maternal care utilisation while controlling for socio-economic characteristics (age, wealth, education and parity) of women. Additionally, a study by Ensor and colleagues (2014) on mothers' groups to improve both understanding of maternal health and access to maternal healthcare services, where they sought husbands' and community leaders' approval for care-seeking. They also used a DiD approach in two cross-sections and corrected for education (i.e., highest level achieved), household wealth, the woman's age and parity and the distance of the woman's home from the health centre (Ensor et al. 2014).

Whilst in Nepal, a cluster-RCT found that providing vitamin A supplements through these groups reduces mortality (West et al. 1999). Women's groups assessed in cluster-RCT and qualitative studies respectively have been found to reduce mortality as the groups increased awareness of maternal health collectively and build social capital for support networks, promote hygiene, and prevent the delays in seeking care (Manandhar et al. 2004; Rosato et al. 2006). For instance, a cluster-RCT community trial in Pakistan has shown that attendants can promote good perinatal hygiene and reduce mortality (Jokhio et al. 2005). Moreover, in India, a cluster-RCT found that women's groups have had a positive effect on birth outcomes (Tripathy et al. 2010). Often not evaluated is the tenuous link between health promotion and improved health outcomes (Section 1.3); although studies in Mexico and India have found that an increase in health promotion of exclusive breastfeeding reduced morbidity related to diarrhoea (Morrow et al. 1999; Bhandari et al. 2003).

In general, community interventions in maternal health, in LMICs, and those based on women's groups have had considerable success. These innovative community-based strategies, combined with health systems strengthening, may improve childbirth care for the rural poor, help reduce inequities in maternal and newborn survival and stillbirth rates, and provide an effective uptake of higher coverage for births attended by an SBA (Darmstadt et al. 2009). These studies discuss that community groups are of benefit as there is a scarcity of resources for delivery of health services to the population (Judd et al. 2001; Duflo 2004). It is difficult to deliver health interventions with low human resources, barriers to
access, illiteracy, and poor compliance. These groups in addition provide a culturally relevant solution, and perhaps a cost-effective and sustainable way to respond to needs (Akhund and Yousafzai 2011). Thus group formation has the potential to enable women to achieve the intervention goals. Groups also come together for non-health activities, for example, social capital, including saving activities (Minkler 1997; Goodman 1998). Women's groups respond best to participatory non-didactic approaches that encourage group members to share knowledge and work together (Akhund and Yousafzai 2011). These evaluations also found that groups are successful if collaboration with the community was attained at the conceptualisation phase of the interventions, by acknowledging the community's contributions, creating a sense of ownership of the group, empowering group members with clear communication of activities or skills needed in the task (for example, describing visual cards), and addressing issues of trust, respect, conflict and power dynamics (Parker et al. 1998; Koné et al. 2000; More et al. 2008). In a few studies, success was also found if group homogeneity was found to be beneficial (Chowdhury et al. 1988; Green 1998). On the hand, others found that group ethnic and social diversity was key (Asthana 1996). Also, the support of husbands was a necessary factor for the group's success (Manandhar et al. 2004). Finally, it was important that the health promoters/supervisors had support from the community (O'Rourke et al. 1998).

As a result of participation in the group, non-members also gained from a positive spillover effect (Section 2.2) due to the activities. Examples of group and mobilisation benefits include knowledge of health, personal skills development, empowerment, emotional and financial support, and lessening of stigma (Minkler 1997; Goodman 1998).

A few studies found that the men and mothers-in-law were the main decisionmakers in maternal health (White 2009; Simkhada et al. 2009). Yet few studies have evaluated the impact of their involvement on maternal health outcomes in LMICs. The studies found that male involvement was associated with improved utilisation of maternal health services (SBA at birth and PNC) and that male involvement was associated with improved maternal health outcomes (Yargawa and Leonardi-Bee 2015) and with positive effects on decision-making in women's reproductive health and family planning in Nepal (Mullany et al. 2005). Finally, Sternberg and Hubley, (2004) evaluated an intervention that targeted heterosexual men. There was evidence that the use of media approaches was a successful strategy, yet there were issues in applying behaviour change approaches. As few interventions have targeted heterosexual men, the latter finding suggests that there is a need for more interventions and/with better evaluations, which would examine not only the process of men's involvement, but also their impact on the lives of both the men themselves and their families. Therefore, one goal of the qualitative part of this PhD evaluation is to determine, to an extent, how the men and mothers-in-law's roles determine women's access to maternal health services.

2.7 Summary of literature review and next steps

These studies have found that there is growing evidence that better utilisation of maternal healthcare services depends on mobilising the entire community and encouraging participation in the group activities. For example, in a programme for improving birth preparedness in Nepal that focused only on women, knowledge of obstetric danger signs increased but there was little change in the proportion of deliveries involving a SBA. It was suggested that the lack of progress occurred because education was provided only for women and not for the whole community (including the men) and because other barriers to healthcare, such as the cost of getting to a facility, persisted. A gap still exists; therefore the next chapter discusses evaluations and the evaluation of health promotion, evaluating non-health outcomes, and the need to evaluate to obtain evidence-based health promotion. The evaluation, in this thesis, was done in order to determine the intervention's effectiveness or impact: the latter is considered an improvement in the outcome as a result of women's group intervention and continuing activity of the group even after the end of study period. To this end, the GTN data were tested for the association between the intervention and attendance (Sections 3.2.3.2 and 4.3.9.1.4), how to identify the gap, and what worked in this setting in rural Nepal using qualitative research (Section 4.3.8).

Chapter 3 Evaluation: its nature in health promotion

3.1 Evaluation

This chapter has been divided into four sections. The first section discusses the underlying philosophies of evaluation in general and its approaches. The second discusses its practice in health promotion. The third elaborates on the process, impact, and outcome evaluation of this thesis. Finally, the fourth section presents the aims and objectives of the impact evaluation of GTN.

It is worth remembering why we evaluate namely to see if something "works". Evaluation is a critical part of the development process for health planning to assess:

- the need for the programme;
- the programme design;
- the way the programme is being implemented (i.e., is the process going according to plan? Are programme's processes maximising possible outcomes?);
- outcome or impact (i.e., what it has actually achieved or using indicators);
- cost and efficiency (cost-effectiveness) (Craig et al. 2006; Khandker et al. 2010).

In essence, indicators are variables used in the evaluation process that help to measure changes in the outcomes or observable characteristics at issue. In addition, an evaluation has to be supported by valid, reliable, and sensitive information/data (Hawe et al. 1990). The evaluation process is intended to be used in a flexible manner and must be adapted to the circumstances in which it is to be used. Perkins and colleagues (1999) discuss the idea that it is the phases of change of a project that will help to design the evaluation, the type of evaluation that is appropriate, and the sorts of questions that are appropriate to ask.

As stated in Chapter 1, there is a critical need to measure the outcomes of health programme efforts and then apply that knowledge for best practice in future interventions. Research ideally aims to be valid to the population under study, i.e. accountability. If applicable to a wide variety of settings, then the research in question is generalisable. Evaluation research is thus concerned with accountability, generalisability, i.e., for the potential of evaluation as a basis for scaling-up (or transferability or external validity) and effectiveness. The main threats to validity are chance, bias, and confounders (Section 4.3.9.1.3) (Altman & Bland 1998; Clancy 2002; Waters et al. 2006).

It should be noted that a trade-off occurs in any study. For instance, in ensuring generalisability, internal validity may be compromised. An example pertinent to this thesis is that study participants in rural Nepal may demonstrate preference or have existing relationships that influence research outcomes. Therefore, to some extent, the definition of "effectiveness trial" describes the necessary trade-offs between generalisation and internal validity (Altman & Bland 1998).

There exists, however, considerable debate over how to evaluate programmes, i.e., how to measure the evidence for the effectiveness of interventions. In health promotion interventions, evaluation involves making judgments about the achievement of the health promotion activity in question by comparing it with some criteria that are considered to be an indication of good performance (for example, indicators used by comparable studies). These criteria are usually derived from the aims and objectives of the project. Therefore, the planning of both the aims and objectives and the actual evaluation is essential (Perkins et al. 1999). Moreover, an activity or programme needs to have a clear rationale with both long and short-term objectives, or it becomes extremely difficult to evaluate (Perkins et al. 1999).

As discussed in Sections 2.1 and 2.2, health promotion requires a mobilisation of appropriate strategies that bring about measurable organisational and community change (Flynn et al. 1994; Tones 2000; Cresswell et al. 2003; Whitehead 2003). Thus, an evaluation of health promotion programmes should take into consideration the needs of the target group and the best current knowledge (evidence) as to how to meet these needs. Evidence in health promotion is discussed in Section 3.2.3. Additionally, an intervention ought to deliver the most appropriate or beneficial programme within the resources

76

available. Therefore, an appropriate choice of quantitative, qualitative, or mixed methods is needed for evaluation (Hawe et al. 1990).

Finally, in evaluation and as also seen in Sections 1.2 and 2.5, the last decade has seen a growing number of collaborations between NGOs and academics to help evaluate the impact of the former's activities in behaviour-change, community-based education and maternal health interventions (Kremer 2003; Banerjee et al. 2007; Baqui et al. 2008). The evaluation of the GTN intervention was discussed in Section 2.5.4.2. The next section introduces broad concepts of evaluation.

3.2 Evaluation, philosophical underpinnings

This section discusses the underlying notions of evaluation and evaluation approaches. There are several types of evaluations that can be conducted, including formative, process, impact, and outcome evaluations (CDC 2013). These types of evaluations will be elaborated on in Section 3.2.3; in particular, impact evaluation (Section 3.2.3.2), as this thesis evaluation was concerned with the latter.

Evaluation is research with a specific purpose (Section 1.2), namely to provide accountability and assess the worth or value of a project, a programme or an intervention (Suchman 1968). In 1978, the WHO drafted guidelines for health intervention evaluations for use with interventions and activities. It summed up the purpose of evaluation as:

"To improve health interventions and the health infrastructure for delivering them to guide the allocation of resources in current and future interventions." (WHO 1998, p5)

The guidelines add that evaluation is a systematic process of learning from experience. The lessons learnt ought to be applied to improve current activities and to promote better planning for future action, for example, for future programmes organised by the NGO in this case, GTN. Also used are criteria, or standards, which may be pertinent questions asked of the activity under evaluation, as seen in Section 3.2.3 (WHO 1998). Here, the indicators used in this study were defined by WHO guidelines, DHS, and the literature review of evaluations in maternal health promotion/groups in LMICs (Sections 1.4, 2.6.1 and 4.3.9.1.4).

As previously mentioned, evaluation aims to provide evidence of the effectiveness of the programme in question. Interventions tend to be complex and context dependent (White 2009; Smith & Petticrew 2010), therefore, the evidence for their effectiveness must be sufficiently comprehensive to encompass that complexity (Rychetnik 2002). Smith & Petticrew (2010) posed three challenges for the evaluation of social interventions: (1) that a broader range of outcomes were needed to evaluate the broader range of interventions; (2) that a broader range of evaluative methods were needed to deal with complexity and multi-sector evaluation; and (3) that a broader range of evaluative methods were needed to meet the needs of multiple stakeholders (Smith & Petticrew 2010). They argue that many public health and health promotion interventions are complex due to the multiple components, outcomes, and externalities that exist, such as unintended consequences. The use of a conceptual framework as described in Section 4.3.2 can help identify these challenges (Craig 2013).

It is important to note that evaluations can be negatively perceived as a means of cutting back on the scope of a programme or as only keeping the "good" aspects of a "bad" programme. Therefore, in order to make the best judgment, data are collected not only on the intervention, such as health outcome or opinion data, but also on the changes the programme has made, or "whom" the programme has reached, and its long-term effects. This is termed 'evaluative research' (Suchman 1968). Evaluative research is more than just making observations and measurements and then assessing what one observes based on some criterion or standard of what is considered to be an indication of good performance (Deniston 1980). It aims to define 'what works best for whom and under what circumstances' (see Section 3.2.3.2). Therefore, evaluative research has several components, such as specifying the subject for evaluation, ensuring

data availability, verifying the relevance, assessing measures of adequacy, progress, efficiency, effectiveness, impact and drawing conclusions for future action (WHO 1998; Hawe 1991). In other terms, evaluative research has been defined as the rigorous and systematic collection of data to assess the effectiveness of a programme in achieving predetermined objectives (Bowling 1997). Thus effectiveness is influenced by the participants (through appropriate targeting), to the exposure of the programme or intervention, resources available, quality of delivery (including training and enthusiasm), and intervention contamination (Waters et al. 2006).

3.2.1 Evidence of effectiveness

As public health interventions are often complex, the methods to provide evidence of effectiveness must be comprehensive (see Sections 1.2, 2.2 and 3.2) (Rychetnik 2002). The Cochrane Collaboration states that the definition of evidence-based care/practice standards requires effectiveness studies with a case control design (Sections 1.2 and 2.2). Therefore, the methods eligible for inclusion in Cochrane systematic reviews are generally non-RCT, interrupted time series designs, RCT, and guasi-RCT. Uncontrolled studies are generally not included in reviews as it is difficult to distinguish the effects of the intervention from the Hawthorne effect, an effect produced when participants know they are being observed, or from what would have occurred naturally over time (Bowling 2014). However, in many areas of public health, RCTs may be impossible, as RCTs tend to be suited to more "simple" and straightforward interventions or efficacy trials (Waters et al. 2006). Non-randomised controlled studies, for example, controlled before-and-after studies, are study designs where participants or populations are not randomly assigned to an intervention or control group. As in the GTN intervention, the outcomes of interest are measured both at baseline and after the intervention periods in both intervention and control groups (Deeks et al. 2003). However, as it was not possible to randomise a community at a familial level, the wider impact of the intervention on the community was analysed. The lack of randomisation in these types of studies may result in baseline differences between treatment and control groups, as randomisation is the only way to control for confounders (Section

4.3.9.1.3) that are not known or not measured (Clarke & Oxman 2003; Waters et al. 2006). First, interrupted time series (ITS) designs use "multiple observations over time that are 'interrupted' usually by an intervention or treatment" (EPOC 2005). These designs may or may not include a control group (Clancy 2002; Waters et al. 2006). Secondly, *RCT and quasi-RCT* refer to trials where participants or populations are randomly allocated (for example, via computer generated randomisation or a random number table) to an intervention or control/comparison group and are followed up over time to measure differences in outcome rates (Waters et al. 2006). A quasi-randomised trial uses a slightly "diluted" method of randomisation or allocation for methodological, for example, allocation by date of birth, alternate allocation, or pragmatic and policy reasons, for example, allocation by housing sector (Waters et al. 2006).

In evaluation, the ideal study design is an RCT or a quasi-experimental design, as they control for confounding variables (Section 4.3.9.1.3), achieve randomisation, and provide certainty in the causal relationship showing an unequivocal evidence of effectiveness (Macdonald et al. 1996). However, controlled studies of discrete non-complex interventions (RCTs) may exclude important parts of the "real world setting" and of the evidence base relating to complex health promotion interventions. This is due to the multi-faceted and context-dependent nature and delivery in multiple settings in the real world (Craig 2013).

Effectiveness trials measure the degree of beneficial effect under "real world" (pragmatic) settings (Gartlehner et al. 2006). In evaluation, examining evidence from community health promotion interventions can lead to a better understanding of the interaction of factors that are inherent between study design, evaluation methods, programmatic strategies, and context. These factors all influence the outcomes and effectiveness of community-based health promotion efforts (Merzel and D'Afflitti 2003). Once evidence is obtained on an intervention's effectiveness, practitioners or programme implementers need to know what works, how it works and under which conditions it works in order to be able to continue programme activities, or implement it in other settings. However, if evaluations only examine the process, the risk is that the results are rarely disseminated beyond the local area and valuable experience is 'lost'

80

(Campbell et al. 2000). Thus, if evidence-based health promotion is to progress, there needs to be practitioners' and service users' accounts of the implementation (process evaluation) as well as evaluation of the outcomes (outcome evaluation). These are expanded upon in Section 3.2.3.

An assessment of how the effects of a programme were achieved is valuable in determining impact of, or effective interventions (Section 3.2.3). An effective evaluation can aid in the (continued) development of the programme by providing healthcare planners and professionals with satisfaction that the objectives were met. Moreover, monitoring attendance and reasons why people do not attend the programme is also important (Perkins et al. 1999; Creswell 2003; Bowling 2014). In summary, to aid public health decision-making, evaluations must be conducted for reasons of evidence, effectiveness, economic, ethical, and accountability (Gartlehner et al. 2006; Waters et al. 2006).

3.2.2 Realist evaluation

In complex community-based evaluations, one should be looking for a realist approach, as evaluations should take account of the context and not only measures of success/performance, in which the intervention occurs. Of the different evaluation philosophies, the realist philosophy is most appropriate to provide context when assessing whether a hypothesis will work or not as several stakeholders (participants) make particular decisions in response to changes (i.e. the intervention). The stakeholders' reasoning is a response to the opportunities (resources) offered by the intervention and is what causes or leads to the (positive or negative) outcomes (Vogel 2012). Pawson and Tilley's seminal work on realist evaluation argued that in order to be useful for decision makers, evaluations need to identify "what works in which circumstances and for whom?" rather than "does it work?". Thus, the complete realist question is: "What works, for whom, in what respects, to what extent, in what contexts, and how?" (Pawson & Tilley 2004: 2). In order to answer the former question, realist evaluators aim to identify the underlying/generative (such as socio-economic status, ethnicity, and rural/urban residence) mechanisms that explain "how" the

outcomes were caused and the influence of context using process evaluation methods. Realist approaches to evaluation assume that nothing works everywhere or for everyone: context really does make a difference to programme outcomes (Evans and Killoran 2000).

In practical terms, impact evaluation is the most appropriate realistic evaluation approach for evaluating new initiatives or programmes that seem to work but for which "how and for whom" is not yet understood. This applies to programmes that have previously demonstrated mixed patterns of outcomes or those that will be scaled-up in order to understand how to adapt the intervention to new contexts. An impact evaluation is an evaluation that examines direct and indirect contributions (often unintended consequences) of an intervention to changes in people's lives, especially mid-term to longer-term changes (Westhrop 2014), as discussed in Sections 3.2.3.2 and 3.2.3.3.

Realistic evaluation assesses social programmes on the basic hypothesis of social betterment. However, the social systems in which these programmes take place are complex: there exist inter-relationships between the programme, stakeholders, behaviours, events on the ground, and social conditions (Pawson & Tilley 2004; Judd et al. 2001; Duflo 2004). Evaluations therefore test the underlying objectives of the programme or intervention to determine whether and how the programme worked (what impact it had) in a particular context. As interventions never work indefinitely in the same way, in all circumstances, or for all people (Pawson & Tilley 2004), thus describing context is useful in any evaluation (Waters et al. 2006): context refers to the social, organisational, and political setting in which the intervention is implemented. Examples of contextual factors that may affect intervention effectiveness include literacy, income, cultural values, power relations, access to media, and health services (Irwig et al. 1998). For example, there may be different social beliefs about the roles and responsibilities or decision-making of women and men in different cultures, which may affect how women respond to a maternal health promotion programme. Whether communities are responsive to the intervention will also depend on a range of factors from the time they have available to attend, their own beliefs about pregnancy, or their beliefs about their reproductive health. Therefore, the context may provide alternative explanations of the observed

outcomes, and these need to be taken into account during the analysis (Pawson & Tilley 2004). The context of this evaluation with respect to maternal health, Nepal, and GTN was detailed in Sections 1.4, 1.4.3 and 2.5, respectively.

3.2.3 Evaluation in health promotion

At a global level, the need for evaluative research in health promotion and community development initiatives for health has been advocated (Hawe et al. 1990; Beattie 1995a; WHO 1981). For health promotion initiatives, evaluation contributes to theory building by providing clear implications for the practice of health promotion. Evaluations are needed as they provide evidence on health promotion effectiveness to:

- identify the best possible ways to promote health;
- make decisions for policy development and funding allocation;
- demonstrate to decision makers that health promotion works and is an effective strategy in public health;
- support practitioners in project development and evaluation;
- show the wider community the benefits of health promotion actions;
- advocate for health promotion development (WHO 2015).

Evidence provides a clear rationale for evaluation. It sets boundaries to evaluation studies, focuses future implementations, and acknowledges the importance of context and settings of the study and the use of evaluation methodologies. The purpose of evaluation is therefore linked to the broad values and goals of health promotion, and at the level of practice, with personal beliefs, values, and perceptions of stakeholders (Perkins et al. 1999). Evidence of health promotion effectiveness came to the forefront of global priorities as seen in Section 2.2. Evidence and effectiveness in health promotion are an integral part of the Ottawa Charter. Health promotion is a different endeavour to clinical practice as it seeks to permit multiple perspectives rather than focusing on a single goal or desirable outcome (Webb & Harinarayan 1999). It is concerned with enabling and empowering individuals and communities to increase control over, and thereby improve their health. This means that a narrow focus on health outcomes borrowed from evidence-based medicine does not adequately encompass all, or even most, of the questions that evaluation in health promotion should eventually address. Many have argued that health promotion is equally concerned with evaluations that focus on issues such as equity, empowerment, access to health services, community involvement, and health public policy (Macdonald et al. 1996; Whitehead 1991; Ziglio 1997). Yet at a practitioner level, an evaluation of the long-term health outcomes of a health promotion initiative is rarely feasible or even appropriate as seen with the MMR indicator (Section 1.3). Thus standard scientific approaches to evaluation have limited value for health promotion (Milburn et al. 1995; Nutbeam 1998; McQueen 2001). A health promotion evaluation should take into account the cost (Section 3.4), time, and resource constraints (Section 3.5). These impose considerable limitations on the type of work that can be attempted. Hence, as a result, a pragmatic approach to evaluation, as in this thesis, is often conducted. This approach takes into account which questions can be answered within the available resources (data, travel, skills, funding, time, etc.). The latter is central to achieving a useful evaluation and to ensuring that the evaluation is not a "failure". Furthermore, if there are not enough time, skills, and funding, or the evaluation has been done "elsewhere", i.e., the successes and failures are well documented and the reasons clearly understood, or the results are not considered for future implementation of planning and support is not available from the programme (managers, participants, etc.) (Wright 1999), here perhaps, a monitoring and evaluation (M&E) exercise will be sufficient (see Section 3.2.3.2).

Health promotion evaluation in theory is commonly divided into three subsections: (a) process; (b) impact; and (c) outcome evaluation (Hawe et al. 1990). This evaluation is similar but not a straightforward application of Donabedian's clinical model (Donabedian 1988). The Donabedian model proposes a framework for evaluating quality of healthcare using three categories: structure,

84

process, and outcomes (e.g. haemoglobin, risk factor for blood pressure, lowdensity lipoprotein cholesterol and end stage renal disease, Figure 11). The latter model is, however, too clinical a model and does not separate impact and outcome models – which health promotion attempts to. In a clinical model, health system organisation is widely known and understood. While in a health promotion model, the organisation is not as rigid and easily measurable. The evaluator community here works with the wider community, which is the system referred to in Donabedian's model. To guide a mixed-method approach, one uses the qualitative research (process) to explain the quantitative assessment (outcomes) to determine impact/effectiveness of a programme. Finally, in this evaluation, a variant of realistic evaluation was applied given the philosophy of the GTN intervention, which was context specific as outlined in Figure 4, in Section 2.5.2.





cholesterol (LDL-c), end stage renal disease (ESRD)

Figure 11 Donabedian model

(Source: Selby 2010)

It should be noted that a formative evaluation can be conducted with preprogramme implementation, which was conducted by GTN (Section 2.5) in the first instance. It aimed to ensure that the (new or adapted) programme was feasible, appropriate, and acceptable before it was fully implemented. This thesis was concerned with a process, impact, and outcome evaluation, as an intervention ought to be evaluated to ensure that it is having the desired effect (Sections 3.2 and 3.3). Therefore, the evaluation assessed the results of the intervention and determined whether the objectives of the health promotion programme were met. As commented upon in Section 1.2, in LMICs, too many programmes are not evaluated or are inappropriately evaluated.

Health promotion has its own methodologies for evaluation, which take into account the aims, values, and processes that distinguish health promotion from other healthcare disciplines as discussed in the second part of this chapter. Speller and colleagues (1998) discussed this complexity of evaluating health promotion: as health promotion research does not have the benefit of clinical studies on safety and feasibility (with several phases of drug development, for example, phase I and II in drug development), researchers must therefore evaluate the intervention itself (Speller et al. 1998). In a field such as health promotion, this proves complex as most health promotion interventions involve individual behaviour change with attempts to intervene at the community level (Britton et al. 1998). Britton and colleagues (1998) suggest integrating evaluation methods will improve understanding of interrelations between behaviour and social structures, and the inclusion of a qualitative process leads to a more robust evidence base for health promotion.

In the evaluation of health promotion, the "classical" methodological and research designs are established on psychosocial and epidemiological research (Waters et al. 2006). However, most readily measurable constructs in health promotion are found in the personal skills (Section 2.2), while social and biomedical researchers have for many years been creating measures and methods that lend to an empirical approach (Waters et al. 2006). Britton and colleagues (1998) caution researchers to avoid viewing non-randomised studies as 'inferior', as Speller and colleagues (1998) question whether RCTs are always the best or most appropriate method of evaluating health promotion. The authors also mention that the goals of evaluation are: (1) attribution of the effects of an intervention and (2) the relative costs involved. Thus, lack of evidence from RCTs should not be viewed as a failure in the quality of research; rather, more attention should be given to refining and

86

strengthening other trial methodologies (such as community trials or before-andafter trials). Additionally, the focus on only effective outcomes too often ignores the process of an intervention. While insisting upon RCTs in low-income countries ignores some of the unique features of health promotion interventions, which often take place at a community level. The expected proportional benefits to individuals can be small, and beneficial outcomes are delayed (Britton et al. 1998). Thus, this thesis combines a quantitative outcome of mid-goals evaluation (impact) with qualitative process, and evaluation to understand the interrelation between people's behaviour and the social structure in which they live.

In implementing a health promotion programme, caution is recommended. As, in health promotion, potential contamination and confounding factors (Section 4.3.9.1.3) may mean that attribution can rarely be a certainty, and even when it can be, the possibility of replication of the intervention may be limited due to the differing contexts in which an intervention is implemented (Britton et al. 1998; Waters et al. 2006). The issues and difficulties that need to be taken into account in health promotion evaluation include:

- 1. focus on populations and communities rather than (only) individuals;
- 2. difficulties characterising and simplifying complex multi-component interventions rather than single interventions;
- 3. analysis of process as well as impact or outcome measures;
- 4. effect of involvement of community members or potential participants in programme design and evaluation;
- 5. effect of using health promotion theories and beliefs;
- analysis of use of different types of both qualitative and quantitative research;
- 7. need to account for the complexity and long-term nature of public health intervention outcomes (Jackson et al. 2001);
- integrity of the intervention highlighting what factors may have influenced the (in-) effectiveness of the intervention, such as participation (including appropriateness), exposure of programme or intervention, resources, quality of delivery (e.g. training, enthusiasm), and limiting intervention contamination (bias), (Dane & Schneider 1998; Waters et al. 2006).

For these reasons, analysing all the available evidence can be a complex task, requiring researchers to have (or have access to) sound data and methodological knowledge. Yet a well-conducted process, impact and outcome can offer valuable insights for the success of a particular project as they take into account the various types of data and methods (Creswell 2003; Duflo 2004).

3.2.3.1 Process evaluation

A process evaluation, one of the sub-sections of a health promotion evaluation, determines whether programme activities have been implemented as intended. This process evaluation is the measurement of the running of the activity or intervention. It assesses how well the latter was implemented and helps to judge whether the "vehicle is suitable for the journey". Process evaluation is used to document a programme implementation and assess how well this has been done (Saunders et al. 2005). Process evaluation can increase our understanding of the relationship between specific intervention elements and the overall impact or outcome. It can help unravel the factors that are responsible for successful outcomes, implementation of the intervention, and intervention reliability (truth). The process data assists with characterising "failure to achieve success" (Steckler & Linnan 2002; Waters et al. 2006). These process evaluations are likely to be more feasible for health promotion practitioners than formal outcome studies due to the recognition of the process of change. Where local evaluation studies have explicitly aimed to identify features of settings that might affect the process or outcomes of a health promotion programme, they have produced evidence that has had important implications for practice. For example, all programmes should be subject to process evaluations to ensure that funds (cost) are spent as intended and to receive feedback from stakeholders on how programmes could be improved (Duflo 2004). On the other hand, outcome evaluations, which provide evidence on the results (effects) of the intervention, are usually specific to a particular context and setting, and may provide little guidance about transferability. In other words, rolling out the programme in certain settings will provide little indication of potential progress (Wright 1996; Perkins et al. 1999).

Process evaluations, such as qualitative studies, are often published separately from outcome evaluations. Yet they can be published (in systematic reviews for example) alongside quantitative studies to assess the adequacy of the delivery of the intervention and the context in which the intervention was evaluated (Moore et al. 2015; Waters et al. 2006). Moreover, process data have conventionally been drawn from observational quantitative research but increasingly use qualitative and quantitative research methodologies where appropriate (Waters et al. 2006). However, while process evaluations are necessary, they are insufficient to determine programme impact. Therefore, a process evaluation is useful alongside an impact or outcome evaluation.

3.2.3.2 Impact evaluation

Impact is the overall effect of a programme on health and related socioeconomic development (Rootman et al. 2001). Impact is the treatment effect. Therefore impact evaluation requires an appropriate comparison group for counterfactual analysis, using either prospective (ex-ante) or retrospective (expost) evaluation design (Gertler et al. 2011). Prospective evaluations begin during the design phase of the intervention, involving collection of baseline, midline and endline data from intervention beneficiaries (the "treatment group") and non-beneficiaries (the "comparison group") (Duflo 2004; White 2009; Khandker et al. 2010). A well-designed impact evaluation covers both process and outcome evaluation questions. As the study can address questions of why, or why not, an intervention had the intended impact, not just whether it did. Evidence-based policy making is thus enhanced by an impact evaluation (White 2009).

An impact evaluation assesses the medium or longer-term changes of selected outcomes that can be attributed to a particular intervention, such as a project, programme, or policy, both the intended ones as well as ideally the unintended ones (Fitz-Gibbon & Morris 1987; Christiansen 1999; IMS 2001; Rootman et al. 2001; Khandker et al. 2010; Bhutta et al. 2011; Baqui et al. 2015). Both impact and outcome evaluations test the causal chain of events that has been

postulated by the programme (White 2006; Gertler et al. 2011). The assumption is that changing knowledge (e.g. about the benefits of iron intake in pregnancy) will lead to a change in certain behaviours (e.g. increased uptake of iron) and subsequently a reduction in maternal morbidities (e.g. anaemia). The difference between impact and outcome evaluation does not depend on what is measured. but is defined by the sequence of measurement of outcomes (medium outcomes measuring maternal services attendance versus long-term avoiding morbidities). It depends entirely on the causal chain of events that has been proposed and where the intervention stands in relation to these events. Thus evaluations can range from relatively short intervals, for the assessment of progress and efficiency, to much longer intervals for the assessment of effectiveness and impact (Hawe et al. 1990). The notion of time is further explored in Section 3.5. In addition, impact and outcome evaluation both involve the assessment of programme effects, but at different levels. In this way, some of the outcome evaluations may incorporate notions of effectiveness as well as costeffectiveness in order to assess the efficiency of a programme (Duflo 2004).

Traditional approaches to evaluation also include M&E. For example, a community-based design that includes results about the needs of a community can subsequently inform the design of further phases of the project (Creswell & Plano-Clark 2007; Plano-Clark & Creswell 2010). M&E are a set of management tools used to audit project implementation on the ground, to assess whether the project is developing according to plan and to make amendments where needed. M&E are often either a simultaneous or short-term assessment of activities and procedures, the use of resources and associated costs. It lends to that an effective impact health promotion evaluation should therefore be able to assess precisely the mechanisms by which beneficiaries are responding to the intervention. The question of causality, or changes in outcome that are directly attributable to a programme (Gertler et al. 2011), makes impact evaluation different from M&E and other evaluation approaches. It should be noted that the main question of impact evaluation is one of attribution: isolating the effect of the programme from other intervening factors (confounders) and potential selection bias of participants (Fitz-Gibbon & Morris 1987; Christiansen 1999; Khandker et al. 2010; J-PAL 2015). Impact evaluation answers the question of causality between the intervention and the outcomes observed, wherein lies the

difference from M&E (see Section 3.2.3.2), (Fitz-Gibbon & Morris 1987; Christiansen 1999; Khandker et al. 2010; WHO 1981). Therefore, impact evaluation is concerned with the assessment of the mid-term effects of the programme and usually corresponds with the measurement of the programme objectives (outcomes), i.e., has there been necessary change as a consequence of introducing an intervention in increasing overall health or socioeconomic development. Hence, it may not be appropriate to draw conclusions on the effects of a project until a process evaluation suggests that they have been successfully adopted, implemented, and are running smoothly (WHO 1998; Perkins et al. 1999).

In contrast to an outcome evaluation, which examines whether targets have been achieved, impact evaluation is structured to answer the question: how would individuals (who did benefit from the programme) have fared in the absence of the programme? Or, how would those who did not benefit have fared if they had been exposed to the programme? This involves a counterfactual outcomes analysis (that is, outcomes for participants had they not been exposed to the programme). Impact evaluations can be rigorous in identifying programme effects by applying different models to survey data to construct comparison groups for participants (Duflo 2004; Khandker et al. 2010). Impact evaluation designs are identified by the type of methods used to generate the counterfactual and can be broadly classified into three categories: experimental (RCTs); quasi-experimental (difference-in-differences analysis or DiD); and nonexperimental designs (before-and-after or interrupted time series with no comparison group). This study conducted an evaluation based on a quasiexperimental design (see further in Chapter 4).

Evaluation designs vary in cost, feasibility, and involvement during the design or after implementation phase of the intervention, as well as degree of selection bias (White 2006). Impact evaluation using an appropriate counterfactual is a key component. It is defined with reference to a control group. The latter is identified to avoid selection bias with the use of either experimental or quasiexperimental approaches. The design is further strengthened with the use of a control/baseline (White 2009). In addition to selection bias, important issues to consider in the design are spillover effects (the control group is affected by the intervention) or contamination (the control group is affected by other interventions). A before-versus-after analysis yields a valid counterfactual in this case, and a treatment versus control analysis of hospital improvements and resource/materials acquisition is likely to be a stronger design (White 2009; Gertler et al. 2011).

It is difficult, however, to answer these counterfactual questions, as at any given point in time, an individual can only be observed either exposed to the programme or not. Comparing the same individual over time will not, in most cases, provide a reliable estimate of the impact the programme had on the individual, i.e., as in an interrupted time-series or before-and-after, as many other things may have changed at the same time that the programme was introduced. Evaluations cannot obtain an estimate of the impact of the programme on each individual. They obtain an average impact of the programme on a group of individuals by comparing them with a similar group that was not exposed to the programme. As referred to in Section 3.2.1, the critical objective of impact evaluation is to establish a credible comparison group. This is a group of individuals who, in the absence of the programme, would have had outcomes similar to those who were exposed to the programme. This is done with either an RCT or DiD (Duflo 2004; White 2006).

Lastly, evaluation does not impose the use of particular methods. The choice of data collection, analysis methods, and tools are guided by the types of data that are needed to answer the evaluation questions, or more specifically, to test the aim in all its dimensions. Usually, both quantitative and qualitative data are collected in an impact evaluation, often with quantitative data being focused on context and outcomes, and qualitative data on generative (process) mechanisms (Pawson & Tilley 2004). A review of quantitative methods and models of impact evaluation can yield measured changes in wellbeing that are attributable to a particular project or policy intervention (see Section 2.3). In this thesis, a mixed-methods impact evaluation (Section 4.2) was used to assess the programme effectiveness in achieving its ultimate goals, which are the aims and objectives of GTN (CDC 2013). Section 3.3 discusses the mixed-methods impact evaluation approach.

92

3.2.3.3 Outcome evaluation

An outcome evaluation measures the intervention's effects on the target population by assessing the progress in the outcomes or outcome objectives that the health promotion programme has achieved. Outcome evaluation is concerned with the subsequent or longer-term effects of the programme, which usually corresponds to its aim. For example, a community-based healthy-eating project may use impact evaluation to assess changes in dietary habits (for example, for a reduction in anaemia) and outcome evaluation to assess the incidence of a related condition (here, improvements in diet over the long-term) in that community.

An outcome evaluation on its own provides no information about how the results were achieved, thus limiting implementation, repetition of success or rejection of programmes that have not been effective (Macdonald et al. 1996). Therefore, as discussed in the previous section, conducting a process and impact evaluation alongside an outcome evaluation is common.

In this thesis, a quantitative study was applied to measure the outcomes of the intervention (Section 4.3.9), i.e., the uptake of maternal health services (see Sections 1.4 and 2.5).

3.3 Evaluation methodologies & mixed-methods

According to Patton (1996), the key issue for health promotion practice is the usefulness of data, not the method by which it is obtained. This means that evaluation methodologies should be chosen to support the intended use of the evaluation by its intended users (Patton 1996). If practitioners adopt a participative approach when planning evaluations, this process helps define the questions (aims and objectives) that the evaluation should answer. These together with the time and resources available guide the choice of methodology. Denzin and Lincoln (1994) refer to a *"quiet methodological revolution"* that has taken place in evaluation studies (page vii). There is a considerable body of evaluation practice that has moved beyond scientific, quantitative methods to also embrace more interactive, qualitative approaches (Perkins et al. 1999).

This is relevant, as impact evaluation aims to understand not just what works, but why (White 2009). Qualitative methods can answer the "why" and quantitative "how much". Thus, both types of methodologies are useful and can be combined or "mixed" (Brannen 1992; Milburn et al. 1995). This mixedmethods combination provides an illumination of the process, the outcomes, and thus the impact achieved (Perkins et al. 1999).

Finally, the evaluation designs that provide useful evidence for practice are those that combine several methods in relation to evaluating community development for health programmes (Beattie 1995b). However, the combination of such methods should not be used uncritically (Sanford 1981; Mays and Pope 1995; Milburn et al. 1995a; Patton 1996a). Mixed methods are further expanded on in Chapter 4.

3.4 Costing of evaluations

The world is dominated by costs and evidence. Consequently, there is also a need for an evaluation to detail how funds are spent (Section 3.1). The WHO's definition of evaluation (Section 3.2) emphasises this by saying that costing an intervention in a process evaluation may help guide the allocation of resources. In this instance, it is beneficial to account for costs of implementation and evaluation to conduct a cost-effective exercise (Duflo 2004).

The main cost of evaluation is the cost of data collection/analysis (evidence), as health promotion activities seek to use evidence-based practice in order to deliver public health policy and to show progress towards meeting public health targets (Levin & Ziglio 1996). Yet, few research studies have looked into practical guidance regarding the cost and duration of evaluation studies (Hulton et al. 2007). The Makwanpur trial (Nepal) suggests that community mobilisation through women's groups is a cost-effective approach to improve birth outcomes and reduce neonatal mortality rate in hard to access villages compared to home visits by outreach workers (Manandhar et al. 2004; Borghi et al. 2006). Therefore, a cost-effective exercise helps to account for resources and accountability in order to scale up (Van Lerberghe & De Brouwere 2001; Koblinsky et al. 2015). In addition to process evaluation, cost evaluations are

likely to be more feasible for health promotion practitioners than formal health outcome (MMR) cost-effective studies. Aside from the cost and the methodological difficulties of these outcome studies, recognition of the process of change means that it is not even appropriate to attempt to measure a project's outcome until it has been successfully adopted, implemented and has been running smoothly for some time, usually several years. As seen in Section 1.2, there is a danger of implementing an ineffective or perhaps harmful intervention. The cost of the evaluation is described in Section 4.3.9.1.7.

3.5 Time & indicators

The development of useful evaluation criteria consists of selecting indicators (here, outcomes) for measuring change at different stages in a project's development and over different periods of time. These will link immediate objectives (focused on implementation), with intermediate goals (focusing on impact), and long-term objectives (focused on outcomes). There is a wide range of "off the shelf" indicators available. In particular, the global agencies have put considerable efforts into developing health promotion indicators, such as the WHO (or MDGs) indicators of uptake of maternal health services, used in LMICs, as discussed previously in Section 1.4.

Indicator data helps to measure knowledge, motivations, skills, behaviour and health status (WHO 2000). They can be used to evaluate certain health promotion projects (see Section 4.3.9). They are also transferable to the evaluation of local (community) level health promotion projects, as they replicate work done at a national level (Section 1.3.1). The use of standardised indicators can be helpful for comparison with other settings or transferability of the project, such as access to maternal health services in LMICs (Sections 1.3.1, 4.3.3 and 4.3.9.1.4).

3.6 Evaluations strengths & weaknesses

Limitations of the more traditional methods of evaluation are discussed in this section. There exists a risk of implementing an intervention that is ineffective or even harmful, as seen in Section in 1.2. Moreover, evaluation is sometimes

viewed as "negative/castigatory" or as a measure of performance or way of determining whether standards were met. It can also be unfairly conducted if subjected to time and funding limitations (Perkins et al. 1999).

Furthermore, the research literature is lacking in practical guidance on how much various kinds of evaluation studies cost and how long they take (Perkins et al. 1999). Evaluation is also difficult when project managers, funders, and staff disagree about the project's aims and objectives, about the relative importance of different objectives, if the project shows weak definition of goals and objectives, planning or methodology, etc. Another potential difficulty is the influence of donors or external partners on the project trajectory and resulting implications on the evaluation process. For example, outcome evaluations that are imposed on health promotion projects by external decision-makers and funders have been known to select outcome criteria that were not the stated goals of the project. Thus, the evaluation process will usually reveal any differences between the various stakeholders' expectations about what the project and evaluation should achieve.

Another limitation is that despite the increased acceptance of the health promotion's broad goals, nationally defined indicators continue to be mainly set within a clinical health gain framework (Section 3.2.1). Relying solely on the rigorous standards of evidence (for example, RCT) and inclusion criteria adopted by effectiveness reviews (systematic and Cochrane reviews) has excluded a lot of evidence that is useful to health promotion practice (Perkins et al. 1999). Due to the difficulties of meeting the stringent requirements of the randomised control trials and "scientific outcomes" in health promotion, the health promotion evaluations may be deemed weaker evaluations designs, such as quasi-experimental designs. As a result, health promotion evaluation is subjected to both design and validity criticisms from those committed to the RCT "gold standard" and complaints from practitioners that the results are not useful for clinical application, because they do not contain sufficient information (Macdonald et al. 1996). In addition, the naturalistic (context specific) and multifaceted nature of health promotion programmes means that it is difficult for scientific outcome designs to meet/converge.

Furthermore, in many circumstances, there are limits on the range of other potential causes that can be monitored (or controlled for). Where this is the case, the outcomes frameworks (such as the Donabedian model – Figure 11) provide a structure for assessing whether the planned outcomes were achieved, whether the interventions designed to achieve them were implemented as planned, and whether there may be other explanations for the outcomes observed (Craig et al. 2011).

There is also the risk of failure, which is related to the evaluation planning and to the evaluator bias. For example, an evaluator may set aims that are too ambitious (outnumber the programme aims) or a set of shared aims with the NGO/client or evaluators may impose predisposed notions and definitions of evaluations on clients, and fail to incorporate the cultural differences of individuals and programmes within the evaluation aims and process (Reeve & Peerbhoy 2007).

This Ph.D. study, therefore to mitigate the above, uses a comprehensive mixedmethods evaluation approach. With regard to the quantitative research, the underlying hypothesis was that the measured aspect (outcome of the evaluation) of the GTN intervention of health-seeking behaviour should improve in the intervention area relative to the control. In order to evaluate the GTN intervention for effectiveness (impact), a specific set of indicators of progress (i.e., uptake of services) was chosen based on the literature on maternal health access (Section 1.3). Further complimenting the quantitative research was the cost of providing health promotion. The qualitative research or process evaluation aimed to identify users' perceptions (knowledge, attitudes, and beliefs) regarding the effect of the intervention, barriers, and facilitators to the uptake of services and changes in decision-making of those that were part of the intervention compared to those who were not. Together the data from the two approaches assisted in determining the impact of the intervention and how to evaluate maternal health promotion interventions in LMICs. The aims and objectives of the evaluation are detailed next.

3.7 Aims & objectives

The aim of this doctoral research was to compare the effectiveness (here, impact) of GTN's health promotion strategy to the existing level of health promotion given to mothers in a LMICs community setting with defined indicators.

The objectives of this evaluation are to:

- measure the uptake of maternal health services by comparing the baseline (2007), midline (2010), and final (2012) data between intervention and control communities regarding ANC, delivery care, and PNC, and to evaluate the effectiveness of the Nepali intervention quantitatively as a "before-and-after" cross-sectional study;
- 2. assess the usefulness of the chosen regression analysis (difference-indifferences) as an analytical tool;
- measure perception of changes in the community, around maternal health;
- assess users' and healthcare providers' perceptions of the effect of the intervention and factors that influenced uptake; and measure the barriers and facilitators to uptake;
- measure perceptions of the intervention decisions around seeking delivery care in the community;
- 6. account for unintended consequences (methods) and;
- 7. measure the cost of the intervention.

3.8 Summary

There is enormous diversity in the approaches to evaluation and health promotion evaluation (approach). Evaluation aims to provide a "rounded" picture of the problem at hand - in this case, the intervention. Too many interventions in low-income countries are not evaluated, thus running the risk of supporting potentially wasteful or less successful interventions at the expense of more effective ones. Therefore, at the outset, it is worth investing time and effort in establishing why an evaluation is needed and negotiating this with the stakeholders who will have an interest in its results. Thus, interventions should be clearly and fully described (implementation and process) and they should be assessed against agreed criteria, including indicators of acceptability and implementability (based on the literature), appropriate outcome measures, and research methodologies (mixed methods). Impact evaluations are increasingly seen as beneficial as they account for the counterfactuals and unintended consequences of interventions. Lastly, the limitations and strengths of health promotion evaluations should be accounted for.

Chapter 4 Methodology & Methods

4.1 Introduction

This chapter will discuss the methodology and methods of this mixed-methods evaluation study design. A mixed-methods approach was chosen to:

- determine the impact of the GTN intervention in contributing to improved access, knowledge, attitudes, and beliefs of maternal health services (determined by multiple regression analyses on secondary data);
- assess women's control over the decisions in regard to their health and the community's perspectives of the intervention (obtained from qualitative interviews with primary data).

A combination of qualitative and quantitative approaches offered a thorough evaluation of the impact of the programme. Together the two approaches provided a deep insight into the key maternal health issues, examining whether there was a change in women's behaviour towards maternal health when community health promoters carried out health promotion activities. The quantitative data were triangulated, or "mixed", with qualitative data consisting of interviews and focus groups with health service providers and users (women) and their families. This mixing was done using an appropriate qualitative methodology to investigate the changes found (or lack thereof) between the data collection points. The use of the complementing and the "mixing" of methods are discussed in this chapter.

This section of the thesis addresses: (a) the mixed-methods approach; (b) qualitative interpretivism; and (c) quantitative positivism. Also outlined in this section are the strengths and weaknesses of the use of primary (qualitative) and secondary (quantitative) data, and the methodological strengths and weaknesses.

4.2 Rational of mixed methods

Mixed methods fit into a particular set of philosophies. It is worth noting that some purists, both from a quantitative and a qualitative perspective, argue that mixed-methods approaches are "incompatible", i.e. mixed methods approaches cannot be done because qualitative and quantitative approaches are unsuited to one another (Howe 1988). Obviously, this researcher supports the belief that quantitative and qualitative can and should be mixed. With this in mind, the following definition of mixed-methods research has been adopted:

"Research in which the investigator collects and analyses data, integrates the findings, and draws inferences using both qualitative and quantitative approaches or methods in a single study or a programme of inquiry" (Tashakkori & Creswell 2007, page 4).

In mixed methods, each project is reported separately as a distinct study, but, overall, the programme of inquiry is mixed-methods research (Baskerville et al. 2001). The use of qualitative and quantitative approaches gives the researcher an ability to use both numbers and words to combine inductive and deductive thinking to address the research problem (Creswell & Plano Clark 2007). Furthermore, using mixed methods addresses different questions within the same study. The process is essentially an amalgamation of two different research philosophies that offer the "best" in realist evaluation and philosophical underpinnings of monitoring and evaluation (Bamberger et al. 2010; Westhorp 2014).

Mixed-methods researchers, in practice, use and make explicit use of data from diverse philosophical positions. These positions are often referred to as dialectal (opposite) stances that bridge post-positivist worldviews with social constructivist worldviews and complemented by transformative perspectives (Green et al. 2007). In order words, mixed methods "mix" scientific methods of inquiry to critique the knowledge/evidence obtained where the researcher can influence what is observed yet he/she is conscious of biases with the knowledge/evidence stems from interactions in the social "human" world, namely pragmatic all the while acknowledging that change is possible in the self, beliefs and in lifestyle choices. These are a number of reasons for undertaking

101

mixed-methods research. Essentially, mixed-methods research is concerned with "what works". It represents an opportunity to transform different methods into new knowledge through a dialectical discovery: i.e. a discourse between two "people" each holding different points of view about a subject. Yet through this dialogue the researcher wishes to establish the truth through reasoned arguments (i.e. research). In general, studies draw upon one or more theoretical frameworks from social, behavioural, or biological sciences to inform all phases of the study; however mixed-methods studies integrate a variety of theoretical perspectives, as previously seen (Pasick et al. 2009). The mixed methods used in this thesis included a pragmatic approach ("doing what works best") that draws on employing "what works," using diverse methods, giving importance of the research problem and question, and valuing both objective and subjective knowledge, i.e. that which can be "observed" or reproduced considered alongside the researchers' personal perspective and belief (Morgan 2007). Moreover, mixed-methods design should be informed by a theoretical and conceptual framework, the latter model is intended to assist the research community in evaluation and to bring about change (Meissner & Sprenger 2010; Meissner et al. 2011).

Mixed methods permit the researcher to "view problems" from multiple perspectives to enhance the meaning of a singular perspective or to contextualise the information at the macro or micro level - either to obtain a picture of a system such as a hospital or add in information about individuals (e.g. attending care). Mixed methods complement the picture of an evaluation by merging quantitative and qualitative data to give a more complete understanding of the problem by comparing, validating or triangulating the results. Furthermore, the use of mixed methods allows the researcher to examine (alongside) the outcomes, processes, and illustrate the context for intervention trends/changes (Plano-Clark and Creswell 2010).

Additionally, mixed methods may use both primary and secondary data. The two main approaches in the primary and secondary research are qualitative and quantitative. Primary data sources are those collected directly from the original or primary source by researchers (Patton 1996; Dawson 2009). Primary data can be seen as collected first-hand data by the researcher. Therefore primary

research is 'new' research (i.e. collecting new data), carried out to answer specific questions using qualitative, quantitative or mixed-methods approach (IWH 2015). The quantitative methods may use a questionnaire as one research tool (survey questionnaire). Meanwhile, the survey is often based on personal interviews using questionnaires at or near health points. Therefore, the questionnaire surveys are very effective in systematically collecting data from a great number of people and at a low-cost, in order to produce summarised and quantitative descriptions.

Secondary data, on the other hand, often include information from the national population census and other government information. One type of secondary data that is used increasingly is administrative data. This term refers to data that are collected routinely as part of the day-to-day operations of an organisation, institution or agency, (e.g. NGO). There are many examples: motor vehicle registrations, hospital intake and discharge records, workers' salary, and more. Compared to primary data, secondary data tends to be readily available and inexpensive to obtain. In addition, administrative data tends to have large samples, as the data collection is routine and comprehensive, and are collected over a long period. That allows researchers to detect change over time (months or years). Secondary data can be examined in addition to the information provided by primary data (i.e. survey results/focus groups) by 'mixing' to provide a more rounded interpretation of the findings (IWH 2015).

4.2.1 Mixing the data & data collection using different designs

Morse and Niehaus (2009) state that there is a point where mixing occurs and the latter may differ depending on the mixed-methods design. Mixing may occur during data collection (for example, when both quantitative items and qualitative open-ended questions are collected on the same survey), during data analysis (for example, when qualitative data are converted into quantitative scores or constructs that are compared with a quantitative dataset), and/or during data interpretation (for example, when results of quantitative analyses are compared with themes that emerge from the qualitative data), (Morse & Niehaus 2009).

Therefore, in a single study of a multiphase programme of inquiry, some projects employ a design that is known as a "stand-alone" design, while other

studies use a design with phases that build on each other and contribute to an overall programme objective. The study designs or approaches that can be used include: a) a convergent (or parallel or concurrent) designs; b) sequential (or explanatory sequential or exploratory sequential designs); c) embedded (or nested) designs; and d) multiphase designs (Creswell and Clark 2011). However, more complex designs exist and are driven by specific questions and aims in particular investigations (Morse & Niehaus 2009), as detailed below.

Creswell (2003) stated that the mixing of data is a unique aspect: by mixing the datasets, the researcher provides a better understanding of the problem than if either dataset had been used alone. In single or multiple studies, the data can be mixed by:

- a. Convergent or parallel or concurrent design: integrating multiple forms of data/merging or converging the two datasets by bringing them together. In mixed-methods studies, data are integrated or combined rather than conducting two separate endeavours to collect data. Challenges exist in integrating the data, therefore it is a priority to "maximise" the strengths of both data and "minimise" the weaknesses. Some suggest using systematic integrative procedures by merging data, connecting data, and embedding data (Creswell and Clark 2011).
- b. Connecting data in a sequential (or explanatory sequential or exploratory sequential) design: connecting the two datasets by having one build on the other. Here, integration means to analyse one dataset (e.g. survey data) and use the information to inform the design and collection of qualitative data (e.g., interview questions, or identification of participants to interview). The analysis of results begins in the initial phase with the data collection followed by the second phase of research where the analysis is completed (Dawson 2009). A slight variation on the sequential design would be where a study conducts an intervention and embeds qualitative data within the intervention to aid understanding. An example of this would be exploring how participants experience the treatment/intervention.
- c. Embedded (or nested) designs, embedding data: embedding one study or method of data collection provides a supportive role for the

other study, or method of data collection. Here, integration means embedding a secondary priority within a larger, primary design, for example before a trial to structure procedures or post-trial to inform development or results of the trial. This would provide "complementing" qualitative data about how participants felt about a clinical trial during the intervention; Miaskowski and colleagues (2004) conducted an evaluation of both the outcomes and process of the intervention. In the RCT study, the qualitative data were collected with the use of audiotapes of the intervention sessions. The notes from nurses and patients provided a fuller picture of the issues, strategies, and interactions experienced during the intervention.

d. Multiphase designs are frequently used in health sciences. These designs emerge from several/multiple projects conducted over time with sequential elements. They are conducted to develop, test, implement, and evaluate a health prevention programme where the project can be a) qualitative; b) quantitative; and c) mixed. In short, study designs are conducted over time with links in place so each phase builds upon another with the common overall objective of testing the health prevention programme.

In the study in this thesis, both quantitative and qualitative data were collected in a single study rather than in multiple studies over time. As, increasingly used in health research, mixed methods begin with the understanding that investigators wish to understand the social and health world by analysing the data yielded. Mixed methods, or the combination of quantitative and qualitative data, are based on an assumption that to understand the social and health world, one needs to gather evidence on the nature of the research question and theory. Therefore, social inquiry is targeted at and influences a given problem (e.g., policies, organisations, the family and the individual). Mixed-methods research is thus more than just simply collecting qualitative data from interviews or collecting multiple forms of qualitative evidence (e.g., surveys and diagnostic tests). It involves the intentional collection of both quantitative and qualitative data as well as the combination of the strengths of each to answer research questions by mixing the data. Quantitative (mainly deductive) methods are ideal for measuring "known" phenomena and patterns of association, including inferences of causality; while qualitative methods identify unknown processes, explanations of "why" and how phenomena occur, and the range of their effects (Pasick et al. 2009).

4.2.1 Advantages & challenges in mixed methods

Mixed methods have several advantages. For instance, quantitative analysis, as compared with the qualitative approach, seeks to gauge potential impacts that the programme may generate, while the latter seeks to highlight the mechanisms of such impacts, and the benefits, or lack of, to recipients from indepth and group-based interviews. Whereas quantitative results can be generalisable, the qualitative results may not be. Nonetheless, qualitative methods generate information that may be critical for understanding the mechanisms through which the programme helps beneficiaries. Thus, mixed methods allow the researcher to use both numbers and words, and to combine deductive, i.e. the "top-down" approach that uses theory to test the observations/measures in order to address the hypotheses, and inductive or the "bottom up" approach that moves from specific observations/measures to detect patterns, or formulate, some tentative hypotheses to develop broader generalisations and theories. In conclusion, inductive reasoning by its very nature is more open-ended and exploratory, especially at the start of a research project. Deductive reasoning is more narrow in nature, and is concerned with testing or confirming hypotheses to address the research problem (Creswell & Plano-Clark 2007).

There are also challenges with sampling in mixed methods. There are analytic and interpretive issues with specific designs. When the investigator mixes the data in a sequential design; the findings may be conflictive or contradictory. Therefore, strategies to resolve differences between the two methods need to be considered before and/or after gathering more data by revisiting the databases – i.e. there is a "point of interface" where the investigator decides what results from the first phase will be the focus of attention for follow-up data collection (Teddlie & Yu 2007; Meissner et al. 2011). There are challenges and benefits of a team approach to mixed methods, for example, forming the mixed-methods research team that includes both quantitative, and qualitative researchers. The subsequent data collection involves multi-disciplinary, inter-disciplinary, and/or trans-disciplinary teamwork, which means different approaches, and where congruency in design and interpretation takes time and effort to be reached. Therefore, there are methodological challenges in mixed methods, which include issues around teamwork, resources, data collection and analysis, and interpretation. For instance, it takes time and resources, almost double, to carry out data collection and analysis (audio recorders, transport, building infrastructure, and resources) (Tashakkori & Teddlie 2010; Meissner et al. 2011). Furthermore, caution is advised, as there are instances where priority is given to either the qualitative or quantitative research despite the methods being equally emphasised. For instance, priority could be unintentionally placed on the secondary data when it is embedded into a larger, primary design (Cresswell et al. 2003; Plano-Clark & Creswell 2010).

Finally, limitations also exist in the reporting of mixed methods. Many publications, for example journals will have word limitations, and this also affects publication of mixed-methods studies in scholarly journals. Researchers, therefore, need creative ways to present material: quantitative findings are often presented in tables, while the qualitative findings can be illustrated as themes (Stange et al. 2006; Meissner et al. 2011).

4.2.1.1 The critique of quantitative and qualitative methods

4.2.1.2 Quantitative research & its evidence

Quantitative research is a mode of deductive inquiry used to test theories or hypotheses, gather descriptive information, or examine relationships among variables, which are measured and yield numeric data that is statistically analysed (such as maternal health services uptake outcomes). Unlike qualitative data, quantitative data provides measurable evidence to help establish probable causes and effects. Efficient data collection procedures can be more likely to replicate and generalise a population, to facilitate the comparison of groups and provide insight into a range of experiences. The approaches used in health sciences include descriptive surveys, observational studies, case-control studies, randomised controlled trials, and time-series designs (Pasick et al. 2009).

Strengths and weaknesses of the quantitative approach

Of the three levels of quantitative research (descriptive, correlational and causal), each has its own individual merit, starting with descriptive studies, which give an indication of the frequency with which something occurs, while correlational studies investigate a relationship between variables (e.g. age, size, etc.). Finally, the strength of causal (or experimental) research lies in its exploration of the relationship between variables once an intervention is introduced, i.e. establishing a relationship between cause (independent variable), and effect (dependent variable), whilst attempting to hold extraneous variables constant (Walker 2005), such as applying a control. Using a control strengthens quantitative research - first, it acts comparison group, and secondly reduces systematic bias and erroneous conclusions of a study hypothesis (Section 3.2.3.2). Standardised statistical analysis lets us derive important information from research data, including trends, differences between groups, and demographics. With regression analysis for instance the number of characteristics can be controlled for in order to reduce variation among cases (Duflo 2004). Therefore reliability/interval validity and generalisability beyond the study sample can be ensured (Duffy 1985). Additionally, guantitative estimates can be obtained of the costs and benefits of interventions (Colburn et al. 2015).

Quantitative research also provides information regarding the relationship between the variables of interest to predict future outcomes. The latter is possible as the researcher is able to "manipulate" an independent variable in order to study its effects on the dependent variable, for example controlling for gender or age and measuring changes in school attendance with (and without) the provision of "free" school meals (Corner 1991). The strength in producing numbers lies in assisting health policy-makers and managers in prompt decision-making on application of resources (e.g. equipment, staff, beds, etc.), and cost-effectiveness of discharge planning and length of time a patient stayed out of hospital (Carr 1994).
Quantitative studies often require as a prerequisite appropriate sample selection of the population under study, and in experimental studies, conducting a power calculation to determine at what size of the population we will see an effect (Walker 2005). Clear documentation needs to be provided regarding the content and application of the survey instruments so that other researchers can assess the quality of the data and the validity of the findings. As samples of individuals, communities, or organisations can be selected to ensure that the results will be representative of the population studied- the principal strength of the quantitative approach is that findings can be generalised to the population about which information is required (Duffy 1985; Walker 2005).

Weaknesses of quantitative methods include that the administration of a structured questionnaire creates an "unnatural" situation that may alienate respondents. While in some instances, the studies are expensive and timeconsuming, and it takes time to obtain the preliminary results. This may be a concern if results are promptly needed to make a decision to implement, continue or upscale an intervention (Choy 2014). Moreover, the why-question is often not answered in quantitative research, for example, why individuals have a preference for one choice over another, therefore a qualitative approach is complementary (Johnson and Onwuebuzie 2004). Furthermore, self-reported data obtained in a questionnaire may be inaccurate or incomplete (Walker 2005). Recruitment and attrition are common problems in sampling, and participants' dropping out of an intervention limits the generalisability (Walker 2005). Often there is no information on contextual factors to interpret the data or explain the variations in behaviour between participants that have similar demographic and socio-economic characteristics. Furthermore certain groups, such as people who are illiterate, or ethnic minorities may be harder to access using quantitative methods. Another concern in quantitative methods is that its tools are 'inflexible' as the instruments cannot be modified once the study begins (Choy 2014).

There may occur weaknesses in sample-size calculations as they rely on an estimation of the expected degree of change in the dependent variable and are therefore limited to studies where research on the subject already exists

(McMahon 1994). Moreover, in statistical analysis, there may be issues in the data analysis/interpretation for instance in the case of "missing" data. While, the correlations produced may mask or ignore underlying causes/realities, or that the results may be due to "random" events, hence the need for a control in time and area. Additionally, errors in the hypotheses tested may yield a misinterpretation of programme quality or influential factors, and errors in the selection of procedures for determining statistical significance can result in erroneous findings regarding impact. Finally, a weakness in health evaluation studies (dealing with holism) includes the diminishing of experiences of the individual as quantitative analytical methods may regard human beings as merely reacting and responding the environment (Corner 1991; Carr 1994).

4.2.1.3 Qualitative component of mixed-methods study

Qualitative research in evaluation can contribute to assessing interventions by illuminating processes, exploring diversity, and developing new theories. There are many methods employed to collect primary data such as structured or unstructured interviews, questionnaire surveys, and case studies. Qualitative methods are most appropriate for exploring complex phenomena or areas not (easily) amenable to quantitative research (Tashakkori & Teddlie 1998; Campbell et al. 2000). They are a distinctive approach to research in their own right (Bryman 2007). Qualitative interviews can enable respondents to express themselves in their own words (Gill et al. 2008; Ulin et al. 2012). Clearly the study of perceptions, awareness, and views lends itself very much to a qualitative approach. Therefore, the approaches in qualitative health research systematically and rigorously investigate (theory), and include a broad range of methods such as in-depth interviews, action research, participant observation, conversation analysis, case study, ethnography, phenomenology, review of documents, conversation analysis, grounded theory, and/or a narrative study (Al-Busaidi 2008; Lewin et al. 2009; Pasick et al. 2009). Qualitative research focuses on the meaning and experiences of participants in the context or inductive of theory-development driven research. Qualitative researchers aim to understand processes, sometimes emerging over time. The data (quotes) can

set a context of the participants 'voice'. Therefore, the collection of data provides an in-depth understanding of concepts (Ezzy 2013).

Strengths and weaknesses of the qualitative approach

There are benefits of a qualitative descriptive approach. In this section, the strengths and weaknesses of this approach are detailed alongside examples of use. Qualitative description is a useful qualitative method in medical research, bearing in mind there are limitations of the approach which ought to be considered when undertaking the research. It is especially relevant in mixed-method research, in questionnaire development, and in research projects aiming to gain knowledge of patients', relatives' or professionals' experiences with a particular topic. Qualitative analysis, as compared to the quantitative approach, elucidates the programme's impact, their mechanisms, and the benefits to the participants using in-depth and group-based interviews. Whereas quantitative results can be generalisable, the qualitative results may not be. Nonetheless, qualitative methods generate information that may be critical for understanding the mechanisms through which the programme helps beneficiaries (Denzin and Lincoln 1994; Lewin et al. 2009).

The human element in qualitative research is both its strength and weakness: one strength is that it provides an understanding of human insight and experience, which influences interaction in the physical setting, and on the other hand, a weakness as qualitative research is subject to the researchers' skills and training (Patton 2002). Therefore, the role of the researcher is in accordance with the research tradition used. A researcher aims to minimise their personal assumptions (bias) when collecting data. Yet, there is no avoiding the "effect" the researcher has on the interview and in defining the shape of the study. Here, for example this had an implication as the perception of being an "upper caste" researcher and perceived as 'better' (more intelligent or wealthier) may have influenced responses from participants who are classed in Nepal as "lower" castes. Britten (1995) suggests that qualitative researchers ought to consider how they are perceived whether it be due to their social category/class, caste/race, and sex during an interview. As a result, participants may try to give a "desirable" response, aiming to please the interviewer (Britten 1995). The latter is defined as *social desirability response bias* (Randall, 1991), where participants may feel a need to justify their responses. Therefore, the objective of a qualitative research interview is to aim to discover the interviewee's own meaning and avoid prior assumptions with "preset" categories, such as those above. Listening and observation are useful skills. Furthermore, a good level of self-awareness is necessary in the researcher in order to reduce the aforementioned possible biases (Britten 1995).

4.2. Summary of methodology

The first part of this chapter has delved into mixed methods and its two main methodologies – the quantitative approach and qualitative approach. Mixed methods research generates questions and hypotheses that form the basis of decision-making or further research. Despite the application of rigorous procedures, including measures to control systematic error and bias, the use of mixed methods is subject to a number of methodological and ethical concerns. The previous section has stressed that neither approach is superior to the other: quantitative methods facilitate the discovery of quantifiable information, and qualitative research is useful for the exploration of subjective experiences of participants. In research, combining both approaches in a mixed methods study, if time and money permit, is valuable in evaluation studies for evidence-based decision-making. The next section expands on the methods used in this evaluation.

4.3 Mixed methods in this thesis

It is clear the aim of this thesis (Section 3.6.1) requires a variety of different research methods. The primary aim of this study was to first address positivist (deductive, i.e. quantitative methods) effectiveness questions, while the secondary aim was – to explore sequentially interpretivist/realistic (inductive and deductive, using qualitative and quantitative methods) explanatory questions. Therefore, the thesis adapted a pragmatic mixed-method approach. In this study, qualitative and quantitative data were collected, entered, and analysed as part of the process and impact evaluation. The integration of process evaluations to an impact or outcome evaluation enables the external validity of the findings to be assessed. This may help in the successful replication of the programme. Process data (qualitative) allow the researcher to explain the results obtained (quantitative), while impact data (mixed data) provide details to the results obtained. This is an impact evaluation due to only five years of data and no health outcome data (Section 1.3).

The next section outlines the individual methods applied in the thesis. These include:

- qualitative: (a) focus group discussions; and (b) face-to-face interviews;
- quantitative: (a) survey method; (b) Difference-in-Difference regression (DiD) analysis; and (c) the cost study.

There will be an introduction to more generic issues concerning all of these methods, including the selection of areas (locality), translation issues, triangulation of findings, and research ethics. In addition, for each of the methods there will be details on sampling frame, the sampling process, data collection, analysis, and reporting.

4.3.1 Mixed-methods evaluation

The mixed-methods approach reported in this thesis combined the strengths of the two different methodological approaches in four phases starting with:

- review of the literature on evaluation of maternal health promotion/group interventions in developing countries (Chapter 2);
- 2. quantitative data (maternal health services uptake) analysis of the baseline, midline and final datasets (Chapter 5);
- 3. cost data analysis (Chapter 5);

4. qualitative data analysis of face-to-face semi-structured interviews with health professionals, women and men (Chapter 6).

In this evaluation the studies were mixed sequentially, where the qualitative followed quantitative study. The latter explored various barriers that could have hindered antenatal, delivery care and postnatal care utilisation, and to investigate the changes found (or lack thereof) between the data collection points. The mixing of methods is schematised in Figure 12. Once the quantitative and qualitative analysis was completed, the data sets were mixed to examine if there is a change in uptake, knowledge, attitudes, beliefs, and behaviour when health promotion activities are carried out by community health workers (auxiliary midwives) with women of childbearing age (with children <2 years old) and their families (typically their mothers-in-law) compared to the control area. These are presented in Chapter 5 and 6. The quantitative data is presented in tables, while in the qualitative component the participants' details are in tables and the findings illustrated as themes accompanied by examples (quotes).



Figure 12 Overview of mixed methods

In this thesis, the findings were mixed or triangulated in a particular way to provide a fuller picture or better understanding of the impact of the intervention on the community by elucidating the mechanisms of change (i.e. uptake), see Chapters 5

and 6. Here, the data analysis consisted of combining the data and comparing the two sets of data and results using a sequential study design. In sequential design one method of inquiry follows the other: that is, qualitative exploration is followed by quantitative or *vice versa*. The in-depth qualitative data can help explain the underpinnings or the responses of the underlying quantitative results. Here, the exploration the quantitative data collection was followed by the design of a qualitative instrument, and then administered to a sample of the target population. Figure 13 depicts the sequential nature of this thesis research.



Figure 13 Merging the data in a single study

It was useful in this evaluation to combine the qualitative data in the form of texts or images with the quantitative data in the form of numeric information. This integration means that results are reported together - first the quantitative statistical data, followed by qualitative quotes or themes that support or refute the quantitative results. Sandelowski (2000) refers to transforming datasets e.g. counting the occurrence of themes in a qualitative dataset or, and through tables or figures displaying both the quantitative and the qualitative results (i.e. data displays).

In this evaluation, the survey outcomes fed into the focus groups and individual interviews with the health promoters, eligible participants, and health workers from the experimental group to help determine why the programme "worked" (Section 3.2.3.2). These results were then compared to the similar individuals in the control

group. The GTN evaluation was a three-stage design consisting of a controlled, non-randomised, repeated quantitative cross-sectional study of the GTN community-based health promotion intervention, mixed with qualitative data that explored the changes if any post-intervention. The evaluation was complemented with a costing of the intervention.

4.3.2 Maternal health services uptake conceptual framework

In the mixed-method evaluation, a conceptual framework to explore factors contributing to (non-) attendance was needed. A literature review guided the choice of the conceptual framework for this evaluation in order to find key elements of evaluation methods to assess the effect of community-based maternal-health promotion interventions in low-income countries. The full details of the literature review were described in Section 2.6.1.

A conceptual framework supports the theory of research, i.e. to support to what is investigated, by providing a strategy to evaluate an intervention's effect on chosen outcome indicators (Graham and Kelly 2004). To conceptualise the analysis for the study, the framework used by Dharmalingam and colleagues (Dharmalingam et al. 2010) was adapted (Figure 14). The latter conceptual framework was adapted in order to expose the causal link of socio-demographic and maternal health service factors to maternal health services uptake. Dharmalingam and colleagues (2010) suggest that the likelihood of uptake is directly or indirectly caused by two major factors: underlying factors (maternal socio-demographic characteristics, such as family's economic status, husband's education, residence, decision-making, etc.), and proximate factors (maternal characteristics such as body mass index, service use, birth interval, smoking, type of cooking fuel used, etc.),



Figure 14 Conceptual framework used in the evaluation of GTN

In this thesis' conceptual framework, there exist two major independent groups of factors (variables). The first group includes the underlying (confounding) factors that an RCT controls for, including community economic status and education, decision-making for health, socio-demographic characteristics, such as ethnicity, rural/urban residence, development and ecological region – i.e. how developed the area is and climate effects on crop and food yield. The second group includes the proximate factors that the researcher in a quasi-experimental study chooses to control for. The choice of these proximate factors/independent variables was based on data from previously published DHS studies (Khanal et al. 2013b), and DHS datasets from other LMICs (Sections 1.4 and 2.6.1). They included time, area, and maternal characteristics such as mothers' age, education, wealth, parity, and health service use/uptake. These latter outcomes included service use variables including the uptake of ANC, iron consumption during pregnancy, uptake of skilled attendant at birth, and PNC uptake (Dharmalingam et al. 2010; Wallerstein 1992; Nair et al. 2000; Sreeramareddy et al. 2011).

By using the above conceptual framework (Figure 14) in a mixed-methods approach, the study was able to explore how proximate factors influenced maternal healthcare attendance (outcomes) during women's most recent pregnancies while taking into account the underlying factors (i.e. context of rural Nepal study in which women live in rural areas, are not necessarily the main decision-maker, pay for/ walk to services etc.). The latter were taken into account in the qualitative component of the study.

In the quantitative component, the adapted maternal health attendance conceptual framework was used to explore factors contributing to (non-) attendance (n = 1,236) from the survey, as seen in Figure 12. Attendance was compared in multiparous women aged 14 years or older (n = 621) with that of control group (n = 615). In the qualitative component, focus groups and interviews were carried out using non-probability/purposive sample (n=47). The sample for the mixed-methods study consisted of 1,283 participants.

First, changes in health services' uptake were compared by using multiple regression analysis (or DiD) and second with the women's perspectives and current knowledge on the intervention topics all while considering the underlying factors. These (latter) qualitative methods, which looked at some underlying factors, are detailed in Section 4.8.

4.3.3. GTN survey & survey population: sampling

This section describes both the qualitative and quantitative sampling methods. Sampling can be described as a process, or a technique of selecting a suitable sample, or a representative part of a population for the purpose of determining parameters or characteristics of the whole population. Two sampling procedures were used for primary (qualitative) and secondary (quantitative) data. For the latter, this delves into how the baseline and subsequent data (midline and final) was collected.

4.3.4.1 Qualitative Interview sampling

Both probability and non-probability sampling are used in qualitative research. Nonprobability sampling is often referred to as purposive sampling, as one may be purposely inclined to obtain information from a specific group. In this case, the researcher assembles individuals with known or demonstrated experience and expertise in the area being researched. Therefore, in this study, the gualitative sampling was both purposeful and homogenous (non-probability), which strengthened the study (Thwala et al. 2012). Data collection took place over one month in 2012 (Sharma et al. 2016b). Participants were women with a recent pregnancy, their mothers-in-law, the husbands, local rural healthcare providers, and the GTN health promoters. Face-to-face interviews were conducted with selected healthcare professionals as identified by the sampling methodology (Kitzinger 1995; Coyne 1997). The qualitative research included face-to-face interviews and focus group discussions. The focus group participants were recruited purposively from existing mother, men and mother-in-law GTN groups, and non-GTN groups (saving/literacy groups). They took place in a range of community settings including fields, village halls and schools. In two locations, local link community health workers (maternal and child health workers and auxiliary nurse-midwives) helped to arrange the focus groups, and these were held in the premises of the local community hospital and healthpost. The interviews were conducted in the local health post, hospital, and GTN offices for the healthcare providers. As this was an evaluation, some of the participants were from the GTN health promotion groups (purposive sampling); this was done in order to determine: a) the exposure to GTN activities; and b) the level of maternal health knowledge in each area.

It should be noted that there are issues with the sampling and sample size that ought to be considered; for instance, women who opt out of the intervention, (may) also opt out of the research (Section 4.2.1). The author is conscious that two types bias may occur during the qualitative component of this evaluation. The first is social desirability response bias (Randall 1991). The author is aware that her surname is a Brahmin surname (Sharma), that she is a "foreigner" (from the U.K.), and that of the qualitative interpreter is a Newari surname (not disclosed). Both these are considered upper castes in Nepal, and these may be considered a potential bias and weakness that may arise from the author or qualitative interpreter. Furthermore, based on past experience from forming health promotion groups (where men would not speak to the female health promoters), the researcher decided that a male interviewer should conduct the focus group with the male participants. In the evaluation of the programme, a Nepali and Newari qualitative female researcher and Nepali male researcher worked with the author, and they also acted as translators. Furthermore, the team of qualitative researchers

119

were perceived as "different" (being foreign/from the city) from the rural/village participants, which may have influenced the responses in this study, as participants may have felt a need to justify their choices or may or may not have interacted fully (openly/honestly etc.) with the interviewer. Finally, as some of the women were from the health promotion groups - this may have led to selection bias (Sandelowski 1995; Higginbottom 2004).

4.3.4.2 Quantitative sampling

In the quantitative study, the GTN surveys were not based on random or opportunistic sampling; it was methodologically much stronger as it was based on total community sampling (Hultsch et al. 2002). The data collectors attempted to include/cover all eligible women in the community in the survey. They were identified through house-to-house recruitment, and the surveyors reported that there were very few refusals to participate. A sampling issue that should be noted is that there is no unique identifier for individuals surveyed. Thus it was not possible to identify overlap between the two surveys – that is, the people making up the population "technically" differed or may not have differed, i.e. it was women with a baby in the two years prior to the study each time. This may have led to recall bias (McColl et al. 2001).

4.3.4.3 GTN survey

The final copy of the questionnaire was in English based on the DHS and literature. It was then translated into Nepali for data collection; and during data entry translated into English. Questionnaires were refined after a pre-test/pilot; and the surveys took place over a two-week period in 2007, 2010, and 2012 (Appendix III).

Women were interviewed in their homes by trained fieldworkers. If they were not available, interviewers returned on several occasions. Following a third attempt, the women were dropped from the study participant list (Simkhada et al. 2009). Health and socio-economic data were collected at the individual, household, and village level using a structured questionnaire. Data was obtained on awareness and

utilisation of maternal and child health services (antenatal, delivery and postnatal), decision-makers in the household, data on knowledge and attitudes of maternal health, women's background characteristics (education, age, marital status, parity, etc.), pregnancy history, socio-economic, caste status, and the population-based information on intake of iron and folic acid. Each cross-section survey consisted of individual women interviewed in four Village Development Communities (VDCs), two in the intervention and two in control area among women, pregnant or not, aged 15-49, and children under 2 years of age.

There are some similarities with the DHS survey as some DHS questions were included in the GTN survey (Section 1.4.4). However, the DHS survey is in selected parts of Nepal and no individual areas (districts) are identified in the survey. Therefore, there is no data comparison possible between Pharping, and DHS data; also, census (district) data is not reliable. DHS data is not divided by district, hence the need for a control group. It was necessary to have a control group for comparing changes across time (Simkhada et al. 2009).

In addition, having data from the baseline, midline, and final evaluation allow for comparison over time on demographic and health-related uptake behaviour correlation. The survey data helped GTN for planning, monitoring, and implementing the programme (Sections 2.5.2 and 2.5.4). It should be noted that a small number of question between the surveys were changed based on the responses, new questions were added by the midline and some were removed by the third survey.

4.3.4.3.1 Training of field staff for sampling

Field research included the supervision of training, data collection, and data entry during the final survey in 2012. Prior to sampling, a team of 15 enumerators and two mappers were recruited and trained (role playing, class demonstrations, and field practice), and instructions were given. Field staff was recruited on the following criteria – a degree in the field of health, experience in fieldwork data collection, entry and statistical analysis with fluency in Nepali and English. The training on paper questionnaire took place over three days prior to each survey. The training included

theoretical and practical sessions such as practical demonstrations, practice interviewing in small groups, and several days of field practice. During training, enumerators were encouraged to ask questions to better understand the questionnaire. Mock interviews focused on questions related to sensitive topics (of reproductive health). It was noted that enumerators had perceived certain topics too sensitive. Enumerators found it hard to ask questions on mortality, abortion, and domestic violence, yet respondents were comfortable in answering them.

Furthermore, the enumerators were confident in carrying out field tasks, and they had a good rapport with the quality control team and trainers. Despite their young age/experience, they were able to recommend alternative phrasing, formatting of the questions, or order of questions. The enumerators were a mixture of men and women, as certain respondents preferred a female interviewer. The enumerators were divided into two teams each day for various wards and given gifts like toothbrushes/soap. Nail clippers and toothbrushes were given to respondents as a thank you for their time. A trained reproductive health researcher, who also acted as the programme manager for GTN, supervised the training, data collection and entry. This individual, as well as the two-field coordinators, monitored data quality (data quality team n=4) and feedback was provided to the enumerators. The communication was regular and via mobile phones.

4.3.4.3.2 Data management

The data processing and entry was done once the research team agreed on the codes and categories. The data management consisted of questionnaire responses checked by the field coordinators and re-checked by the data auditor before the data was entered in the GTN offices. Because the volume of data was considerable (~300 variables and each round of survey with ~400 respondents) checking data entry quality was essential. This was ensured through observation by the data quality team including the author and then random checks from the hardcopy questionnaire. They observed 10% of survey interviews. Hard copies of the records were stored in a filing system in a lockable room, while the electronic output was anonymised. All data were recorded and analysed anonymously during fieldwork. The three sets of survey data were coded and entered with STATATM (version 11.0, Stata/SE 11.0s Stata Corporation, College Station, TX, USA).

4.3.5. Study area & population

Control area/area selection

GTN selected two districts in Pharping with similar socio-economic characteristics, one north and one south of Kathmandu, each with a total population of fewer than 9,000 inhabitants. Pharping is in the mountains of Kathmandu Valley (Figure 15); it is a fertile land near a tarmac road to Kathmandu (Simkhada et al. 2009; GON 2014) (Table 2). There were four VDCs in the study area. All four VDCs contained nine wards each. Two of the four VDCs were used as the control community (VDC A and VDC B). The control community is located 20 kilometres northeast of Kathmandu Valley while the intervention community is located 20 kilometres southwest - reducing any possible contamination bias from the intervention site, also consisting of two VDC areas with rural small towns and villages, VDC C and D. Some of the wards of VDC D are connected by road to Kathmandu: VDC D.



Figure 15 Map of area as part of Kathmandu district

(Pharping: bottom left hand corner of Kathmandu district, Source: HRRP, 2015).

The control (C) community was selected on the basis of its location, population composition, ethnicity, and literacy rates (Table 2), facilities available and its similarity to the intervention community (Section 2.5.3). In the control area, there are two health posts and a primary care centre nearby (similar to the community hospital in the intervention area) (Table 2). In the control community, there are a total of 1,574 households with a total population of 8,292: 4,111 are male and 4,186 are female (CBS 2001). On the other hand, in the intervention community, there are 1,646 households and a total population of 8,569 with 4,142 males and 4,427 females (CBS 2001).

Table 2, compares the intervention and control communities: the population size and the number of households were similar in both communities, although slightly higher in the intervention (I) community. Ethnicity was evenly distributed in both communities, with a majority of Tamang and Brahmin-Chhetri (CBS 2001; Simkhada et al. 2009). Health services centres were comparable in the two areas (Section 2.5.3).

The control and intervention area were chosen from a few pre-selected districts (Section 2.5.3). Of the pre-selected districts, the intervention area scored the highest for implementation: GTN was able to recruit health promoters from that area and build upon existing groups and had support from the community hospital, and local health post (Sharma et al. 2016a).

Characteristic		Intervention	Control
Households		1646	1574
Total population (Census 2001)		8569	8292
	Male	4142	4111
	Female	4427	4186
Literacy (%)		59.75	51.34
Ethnicity (%)			
	Brahmin	10.3	8.7
	Chhetri	19.6	21.1
	Tamang	32.9	30.4
	Newar	31.4	33.5
	Dalit/Rai	1.8	2.8

Table 2 Characteristics of intervention and control community

Source: Census (2001)

4.3.6 Baseline data

The project timeline was over six years, from November 2006 to September 2012. The prospective baseline enrolment took place from November 2006 to September 2007. This was done to ensure the baseline data were collected prior to the start of the intervention. Finally, the intervention ran from 1st January 2007 to 31st December 2012 (5 years).

4.3.7 Study population

The study population included all married women in the reproductive age group 15-49 years old. These women were residing in the study area and delivered their last baby within the last 24 months preceding the study. Since there were no accurate data recorded at the local level (VDC SHPs) to act as the sampling frame (Simkhada et al. 2009), the study population was calculated on the basis of the last census data (prior to the baseline in 2001). According to the last census data (CBS 2001), it was estimated that the two-year olds comprised 4.2% of the total population. On this basis, the sample size was roughly calculated from all women with a child of less than two years. The total sample (population) size for the four VDCs was 708 women with a child under the age of two [4.2% * 16,861]. In 2008, using trained Nepalese fieldworkers; GTN subsequently visited every household in each VDC over a two-month period to collect baseline information.

Having visited all households in all four VDCs, GTN could only find evidence of 485 women with a child under the age of two. Of these 485 women, 412 women agreed to complete the survey, 36 declined to participate, and 37 could not be found despite several visits to their homes. The women who refused to participate or who could not be found were reported to be reasonably well distributed across the four VDCs.

There are several possible logical explanations for this discrepancy in population numbers:

- 1. both data sets are right but there has been a change in the population over time;
- 2. the way census data are amalgamated introduced anomalies (Simkhada et al. 2009);
- 3. the census data are imprecise or incorrect;
- 4. the GTN data might be incorrect, although this is unlikely as all houses were visited in person;
- finally, it is possible that both data sets are incorrect as there are 3 million Nepalese working abroad (Kollmair et al. 2006);
- 6. this data discrepancy has meant an over-estimation of women that were eligible to participate in the study.

4.3.8 Qualitative methods

The semi-structured interviews and focus groups were conducted with the assistance of a Nepali translator and a maternal health qualitative researcher (Pitchforth & van Teijlingen 2005; Pitchforth & van Teijlingen 2006). The three

facilitators (the author, the maternal health researcher and the male interviewer) used a semi-structured topic schedule, developed in English and informed, in part, by themes emerging from the literature, (Sections 1.3.2, 1.4.3. and 2.6.1) around barriers from women's perspectives and discussions within the research team. The interview schedules also took on board the initial review of the quantitative analysis.

The interview and focus groups schedule were piloted on a number of Nepalese students at BU. Pilot studies can help: (a) identify practical problems in the research protocol; (b) develop and test the adequacy of research instruments; and (c) determine the feasibility of a full-scale study (van Teijlingen & Hundley 2001). Openended interviews were also conducted and translated into Nepali once the pilot had been conducted.

Focus group interviews took place in a neutral meeting place. They were conducted by the Nepali researcher, lasted no longer than 40 minutes, and were digitally recorded, with consent. Any identifiers were removed on transcription (Thomson et al. 2005). The qualitative open-ended interviews were conducted in Nepali with the help of two local translators (one female and one male) who were familiar with the subject matter, since cross-cultural qualitative research is difficult in a language other than the researcher's primary language and reliable and valid information can be "lost" when conducting the research. Furthermore, no standards exist for translation of qualitative research (Lopez et al. 2008). Pitchforth and van Teijlingen (2005) have discussed the challenges of language barriers and working with interpreters in qualitative research. They suggest that an effective relationship be developed with the interpreter and awareness be maintained of the interpreter's impact on the research process and how accurate the translation is. The final study's Nepali transcripts were then translated into English. Two students of Nepali origin from Bournemouth University individually transcribed four of the interviews.

This study (Section 3.6.1) aimed to explore the enablers and the barriers to the uptake of health promotion initiatives for improved pregnancy outcomes in rural Nepal and the influence of these barriers on maternal service uptake, especially in the rural areas. This study also aimed to analyse the improvement in the intervention area (in the last five years, since 2007) with a particular focus on change in health and maternal health behaviour and uptake, and what could have led to such improvements compared with the other/control VDCs. This might

elucidate health inequality for various reasons including those mentioned above, and provide an understanding of social complexities and changes since 2007 within the study areas.

Also included in the research was the women's ability to make healthcare decisions. This was taken as an indication of maternal autonomy. The factors that were taken into consideration were the women's previous experience, the choices they were given about place of delivery, the factors they considered when making their decision, and who or what had influenced the decision-making process. The questions were categorised into (i) women, (ii) women and husband together, and (iii) husband and others (family members). The focus group discussions were planned to examine the "group dynamics" within each group and between groups (i.e. daughters- and mothers-in-law, wives, or husbands) to identify the constructs that facilitate attending health services. See Appendix V for the interview schedules.

4.3.8.1 Interviews data collection

4.3.8.1.1 Focus groups

The primary method of data collection among the various sub-groups of the general public was conducting focus groups (van Teijlingen & Pitchforth 2006). The discussion of the focus groups should be in a safe and quiet "controlled environment". For instance, in Nepal, where women are not the main decision-makers for their reproductive health, it should mean a place where women can freely discuss these issues (van Teijlingen & Pitchforth 2006; Pitchforth et al. 2008). The interviews (ID) and focus groups (FG) typically lasted 40 minutes and were recorded with two digital recorders. The demographic data and recording was then double-checked and any last questions were answered. In total, 18 groups were approached to reflect the range of people (listed in Table 3), taking into account that not all of them may consent to be a part of the study.

Control area	Intervention area	Not eligible
2 groups of pregnant	2 groups of pregnant	1 group of women (15-
women;	women;	49) with no children
2 groups of women with a	2 groups of women with a	
child up to the age of 2;	child up to the age of 2;	
2 groups of women of	2 groups of women of	
childbearing age (18-49	childbearing age (18-49	
years), not pregnant;	years), not pregnant	
2 groups of mothers-in-	2 groups of mothers-in-law	
law;		
	1 group of men (married)	
1 group of men (married)	18-49 years	
18-49 years		

Table 3 Focus group target population

4.3.8.1.2 Semi-structured interviews

A semi-structured interview is described as a structured conversation. These interviews were driven by the questions set in this study that were used for FG (See Appendix V Annex-1: qualitative topic guide, Intervention area and Appendix V Annex-2: qualitative topic guide, Control area). Open-ended questions were created to investigate knowledge, attitude, and beliefs towards maternal care (See Appendix V, Annex-1: qualitative topic guide, Intervention area and Appendix V, Annex-2: qualitative topic guide, Control area). This part of the study ran for a one-month period (June 2012) with the analysis being conducted as the transcripts became available. Thus, the interviewer pursued in-depth information around the topic, and this was useful to further investigate their responses.

4.3.8.2 Qualitative data analysis

Description, field analysis & observation

Qualitative research can produce vast amounts of data (Pope et al. 2000), particularly here when the ID and FG were conducted in Nepali, and Newari (a

dialect of the Newar caste). The research anticipated for these two eventualities. In addition to the verbatim notes of the transcribed recordings of interviews or focus groups, more detailed "field notes" of observational research and the researcher's reflective notes (observations) were collected during the research as well as once the ID and FG were completed with the translator. As such, transcripts and notes are complementary to provide explanations since the researcher has to make sense of the data by sifting and interpreting them (Pope et al. 2000). Therefore, the data analysis was concomitant with data collection to continuously refine questions and pursue new themes (groups). In addition, because qualitative research uses analytical categories to describe and explain social phenomena, qualitative methods therefore need critical and creative thinking when conducting a study and interpreting its results in a balanced manner (Patton 2002).

Here, a qualitative approach was in addition necessary to ascertain whether there might be some information that requires an update - as the situation of the respondent may have changed since the last survey. Also, emerging insights, which could have been missed out by quantitative analysis alone, cannot be completed with qualitative methods. In general, qualitative research does not seek to quantify data or propose causality. In practical terms, the data was read and reread by two researchers to identify and index themes or categories, which may centre on particular phrases, incidents, or types of behaviour. The themes and categories were added to reflect as many of the nuances in the data as possible, rather than reducing the data to a few implicit or simple numerical codes (Mays & Pope 2000). Indeed, quantitative analysis provides a statistically representative set of respondents and may provide a useful summary of some aspects of the analysis. Yet, results in relative frequencies may be misleading, and simple counts are used (Mays and Pope 2000).

Usually in qualitative research analysis, the data is preserved in its textual form and "indexed" into analytical categories and theoretical explanations. Field notes and transcripts were used as textual data for content analysis in parallel with quantitative data since this permits the elucidation of those data parallel or opposite to the emerging hypotheses (Mays & Pope 2000). For instance, sometimes interesting or

unfamiliar terms used by the group studied can form the basis of analytical categories (groups). These discrete incidents may include multiple themes particular to rural Nepal, where new themes arise due to the new or unfamiliar context. With analytical and theoretical ideas developed during the research (using a conceptual framework, see Section 4.3.2), these categories were further refined into groups (Mays & Pope 2000). Thus the groups or patterns across the data sets are important to the description of the phenomenon (event or idea); and here they were associated with specific research questions (Section 3.7).

The transcripts were first coded by hand, using a form of inductive thematic analysis (Mason 2002; Forrest Keenan et al. 2005) to ensure the codes were agreed upon for all the interviews conducted by the research team (Section 4.3.8). Inductive thematic analysis typically involves six phases: familiarisation with data; generation of initial codes; searching for themes among codes; reviewing themes; defining and naming themes; and producing the final report (Thomas & Harden 2008). Therefore, all the data relevant to each category was then identified and examined using a process called constant comparison, in which each item was checked or compared with the rest of the data to establish analytical categories. This requires a coherent and systematic approach so that every sentence/line of text was coded once the key words, concepts/images came to light (Thomas & Harden 2008). This particular approach led to the elucidation of themes by a rigorous and systematic classification process of coding to identify patterns/codes that emphasise the reliability and replicability (meaningfulness) with these specific units of information. By using thematic analysis to "distil" data, first broad patterns are highlighted that permit "granular" themes to be elucidated, i.e. narrowed down in a more finegrained analysis. In practical terms, the pilot and initial interviews assisted in defining the categories or codes from broad patterns to fine-grains. In this type of analysis - the data itself is used to derive the structure of analysis; and guided by the conceptual framework (Section 4.3.2). In other words, the themes emerge from the data that is gathered and are not imposed or predetermined by the researcher. This type of analysis permits "flexibility" of analysis, and the themes are strongly linked to the data since they emerge from it – reducing bias or a limited interpretation of the data. This approach is comprehensive and therefore timeconsuming and is particularly useful when little is known about the event or topic under study.

131

Finally, the methods of analysis were discussed and the use of mixed methods was critically examined with regards to the qualitative approach. The software Nvivo[™] (Version 18) was for coding the qualitative analysis. Two team members independently analysed the transcripts to discuss the emerging analysis and major themes and ensure a degree of quality control. The team of three researchers coded all of the transcripts, using the thematic approach, independently. They then discussed with the researcher (SS) and agreed upon emergent themes and extracted quotations from the participants' transcripts to illustrate the themes (Chapter 6).

4.3.9 Quantitative study of this mixed-methods study

In order to conduct the evaluation of the GTN intervention and determine if it was effective and why, a quantitative analysis of the evaluation data and design was conducted. The quantitative study was a controlled before-after, cross-sectional, and non-randomised study. Women were interviewed using a questionnaire in the four VDCs, including measurements at three points in time, from the three surveys in 2007, 2010, and 2012 (van Teijlingen et al. 2012).

The baseline data collection began in 2007 (time 0 or T0) using the health behaviour and socio-economic questionnaire in the intervention and control area (Appendix III - Questionnaire for Women). Similarly, a midline (time 1 or T1, 2010) and a final round of data collection (time 2 or T2, 2012) used the same questionnaire to see if there was an increase in the uptake in maternal health services from T0 compared to T1, and then to T2. The notion of "time" (T) in the DiD method is referred to in the text as T1 or T2, and in the regression as 'after' (TA1) and 'afterafter' (TA2), respectively. At this point, the analyses of changes in health behaviour were done (Sections 4.3.9.1.4 and Table 4), using the variables of maternal health attendance: i) between the baseline and the midline; ii) between the baseline and the final; and iii) between the midline and the final. The inclusion of a midline survey as the data analysis from the midline permitted the researcher to determine the effects of time on the intervention. As discussed in Section 1.3, negative health outcomes such as mortality are fortunately very rare and therefore are hard to evaluate. This intervention focused on proxy outcomes, in this case nonhealth outcomes as they have an impact on obstetric (health) outcomes (Bhutta et al. 2005; UN 2011a; UN2011b). The outcomes were: (1) antenatal clinic (ANC) attendance at least once during the whole pregnancy and (2) during the first trimester and (3) number of ANC visits, (4) consumption of iron and folic acid during pregnancy, (5) presence of a skilled birth attendant at birth (SBA), (6) birthing in an "institution", that is a hospital or clinic (ID), and (7) attending postnatal care (PNC).

The evaluation aimed to capture positive "spillover" or herd effect since the programme can have an impact on not only women receiving directly the intervention but also the overall community (Baez 2007; White 2013; de Heer et al. 2011; Vanderweele et al. 2013). Therefore, women were surveyed independently from their participation in the intervention groups.

133

4.3.9.1 Data analysis: secondary analysis, GTN

Data was analysed with STATA[™] (version 11). The statistical value was calculated and tested for significance at the 5% level. The determined p-values were considered statistically significant when 'p' was equal to, or less, than 0.05. In statistics for the majority of analyses a value of 0.05 is used as the cut-off for significance of effect of the treatment/intervention. If the p-value is larger than 0.05, it cannot be concluded that a significant difference exists between the means (STATA 2017).

Descriptive statistics (i.e. frequency analysis) measured demographic, cultural and socio-economic characteristics such as caste and socio-economic indicators (wealth, age, and education, etc.) as well as chosen indicator/outcomes data (summarised as percentages). Analysis measured the impact or the individual probability of engaging in one of the intervention binary outcome variables (measured at baseline then at 30 and at 60 months after the intervention). Cross tabulation, the Chi-Square test (Pearson's Chi-Square test) and Difference-in-Differences (DiD) were applied for continuous and categorical data (binary or dichotomous variables) to assess the association between variables (detailed in the following section). For instance, the attendance data was presented in proportions, percentages, odds ratios, and confidence intervals (CI). The analysis at the level of the individual was done using logistic regression models, DiD analyses, on dependent variables. These dependent variables were adjusted for independent variables, as they have an impact on the selected outcomes.

4.3.9.1.2 Dependent & independent variables

The following variables were considered potential independent variables: age (continuous), wealth (categorical), parity (categorical), woman's education level (categorical), time, and intervention.

Binary dependent variables were the following outcomes: ANC attendance (1 visit, 4 visits, and 1 visit in the first trimester), examination, taking iron or folic acid, institutional delivery, skilled attendant at birth, and postnatal care.

4.3.9.1.3 Chi-square test

A Chi-square test was used to determine a relationship for selected confounding factors (covariates with an association) variables, for example, between taking part in the intervention and seeking maternal care. A confounding variable is a perceived relationship between an independent variable and a dependent variable that has been misestimated; this failure to account for a confounding factor is termed omitted-variable bias. They are factors that aim to make a link between attending health services and living in the intervention area. For example, in the case of researching/conducting risk assessments that evaluate the magnitude and nature of risk to human health, it is important to control for confounding factors, i.e. those covariates that have an association to the outcome of interest to isolate the effect of new "treatment", i.e. the intervention. For prospective studies, it is difficult to recruit and screen for volunteers with the same background (age, diet, education, geographical location, etc.); and in retrospective or historical studies, there can already exist similar variability. Due to the inability to control for variability of volunteers and human studies, confounding is a particularly big challenge. For these reasons, a Chi-square test is used to see if there is a relationship between two categorical variables, here the test was applied to check for an association. It is a two-tailed test with a 5% significance level. First, the association between ANC, ID, SBA at birth, iron/folic acid supplementation and the intervention were measured. Secondly, the association was tested between the intervention and categorical independent variables of interest (age, wealth, parity, education etc.). Both were determined by using first a Chi-square test (s²) (the BMJ 2015), then with the Difference-in-Differences analysis.

4.3.9.1.4 Difference-in-Differences analysis

Difference-in-Differences (DiD) addresses a gap in the evaluations of community maternal health promotion using longitudinal analysis on programmes with a control to measure intervention effect (impact) on health services uptake behaviour (Alderman et al. 2009). DiD is a technique often neglected in the evaluation of health promotion programmes. Therefore, it was applied as an evaluation tool (Bonell et al. 2011). DiD analysis measured the impact of the intervention on the individual probability of engaging in one of the intervention binary outcome variables chosen (Vyas & Kumaranayake 2006; Howe et al. 2008; Liu et al. 2010). DiD measured the difference in each outcome between intervention and control groups, and before and after treatment while controlling for potential confounding factors. Differences between outcomes were valued over three periods for two groups, at the baseline (2007), midline (2010), and final (2012) survey. One of the groups was exposed to a treatment (intervention) in the second and third period but not in the first period (baseline). The second (control) group was not exposed to the treatment during either period. With repeated cross-sections, the regression model with the intervention, time, and their interaction were determined. Control variables (detailed below on Page 136), in addition to the ones representing the impact of the intervention, were chosen based on the literature (Sections 1.4 and 2.6) and previously published DHS data based studies (MOHP, New ERA & ICF International 2012).

For a binary outcome variable of maternal health uptake behaviour, the DiD estimate is the difference in 2010 (midline or TA1) and 2012 (final or TA2) in changes from 2007 (baseline or T0) in the proportion of women having an event, i.e. attending health services – this is denoted as a 1. Consequently, the zero refers to not attending health services (Table 4). The DiD estimator (*treatafter or TA*) is the coefficient of the interaction term between the intervention and time in a linear regression model with intervention, time, and their interaction as covariates (Liu et al. 2010). Here, the estimator is presented as an odds ratio (OR). In essence, the *treatafter* estimator represents the difference between the pre- and post-intervention respondents' differences in the treatment and control groups (or T0, TA1 and TA2).

<i>treatafter</i> (μ) coefficient – post-intervention for outcome variables	2007 (Baseline, T0)	After 2010 (Midline, TA1)	Afterafter 2012 (Final, TA2)
Intervention (I)	0	1	1
Control (C)	0	0	0

Table 4 DiD Estimation

The analysis first measured the impact of the intervention on the binary outcome variables, which were first summarised by the percentage of women who had an outcome (maternal health uptake). Then, chi-square tests and bivariate regressions followed to explore the determinants of the indicators of interest/independent variables (time, area, age, education, parity, and wealth). Only the variables that had a significant relationship with the response variables at the P < 0.05 level were then entered as independent variables to be included in each final regression model. The criterion for removal in the regression analysis was P > 0.05. The DiD multivariate regressions were then applied to determine the factors that were most strongly correlated with the outcomes of interests presented as DiD (OR). Thus, the DiD estimation was used to assess the effects of intervention on the outcome variables (for example, the number of ANC visits) while controlling for the following covariates: socio-economic and other personal characteristics such as parity, age, wealth, and level of education women in the framework of the linear regression model (Appendix IV – Variables Description).

Marital status and religion were not adjusted for, since all the women in the sample were married and Hindu. There was no need to adjust for the *Aama Surakchhya* maternity incentive programme as it was operating in both the intervention and control areas.

The DiD estimation strategy can be used to analyse these cross-sectional data even though they are not repeated observations. Had they been repeated observations, i.e. balanced panel data, a correction for correlations around the same unit of observations would be needed here. One of the limitations of DID is when something other than the treatment (intervention) changes in one group but not the other at the same time as the treatment (Bertrand et al. 2004; Imbens & Wooldridge 2007).

With the aim of evaluating the intervention at two points in time, two different types of regressions were estimated: (a) regressions on the sample constituted by women interviewed at baseline and at midline; (b) regressions on the sample constituted by all women in the sample, including baseline, midline, and final evaluation. The former (point - a) permitted the evaluation of the impact of the intervention after 2.5 years and the latter (point - b) the evaluation of the overall impact of the intervention after 5 years from the start. The two regressions were run at both time points (midline and final) and not as one regression with one time point, since time was treated as a continuous variable in the regression. For instance, what is captured in the second regression is an overall effect of the intervention from baseline to the final. Because the intervention is not run in stages or steps, the evaluation cannot be conducted as distinct points in time (T0, T1 and T2).

The linear regressions from baseline to midline were the following:

 $\begin{aligned} Y_{ni} &= r_0 + \beta_1 \text{ intervention}_i + t_2 \text{ after}_i + \beta_3 \text{ after}^* \text{intervention}_i + e_4 \text{ age} + \beta_5 \text{ age}^2_{i+1} \\ \beta_6 \text{ wealth index}_i + \beta_7 \text{ education}_i + c_8 \text{ parity}_i \end{aligned}$ [1]

The regressions from baseline to final evaluation were the following:

 $\begin{aligned} Y_{ni} &= r_9 + \beta_{10} \text{ intervention}_i + t_{11} \text{ afterafter}_i + \beta_{12} \text{ afterafter}^* \text{intervention}_i + e_{13} \text{ age} \\ &+ \beta_{14} \text{ age}^2_{i+} \beta_{15} \text{ wealth index}_i + \beta_{16} \text{ education}_i + c_{17} \text{ parity}_i \\ \end{aligned}$

In the above, i indicates women participating in the surveys. Since there were seven outcome variables, many regression analyses were necessary with the combined dependent variable. The attendance variables taken as Y_n were the binary response variables. The binary outcome (dependent) variables (n) were the following:

(n=1): antenatal clinic attendance at least once during the whole pregnancy;

(n=2): antenatal clinical attendance during the first trimester;

(n=3): at least four antenatal care (ANC) visits during pregnancy. The WHO recommends a minimum of four ANC visits and that the first ANC visit should be within the first trimester of pregnancy (AbouZahr & Wardlaw 2003);

(n=4): the presence of a skilled birth attendant (SBA) at the time of delivery. SBA in Nepal are defined as nurse-midwives, auxiliary nurse-midwives, obstetricians and gynaecologists. The following groups were excluded: traditional birth attendants, health attendants, medical students since they are not classified by the WHO as SBA (WHO 2004b);

(n=5): institutional delivery (ID), including delivery in a hospital, primary health centre, private hospital or clinic. This was chosen as an outcome because it is recognised as a strategy to improve maternal and child health outcomes (Kesterton et al. 2010; Asefa et al. 2013; Kestler et al. 2013);

(n=6): attending postnatal care (PNC). PNC was defined as the mother and newborn being seen within 24 hours of delivery. This outcome was included based on the evidence that 60% of maternal deaths in the low and middle-income countries occur postpartum (Middleberg 2003; WHO 2013; Li et al. 2014);

(n=7): taking iron or folic acid. In Nepal, iron and folic acid supplementation is provided at government health facilities throughout the country (MOHP 2012) as a measure to prevent anaemia and neural tube defects (WHO 2012b).

Control variables were:

Intervention = {equations [1] and [2]} denoted the observations of two groups: intervention and control.

After, or A1 = {equation [1]} denoted time (treated as categories), before (baseline or T0), and after (T1 and T2) the intervention started.

After*intervention, or TA1 = {equation [1]} is the variable that identifies the group of observations belonging to the intervention group after the intervention started as compared to the remaining observations (namely all the observations belonging to the control group and the observations of the intervention group before the programme started). Its estimated coefficient, $\beta_{3 hat}$ represents the impact of the intervention.

Afterafter, or A2 = {equation [2]} identifies the observations collected both at the midline, and in the final evaluation as compared to baseline.

Afterafter*intervention or TA2 = {equation [2]} was the variable that identified the group of observations belonging to the intervention group at the midline (=1) and at the final (=2) evaluation as opposed to the baseline and to all the observations in the control group at any time (=0). Its estimated coefficient, $\beta_{12 hat}$ represented the impact of the overall intervention.

Age = {equations [1] and [2]} was a continuous variable representing the age of the individual at that point in time.

Wealth index = {equations [1] and [2]} was a categorical variable extracted from a series of assets owned (details on the construction are in the next section). The inclusion of this variable aimed at testing the hypothesis of attendance depending on the women's socio-economic status (Tuntiseranee et al. 1999; Simkhada et al. 2008; Ahmed et al. 2010). This variable was included instead of caste in force of a statistically significant relationship between the two variables (Pearson chi2(14) = 326.15; Pr = 0.000).

Education = {equations [1] and [2]} was a categorical variable indicating women's level of education and taking values 0 =none, 1 =primary school and lower, 2 =secondary school or higher.

Parity = {equations [1] and [2]} takes values 1 if the women are primapara, 2 if they have two children, or 3 (or more) if they are multipara.

4.3.9.1.5 Wealth index construction

DiD estimation was used to assess the effects of intervention on the outcome variables while controlling for a constructed wealth index and other personal characteristics, such as parity, age, and level of education (Vyas & Kumaranayake 2006; Howe et al. 2008; Liu et al. 2010). Since attendance depends on women's socio-economic status ("wealth") (Tuntiseranee et al. 1999; Ahmed et al. 2010; van Teijlingen et al. 2012), a wealth index was constructed using Principal Component Analysis (PCA). Using PCA women were "placed" into socio-economic tertiles (not weighed according to a standardised socio-economic index). PCA is commonly used to construct socio-economic indices when household expenditure or income data are not available (Filmer & Pritchett 2001).

PCA can be used to create a wealth index score for each respondent (Vyas & Kumaranayake 2006). It was constructed here using a number of variables such as assets owned by women's families (Vyas & Kumaranayake 2006; Howe et al. 2008). The method used to construct the PCA was based on the World Bank (1994) paper written by Filmer and Pritchett (2001) on how to construct socio-economic indices on non-expenditure data. The methodology can be used as a proxy for expenditure (Filmer & Pritchett 2001). In this evaluation, the marital status variable was not used since all females in the study were married, as expected in Nepal. PCA assets for the wealth variable construct were the following household assets/components: 1) dummy variables (bicycle, motorcycle, goat, land, and car), and 2) categorical variables - type of access to hygienic facilities (sources of drinking water, types of toilet), number of rooms in the dwelling, and construction materials used in the dwelling (materials used for flooring, walls, and roofing),

(Pitchforth et al. 2007). Land was also included. Land in Nepal is measured in "ropanis" and one ropani in the hill area is 0.05 hectares. The respondents' land ownership was distributed as wealthier for those with more than 3 ropanis (3 ropanis representing the median of surveyed population), and poor for those with less than 3 ropanis. A description of variables included in the PCA is provided in the Table S1 in the Appendix (Appendix IV – Variables Description).

In STATA[™], the combination of the variables produced a wealth index score. The first component extracted explained 20% of total variability: a potential difference explained in the sample 1/5 of the differences seen. The scores based on the first component were grouped into tertiles, with the lowest (Group 1) representing the poorest and the highest (Group 3) representing the richest women in the sample. This score was then used to divide the respondents into wealth categories (from low to high) for inclusion in the regression analysis.

4.3.9.1.6 Caste

Caste and the constructed wealth index were compared to ensure they are comparable. First, caste/ethnicity was classified as (i) relatively advantaged -Brahmin, Chhetri, Thakuri, Gurung and Newar; (ii) relatively disadvantaged -Janjati including indigenous groups; and (iii) relatively disadvantaged - Dalit, the lowest caste (or untouchable) (Section 1.4). Among all caste groups, Dalits have traditionally experienced high levels of social exclusion and marginalisation in Nepal (Khanal, Sauer, et al. 2013b). Secondly, the distribution was split according to high, middle, and low castes, according to the published definition of caste (MOHP 2012, Government of Nepal; National Planning Commission Secretariat 2014b). Distributions per caste are high caste (=1): Brahmin, Chhetri, Newar; middle caste (=2): Tamang; or low (=3): Newar Dalit, Balami, Dalit, and others (Christian or Muslim). Finally, caste was compared with the wealth index to ensure that they were correlated using Chi-square (χ^2) (see Section 4.3.9.1.3) and then presented visually as histograms in Figures 16, 17, 18 and 19 (Section 5.8.1). Wealth was strongly correlated to caste (χ^2 = 383.0, p<0.05) and the latter was therefore not included in the regression as a more precise proxy (score) than caste.

4.3.9.1.7 Cost analysis extrapolation

A cost analysis was conducted as outlined in Section 3.4. It has been shown that community mobilisation through women's groups is a highly cost-effective and an affordable strategy to reduce maternal and neonatal mortality (Prost et al. 2013). Combining community mobilisation with quality improvement of health facilities is more effective and expensive, but also highly cost-effective and potentially affordable in this LMICs context (Colbourn et al. 2015). For any intervention aimed at improving maternal healthcare, it is important to know whether it is scalable and cost effective (Ensor et al. 2014). Therefore, the effectiveness of community mobilisation through women's groups to use and understand the health facility quality plays a role in reducing maternal morbidity and mortality. In order to conduct effective and cost-effective analysis, health outcomes are needed. In maternal health, they are typically taken as maternal mortality, prevalence of low-birth weight, neonatal mortality, infant mortality, and any kind of medical complication during delivery. In the GTN datasets, there were no data on these health outcomes. Thus as cost-effectiveness analysis needs such outcomes, a cost analysis was conducted. The difference between the two types of analyses is that costeffectiveness measures outcomes against inputs (money, staff or resources), whilst cost analysis calculates the cost of the implementing and running of the intervention (Bhattacharya et al. 2013). The cost data were collected from hand-written and computer records from the office of GTN and its field workers. The substantive cost data of the intervention were "cleaned," i.e. categorised and allocated to two cost centres: a) implementation costs, and b) programme running costs. This is necessary to be able to allocate the accumulated cost data to the appropriate cost centre. The cost centres had missing data. In order to extrapolate for the missing costs, the mean costs were calculated for each year using the recorded months' average. This monthly average was then multiplied by the number of missing months and the total was then added to the recorded months' total. The annual costs were entered by "year" as defined by GTN. The cost data was converted from Nepali rupees (NRs.) to British pounds (GBP) for this U.K. university thesis using the median of the annual exchange rate and for the baseline, midline and final surveys the conversion rates for each survey year were the following, (Oanda 2015):

Average/Median 2007, GBP 1=NRs. 130.51 Average/Median 2010, GBP 1=NRs. 111.63 Average/Median 2012, GBP 1=NRs. 137.35 Finally, the GTN intervention cost on likelihood of attendance is ascertained by dividing the total cost by a DiD proportion change for a population of 8,569

accounting for the spillover effect (Section 2.2).

4.4 Validity, reliability and goodness of fit of study methods

Validity and reliability in research refer specifically to the measurement of data that will be used to answer the research question; the collected data can only be useful (reliable) if it is measured through a good instrument such a survey or interview schedule which is designed based on evidence, previous studies and is pilot-tested (Ryan et al. 2001). Validity can be internal or external. Internal validity relates to conclusions warranted from the observations (data), and external validity refers to the replicability/generalisability of a study (Clancy 2002; Steckler & McLeroy 2008). According to Ryan and colleagues (2001) validity in qualitative research involves determining the degree to which the researcher's knowledge matches the reality. Reliability, on other hand, relates to credibility, trustworthiness, consistency and dependability (of the data). Reliability is important in research because it ensures the researcher's confidence in the measure taken. Similarly, validity is important because it tells the researcher that the chosen measure will measure what it is supposed to and not something else (Ryan et al. 2001). With regard to this study, the researcher ensured the validity and reliability of the data collected. For the qualitative study, first the validity of the methods was ensured via constructive feedback from experts of Bournemouth University and who had experience and expertise in maternal health. The methodology was revised and improved according to the advice given and suggestions made. The reliability of the instrument was improved through piloting. The qualitative approach was used to collect the primary data through interviews. The Ph.D. student first piloted the interview/FG schedules then conducted all face-to-face interviews and then data was analysed (by two researchers) to see whether or not this technique is reliable to answer the research question. The qualitative data was analysed using a thematic approach by two researchers to minimise bias and to ensure the reliability of data (Forrest Keenan et

144
al. 2005). The two researchers compared notes and agreed (or disagreed) on themes that arose from the interviews and focus groups to ensure consistency.

For the quantitative component, a pre-test was conducted by GTN among nine women who have a child of less than two years. The women were interviewed in both the control and intervention area. The main purpose of the pre-test questionnaire was to find out its appropriateness, obtain clarity and determine the length of time needed to complete the questionnaires. Some corrections to the questions were made after the pilot study and inappropriate questions were excluded. Second, the investigation method was chosen based on the literature which suggested that DiD is highly suitable for cross-sectional data (Section 4.3.9.1.4).

In addition, for DiD, predicted probabilities were used to ensure the "goodness of fit" of the DiD regressions (Section 5.10). Goodness of fit in logistic regression assesses how well a model "fits" the data. It is applied once a "final regression model" has been selected. Of course, any selected regression model aims to contribute towards final conclusions/inferences. For instance, predicting probabilities refers to measuring the "specificity" (i.e. how many true responses: 'Yes, I attended ANC = 1') of a diagnostic tool/regression model to detect positive (=1), and negative (=0) cases for predicting probabilities of attendance of maternal health services. The function 'predict' in STATA[™] is applied to see if the predicted association is random between the dependent variable (Y) and the intervention. We remind the reader that the dependent variable (Y) was classified as at least once attending ANC (=1) etc. Y, thus, is a predicted function (i.e. regressed) of being in the intervention/control area in function of time, age of woman (of the attendance variable), etc. Running a logit of a combination of these factors generates an OR using DiD. As stated previously, the chosen regression model DiD is applied in order to determine the impact of each factor (age, parity, education, and SES) on attendance likelihood (detailed as an OR), i.e. will an increase in age will have a consequence impact on the OR in question. A goodness of fit models was applied, as errors exist. In the regression, there is an error term (E). The inclusion of E accounts for those factors that cannot be included in a regression or those that have been but have a degree of error, e.g. confounders and those included variables that are measured as proxies. For example, all the independent variables such as the variable wealth are constructed as a proxy, so there may be missing variables/information or the latter

145

is a proxy based on assumptions or how the data is design/assigned (Sections 4.3.9.1.4 and 4.4). Therefore when the DiD model is applied, it calculates the 'fitted' value of the regression. In order to classify, a cut-off value was chosen on the probability scale, e.g. 50%, this helps classify all predicted values above that as a 'predicting' an event, and all below that cut-off value as not 'predicting' the event (i.e. attendance).

4.5 Ethical Considerations in mixed-methods

Ethics are essential to ensure the rights of participants are maintained (Orb et al. 2001). Furthermore, ethical approval in developing countries is necessary. It raises issues of registering health research, protecting participants and their privacy (Clinton 2010; van Teijlingen et al. 2012). The ethical approval letter can be found in Appendix I. Ethical considerations in qualitative research are complex since some issues in qualitative interviews are unique (Corti et al. 2000). First, the attempt must be made, at all times, to guarantee promises of confidentiality made to research participants, where possible. For example, if data files are stored in a secure manner in archived data, it helps maintain the informed consent agreements for confidentiality purposes.

The most important ethical issue in both quantitative and qualitative research is informed consent - that must be obtained from the interviewee after they have been carefully and truthfully informed about the nature of the research. Consent needs to be obtained while confidentiality is ensured since the respondent's anonymity needs protection, and their identity or any other personal information will be excluded from the research (Shaw 2003). Prior to the collection of data for the project, authorisation was sought from the Nepal Health Research Council (NHRC), relevant VDCs local authorities, and the ethics committee at Bournemouth University for the fieldwork, survey data collection and qualitative in-depth interviews. The study was approved by the NHRC on 1st August 2011 (Reg. No. 37/2011), and by the Bournemouth University Ethics Committee on September 13, 2011. For the quantitative study, informed consent was obtained from each individual participant, and participants were guaranteed anonymity and confidentiality and assured that they could withdraw, if they so wanted. All names and other forms of personal identification were omitted in all reporting. The survey, semi-structured, and focus

146

group interviews took place in a neutral meeting place to guarantee the aforementioned.

In practical terms for the qualitative study, the study was explained and participants were asked to sign an informed consent form. The consent process was explicitly and clearly detailed in Nepali. Once the anonymity and confidentiality was discussed, the semi-structured interviews were conducted in participants' homes, open fields, or the village health post. Participants could withdraw from the study at any time. The interviews and focus groups lasted approximately 40 minutes and were digitally recorded with permission. Afterwards, they were transcribed and translated. Being a mix of female and male research team allowed for certain issues to be explored more in-depth with regards to maternal practices with the various participants (Sein 2013), and male participants felt comfortable being interviewed by a man. Participants were given the opportunity to ask the researcher questions. Accordingly, the researcher has not mentioned any of the personal information of the interviewees. Finally, topics such as stillbirth or death of a child or power relationships (such as a male relative coming to listen in) might arise. Thus, respondents were reminded that they could stop the interview at any time (van Teijlingen et al. 2006). Finally, the rich nature of qualitative data lends itself to descriptions of the interviewees, their lives and their surroundings, and as such dilemmas are presented to the researcher in how much detail to reveal, for example, how to completely disguise a workplace or a village for confidentiality reasons. The situation may become more complex if the data is re-used. Therefore, anonymisation must be stringent so that respondents are suitably protected using qualitative data archivists (Corti et al. 2000). The qualitative data was accordingly protected, the tapes were kept securely, and the transcripts were not identifiable by participants' names.

In this thesis, in the sequential design, the findings are presented in Chapter 5 and 6; the data was integrated in Chapter 7 for discussion.

4.6 Summary

This methods chapter discussed the following:

- the research methodology and the subsequent methods used in the impact evaluation research;
- the difference between qualitative and quantitative research;
- the choice of the qualitative approach as a tool for primary data collection;
- the quantitative approach as a tool for secondary data collection and;
- the way the data was "mixed" taking into account issues of validity, reliability and ethics.

Chapter 5 Quantitative study findings

5.1 Introduction

This chapter presents the results of the study and describes the impact of the intervention on the utilisation of antenatal care, delivery care, and postnatal care among the rural women in Nepal. The relationship between socio-demographic and socio-economic variables and the epidemiological data analysis of the baseline (2007), midline (2010), and final (2012) datasets are presented. The validity and reliability of the regression analysis and cost data of the intervention are also presented.

5.2 Findings

A total of 1,236 women (611 in the control and 625 in the intervention area) completed the surveys, with an overall average response rate of 84% (Table 5).

Demographic and Health Survey data from the Central hill area are similar to the study site where the median age at first marriage was 19.4 and where there was a rising age of marriage in the country. With regards to women having their first child the median age was 21. Furthermore, literacy rates were steadily increasing (MOHP 2012). In the GTN study, a large majority were Hindu, as is typically representative of Nepal (in both areas over the past 5 years it was 71.50%). There were some similarities with the DHS survey. However, as DHS is in selected parts of Nepal and no individual areas are identified in the survey, DHS data is not divided by district, hence the need for a control group (Section 4.3.7).

Survey (year)	Baseline (2007)	Midline (2010)	Final (2012)	
Control area	204	204	203	
Intervention area	208	217	200	
Total (n= 1236)	412	421	403	

Table 5 Number of surveyed women with last child <2 years by area and year

5.3 Socio-demographic & socio-economic characteristics of samples

At the baseline, the control and intervention samples were similar in terms of age, religion, and ethnicity (Table 6). In addition, the baseline characteristics (wealth and maternal health indicators) were not statistically different between the intervention and the control district; therefore the groups can be said to be comparable (Table 6). The majority of women were from the Tamang caste (40.67% overall). The minimum age was 15 years and the maximum age was 49 years. The overall mean age of respondents was 25.35 ± 5.08 years, the mean age of marriage 19.59 ± 3.33 years, and the mean age of first pregnancy was 20.86 ± 3.27 years. At baseline, the majority of women in both groups were married before the age of 20; this proportion dropped to 41.38% in the control group and 48.50% in the intervention group in the five years period. The reported age of first pregnancy was lowest at 14 and the oldest reported age was 49. The main occupation of respondents was either housewife or farmer (89.56%). Most of the women were multigravida (56.07%) and the second largest group was primigravida (43.93%).

As religion, caste/ethnicity, age of 1st marriage, age of 1st pregnancy, literacy, and education was not significantly different based on the control and intervention sites. The choice of study sites and the survey data on these variables confirmed that the control and the intervention group women were similar at baseline (Table 6).

		Control Intervention							Difference in baseline characteristics between control and intervention area	
Characteristic used as denominator	n	Baseline 2007	Midline 2010	Final 2012	p- value*	Baseline 2007	Midline 2010	Final 2012	p- value*	p-value*
Religion	1236		%		0.001		%		0.004	Pearson chi2 = 1.1 p = 0.8
Buddhist	323	34.80	18.14	31.03		31.73	22.58	18.50		
Hindu	884	62.25	80.39	67.98		66.35	74.65	77.50		
Other (Christian, Muslim)	29	2.94	1.47	0.98		1.92	2.76	4.00		
Caste/Ethnicity	1236				0.21				0.003	Pearson chi2 = 5.9
Brahmin	323	10.29	10.78	6.90		19.23	13.82	13.50		
Chhetri	187	20.59	18.14	17.24		14.90	7.83	12.50		
Tamang	504	38.24	35.78	38.42		39.90	51.15	40.50		
Newar non Dalit	258	25.49	27.45	26.11		14.42	13.36	19.00		
Newar Dalit	20	1.47	2.45	2.46		1.44	0.46	1.50		
Dalit	29	1.47	1.96	2.46		1.92	3.23	3.00		
Balami	39	0	0	0		6.25	5.53	7.00		
Other (Gurung etc.)	45	2.45	3.43	6.40		1.92	4.61	3.00		
Age of marriage (yrs ¹)	1236				0.01				0.01	Pearson chi2 = 25.4 Pr = 0.2
10-19	648	50.00	53.94	41.38		60.58	60.37	48.50		
20-24	493	38.42	40.69	45.32		37.02	35.02	43.50		
25-29	82	10.78	5.39	10.34		1.92	4.15	7.50		
30 and above	13	0.98	0.98	2.96		0.48	0.46	0.50		

Table 6 Characteristics of the Control and Intervention area

Age of 1 st pregnancy	1236				0.01				0.0001	Pearson chi2 = 24.9 p = 0.2
14-19	461	37.75	37.44	27.59		51.44	37.79	31.50		
20-24	619	47.06	51.96	52.71		42.79	52.53	53.50		
25-29	132	13.73	8.82	15.27		4.81	8.76	13.00		
30 and above	24	1.47	1.96	4.43		0.96	0.92	2.00		
Literacy	1236	64.22	76.96	70.44	0.01	66.35	73.27	81.00	0.0037	Pearson chi2 = 0.2 p = 0.6
Education	1236				0.12				0.0025	Pearson chi2 = 0.7 p= 0.7
None	424	43.63	31.37	33.00		39.90	33.18	24.50		
Primary	448	28.43	41.67	34.48		31.73	42.86	38.00		
Secondary and higher	364	27.94	26.96	32.51		28.37	23.96	37.50		

¹ Yrs – Years, *p-values are based on Kruskal Wallis test to compare each categorical variable across time.

5.4 Household decision on the utilisation of maternal care

Women were asked who made the decision with regards to them obtaining care (Table 7). In terms of decision-making, the proportion of women who reported to have autonomy on maternity care and place of delivery decisions increased over the duration of the study period. The increasing trend was for all the family members to make the decision about ANC, place of delivery, and healthcare jointly. In both areas, the patterns for decision-making in seeking ANC and delivery care were similar (Table 7). Finally, the majority of women said that their most recent pregnancies were planned.

		(Control			Int			
Outcome	n	Baseline 2007	Midline 2010	Final 2012	p- value*	Baseline 2007	Midline 2010	Final 2012	p- value*
			%				%		
Planned pregnancy		69.10	81.30	82.20	0.5	74.5	73.73	86.50	0.0001
Decision- maker	1090				0.1				0.002
ANC ¹					0.1				0.0001
Myself	518	39.02	41.38	54.1		39.20	58.50	50.00	
Husband	346	41.10	37.36	25.3		37.50	25.71	25.51	
Mother-in-law	85	13.41	6.30	3.5		15.34	6.67	2.55	
All Family Members/Jointly	141	6.10	14.93	17.0		7.95	9.05	21.90	
Place of delivery	1232				0.4				0.0001
Myself	569	42.16	42.16	52.00		42.31	58.06	39.70	
Husband	378	30.39	39.71	25.50		36.06	25.81	26.63	
Mother-In- Law/Grand	138	24.02	8.82	4.00		16.35	9.68	4.02	
Mother-in-law Father-in-law	141	2.94	-	-		1.44	0.92	-	
All Family Members/Jointly	6	0.49	9.31	18.50		3.85	5.53	29.65	
Healthcare in family	1235				0.002				0.6
Myself	459	38.24	24.51	36.14		37.02	41.94	45.00	
Husband	440	38.24	44.12	32.18		41.35	32.72	25.00	
Mother-in-Law/ Grandmother- In-Law	141	17.16	6.37	6.44		19.71	14.29	4.00	
Father-in-law	65	5.88	15.20	2.97		0.96	5.07	1.50	
All Family Members/Jointly	130	0.49	9.80	22.28		0.96	5.99	24.50	

Table 7 Decision-maker for healthcare in the household: percentagechanges over time & area

¹ANC-Antenatal care, **p-value*** are based on Kruskal Wallis test to compare each categorical variable across time.

Table 8 reports the details of household assets used to construct the wealth index. The majority of respondents lived in their own dwelling on their own land and had their own piped or common piped water and a flush toilet. Most respondents owned electric goods such as a television and radio as they had electricity. Few respondents owned a fridge. With regards to the road access, the majority of women lived less than a 5-minute walk away from a road. There was no data on the distance to the facility. By 2010 and 2012, all respondents owned one or more goats, toilets, land, and piped water; perhaps an indication of the cohort getting richer.

Table 8 Household Assets of Respondents

		Co	ontrol		Intervention			
Household assets	n	Baseline 2007	Midline 2010	Final 2012	Baseline 2007	Midline 2010	Final 2012	
Own House	1236	94.61	91.67	87.19	90.38	92.63	94.00	
Dwelling roof	1236	28.92	25.00	27.09	34.62	35.48	49.00	
Water	1212							
Own piped	531	43.14	36.76	42.36	48.08	36.87	51.00	
Non piped source	72	9.31	6.37	9.36	4.83	1.39	4.02	
Common pipe	609	48.0	48.5	49.8	48.1	55.3	45.5	
Toilet	881							
Flush Toilet	739	50.00	62.75	62.56	58.17	58.53	67.00	
Pit Latrine	142	22.06	8.82	5.91	11.06	12.90	8.00	
More costly items								
Radio, cassette tape	1236	77.94	76.47	61.08	78.37	68.66	58.00	
Television	1236	81.47	79.41	80.79	77.88	73.73	79.00	
Fridge	1236	8.33	15.20	16.26	6.25	11.98	17.50	
Electricity	1236	98.04	99.02	96.55	95.67	96.77	93.5	
Bicycle/rickshaw	1236	4.90	9.80	4.93	1.92	6.91	4.00	
Ownership of agricultural land	1236							
Own less than 3 ropani (<0.27 hectares)	577	46.57	53.92	34.48	44.23	53.00	47.50	
^Median is = 3 Ropani								
Own between 3-20 ropani (0.27-0.54 hectares	659	53.43	46.08	65.52	55.77	47.00	52.50	
Road <5 minutes away	1236	52.45	91.18	70.44	74.52	94.01	78.00	
Goat	1236	55.88	100.00	100.00	44.71	100.00	100.00	

*1 ropani = 0.05 ha (Hills)

5.5 Confounding factors

A Pearson's Chi-square test was performed to explore whether living in the area where the intervention took place was a confounding factor using the baseline survey data. In this analysis, there is no statistically significant relationship between living in the intervention area and (a) seeking ANC, (b) seeking ANC in the first trimester, (c) having four ANC visits, (d) having an institutional delivery, (e) seeking a skilled birth attendant, (f) attending PNC, (g) taking iron and folic acid, (h) age, (i) wealth, (j) parity, (k) education, and (l) time (Table 9).

	Intervention								
Outcome	N	Baseline (2007) X ²	Р						
Seeking ANC ¹									
At least once	412	1.27	0.26						
In the 1st Trimester	334	2.09	0.15						
4 or more visits	412	2.83	0.09						
Taking Iron/Folic Acid during									
pregnancy	331	0.57	0.45						
SBA ⁴	412	0.75	0.39						
ID ³	412	1.14	0.29						
Seeking PNC ²	408	2.83	0.06						
Age	412	31.23	0.26						
Wealth	412	0.50	0.79						
Parity	412	1.05	0.59						
Education	412	0.72	0.70						
Time	412	0.09	0.76						

 Table 9 Correlation of intervention outcomes and living in the intervention area at baseline

¹ ANC - Antenatal care; ² PNC – Postnatal care; ³ ID – Institutional delivery; ⁴SBA – Skilled birth attendant

*Significance is at p<0.05

5.6 Utilisation of maternal care

In Table 10, at the baseline, 80.39% of women in the control area attended ANC compared to 84.62% in the intervention area. In the intervention area, from baseline to final evaluation, the proportion of women who sought ANC at least once significantly increased from 84.62% to 98.00%. The proportion seeking ANC in the first trimester significantly increased from 47.70% to 62.37%; those seeking ANC four or more times significantly increased from 67.31% to 81.00%. In addition, a greater proportion of women reported taking iron/folic acid (from 86.54% to 95.98%) and seeing a SBA (from 60.60% to 82.00%). Significant increases were also seen in seeking an institutional delivery (from 60.58% to 76.00%) and PNC (from 52.20% to 85.86%). Use of safe delivery kit significantly increases from 5.00% to 34.29%. Improvements were also registered in the control group but not all were significant.

Table 10 Maternal Health Uptake (%) of health services in the intervention and
control area

Contro			ol	Intervention			ntion		
Outcome	n	Baseline 2007	Midline 2010	Final 2012	p-value	Baseline 2007	Midline 2010	Final 2012	p- value
			%				%		
Seeking ANC ¹									
At least once	1236	80.39	85.29	88.67	0.06	84.62	96.77	98.00	0.0001
In the 1st Trimester	1086	55.62	68.32	61.10	0.05	47.70	61.17	62.37	0.007
4 or more visits	1236	59.31	64.22	69.95	0.08	67.31	86.18	81.00	0.0001
Iron/Folic Acid during pregnancy	1236	76.35	79.90	79.31	0.6	86.54	94.47	95.98	0.0006
SBA ⁴	1236	55.39	63.24	75.37	0.0007	60.60	70.05	82.00	0.0001
Institutional birth ³	1234	55.39	54.68	71.43	0.0006	60.58	66.67	76.00	0.003
Seeking PNC ²	1234	42.86	61.76	73.82	0.0001	52.20	76.85	85.86	0.0001
Use of safe delivery kit	388	11.49	17.11	11.63	0.5	5.00	40.30	34.29	0.0001

¹ ANC - Antenatal care; ² PNC – Postnatal care; ³ ID – Institutional delivery; ⁴SBA – Skilled birth attendant *significance is at p<0.05

5.7 Impact of the intervention

Tables 11 and 12 show the estimated odd ratios for the midline and overall evaluations, respectively. From baseline to the midline, there is a significant increase women's likelihood of attending ANC at least once during their pregnancies [OR=6.96, 95%CI (2.26; 21.39)] in the intervention group. A significant increase was also seen in the probability of taking iron/folic acid [OR=3.03, 95%CI (1.16; 7.87)]. The probability of seeking four or more antenatal care visits was doubled, [OR=2.15, 95%CI (0.99; 4.67)] (Table 11).

Over the five years (from baseline to the final term), women were three times more likely to seek ANC at least once [OR=2.97, 95%CI (1.52; 5.81)]. Women were nearly twice as likely [OR=1.89, 95% CI (1.12; 3.18)] to take iron/folic acid, and one and a half times as likely to attend for postnatal care [OR=1.50, 95% CI (1.05; 2.15)].

Over the five years, the intervention had no impact with regards to women attending ANC in the first trimester, or on SBA at birth, or on ID either in the midline, or in the overall evaluation. Impact was seen on four ANC visits only from the baseline to the midline. Three of the outcomes improved (as seen in Table 12) that were clinically relevant (but not statistically significant); there were an increase in the proportion of women with a prenatal visit in the first trimester (47.70% to 62.37%), increases in institutional deliveries (60.58% to 76.00%) and the proportion of women who had a skilled attendant at birth (60.60% to 82.00%) in the intervention area (Table 10).

	Seeking ANC ¹ at least once	Seeking ANC in the 1st Trimester	Seeking ANC 4 or more times	Taking Iron/Folic Acid during pregnancy	SBA ²	ID ³	Seeking PNC ⁴
Observations	832	714	832	831 832		830	832
Treat	1.33 (0.72; 2.44)	0.67 (0.42; 1.07)	1.21 (0.72; 2.04)	2.26 (1.24; 4.09)**	1.19 (0.74; 1.90)	1.24 (0.79; 1.96)	1.46 (0.94; 2.27)
After	1.16 (0.61; 2.18)	1.59 (0.98; 2.57)	0.87 (0.52; 1.45)	0.99 (0.56; 1.73)	1.26 (0.78; 2.03)	0.80 (0.50; 1.27)	2.28 (1.45; 3.56)**
Treat-after	6.96 (2.26; 21.39)**	1.22 (0.63; 2.35)	2.15 (0.99; 4.67)*	3.03 (1.16; 7.87)**	1.45 (0.74; 2.83)	1.72 (0.90; 3.30)	1.64 (0.86; 3.12)
Wealth							
Wealth 2	3.38 (1.91; 5.99)**	1.65 (1.11; 2.46)**	1.24 (0.81; 1.90)	2.59 (1.54; 4.36)**	2.69 (1.85; 3.90)**	2.21 (1.53; 3.20)**	1.70 (1.17; 2.47)**
Wealth 3	5.98 (2.27; 15.73)**	3.05 (1.91; 4.87)**	5.24 (2.55; 10.76)**	2.78 (1.33; 5.82)**	11.03 (6.27; 19.39)**	7.59 (4.55; 12.66)**	4.01 (2.49; 6.44)**
Age	0.90 (0.87; 0.95)**	1.01 (0.97; 1.05)	1.00 (0.96; 1.04)	0.92 (0.89; 0.96)**	0.98 (0.95; 1.02)**	0.98 (0.95; 1.01)	0.99 (0.96; 1.02)
Education							
Education 2	5.21 (2.69; 10.07)**	1.53 (1.03; 2.29)**	0.95 (0.61; 1.48)	3.98 (2.28; 6.95)**	1.95 (1.33; 2.86)**	1.95 (1.33; 2.85)**	2.05 (1.40; 3.00)**
Education 3	9.34 (3.11; 28.04)**	2.58 (1.60; 4.16)	2.01 (1.04; 3.87)**	10.10 (3.94; 25.84)**	4.73 (2.76; 8.08)**	4.28 (2.59; 7.07)**	4.47 (2.75; 7.26)**
Parity							
Parity 2	0.67 (0.37; 1.23)	0.62 (0.43; 0.91)**	0.57 (0.36; 0.90)**	0.53 (0.30; 0.92)**	0.52 (0.35; 0.77)**	0.53 (0.36; 0.78)**	0.65 (0.45; 0.95)**
Parity 3	0.48 (0.25; 0.93)**	0.56 (0.35; 0.89)**	0.64 (0.37; 1.10)	0.31 (0.17; 0.57)**	0.62 (0.39; 1.00)*	0.59 (0.37; 0.93)**	0.75 (0.48; 1.17)

Table 11 DiD analysis of maternal health uptake at midline evaluation (OR & CI)

¹ ANC - Antenatal care; ² SBA – Skilled birth attendant; ³ ID – Institutional delivery; ⁴PNC – Postnatal care

***pvalue*<0.05; Afterafter=at 5 years of the intervention, treatafterafter= those exposed to the intervention in 2012, treat=intervention area, wealth 2=middle-wealthy and wealth 3=the wealthiest.

	Seeking ANC ¹ at least once	Seeking ANC in the 1st Trimester	Seeking ANC 4 or more times	Taking Iron/Folic Acid during pregnancy	SBA ²	ID ³	Seeking PNC ⁴
Observations	1235	1086	1235	1233	1235	1233	1235
Treat	1.49 (0.84; 2.64)	0.68 (0.45; 1.03)	1.48 (0.91; 2.40)	2.41 (1.38; 4.18)**	1.32 (0.86; 2.03)	1.52 (1.01; 2.31)**	1.51 (1.00; 2.26)**
Afterafter	1.18 (0.85; 1.62)	1.02 (0.81; 1.30)	1.01 (0.77; 1.31)	0.87 (0.66; 1.15)	1.34 (1.05; 1.72)**	1.26 (0.99; 1.59)**	1.75 (1.39; 2.21)**
Treat-afterafter	2.97 (1.52; 5.81)**	1.21 (0.88; 1.67)	1.06 (0.71; 1.58)	1.89 (1.12; 3.18)**	1.04 (0.73; 1.48)	0.93 (0.66; 1.30)	1.50 (1.05; 2.15)**
Wealth							
Wealth 2	2.49 (1.55; 4.00)**	1.75 (1.27; 2.42)**	1.37 (0.95; 1.96)	2.63 (1.71; 4.04)**	2.55 (1.86; 3.49)**	2.20 (1.62; 3.00)**	1.68 (1.22; 2.31)**
Wealth 3	4.74 (2.15; 10.44)**	2.80 (1.93; 4.06)**	3.33 (2.00; 5.56)**	2.37 (1.35; 4.15)**	9.29 (5.88; 14.68)**	6.97 (4.59; 10.57)**	3.82 (2.55; 5.74)**
Age	0.93 (0.91; 0.97)**	1.00 (0.97; 1.03)	1.00 (0.97; 1.03)	0.96 (0.93; 0.99)**	0.99 (0.96; 1.02)	0.99 (0.96; 1.02)	1.00 (0.97; 1.03)
Education							
Education 2	4.69 (2.72; 8.09)**	1.35 (0.97; 1.87)	1.04 (0.71; 1.51)	3.31 (2.13; 5.13)**	1.90 (1.37; 2.63)**	1.88 (1.37; 2.58)**	2.23 (1.61; 3.08)**
Education 3	11.02 (4.18; 29.04)**	2.24 (1.53; 3.27)**	1.80 (1.09; 2.98)**	9.27 (4.55; 18.90)**	4.55 (2.91; 7.10)**	3.97 (2.63; 5.99)**	4.70 (3.10; 7.13)**
Parity							
Parity 2	0.61 (0.37; 1.00)*	0.72 (0.54; 0.96)**	0.72 (0.51; 1.04)	0.55 (0.36; 0.85)**	0.45 (0.32; 0.62)**	0.48 (0.35; 0.65)**	0.56 (0.41; 0.76)**
Parity 3	0.44 (0.25; 0.78)	0.64 (0.44; 0.94)**	0.69 (0.44; 1.09)	0.32 (0.19; 0.54)**	0.50 (0.33; 0.75)**	0.52 (0.35; 0.77)**	0.66 (0.44; 0.98)**

Table 12 DiD analysis of maternal health uptake at final evaluation (OR & CI)

¹ ANC - Antenatal care; ² SBA – Skilled birth attendant; ³ ID – Institutional delivery; ⁴PNC – Postnatal care

***pvalue*<0.05; Afterafter=at 5 years of the intervention, treatafterafter= those exposed to the intervention in 2012, treat=intervention area, wealth 2=middle-wealthy and wealth 3=the wealthiest

5.8 Impact of covariates on outcomes

5.8.1 Wealth and caste

Caste was first separated into three categories (Section 4.3.9.1.6); Caste 1 = 48.46%; Caste 2 = 40.78%, and Caste 3 = 10.76% (Table 13). Wealth was a significant factor; it explains a high proportion of the variation in all the outcomes both in the midline and in the overall evaluation. In particular, being richer (3^{rd} tertile) compared to being poorer (1^{st} tertile) substantially increased the probability of having an SBA in the midline [(OR=11.03, 95% CI (6.27; 19.39)] and in the overall evaluation [OR=9.29, 95% CI (5.88; 14.68)].

Wealth was also strongly correlated to caste ($\chi^2 = 440.19$, p<0.05).

Caste	n (1236)	%
1	599	48.46
2	504	40.78
3	133	10.76

Table 13 Caste distribution

Wealth quantiles	n (1210)	Overall (%)	2007 (%)	2010 (%)	2012 (%)
1	404	33.39	40.63	36.52	22.11
2	403	33.31	34.06	33.89	31.84
3	403	33.31	25.30	29.59	46.05

Table 14 Wealth distribution

Relative to the construction of the wealth index, the first component (roofing material) extracted explained 20% of total variability in the population (Section 4.3.9.1.5). The distribution of the wealth score based on the first component is shown in Figure 16 (n=1236). Scores based on the first component were grouped into tertiles, with the lowest (group 1) representing the poorest and the higher (group 3) representing the richest women in the sample. The share of women falling into each tertiles were: Wealth 1 = 40.63%, 36.52%, 22.11%; Wealth 2 = 34.06%, 33.89%, 31.84%; Wealth 3 = 25.30%, 29.59%, 46.05% in the first, second, and third survey, respectively (Table 14).



Figure 16 Wealth distribution by caste

The distribution of wealth was also 'split' according to caste (Figure 16) or to high (Figure 17), middle (Figure 18) and low castes (Figure 19), as per the published definition of caste (Ministry of Health and Population 2012; Government of Nepal, National Planning Commission Secretariat 2014b). Distributions per caste are 'high' caste (=1): Brahmin, Chhetri, Newar, 'middle' caste (=2): Tamang middle, or low (=3): Newar Dalit, Balami, Dalit, and others (Christian or Muslim) (Section 4.3.9.1.6). As seen in Figure 17, the wealth is better distributed, the higher the caste.



Figure 17 Wealth distribution in high castes



Figure 18 Wealth distribution in middle castes



Figure 19 Wealth index distribution among lower castes

5.8.2 Age, education & time

Age was a significant factor in determining whether women sought one antenatal visit and took iron/folic acid both at the midline and final evaluation. In both cases, older age lowered the probability of a positive outcome. Having higher education compared to no education increased the probability of a better attendance of all the outcomes considered. In particular, having secondary school or higher-level education increased the probability of attending ANC at least once at the midline [OR=9.34, 95% CI (3.11; 28.04)] and the overall evaluation [OR=11.02, 95% CI (4.18; 29.04)].

In the intervention area, women were more predisposed to seek an ID [OR=1.52, 95% CI (1.01; 2.31)] and PNC [OR=1.51, 95% CI (1.00; 2.26)] at any time. Women in the intervention area were 2.26 [OR=2.3, 95% CI (1.24; 4.09)] times more likely at midline to take iron/folic and 2.41 [OR=2.41, 95% CI (1.38; 4.18)] by year 5. Over time (variable *afterafter*, see Table 12), women become increasingly more likely to take up SBA at birth [OR=1.34, 95% CI (1.05; 1.72)], ID [OR=1.26, 95% CI (0.99; 1.59)] and PNC [OR=1.75, 95%CI (1.39; 2.21)], reflecting background changes. With increasing parity, the ORs for all indicators remain significantly below 1, when referring to a reduced likelihood of attendance the more children a woman has.

5.9 Continuity of care

The uptake of ANC, ANC in the first trimester, and 4 ANC was significantly related to having a skilled attendant at birth (Table 15).

SBA	ANC	ANC in the 1st Trimester	4 ANC
Observations	807	530	689
Percentage of uptake	73.36%	81.79%	78.12%
X ²	170.55	60.38	165.62
	p<0.05*	p<0.05*	P<0.05*

Table 15 SBA uptake post-ANC

5.10 Validity, reliability & goodness of fit using predicted probabilities

In the quantitative component of this evaluation, the method of investigation was chosen based on the literature that DiD is suitable for cross-sectional data. Thus ensuring validity and reliability of the DiD regression analysis (Section 4.4). Secondly, the use of predicted probabilities was used to ensure the 'goodness of fit' of the DiD regressions (Section 4.4). The use of predicted probabilities helps us understand the regression model. The beta value probabilities were calculated to ensure that the regressions applied in the DiD analysis were similar to the actual attendance range (Table 16). For example, an example from the table below can be interpreted as: the predicted probabilities model suggests that the probability of seeking ANC once in the control area was 87.19% (with a range of 85% to 88%), while the actual attendance was 88.67%. As stated in Section 4.4, the calculated value for predicted values, with error or the error term (E), above were 'predicting' an event and all below that cut-off value as not 'predicting' the event (i.e. attendance). This exercise is a measure of how well the modelled probabilities fit, with error as not all confounding factors have been taking into account, as the "actual" proportions of the percentage uptake of health services in 2012 match are similar to the predicted probabilities.

168

Table 16 Predicted probabilities of DiD analysis of maternal health uptake changes in intervention & control area, finalevaluation (%)

	Seeking ANC ¹ at least once	Seeking ANC in 1st trimester	Seeking ANC 4 or more times	Taking Iron/Folic Acid during pregnancy	SBA ²	ID ³	Seeking PNC
Uptake of Health Services Control 2012 (%)	88.67	54.19	72.41	73.82	75.37	71.43	79.31
2012 Control Predicted probability, % of women (with a lower and upper cut off)	87.19 (0.85-0.88) (85-88%)	62.42 (0.61-0.64) (61-64%)	81.28 (0.80-0.82) (80-82%)	79.80 (0.77-0.82) (77-82%)	68.71 (0.66 -0.71) (66-71%)	63.16 (0.61-0.65) (61-65%)	63.25 (0.64-0.68) (64-68%)
Uptake of Health Services Intervention 2012 (%)	98.00	60.50	86.00	95.98	82.00	76.00	85.86
2012 Intervention Predicted probability, % of women (with a lower and upper cut off)	97.36 (0.96-0.97) (96-97%)	61.14 (0.59-0.62) (59-62%)	88.72 (0.88-0.89) (88-89%)	95.20 (0.94-0.95) (94-95%)	75.05 (0.73 – 0.77) (73-77%)	71.17 (0.69-0.73) (69-73%)	81.77 (0.80-0.83) (80-83%)

5.11 The cost of the intervention

This section presents the cost analysis of running women's groups. The intervention lasted 60 months, and nine months were considered the implementation period. Starting with the cost of the surveys, the baseline cost per survey was NRs. 180,000 (GBP 1,379.20). This included enumerator allowance, subsistence transport, salary printing, questionnaires, mobile phones, and the cost of training (U.K.). The second survey was NRs.188, 000 (GBP 1,684.10) and the third survey NRs. 194,000 (GBP 1,412.40) which was the most expensive in Nepali rupees due to the inclusion of this PhD thesis research (Table 18). The disparity in GBP is due to the fluctuation in exchange rate over the years (Section 4.3.9.1.7). The three surveys totalled NRs. 562,000 (GBP 4,091.70). The GTN intervention cost included start-up costs and recurrent costs. The former were for recruiting and office rental. The main recurrent costs were for training, group activities (incentives), and transportation.

The total cost of implementing (and running) the intervention was NRs. 5,210,974 (GBP 42, 950.19) over 60 months (Table 17). The evaluation total was GBP 17,986.48 (Table 18). The total cost of the intervention and the evaluation was GBP 60,936.67. The cost per year was NRs. 1,042,194.80 (GBP 8590.38). Table 17 shows that the cost increased in the fourth and five year, which can partly be explained by the inflation in Nepal and largely by the increased numbers of groups running (Section 4.3.9.1.7).

Year	Description	Months	Nepali Rupee (NPR)	1 GBP=NPR	GBP
1	1 st January 2007 to 29 th February 2008	14	753410	127.66	5901.69
2	1 st March 2008 to 31 st July 2009	17	764772	128.47	5952.92
3	1 st August 2009 to 28 th February 2010	7	487384	126.24	3860.77
4	1 st March 2010 to 30 th April 2011	14	1776452.00	116.04	15308.96
5	1 st May 2011 to 1 st January 2012	8	1428956	119.82	11925.85
For total of the intervention (60 months)	1 st January 2007 to 31 st December 2012	60	5210974		42950.19
Per year (Considering 12 months of the intervention	Yearly average		1042194.80		8590.38
Per month (Considering 60 months of intervention)	Each month (average)		86849.57		715.84

Table 17 Cost of the implementation of the intervention

Items	Details	Nepalese rupees	GBP
Total cost of Surveys 3 rounds	Breakdown of surveys cost: enumerator allowance, subsistence Transport, Salary Printing questionnaires, Mobile phone 3*2 months of costs for surveys	180000 (June 2017)	1379.20*
month each in both control /intervention (transport, subsistence, photocopy)		188000 (June 2010)	1684.10*
photosopy)		194000 (June 2012)	1412.40*
Total		562000 (June 2012)	4091.70*
Incentives, gifts and assistance	Baby blanket, stretcher, sustenance, transports, stationary, communication, bed and stretcher for clinic 422 blankets * NR300	126600	1023.86**
	Per respondent subsistence NRs. 20*731 people=NPRs.14, 620	24720	199.92**
Cumulative Total		165881.74	1341.54**
Training and flights	External trainers 3 Flights (1000 each) + GBP 1000 per day (5 days)		8000.00
	Needs assessment - intervention area Translator, subsistence (NRs. 20,000 converted using June 2017 exchange rate)		153.24*
	(2 MSc students, 1 PhD student) Flights (3) @1000		3000.00
	Mid-term qualitative evaluation, that inform post-midline HP (S, B &S) 2 qualitative interviewers		1400.00
Total			12553.24
Evaluation total			17986.48

Table 18 Cost of the evaluation of the intervention

* Exchange Rates: June 2007, GBP 1=NRs. 130.51, June 2010, GBP 1=NRs. 111.63 and June 2012, GBP 1=NRs. 137.3.

** Exchange Rate on date of entry into GTN budget, GBP 1= NRs. 123.65

Considering the total recorded attendees in the groups by year 5 were 731 women, the cost per person (attendee in a women's group) was GBP 11.75 over the five years (60 months).

Finally, the GTN intervention cost on likelihood of attendance is ascertained by dividing the total cost by a DiD proportion change for a population of 8,569. For example, taking the outcome attending ANC (at least once), the cost per percentage point increase, i.e. to increase ANC uptake by 1% in population from the baseline to the final evaluation, a 13.38% increase was seen in the intervention area (Section 5.6, Table 10) and the cost over 5 years was GBP 5746.73. For the DiD analysis for the cost for a 12.74% likelihood of increase over 5 years, it was GBP 5,471.85 (Table 19).

Outcome Attending ANC at least once	Over 5 years: Baseline to Final in a population of VDC 8,569	GBP	
% Change	13.38	5746.73	
% Change due to DiD	12.74	5471.85	
Total cost per year	GBP 42950.19		

Table 19 Cost of the likelihood of ANC uptake for the population in the intervention area

5.12 Summary

This chapter has discussed the research findings of the quantitative research. There were increases in the attendance indicators for both areas. The use of DiD and PCA on the secondary survey data permitted the approximation of the impact of the intervention on attendance outcomes. They showed that certain ANC outcomes and PNC attendance improved compared to delivery care outcomes as a result of the intervention. The DiD "goodness of fit" test was presented. Finally, this chapter also presented the cost of the intervention. The chapter that follows presents the qualitative findings.

Chapter 6: Qualitative study and findings

6.1 Introduction to the qualitative findings

This chapter presents the qualitative findings of this study. First, the description of the characteristics of the focus groups and interview respondents are detailed. Secondly, the themes of the focus groups, interviews, and key informant interviews are presented. Third, the themes of the focus groups and interviews, as well as the focus group thematic analysis and findings of the study population, are presented with quotations extracted from participants' transcripts to illustrate context and themes. Finally, a reflective section is offered.

6.2 Participants' characteristics

A total of 14 focus groups (FG) were conducted ranging from 3-12 participants. Two sets of qualitative data from the key informants (the two health promoters) and from the participants were separately analysed, as they are separate perspectives of the intervention. They included nine groups of women aged 17 to 62 (separated into groups of recent mothers and mother-in-laws to extract authentic results); two groups of men aged 35 to 66; two groups of female community health volunteers (FCHV) aged 32 to 36 and the other aged 26 to 48; and one group of maternal and child health workers (MCH) (Table 20). All of the men and two-thirds of the women were literate, at least at a primary school level. All participants were married. Additionally, three individual interviews and three joint interviews were conducted with the two health promoters, aged 30 and 40, and seven health workers aged 25 to 52 (Table 20). The total number of participants in the focus groups was 38 individuals, and 9 individuals participated in the interviews (Table 21).

Focus group (FG) number	Group Interviewed	Age	Intervention or Control Area
FG1	Mothers with children <u><</u> 24	25-35	
	months		
FG2	months	21-28	I
FG3	Mothers with children <u><</u> 24 months	17- 23	I
FG4	Mother-in-law groups	37-55	I
FG5	Mother-in-law groups	40-62	I
FG6	Mixed type of Mothers having either <u><</u> 24 months children and non GTN members	26-48	I
FG7	Female Community Health Volunteer (FCHV)	26-48	I
FG8	FCHV	32-36	С
FG9	Mother-in-law groups	55-60	С
FG10	Mothers with children <u><</u> 24 months	22-28	С
FG11	Mothers with children <u><</u> 24 months	28-34	С
FG12	Maternal and Child Health workers (MCH)	24 and 31	I
FG13	Extended family groups (Husbands/Father-in-law) Group Interview	29-47	I
FG14	Extended family groups (Husbands/Father-in-law) Group Interview	46-66	С

Table 20 Focus Groups

Interview number	Group Interviewed	Age	Intervention or Control Area
IDI1	Hospital staff (2)	25 and 28	I
IDI2	Health worker, Sub- Health Post In-charge.	40	I
IDI3	MCH worker (Outreach clinic)		С
IDI4	Health promoters GTN staff (2)	30 and 40	I
IDI5	Health workers, PHN and MCHW (2)	42 and 52	С
IDI6	FCHV	42 and 52	С

Table 21 Interviews

From the designed sampling frame (Section 4.3.8.1), six interviews and fourteen focus group discussions were conducted with an independent Nepalese facilitator and interpreter in order to explore (1) changes in the last five years, (2) the reasons underlying the (non)-attendance of women at the Green Tara Nepal (GTN) groups (barriers and facilitators to uptake of services), and (3) users' perceptions regarding the effect of the intervention. The aims of the qualitative study were to document changes starting in 2007 up until 2012 in both the intervention and control areas, and to determine whether such changes are attributable to said intervention (Section 3.6.1).

When discussing changes in the last five years, the author distinguishes between the control or intervention area unless otherwise noted. In this chapter, the Tamang caste (as well as the caste system in general) has been discussed in detail in order to add useful information/'colour' to the quotes. Information provided by the interviewees and group participants in villages where the Tamang dominate, and might be useful for the reader to have/to process before reading the sections.

The findings are presented in two sections (6.2.1 and 6.3), one from health promoters (key informants) and the other from the participants (service users), because participants in each are very different, i.e. it is expected they would

respond very differently and would have very different perspectives on the issues, if only because of expert and lay distinction. A further reflective section considers other possible drivers of the changes observed.

6.2.1 Health promoters' evaluation, key informants

Part of the evaluation involved in-depth discussions with the health promoters (HPs). Below is a summary of the interviews and the observations on the HPs' activities in the field. The first health promoter is 26 years old. She is an auxiliary nurse-midwife (ANM). Her colleague (the second health promoter, 36 years old) reported that she brought 12 years of work experience as a community medical assistant (CMA, the equivalent of a health visitor in the NHS). Both health promoters are *Bahun Brahmin* and live in the same community where they work. They said that GTN wanted to work with health promoters "locally", as a person from the same area would be more likely to be trusted by their own community and were more likely to stay for the duration of the intervention.

During the interview and observations, the HPs mentioned that they were initially uncertain about how to organise the groups for pregnant mothers and new mothers in 2007, as the concept of groups where participants "just talked" was perceived as "foreign/odd". They recalled that the women who were targeted did the majority of the house and fieldwork and did not have the "time" nor the permission from their family, particularly their mothers-in-law, to attend group meetings. They emphasised that the groups were mixed: mothers-in-law with their daughters-in-law. Yet the mothers-in-law were one of the main barriers to access to care. Therefore, they decided to separate the groups once "permission" was obtained from the mothers-in-law. However, even at the midline evaluation in 2010, the pregnant mothers and new mothers reported that sometimes the mothers-in-law. It seems that this notion of "control" still existed.

When interviewed, the staff reported that they started the groups by focusing on the health needs of the community, in other words, by addressing the needs of mothers and informing them of the care of their newborns. This led to behavioural change via the health promotion of women's groups. The staff addressed this health promotion education in areas where it was not demonstrated by the mothers-in-law and with particular emphasis on the pregnant women and mothers.

179

Since 2007, they sought out and informed all pregnant women and mothers in the intervention area to attend the GTN women's groups. They felt that there was resistance to form and attend groups, as the maternal health promotion group concept was alien to them. Thus, they kept returning to the area, providing tea, biscuits, and blankets as incentives to attend. An interesting approach to engage participants was to "gift" a baby blanket to women who attended four group meetings while pregnant. The reasoning was to prevent hypothermia at birth and to keep the child warm.

The staff commented that the groups were formed and continued only once the community trusted them and due to the incentives. Trust, they said, was gained due to their continuous presence, the fact that they were local, and the fact that they were qualified as ANMs. They recalled that in 2009 (almost two years after the start of the intervention) women in the antenatal period were not always included, so they sought antenatal women and invited them to join the GTN groups. Furthermore, the staff said that they felt they required additional training on intimate partner and sexual violence. The GTN HP staff seemed proactive in their approach, even raising funds for GTN:

"We celebrated "[Teej]", and raised 50/60 thousand rupees." ID4, GTN staff, Bahun.

In order to improve attendance, a new strategy was developed by GTN and the HPs. They formed separate groups for mothers-in-law, pregnant women, and first-time mothers with children aged less than 2 years. By year three, 23 new groups were formed, and monthly meetings were held with each of these groups. Mothers with children more than 2 years old were excluded (in order to avoid recall bias, see Sections 1.2 and 4.3.4.2). The health promoters were able to recruit women who met the recruitment criteria but for example 'older' women (i.e. mothers-in-law) and men were hard to recruit:

"Old women and men missed in group activities. We were not interested in them either; they were not our target group." **ID4, GTN staff, Bahun.**
Also hard to reach were those who lived in rental houses and those who were Newari (higher caste). In these cases, GTN got help from FCHVs, who would visit the houses.

"Those who are staying in rental houses, such as wives of policemen and typical Newar in Pharping, were difficult to trap for group discussion. In that situation, we do get help from FCHV. They visit them at home and women listened to them. Anyway, we felt our message reached the target group (laugh)..." ID4, GTN staff, Bahun.

The HPs stated that people do not attend if they are busy. If the target group of women did not listen to the HPs, the health promoters would also ask FCHV to encourage them to attend groups. Similarly, the HPs enlisted help from FCHV for those who rented houses and were mobile.

Each new topic for discussion encouraged the women to participate in the group, for example the importance of iron, folic acid, danger signs during pregnancy, and childbirth and breastfeeding. During the intervention, they did not feel that they had to cancel scheduled meetings because of absent members. Some groups have members coming in and going out and occasionally, only one person may be left at a group session. They conduct meetings at convenient times for members during their hectic days. They do conduct meetings both in the evening and in the morning for the convenience of the participants. Initially, tea was offered but most had had tea at home. After a few months, they gave a small amount of cash or a small gift equivalent to an ear pick, brush, soap, etc., in each group sitting:

"That was very nominal (the gift-giving); the most important things were that they were happy in each meeting and that they learned something new each time..." **ID4, GTN staff, Bahun.**

Recently, the HPs reported that the groups became difficult to manage:

"In the past, we accepted all people who came to the groups. It was difficult for us later when many people came to the groups... (the groups were) uncontrollable." **ID4, GTN staff, Bahun.** They did state that during the 5 years, they were not able to form many men's groups. They reported that men "ran" away when they approached or stated that they felt this intervention was more suitable for women:

"We generally do not know men (to come) but they recognise us as "GTN ko didiharu" (GTN Staff)." **ID4, GTN staff, Bahun.**

They felt that the curriculum and this evaluation could be a monitoring tool for future GTN work and future training. The GTN staff kept their own Monitoring and Evaluation (M&E) data. In addition, engaging adolescents would be valuable, and having a Tamang health promoter would be beneficial as they are the dominating group in the area (Table 6). Yet the workload proved to be problematic due to the weather, the volume of meetings, and the unstructured form of some groups:

"Sometime, it was challenging to work in the field. Sometime, we forgot what we discussed in a given meeting as there were lot of meetings." **ID4, GTN staff, Bahun.**

They discussed progress, problems, planning, implementation, and evaluation and the next women and men's group meeting agenda:

"It (conducting groups, doing the health promotion) became easy as we both have a medical background. We teach new topics in the groups after discussing (between two of us) what to teach, and we created a curriculum with the other GTN staff." **ID4, GTN staff, Bahun.**

Yet there were challenges such as the weather, the workload, and members missing sessions. The latter was used as an opportunity to have group members discuss the health promotion activities:

"The problem was they do not come on time for meetings. Then we had to repeat the discussions...we (thought to) ask old members to speak and share what we had previously said to late comers to the meetings instead of repeating ourselves." **ID4, GTN staff, Bahun.**

The HPs reported that almost all new mothers attended GTN. Especially once learning that they are pregnant, women came to ANC clinics and for check-ups.

They found that the women came by their own choice to learn about health; they were eager to obtain information and pregnancy check-ups, too.

"It has been a good trend that new mothers attend the GTN groups. In the past, there were few who participated in the group meetings." **ID4, GTN staff, Bahun.**

When there were a lot of members in the group, they were sometimes not able to address all members since it became very large. They asked one member from each household:

"Now that so many people come to the group discussions, we have difficulty controlling the groups. Now, we have two separate groups, one being mothers with children under 2 years...they all come to the groups..." **R2, GTN staff, Bahun.**

By June 2012, a total of 58 women's groups were running. Two groups had stopped as they only had one or two attendees, and home visits were conducted instead. Home visits were also done for those who could not attend, as they did not have permission from the mothers-in-law or could not take time away from the family, household tasks, or fieldwork. Finally, for those women who were housebound or bedbound during their pregnancy (due to anaemia, hypertension, or poor nutritional status) or postpartum (due to anaemia, sepsis or recovery from caesarean section) home visits were also provided.

B.1 Health promoter conducting focus groups

An interesting validity/reliability exercise of the interview schedule was to ask the HPs to conduct using the qualitative questionnaire the same interviews with the groups they ran, to see if the answers would be pleasing for example. The answers on transcription were very didactic and to the point (short or one-word answers). The HP interviewer felt shy about being recorded as she felt she was being observed. The HP for the majority of the FG answered some of the questions herself as she felt she knew the right answers already or knew her group. They kept repeating the questions. It felt forced, directive, and one-sided:

I: "You have attended meetings so you should tell ANC, MH, and PNC about hand-washing, hygiene, (and) sanitation, problems."

Participant: nod, no answer. Balami and Chhetri mothers, I.

I: "What education have you received (reads list to participant)" Participant: nod, no answer. **Balami and Chhetri mothers, I.**

6.3 Focus group findings

According to the FG participants, GTN health promoters (HPs) conduct an average of four meetings per month. During the meetings, the groups were encouraged to interact with their community to identify barriers to accessing health services and then to develop ways of addressing them. Using approaches in the groups (Sections 2.4 and 2.5.2), the HPs presented and discussed the barriers and corresponding strategies to deal with "problems" with support from the community members. The groups then developed plans to address problems with the aid of discussion and a training manual for the duration of an hour. Meetings were limited to an hour because participants had to return to their field or house work. The participants neither commented, argued with, nor gave specific feedback with respect to what was said during the FG; it seemed that they accepted what was discussed in the GTN group discussions. However, all participants irrespective of area did not seem to be open to discussing abortion (results not presented). They were also initially reticent to speak of "traditional practices" for fear of being judged; however, as the interviews and focus groups progressed, they became more talkative. During focus groups and interviews, participants were generally shy and somewhat reticent to share. They may not have felt entirely comfortable during the interviews and FGs.

However, the moderator tried to encourage participation in the discussion during the interview:

"Yes. Now I would like others to speak. Why isn't anyone else talking?" FG3, Balami and Chhetri mothers, I.

6.4 Thematic analysis of the study population

The themes were generated using a thematic approach (Section 4.3.8.2). This approach refers to examining and recording recurrent patterns or "themes" within the transcripts/data by coding. With specific research questions (Section 3.7) – this particular approach permits the elucidation of themes by a rigorous classification process of coding to identify patterns and codes (nodes). In practical terms, previously defined categories or codes from the pilot and initial interviews were used to classify the content into explicative themes. Using inductive thematic analysis meant the interview/FG data were read by two researchers to identify emerging analysis of individual interviews and observations combined were categorised into three themes: 1) changes, 2) changes due to groups, and 3) barriers and limitations (i.e. barriers to attending groups and health services), then into subsequent 15 sub-themes, listed overleaf (Section 4.3.8.2). The transcripts were coded by the software Nvivo™. Figure 20 depicts an example of a transcript demonstrating the process of analysis using the software, first the transcript was read through and each idea/theme was coded/highlighted according the list of codes (nodes) on the following page. As mentioned in Section 4.3.8.2, each theme was identified using this constant comparison process, whereby each highlighted item was checked or compared with the rest of the data (interviews and focus groups) to establish analytical categories or themes coherently and systematically for all the data. In order to have/complete the final set of themes, the field notes, transcripts were used in parallel as textual data for content analysis. The interesting or unfamiliar terms used by the groups/participants were added to the list of refined themes - here they have been defined in parentheses, for example, Nwaran, a naming and purification ceremony.

186

Themes and sub-themes of study population

1. Changes

- a. Changes in the last five years in the area/community
- b. Changes in attitudes with respect to seeking care
- c. Perceived progress of maternal health
- **d.** Changes in attitudes towards motherhood and female children

2. Changes due to groups

- **a.** Groups working in the area
- **b.** Women and workload
- c. Shyness
- d. Men's involvement in maternal health
- e. Effects of the groups
- f. Health workers working with GTN
- g. Empowerment
- h. Changes in decision-making patterns
- i. Decision-making in PNC

3. Barriers and limitations to attending groups and health services

- a. Socio-economic barriers and limitations
- **b.** Cultural barriers

Home Create Data Analyze Q	uery Explore Layout View				Ψ
× = / 🖻 💑	ut (************************************		E D 🙆 Insert 🔻		
	opy	Reset Settings	Select Find		
w w w w w w w	erge 🔻		The select ring		
Item Clipboard	d Format	Paragraph Styles	Editing	Madified Dec	
SOURCES	Rarriers and limitations	1 5 7 Jul 2016 16:22	SS 13 Jul 2016 13:31	SS	
Externals	Change	1 1 7 Jul 2016 13:25	SS 7 Jul 2016 13:25	55	
Memos	Changes due to groups	0 0 7 Jul 2016 13:44	SS 7 Jul 2016 13:44	SS	
	Home birth	1 9 8 Jul 2016 11:35	SS 13 Jul 2016 12:30	SS	
Nodes	Interruptions	1 1 12 u 2016 01:18	SS 12 Jul 2016 01:19	SS	
Node Matrices	Nutrition	1 9 8 Jul 2016 11:44	SS 13 Jul 2016 13:16	SS	
	Suggestions for GTN cu	1 12 8 Jul 2016 12:13	SS 13 Jul 2016 13:27	SS	
CLASSIFICATIONS		-	_		
	FGD_Interview_compiled				🗌 Edit
D QUERIES	Respondents # 1, 3,	, 5, 2: we learned on uterus prolapsed,	how, what to do etc.		Cha Woi Shy Cul
	Respondents # 3, 4 5	5, 7: Also learned on importance of co	lostrums "bigaute dudha",		ness ricts ceive
	which increased immunity power of newly born baby, fight against disease. In the past, it				
	was thrown away by milking				
	Respondents # 2, 7, 8: we also discussed on family planning and it's temporary				kloa sion
	methods, depo, pills, it's advantages and disadvantages, exclusive breast feeding.				
					litern ding
	14 What barriers are there to women like you to attend sessions? (time				
	cost for transportation, had to work in the field, house and permission				
	from the family)				
	Respondents # 1, 2, 3, 5, 6: Not all came in GT group. About 60 percent did not join in				
	green tara. This village is big Bhattari villageupper village They say that they do				
	not have time, some did not show interest. Some said that "tato na saro" (they will not get				
	any benefit for spend	aing time in green tara).			stion Prog
					III Is fo an
	15. Are vou a	able to communicate freely and hon	estly with GTN staff? What		
OPEN ITEMS	is good, w	what is bad? How can GTN want to i	mprove?		
*	All: Yes we do not h	ave problem			rricu serv
FGD_Interview_compiled					- um
0	SOURCES 🕨 💼 Internals 🕨 📄 FGD_Inte	erview_compiled			

Figure 20 Example of NVivo analyses

6.4.1.1 Changes in the last five years in the area/community

As stated in the methods, Section 4.3.8.1, when asked what changes had occurred since 2007, participants spoke of construction of a water tank, construction of a road, and how road access has improved accessibility to the market in the intervention and the control area.

These changes have led to an increase in housing and roads:

"More houses have been built. There were no houses in this area (He points). Many "gumbas (monasteries)" have been built." **IDI, Hospital staff, I.**

In the control area, more houses were built and the population subsequently increased:

"The population has increased a lot. We have a lot of new houses, and the village, "Sankhu" has increased in size." **FG9**, **Newar mothers-in-law, C.**

Participants also mentioned that they now had fewer children. When discussing population changes in the last 5 years, the respondent stated that there was an increase in population due to migration into the community. Moreover, the community grew from the previously high birth rate, yet it seemed people now were choosing to have fewer children:

"The population has increased a lot in last five years. We have a lot of houses in the village. Now people do not want more children. People in our village have only 2-3 children." **FG1, Tamang, Older mother, I.**

6.4.1.2 Changes in attitudes with respect to seeking care

Respondents were then asked about any changes in their attitude towards seeking care for their delivery. Some respondents referred to the place of delivery. A mother-in-law respondent mentioned how there was a change in attitude in the place of delivery as the women gave birth at home:

"There are a lot of changes. I did not go to a hospital for delivery - I have four children, all were born at home (laughing)...My mother-in-law and neighbours

helped me during my delivery...Now we all go to the hospital for delivery." FG9, Mother-in-law, Newar, C.

A mother-in-law in the control area responded that while babies were delivered at home in the past, institutional deliveries were now more common due to an increase in awareness and the incentive scheme provided by the government:

"People go to the hospital for delivery (the hospital provides 1000 rupees for hospital delivery). In the past, all gave birth at home. People are conscious and do care of pregnant woman and after delivery." **FG11, Older mothers, Tamang, C.**

In the intervention area, a health worker mentioned that women seemed to come regularly to the family planning and antenatal maternal health clinics:

"... They're coming more often for their regular check-up, and they are more aware about their health than before." **IDI3, MCH health worker, Brahmin, I.**

A female participant responded that while she had given birth to all of her children at home, nowadays mothers went to the hospital as they had the funds to do so:

"I delivered 9 children at home itself. I feel today's women have more money, that's why they go to hospital." **FG4, Bahun/Chhetri, mother-in-law, I.**

However, the MCH worker interviewed stated that Tamang women preferred to deliver at home, due to feeling awkward or shy, and that the decision was made by family members, including the women themselves:

"When a woman is having labour pain in the Tamang community, the woman doesn't go to hospital immediately... She and her family just wait...they feel awkward going to a hospital. Women feel happy giving birth to children in home. The decision to remain at home is made by family members and even the pregnant woman." **ID3, MCH worker, Brahmin, I.**

6.4.1.3 Perceived progress of maternal health

6.4.1.3.1 Changes in attitudes towards motherhood and female children

Fertility was reported to have declined as attitudes had changed towards children and the sex of the child. Both family planning changing attitudes towards gender preference in children were mentioned:

"Couples are starting to get operated on for family planning (sterilisation) even if they have only one child. The majority of people in the village now do not continue trying for a son. In the past, people kept trying to conceive a son, and if they did not get son, they did not get family planning services (e.g. contraception)." **ID4, GTN staff, Bahun, I.**

A health worker commented on the fact that women were not having fewer children. They referred to past sentiments, when children were considered a "gift". However, there is now a preference for fewer children.

"But now they give birth to one or two children and visit the health post for family planning, which is now free. In earlier days, people used to give birth to many children thinking that they (children) are a gift from God's. In some houses, we would see 7 children, but now the scenario has changed." **ID3, MCH worker, Brahmin, I.**

With respect to attitudes towards the sex of the child, women previously felt happy to have become a mother, although they mentioned a preference for male children:

"Khusi lagchha ni" (It feels very good to be a new mother). I am happy to have a son." **FG2, Young mother, Tamang, I.**

Currently, other respondents (men, mothers, and mothers-in-law) mentioned that they felt no preference for either sex. In fact, they considered "preference" to be a form of discrimination:

"Our district is not so backwards as it is closer to Kathmandu, so there is no discrimination between son and daughter. If it is first baby then every member in (the) family is pleased." FG13, Men/Father-in-law, I.

In both the control and the intervention areas, the respondents were aware of the fact that the government provided iron and folic acid supplement tablets free of cost to pregnant and postnatal mothers. Also, the respondents knew of the Safe Motherhood incentive programme (*Aama Surakchhya* program, for further details see Chapter 1, Section 1.4.1), which charges 1000 Nepali Rupees for hospital delivery and 400 Nepali Rupees for 4 ANC visits. The VDC healthpost is reportedly registered as an incentivised centre only for the ANC incentives. As the public-private community hospital was not, the respondents had to travel to Kathmandu. A staff member of the community hospital interviewed commented on the incentives available to women:

"Government provides iron/folic acid (tablets) free of cost to pregnant and postnatal mothers. They have to buy calcium. As per the mother incentive programme (Aama Surakchhya program), those women who birth are given 1000 Rupees for hospital delivery and 400 for ANC visits." **IDI 1, Hospital health worker, Bahun, I.**

In addition, mothers, who were also GTN group members, were able to discuss the incentives that they had received in detail:

"Women get 1000 Rupees for birthing in a hospital and 400 Rupees for 4 ANC check-up from health post. If hospital delivery they get (the money) if not they don't." **FG7**, **FCHV**, **I**.

Yet one participant, a non-GTN group member, reported that while there was no cost of ANC services, she was not aware of any incentives:

"There was no need to pay for ANC check-up at the health post. They do not give us money, either." FG6, Non-GTN group, Mother, I.

Despite this increase in hospital deliveries, women continued to use traditional healers:

"More are now delivering in hospitals than at home. People go to the health post and the hospital when they get sick. In the past, they solely depend on "Dhami Jhankri" (traditional healer)". Now, they receive treatments from both the hospital and from "Dhami Jhankri." **ID1, hospital health worker, Bahun (AHW), I.** The GTN health promoters interviewed mentioned that services had improved for women, as had bathing practices, and that they were called to provide mobile healthcare on health matters. They also mentioned that they provided antenatal and postnatal services. However, they also reported that certain castes still followed their own practices:

"Services like depovera (contraception), immunization, and family planning are provided from the Out-Reach Clinic (ORC). GTN staff provides antenatal and postnatal services. The numbers of clients are increasing at the ORC clinic. In the past, newly born babies were given a bath within 24 hours of delivery. (But) In the Tamang community, they give baths to baby at any time." **ID4, GTN staff, Bahun.**

6.4.1.4 Groups working in the area

When asked which NGOs work in the area, the majority of female respondents named GTN. Those in the intervention area also named the community hospital. Male participants seemed aware of the existence of GTN groups:

"...and the Manmohan Community hospital is also a form of government. If you are talking about an NGO, there is only GTN available here; Green Tara is working here (someone talks in between)." FG13, Husbands/Father-in-law, Tamang, I.

A mother who attends groups went on to report that meetings were held twice each month:

"Green Tara is working in our village. We have two meetings per month, and we hold meetings here in this cottage, usually on the 24th day of the month... (and) at the end of the month." **FG1, Mother, Tamang, I.**

Another participant commented that there used to be resistance to attending group meetings. Over time, women's attendance has increased, and they have learned a lot about personal healthcare as a result. The daughters-in-law no longer argue with their elders regarding their attendance at group activities or health uptake, but listen silently as other women start attending health clinics:

"Old people complain that we do not need to this check-up...now, these people (new mothers) have to get check-ups, immunizations, rest...we just listen and we keep ourselves silent." **FG1, Mother, Tamang, I.**

The MCH worker interviewed mentioned that she worked with GTN to co-ordinate family planning and ANC clinics. She added that she was aware of their work with women on maternal health, HIV/AIDS, the importance of a skilled attendant at delivery, and neonate hypothermia and she was able to communicate with them:

"I feel free to talk with them, not in the least bit afraid. I was a bit afraid of the male HP...but everything was good (during our interaction with GTN). The changes are really apparent...Due to their space management...I can check on women in a public building close to their village. Many women were unaware of issues related to maternal health, but due to GTN programme, they know much more about that as well as related issues...The majority of women attend GTN in this area." **ID3, MCH worker, Brahmin, I.**

A member of a mothers' group was also positive about her group participation and mentioned that she could communicate openly:

"I can communicate freely. I really like it. (Laughs)." FG2, Mother Balami, I.

Some participants who felt that there were no barriers preventing them from attending the group meetings and that they could openly talk to GTN health promoters:

"Pregnant women and children are not allowed to touch a woman whose baby has died. My child was died due to tetanus when he was one year old. Local pregnant women and children did not touch me. This is not happening now – this practice has disappeared because GTN has taught us." **FG2**, **Mother**, **Tamang**, **I**.

Another female participant in the women's groups commented on the pre-existing savings group that met after their GTN group meeting, and reported discussing the health of women and children with GTN health promoters:

"One is to collect savings and to discuss the status of savings and credit. Another meeting is for discussing the health of mothers and children, hygiene sanitation, and other health-related issues." **FG1**, **Mother**, **Tamang**, **I**.

Mothers who were interviewed in the control area (Tamang 28-34), mentioned that the only women's groups in their village were saving groups; health matters were not discussed:

"We now have women's groups in the village. I am a member of a women's group; we collect money, save money and use that for opening shops, for our farms, etc. Hum! We do not talk much about health." FG11, Mother, Tamang, C.

Another mother-in-law also mentioned the savings group where she was a member:

"We have a women's group for savings and credit...We give a loan (using that money) for a small entrepreneurship like knitting, goat raising, opening shops, etc." FG9, Mother-in-law, Newar, C.

A FG participant mentioned that although there may be women's groups, she was unsure of their activities:

"Actually, the mothers know more about their health (and how to care for themselves) than us (laughing). They have meetings from time to time for this...there is GTN (somebody came and said, 'what is this?' And sitting nearby and talking with participants; mobile ringing and one of the participant is talking)." FG13, Husband/Father-in-law, Newar, I.

While a male participant felt that the whole village should take part in the meetings, including men:

"...If, in a given village, everyone gets together, then it would make for a good (GTN) meeting." FG13, Husband/Father-in-law, Tamang I.

6.3.1.4.1 Women and workload

The increase in awareness of maternal health seemed to be attributable either to women's decreased workload at home and in fields or to women and health workers learning and gaining increased awareness from group activities (GTN groups). As seen in the previous section, women gained increased awareness due to maternity incentive groups.

When asked about changes in attitudes, mothers-in-law in the control area mentioned the women's workload:

"In our time, we were given heavy work during pregnancy and after delivery...Now there are a lot of changes." **FG11, Mother, Tamang, C.**

Pregnant women are now allowed to rest more, and they were even encouraged to do so by their mothers-in-law:

"Now, mothers-in-law also say that you should rest during your pregnancy." **FG1, Mother, Tamang, I.**

An interviewed health worker also stated that women were advised to reduce their workload during pregnancy:

"They are not allowed to carry heavy loads." ID3, MCH worker, Brahmin, I.

Other respondents who were asked about women and their workloads reported that their hiatus from work varied from only a week to a few months depending on the woman's physical state postpartum:

"Some work 2-3 months, some a month; some may be 5 days. It depends on the time (and her) condition." **FG13**, **Men/Father-in-law**, **Tamang**, **I**.

According to a male respondent, women's workloads were linked to the family norm and the type of delivery they had had. The postpartum resting period ranged from a few days to months based on the time needed to recover: "Is it simple delivery (vaginal) or is there a delivery involving an operation (e.g. C-section)? In the case of an operational (surgical) delivery, that would make it difficult for a woman to work (post-surgery), but if it is a simple (vaginal) and good delivery, they usually work after 7-8 days...It depends on family type. If it is a 'good family', they allow a new mother to rest for 3-4 months both at home and at the parents' home. Sometimes there is difficulty – it may be so hard to sustain daily life in such a family that the mother will have to work by tomorrow (the next day)." **FG13, Men/Father-in-law, Tamang, FG, I.**

However, another male respondent reported that women do have to do strenuous work, and working hard both during pregnancy and after was the norm in their village:

"Actually, we have seen that pregnant women, we say not to lift heavy goods/things, not to do difficult work. But it doesn't happen in this village, because they have to work here. Here in village, people are not so considerate, so some families don't care if a woman is pregnant - she still must work." FG13, Men/Fatherin-law, Tamang, I.

During one FG, some women stated that they had to work in the field and in the house during their pregnancies:

"In the village, we have to work during and after pregnancy (we) continued cutting grass, fetching water, cutting wood in the jungle..." **FG2, Mother, Tamang, I.**

6.4.1.5 Shyness

An interviewed health worker added that women's attitudes had changed with respect to their maternal health. Women used to go to traditional healers or give birth at home or in a cowshed or in the "dark" because they were shy. Now, however, women go to the hospital for a delivery or ask a SBA to assist them at home during birth. Additionally, more women came to the health post for a regular check-up or for ANC:

"There are lots of changes in healthcare. People used to feel shy about visiting the health post; they used to go to "Dhami Jhankri" (Traditional Healers) for disease treatments. Women used to give birth in dark places in their houses thinking people in their community will see them while giving birth to children. Not only this, but they used to give birth to children in cowsheds. This is not the case now. Women feel free to visit the health facility (ANC) whenever they feel they should go. Women go to the hospital to deliver. If they give birth to a child at home, they just (have) SBA over to aid in the delivery." **ID3, MCH worker, Brahmin, I.**

A health worker interviewed mentioned that women and men seemed to be more aware of their general health. They were reported to be less shy and more willing to come the maternal health clinics:

"There are lots of changes in this area. In the past, people from this area were not aware of minor things, and communication was hard. They used to feel shy. But now things have changed. There are changes in the health sector too. Women come to health facilities for their regular check-ups, and they are more aware of their health than before." **ID3, MCH worker, Brahmin, I.**

Male participants, and their partners were perhaps too shy to discuss their health with their husbands also mentioned feelings of shyness. Attending GTN groups therefore helped women to be more open and communicative with their husbands:

"There are many GTN groups, some older some are newer. They meet each month (laughs). It is good to share feelings, ideas, and problems. Some are shy about talking with their husbands, and it helps to open (to) them (GTN)...they share the meeting's discussion with us (smiling)." **FG13, Men/Father-in-law, Newar, I.**

The maternal child health worker (MCH) reported that there were lots of changes in this area. In the past, people from this area were not aware of small "things" they could do for their health, and communicating which services were available was difficult:

"They used to feel shy. Things have changed; women come to the MCH health facilities." **ID3, MCH worker, Brahmin, I.**

6.4.1.6 Men's involvement in maternal health

Most of the men interviewed were somewhat vague when discussing the existence of groups, and they referred to the women who may know more about the programme. They mentioned an existence of women's groups:

"There may be but I forget... yes, there may be women's groups in the village, but I don't know so much about what they are doing." FG13, Men/Father-inlaw, I.

Whereas a few male participants seemed to be aware of the existence of GTN but not what they did:

"My wife went to GTN a few days ago, but I don't know what she did there." (Laughs) FG13, Men/Father-in-law, Tamang, I.

Men commented that they were aware that their wives went to GTN meetings and were somewhat aware of their activities:

"GTN or something, this group is in different places (villages, areas). It works for mothers and babies. The group provides a blanket for the baby after it has been delivered. They come and do health check-ups for women - that's it." FG13, Men/Father-in-law, Tamang, I.

Some men also cited the need for programmes like GTN, as they considered the advice obtained as useful:

"We can't do anything by ourselves - there should be projects like GTN, which go from home to home and village to village, to provide more health information to the mothers and care for their children. This would be very good." FG13, Men/Father-in-law, Tamang, I.

Males in the control area did not seem aware of any maternity incentive scheme or of any group activities for maternal health: "There are no programmes here yet ...Related to health, there is no NGO here." FG14, Father-in-law/Husband, Dalit/Chhetri, C.

6.4.1.6.1 Effects of the groups

The participants reported that during the GTN groups, they discussed handwashing, hygiene, and sanitation:

"Babies need to be kept neat and clean." FG9, Mother-in-law, Newar, I.

Also discussed are problems related to ANC, delivery and PNC, infection, exclusive breastfeeding, skin-to-skin methods, using family planning methods (contraception), keeping the baby clean, taking care of the pregnant mother, delivery and after delivery, uterus prolapse, HIV, family planning, danger signs, and when to go in hospital:

"We discuss health related-topics, like pregnancy, exclusive breastfeeding, and danger signs." **FG1, Mother, Tamang, I.**

They also learned about avoiding preferring one gender to another:

"We talk about not discriminating between having a daughter versus a son." **FG1, Mother, Tamang, I.**

Some of the participants in the groups discussed certain topics, such as safe sex practices, ANC, and good nutrition only shyly:

"(With small voice) ... (we) talked about safe sex. We need to check for pregnancy at 4, 6, 8 and 9 months - at least four check-ups during pregnancy. Mothers need to eat 180 iron tabs from conception up to 45 days after delivery. Babies should be given mother's milk only for 6 months and then we should feed them lito (porridge)." **FG1, Mother, Tamang, I.**

The danger signs during pregnancy were also discussed:

"...It is dangerous if our hands and feet swell, if we experience bleeding, vertigo, white discharge, lower abdominal pain...in that case don't stay at home; go and get a check- up." FG1, Mother, Tamang, I.

Overall a good knowledge of pregnancy, diet, and sanitation were seen. One change observed was that women who had had an institutional delivery were subsequently allowed to rest:

"In the past, pregnant mothers were not taken to the hospital for 2-3 days even though she was in pain. Pregnant mothers should be taken to the hospital for delivery. They need rest and they should be given time for rest after delivery." FG1, Mother, Tamang, I.

A group of GTN attendees mentioned that another change was that women went for antenatal check-ups as a result of participating in the group activities:

"From the time of conception to nine months, women now go for check-ups, but earlier it was not so popular. During pregnancy, women did not go for even a single check-up earlier. In the nine months span, we now go for a minimum of 7 check-ups...Even with the slightest sign we go for check-up... I think even the minimum 4 (ANC) check-up is a big change..." FG3, Mother, Balami/Chhetri, I.

A Tamang participant mentioned that they learnt that mothers needed rest during the postnatal period:

"The postnatal mothers have to work very soon, like after a week (postpartum) in the past. Now we all know that mothers need to rest during and after delivery, so usually within a month, mothers have to begin working again." FG1, Mother, Tamang, I.

6.4.1.6.2 Health workers working with GTN

As a result of working with GTN, collaborations formed among the community health workers. As a result, ANC and PNC services became available to women, the FCHV led the mothers' groups, and contraception became available from these joint clinics. Additionally, the GTN staff provided antenatal and postnatal services:

"FCHVs lead the mother's group...Services like Depo-Provera (injectable contraception), immunization, family planning, are provided from the outreach clinic (ORC)- GTN staff (alongside) provides antenatal and postnatal services. The numbers of clients have increased in ORC clinic". **ID4, GTN staff, Bahun, I.**

A few health workers mentioned what they felt could be learnt from GTN. For instance, the MCH worker mentioned how awareness of general health issues and particularly maternal health could be increased. She also went on to mention how they had a good working relationship with GTN:

"I feel free to talk with them. There was no management of the place (clinic) before. Many women were also unaware of issues related to maternal health, but due to GTN programme, they now know much more on the related issues. Almost all the women attend GTN." **ID3, MCH worker, Brahmin, I.**

6.4.1.7 Empowerment

Changes over the five years included increased notions of empowerment, as women had more autonomy not only with respect to their health but also with respect to driving. Particularly the women who were group members voiced this sentiment:

"Increased number of women driving their scooters in the road." ID4, GTN staff, Bahun, I.

Respondents cited increased feelings of empowerment due to their participation in GTN, including how they were motivated to take up health services, to have no gender preferences for children and the realisation of the importance of immunisation:

"Old village people gossip about pregnancy check-ups, rest during pregnancy, health check-ups, and child vaccinations that (they said) were not needed in their time. We just avoid these types of gossips. We have been here for the GTN group from the beginning and have been teaching in the village that we have to go for check-ups during pregnancy, avoid discriminating in terms of babies' genders, and bring children immunisation." **FG1, Mother, Tamang, I** Similarly, the GTN staff reported that they felt that the GTN group women were more confident in attending groups, that the women themselves took charge of organising the groups, and that they were more involved in discussions:

"Women can keep their voices in front of others...They can share their problems now. They now discuss in groups. In the past, no one came in groups. These days, they are even calling us in groups to share new things...they come themselves to discuss." **ID4, GTN staff, Bahun, I.**

These women seemed more confident as they were able to attend the GTN group meetings once a month with no restrictions on their attendance by their families:

"We all attend the meeting once a month (loudly). We do not face any (form of) objection from home." **FG1**, **Mother**, **Tamang**, **I**.

During a FG of mothers-in-law, participants noted that they felt women were more in charge or made more decisions with regards to their own health, particularly as they had savings groups:

"Women are empowered. We have saving/credit, and we feel women are empowered." **FG4, Mother-in-law, Bahun/Chhetri, I.**

6.4.1.8 Changes in decision-making patterns

When participants were asked who made the decision to seek care (for instance, ANC) the responses varied. It seemed that women did make the decision solely for themselves, especially if they were group members, and that sometimes their husbands played a role:

"Women themselves make their own decision for ANC, and sometimes their husbands also played a role in the decision-making...Basically, family members make decisions." ID3, MCH worker, Brahmin, I.

While in other instances, the mother-in-law played a role during the pregnancy and birth:

"Women themselves...We have to tell (her)...but the grandparents make decisions regarding care after delivery." **FG5**, **Mother-in-law**, **I**.

One of the men interviewed stated that making the decision together arose when problems occurred during labour. Reflecting a general change of no longer delivering at home, the woman went to the hospital, sometimes accompanied by a female relative:

"If there is delivery in the hospital, then health workers take over care at that time, and usually mothers make the decision regarding their care, and after they come home, all (members) cares (for her) in the family." **FG14, Men/Father-in-law, Chhetri/Dalit, C.**

Whereas a few male participants mentioned that it is more of a consultation among all the family, rather than making a decision to deliver at home:

"For this decision, we all consult as a family". FG13, Men/Father-in-law, Tamang, I.

For example, if a woman feels unwell, the family will come together, discuss the matter, and then make a decision to take her to hospital:

"Husband and wife make the decision...where to get treatment." FG14, Father-inlaw/Husband, Chhetri/Dalit, C.

At the time of delivery, a male respondent in the control area also mentioned that the head of the house made the decision with the family:

"Mainly household health, but all makes decision together in the family." FG14, Father-in-law/Husband, Chhetri/Dalit, C.

While a few mothers (Tamang, Balami and Chhetri) said that they themselves decide, here is an example of a quote:

"Myself, ourselves, we discuss with the family and go". FG3, Mother, I.

Mothers-in-law in the control area mentioned that the decision was left to the couple:

"Sons and daughters-in-law knew about it, and they can decide themselves now." FG9, Mother-in-law, C.

Others reported that they consulted with their husbands and the husband's family one example is provided below:

"We decide ourselves to go in for check-ups and deliveries. Most of us are staying with our husbands. We share our thoughts with husbands and they also agree. Some of us are staying with our mother-in-law and father-in-law. They share (the decision-making responsibility) in the family." **FG2, Mother, Tamang, I.**

When asked who made the decision during delivery care, mothers reported that the entire family makes the decision:

"Mother-in-law, father-in-law, entire family, husband". FG3, Mother Balami/Chhetri, I.

Three respondents mentioned that the reason a joint decision is taken with the family is that they live as a nuclear (joint) family. They share that decision with the family with agreement from their husband:

"We ourselves decide for us. Then we share with our husbands. We are living as a nuclear family. So it is easy for decision-making. The husband agrees on it." FG11, Mother, Tamang, C.

6.4.1.8.1 Decision-making in PNC

When asked who made the decision to attend PNC, participants in a GTN group stated they make the decision themselves to attend:

"(If) we have (health) problem in PNC, we decide on our own". FG3, Mother, Balami/Chhetri, I.

They reported that if women feel well in the postnatal period, they do not attend PNC, but are instead primarily concerned with labour and birth. This might indicate a gap in awareness.

"Most of them do not go for PNC check-up. If they have any problem, then only then do they go in for a check-up, otherwise, they do not come out from home. They say that 'there is nothing wrong with me, why should I go to the clinic?'... They go for PNC every? 45 days for family planning. They do not go out if everything is normal. They need to walk, have to take bus, which they don't want to do during this period." **ID4, GTN staff, Bahun.**

It seems that decision-making during PNC is a mix of the women themselves and family and husband:

"We ourselves decide to go for pregnancy check-up. We are living in homes separately from our mothers-in-law. We share our beds with our husbands." FG1, Mother, Tamang, I.

Men who were asked the question regarding PNC attendance felt that women were physically weak and needed help to make a decision:

"At that time in the postnatal period, she can't do anything, we have to care (look after her). First of all, the family members, like mother-in-law, father-in-law, and husband, who also make decisions for them (on their behalf)." **FG14, Men/Father-in-law FG, C.**

Whereas others stated that they did decide themselves:

"No way, if we do not decide ourselves, who will do for us?" **FG1, Mother, Tamang, I.**

6.4.1.9 Barriers and limitations to attending groups and health services

6.4.1.9.1 Socio-economic barriers and limitations

The respondents reported that there were few women with children under 2 years old in the area who did not come to the group meetings. Those who reported not attending were either new mothers or those with too much housework.

When asked why some people might not attend, men cited household work – women do most of the house and fieldwork, which makes it difficult to manage their time, as they are busy:

"Main reason is household work (his baby cries) household works makes it difficult to manage time." FG13, Men/Father-in-law I.

On non-attenders, the attenders mentioned that they were told that there are few who did not attend groups, stating they do not have time:

"They scold us 'timiharu just kaam napakeyo ho ra hami kaam ma janu parch', which means, 'we are not like to you who do not have work, we have to go for work". **FG2, Mother, Tamang, I.**

The GTN staff also reported that they organised home visits for those women who could not attend groups:

"There are some who cannot come to the group. For them we talk to them during home visits ...there is one lady - I told her family where to go for delivery and what to do when I met her family (at home)." **ID4, GTN staff, Bahun.**

The other GTN staff said:

"One who is extremely busy does not come to group discussion, like one who has to go in office, who has to send children to school, who has to cook lunch, who has to raise goats, animals in home etc...For them, we visit them their homes." **ID4, GTN staff, Bahun.** Participants were asked about the barriers preventing them from attending PNC, which included lack of awareness on the need for PNC:

"Women rarely come for a PNC visit. Maybe there is lack of counselling for PNC. I think we should go on home visits and inform them about PNC. We should counsel them very nicely. Even if we tell them to come for PNC and they don't, they'll visit health facilities during vaccination times. Until and unless postnatal mothers get sick or suffer, they won't go for check-ups." **ID3, MCH worker, Brahmin, I.**

In several of the interviews, the respondents cited time, having to work in the field or house, permission from family, and housework as barriers to attendance:

"There are many who don't. Let's look at the community as a whole and think...due to work at home, baby, time constraints...We are asked during our duty time after we leave work, where do we (really) need to go?" FG3, Mother, Balami/Chhetri, I.

For some women, non-attendance was the result of not having the money to attend health services:

"Money is the biggest problem...we if we go to the hospital, we won't have money to even pay the fee." FG3, Mother, Balami/Chhetri, I.

Other women reported that non-attendance was due to being physically weak or mistreated by the family:

"Mothers are weak. They can't even get nutritious food. They are usually engaged in work, so they don't have time to care their children and their health. There are some cases where mothers-in-laws don't give food to their daughter-inlaws and order them to engage in work. Financial problems are also a major problem; as a result, women don't go to health facilities for treatment. So, there are things that should be changed." **MCH health worker, Brahmin, I.**

One of the GTN health workers mentioned that women did not attend health services due to fear or shyness about being scolded for having married and gotten pregnant at such a young age (teenager): "Many married young and do not go for antenatal check-up when they are pregnant because of shyness and because they are afraid that they will be scolded by the health providers for being pregnant at such an early age...Teenage mothers do not attend ANC clinic because they thought that they would be scolded." ID4, GTN staff Bahun, I.

6.4.1.9.2 Cultural barriers

Seemingly, there are seemingly several cultural practices that include isolation, which may have resulted in women not attending PNC. This may mean that such cultural practices also act as a barrier to uptake of the intervention. Another barrier seemed to be the lack of awareness of health promotion activities and services that exist.

Other dates of significance in Nepal include the date when the new mother can leave her in-laws' home (30 days after birth) to go to her parents' home for a period that lasts from a few days to up to a month dedicated to rest:

"The postnatal mother goes to her mother's house within a month of delivery and stays there around one month." **ID1, Hospital staff, AHW, Bahun, I.**

Another respondent reported that women travelled to their parents' homes and stayed for a full month; this may mean they did not attend PNC.

"The postnatal mother is taken to her mother's home within a month following delivery and kept there for a month. It depends on the condition of the house. If nobody's in the home, they are taken to their mother's home early, like within 10-15 days, and they stay more days in their mother's home." **FG9, Mother, Newar, C.**

Respondents were asked about their cultural practices that exist around pregnancy to see whether, if asked in a different way, certain potentially harmful traditions were practiced despite evidence-based health promotion training. Interestingly, the practice of isolating women (in a shed, typically a cowshed – "Chhaupadi") was mentioned in a men's focus group during the men's FG:

"In this matter, there is not so large an influence or effect in this village, but in some places, there is still system of isolating women during menstruation/pregnancy (Chhaupadi). Sometimes women have to stay in the stable, also. In my thinking, in these parts (cough), these are not bad practices. Everyone is doing equity/equal behaviour." **FG13 Men/Father-in-law, Tamang, I.**

Respondents were asked about cultural practices after birth and stated that they followed religious practices and were not permitted to go to the temple. They said similarly when women menstruated they were isolated and were not in physical proximity to anyone:

"Those cultural practices (are done) in (a) religious way. After 2-3 or 4 months of pregnancy, they shouldn't go to (Hindu) temple. Other practices are during that menstruation no one 'touches' (her) but there is the practice of isolation (Chuwachut/Chhaupadi)." **FG13 Men/Father-in-law, Tamang, I.**

Another male respondent stated that although there was no isolation during menstruation or pregnancy, isolation does occur in the postnatal period:

"There is no system of isolating women during menstruation/pregnancy here...yet after delivery there is 3 days and sometimes 5 days (of isolation) to name the baby." **FG13 Men/Father-in-law, Tamang, I.**

Isolation can also be a positive phenomenon where women are allowed to rest after giving birth:

"The postnatal mothers had to go for work very soon, like after a week, in the past. Now, we all know that mothers need rest after delivery (birth), so usually within a month, mothers have to go for work." **FG1**, **Mother**, **Tamang**, **I**.

The mothers group who attend GTN seemed reluctant to mention cultural practices as they felt that the interviews might be shared with GTN and they did not want to be judged. Women reported nothing negative about GTN, which may be a reflection of pleasing/imbalanced views they wanted to give on GTN. Most women reported that, unlike in the past, they were now allowed to rest after birth in their in-laws' home. In the days, before "nwaran" (naming ceremony), however, the new mother cannot leave the room or the house, and no one will touch her or take the child directly from her, as a vaginal birth is deemed "dirty". The mother-in-law will use her old sari or old clothes to pick up the child. After nwaran, the new mother is re-introduced to household activities once again. Until that time, women have other restrictions imposed on them.

Not attending due to the naming ceremony, nwaran:

"It is good, in relation to health, in our culture that after the delivery (up to 7 days post-delivery), the mother doesn't come out from home. When we offer the baby a name, only then can she come out and go here and there. And this is good." **FG13 Men/Father-in-law, Tamang I.**

6.4.1.10 Reflexive section

Overall, when comparing the interviews of the intervention versus the control sites, participants in intervention sites seemed more knowledgeable, confident, outspoken, and expressive than those in the control area, as well as more autonomous in their decision-making ability. Specifically, they seemed more knowledgeable in general health, and maternal and child health issues. For instance, some participants in intervention sites were able to comment in detail on the danger signs of pregnancy, on the amount of iron tablets that ought to be taken during pregnancy, which months and times they should attend ANC to 'achieve' 4 visits, and so on during focus groups. Overall, the responses were more forthcoming and fluid than those from participants in the control sites.

During the initial GTN group meetings held in 2007 and 2009, the participants were shy and reluctant to engage, particularly with respect to cultural practices. The intervention started when the civil war had just ended, and the notion of outsiders involving themselves in "private home" affairs were initially strongly disliked. Over the years and with time, women seemingly became more confident. By June 2012, they seemed more open to communicate. The training sessions the HPs used were based on the curriculum and their own experiences (marriage and childbirth) to encourage the group to participate. They shared that they too were also mothers

211

and thus encouraged exclusive breastfeeding and a balanced diet for weaning children so that they would do well at school. Finally, the health promoters and the health post and hospital health workers felt that the interviews were an evaluation of their job performance. In the analysis of the key informants, they also referred to enablers and barriers, yet they are from a different perspective.

Respondents were asked about cultural practices existing around pregnancy in order to find out, if asked in a different way, if certain traditions, perhaps harmful, were practiced. This highlighted a need to address ineffective practices in the future with evidence-based health promotion training.

On reflection, the use of a Tamang-speaking researcher may have facilitated the group discussions in villages where the Tamang dominate, particularly in three FG. The language barrier required a two-stage interpretation to extract opinions from participants in the groups. There appeared to be interplay between a moderator of a different caste and the participants both when comparing the caste responses and in the manner in which the interviewees expressed themselves. It seemed that the female Tamang respondents were more empowered or self-assured. The Newari moderator/translator used for the focus groups stated in a surprised tone: "*they [the Tamang women] decide for themselves and hold more power in the household than one would expect and when compared to other (upper) castes*".

Chapter 7 Discussion

7.1 Introduction

The quantitative (Chapter 5) and qualitative (Chapter 6) results from the study show changes: i.e. a) improvements in socio-demographic factors (some of which were confounders), and b) attendance outcomes that occurred over time in both intervention and control areas. This evaluation took a pragmatic approach, with elements of realism, and so was only concerned with the changes that were likely to be attributable to the GTN intervention. Therefore, in this chapter, only the evaluation's main findings, their measurement, and the study's strengths and limitations are discussed. This chapter consists of these three interrelated parts, starting with a discussion on the contribution to new knowledge in the field and the substantive findings of the PhD research followed by ways of evaluating interventions in LMICs settings. The final section highlights the limitations and strengths of the research conducted for this thesis and reflections on the evaluation research process.

7.2 Difference-in-Difference

This is the first mixed-methods evaluation in maternal health promotion in Nepal, possibly in any LMICs that used a setting-appropriate methodology difference-in-difference (DiD) with three time points. Here, the value of the DiD analysis was one of effectiveness and attribution; it isolated the effect of the programme from other external and internal factors, and potential confounders (Sharma et al. 2016a). In other words, DiD permitted a closer approximation of the causal effect of the programme, the "treatment" on the outcomes of interest (Alderman et al. 2009a; Liu et al. 2010; Ensor et al. 2014). The DiD analysis provided a more useful method in the impact evaluation of this quasi-experimental study design than a standard beforeafter analysis. A before-after analysis would be concerned with looking at percentage change and thus would have shown greater uptake than should be attributed to the intervention. For example, taking the outcome of attending ANC (at least once) from the baseline to the final evaluation, a 13.38% increase was seen in the intervention area (Section 5.6, Table 10). Whilst the DiD analysis for the same indicator showed that the change was

213

only 12.74% (Section 5.10, Table 16). This method prevents the overestimation of the intervention.

Another key highlight of using DiD is the precision it offered. An increased uptake was seen across the study for the outcomes SBA at birth and institutional delivery (ID) (Section 5.7, Table 12). It is likely, as mentioned in the qualitative findings, that this percentage increase may be due to the maternity incentive rather than the intervention (Section 6.4.1.3). The maternity-voucher-incentive scheme aimed to address cost barriers to the uptake of maternal health services in Nepal (Section 1.4.1). The DiD analysis, however, showed no statistically significant difference between the intervention and control groups (treatafter) as the possible confounders, including the maternity voucher incentive scheme (Section 1.7), were deemed to have been controlled for by virtue of the fact that it was available in both areas since both the control and intervention area received the intervention. In addition, as seen in Sections 2.3 and 6.4.1.6.1, the intervention offered ANC and selected aspects of PNC - it only provided knowledge on the importance of having a SBA at birth or an ID and no additional resources (Sharma et al. 2017). Therefore, it is not surprising that no impact was seen for the intervention group in terms of seeking a SBA or having an ID. This finding is similar to a previous study on women's groups, which was designed to improve birth preparedness (McPherson et al. 2006). The study resulted in an increase in the knowledge of obstetric "danger" signs and little change in the proportion of deliveries involving SBA. The authors hypothesised that this was due to the fact that barriers, such as the cost of getting to a facility, persisted (McPherson et al. 2006; WHO 2014c).

DiD was also valuable in understanding sometimes contrary findings within the study. Two such examples are given below. First, there is evidence that ANC utilisation is strongly correlated with the utilisation of a SBA (WHO and UNICEF 2003; WHO 2009a). This could explain the finding that the less sophisticated before-after analysis (Pearson's Chi-square test) demonstrated a correlation between attending ANC once and having a SBA at birth, in both areas, among the whole population (Section 5.6). These findings are similar to those in a study on the continuum of skilled care (Christian et al. 2003). Based on the assumptions and given the data, the best causal estimation is that implementing an ANC service has the potential to serve as a strategy for increasing the utilisation of SBA (Section 5.9). Women receive advice to seek skilled birth care (Section 2.5.4). Yet, this association would need to be explored whilst controlling for potential confounders when looking at the association between SBA and ANC; and if facility costs are addressed as they represent a significant barrier to attending services (Witter et al. 2011).

Secondly, as mentioned in Sections 1.1, 3.2.3.2, and 7.2.1, it is important to account for counterfactuals/temporal trend comparisons, i.e. changes that happen over time; for instance, in order to provide a comparative trend by linking to national data (Section 1.1). At a national level, births occurring in the presence of a SBA rose by 17.3% from 2006 to 2011 (MOHP et al. 2007; MOHP et al. 2012). Whilst in the similar period to the GTN intervention, it rose by 21.40%, and in the control group by 19.98% (2007-2012). One possible explanation for the rise in SBA at birth is the above-mentioned maternity voucher incentive scheme (AAMA) that was rolled out in 2009 (Section 1.4.1). Or the increase in women's education, a factor linked to increased maternal services use (Section 1.3 and 1.4.4). It is noted that the control area demonstrated a comparable increase to the intervention area for this indicator. More research is needed to ascertain if there was any correlation between the GTN intervention and the AAMA programme.

The DiD analysis also provided details on covariates and barriers to attendance. This study (Section 5.7) saw that a low educational level and low household income were risk factors for non-attendance (Baral et al. 2010). Multiparous women were more likely to attend; unlike other studies multiparity was not a barrier in attending maternal health services. Furthermore, well-documented socio-demographic data indicate that women from relatively poor backgrounds who live in rural areas and/or have low levels of education are less likely to access antenatal services (Abouzahr 2003; Houweling et al. 2007; Simkhada et al. 2008). Other factors, including having a husband with a low level of education, living a long distance from a clinic, and having high parity, have also been identified as barriers to accessing care (Kabir et al. 2005; Trinh et al. 2007; Brown et al. 2008; Bassani et al. 2009). Similar factors emerge in the reviews of barriers to ANC in high-income countries (Rowe and Garcia 2003; Lewis 2011; Downe et al. 2009; Thomson et al. 2013) which suggest that the issues for women

215

who remain marginalised at local, national, and global levels are much the same (Finlayson and Downe 2013; Sharma et al. 2016a; Sharma et al. 2016b).

On the whole, the barriers to seeking a SBA, including socio-economic, financial and geographical, are more difficult to overcome than the barriers to ANC and PNC (Borghi et al. 2006; Furuta and Salway 2006; Choulagai et al. 2013). In Nepal, the shortage of SBAs and perhaps the quality of care provided in institutions is a considerable barrier to this continuum of care (Section 1.4.3) as Nepal does not have midwives (Bogren et al. 2013; John 2015). Furthermore, these factors may explain why the intervention did not have an impact on delivery care. Moreover, the qualitative study suggested that the weight or influence of the family's decision is greater in attending delivery care than it is regarding ANC (Section 6.4.1.8). The explanation behind this may be that the family controls finances and is likely to make decisions regarding place of birth based on cultural preferences of childbirth and that birth is more expensive than ANC (Kwambai et al. 2013).

The qualitative findings also suggested that in certain castes, such as the Tamangs, there seemed to be a preference to give birth at home (Section 6.4.1.2). This highlights an area that needs to be addressed, that of preferences - women in Nepal prefer to have reproductive health services and give birth within their communities. Pitchforth et al. (2008) discussed the concept of "choice" and place of birth in (rural) Scotland. Women engaged differently in the choice process, and health professionals, pregnancy complications, geographical accessibility, and the implications of alternative place of birth all played a role in terms of demands in their social networks (family/community) (Pitchforth et al. 2009). Yet, the provision of different models of maternity services may not be sufficient to convince women that they have "choice" and therefore they may prefer to birth at home, such as in rural Nepal. It would be of interest to know the proportion of Nepalese women who would birth with a SBA if they had transport to the hospital. It may be that they choose to birth at another facility/hospital, thus disputably exercise freedom of choice.

It is important to choose an appropriate research approach for health promotion evaluations. DiD is suitable for the GTN intervention, as the
evaluation was not a clinical trial, it was a complex community intervention (Sections 2.1 and 2.2). Moreover, many researchers have conducted either process only or outcome evaluations of community-based interventions in LMICs (Section 1.2), but these lack a control or comparison group, typically used in social experiments to gauge impact (Rauniar et al. 2012; Sharma et al. 2016a). Others have done significant work with RCTs (Manandhar et al. 2004; Shrestha et al. 2011), but these study designs are often inappropriate for the local setting as they are difficult to organise. For example, the randomisation of clusters/villages due to socio-economic mobility, and the local politics on the ground is challenging and may involve having to renegotiate access to the community at various stages of the trial, which can prove expensive and time-consuming (Rosen et al. 2006; Scriven 2008; Dixon et al. 2013). In addition, for ethical reasons, some interventions cannot be measured using a RCT. For example, a study on home birth is likely to be considered unethical, if the place of birth is randomly allocated in a setting where women are used to having choice. Moreover, there are sample size issues and potential difficulties include getting villages to buy-in to be the "control" (tested but "no reward"). Hence, there may be little incentive to stay in the control community. In the field of maternity care this might mean paying for unintended consequences of an RCT (highest in the hierarchy of evidence). An unintended consequence is defined as the unforeseen or unanticipated consequences of purposive action (Merton 1936). Unintended consequences are not necessarily undesirable. They may be highly beneficial or neutral (Sections 1.2, 3.2 and 3.2.3.2). However, an unintended consequence of an RCT is such that the cost exceeds the intervention (Thompson & Schoenfeld 2007; Bothwell et al. 2016). The DiD helps keep the costs of the evaluation down whilst achieving relatively high precision as a randomised study (WHO 1998; Duflo 2004). This would help to reduce the 10/90 gap burden, where only 10 per cent of health research is devoted to conditions that account for 90 per cent of the global disease burden (Stevens 2004).

In summary, the DiD permitted a more precise evaluation of the programme by adjusting for covariates in order to determine the intervention's effectiveness. A further strength of the method was the examination of the wider confounding factors and highlighting unintended consequences within the mixed-method study design.

7.2.1 Mixed-measurement evaluation

As mentioned, one of the key strengths of the evaluation was the use of the DiD analysis in a community-based LMICs intervention and the alongside qualitative study. As was seen in Section 3.2.2, an evaluation should be context specific. The results are mixed with the qualitative findings to yield richer data on changes over time in maternal health and in decision-making. Few programmes evaluate a five-year project in such detail, and the strength of the research in this thesis is the mix of statistical analysis, qualitative and expenditure data that enabled an evaluation of the cost, time, effect and impact on health uptake/attitude behaviour of the study population, and programme's staff time (Section 3.2.3).

It is estimated that approximately 50% of women in LMICs receive inadequate antenatal care (Finlayson and Downe 2013b). The DiD analysis showed that there was some improvement in the intervention group regarding maternal healthcare uptake, specifically an improvement in antenatal care utilisation for the rural women attending once over the five years and for those who received the WHO recommended four ANC visits only in the first two and half years. The qualitative research also suggested a possible explanation for this increase; women over time: a) were more aware of the importance of attending ANC, b) perceived that there were fewer barriers to attending ANC, and c) seemed to have more autonomy in making a decision to attend ANC (Section 6.4.1.8).

Iron/folic acid uptake significantly increased over both time periods for women living in the intervention area. A possible explanation for this increase could be that women in the intervention group were made more aware of the iron and folic acid supplementation that is provided at government health facilities throughout the country at no cost (MOHP, New ERA & ICF International 2012; Sharma et al. 2016a). However, in this study improvements were not seen for women in their first trimester of pregnancy (i.e. ANC attendance in early pregnancy). The qualitative study did not highlight any particular reasons for this; however women mentioned general notions of shyness/awkwardness (Section 6.4.1.5). The literature reports that women might not know that they are pregnant in the first few weeks and

that there are cultural reasons in Nepal behind why pregnancy is kept a "secret" in the first trimester (Simkhada et al. 2008, 2009; Finlayson and Downe 2013). Perhaps a "first-time" mother might feel unable to attend an ANC in the first trimester if she needs to (a) ask permission or monies from her family (typically her mother-in-law) to attend; (b) find time to travel to the clinic, in Nepal daughters-in-law are responsible for the household and farm tasks and therefore delay the first ANC visit; and (c) where early "disclosure" of pregnancy could lead to unwanted religious or spiritual complications (Simkhada et al. 2010; Puri et al. 2011a; Finlayson and Downe 2013a). The literature (Pell et al. 2013; Sharma et al. 2016a) suggests that pregnant women and mothers are often influenced by the experiences of their immediate social circle (family and friends), and in Nepal, the mother-in-law and husband tend to be the most influential (Simkhada et al. 2006; Lewis et al. 2015; Sharma et al. 2016a).

In addition, the qualitative study revealed that some women missed ANC sessions or health promotion groups due to the demands of daily family life and work (Section 6.4.1.9). Although not all women attended all sessions, this intervention improved ANC attendance. Yet, not all four ANC indicators improved over the five years. If women do not attend complete antenatal care, they are less likely to be prepared for birth and less likely to choose a SBA at the forthcoming birth (Morrison, Thapa et al. 2014). This poses an alarming problem; population groups in LMICs receiving few antenatal visits have been shown to have an increased risk of perinatal mortality and stillbirth (Dowswell et al. 2010). There was evidence that health promotion can make a difference in empowering these women to seek care (Section 7.2.5). The two sets of data suggested that as a result of the intervention, women were more empowered to make their own decisions to attend care, particularly in the antenatal period (Section 6.4.1.8). Empowerment will be explored further in Section 7.2.5. The importance of decision-making by women is highlighted in the literature. Women should be provided with information based on the available evidence and supported to make informed decisions about their care (Sharma et al. 2016a; Sharma et al. 2016b, Sharma et al. 2017). This issue of not being provided with information also seemed to apply to this low-income setting in Nepal and for women in high-income countries (Pitchforth et al. 2009; Watkins and Weeks 2009).

7.2.1.1 DiD & confounding factors

The DiD analysis also permitted the "unpacking" of the effect of each confounding factor independently. This was important as they have varying degrees of effect on the intervention outcomes, but they are frequently reported together in the literature. In the DiD analysis multiparous, more educated and wealthier women were more likely to attend maternal health services. The literature confirms this finding as older age, higher parity, and higher levels of education, and household economic status of the women were predictors of both attendance at four or more visits and receipt of 'good quality' ANC including iron supplementation (Joshi et al. 2014). This is also similar to the findings reported in this thesis, which showed that older women were more likely to attend ANC once or take iron and folic acid. However, the literature is not clear on the direction of the effect. Simkhada et al. (2008) found that in LMICs, women in their 30s attended ANC early and more frequently than teenagers and older women. The expectation might have been younger women in Nepal, with more education, and older women with more "maternal experience" attend services (Khanal et al. 2014; MOHP, New Era & ICF International 2012).

Years of research have suggested that many socio-cultural factors influence maternal healthcare uptake behaviour (Sections 1.4, 1.4.3 and 2.6.1), as was detailed in the maternal health conceptual framework (Section 4.3.2), and therefore these factors were justifiably included in the regression analysis. The results of the evaluation in this thesis confirmed that the combination of factors were as important but the general trend was that wealth, education, and parity have an independent impact on the likelihood of uptake of maternal health services. The fact that they work independently is important, and it suggests that there is a chance of impacting inequalities, for instance through education without increasing wealth (Section 5.7; Sharma et al. 2016b). This was noted in particular with delivery care outcomes of SBA at birth and ID (Section 7.2.1). For instance, the uptake of ID statistically increased, as did SBA over the five-year time, however, as a result of other factors rather than as a direct result of the intervention, wealthier and more educated women were likely to have an SBA at birth or an ID (Section 5.7; Tables 11 and 12). A future intervention may consider

this complexity and investigate whether increasing education and increasing wealth, either together or independently, can improve delivery care outcomes (Section 5.7). In addition to socio-economic indicators, time was also a confounder on the outcomes of the intervention, and this is discussed below.

Time points

Previous studies, conducted by e.g. Ensor et al. (2014), and Liu et al. (2010) referred to in the literature review (Section 2.6.1), only use two time points in their DiD analyses. This is the first study of a community-based intervention to use DiD analysis at three time points (Section 4.3.9.1.4). The inclusion of the third time point enables longer-term effects to become apparent. For example, the intervention had no impact on PNC attendance in the first two and half years when analysed with DiD, but there was evidence of an effect when considered over the five years, at the third time point. This may be due to the lower baseline in the intervention area - it rose from 52.20% (baseline) to 76.85% (midline), and to 85.86% (final). It may also be due to delay and decay effects that occur over time, i.e. it takes time to change and sustain health services attendance behaviour (Clore, & Schnall 2005; Higgins 2014; Sharma et al. 2016a).

Furthermore, the use of three time points enables a more realistic interpretation of the impact of the intervention in the longer term; women were seven times more likely by the midline and three times more likely by the final survey to attend ANC once (Section 5.7). While seeking ANC four or more times was significant from the baseline to midline, yet not in the final survey. This may suggest that the intervention was less effective or that it was difficult to sustain that effect (to change the way the message is delivered as communities get used to hearing the same thing and it becomes part of the "background noise") in promoting antenatal uptake after 5 years than after two and a half years. Thus, health promotion can change certain behaviours within 2.5 years to increase uptake of services in the community. First, the literature suggests that achieving the "last mile" can be difficult. For example, it is easier to achieve an increase from 50% to 55% than from 90% to 95% (Dhaliwal et al. 2011). The challenges for programmes in completing the last mile include the need for extra resources,

the distances (time and transport) that would need to be covered, and a lack of available information in a particular village. It seems health promotion was more effective by year 2.5 then by year 5, particularly for ANC outcomes. This may mean an intervention can be run over a shorter time period in the future. There may be a challenge in sustaining a behaviour-change intervention as initially achieved gains often diminish over time due to lack of resources or motivation (Ory et al. 2010). Moreover, there is a body of literature that shows that the early adoption plateaux of interventions are maintained while the later adopters and laggards are more resistant to change (Rogers 2002). The qualitative study suggests that women were more confident, or empowered, as they participated in the group (Sections 6.4.1.6.1 and 6.4.1.7). Despite this, however, it is not possible to say whether the participants were early or late adopters.

In the next section, the qualitative interpretation of the mixed-methods study is discussed. There is an added value of mixed-methods studies; not only does this study have a sophisticated analysis that provides attribution, as was discussed in this section. It is further complimented by a process evaluation (qualitative) that helped explain some of the key findings or lack thereof (Section 4.3) as will be explored in the next section.

7.2.2 Qualitative

Based on the voices in Chapter 4 from the women, their families, and the health workers, the qualitative study highlighted why the intervention worked and how. Several barriers were highlighted to their roles in preventing access to health services. The qualitative component of the thesis looked at the changes over time, the knowledge of reproductive health, antenatal, delivery, and postnatal care and if there were barriers to uptake of the intervention or health services. The findings were that there was an improved capacity of the community to identify, negotiate, and solve health related problems of maternal and child health and a better understanding of the need for a skilled attendant present at the birth or birth in a hospital (Section 6.4.1.6.1). Despite the rollout of the intervention, there still existed a range of barriers to accessing care (Section 6.4.1.9). As previously seen, there were several issues that come with attending ANC, institutional care, and PNC; these include knowledge, time, cost, distance, workload, and

familial relationships (Sharma et al. 2016b). There is potential for continuity of care during pregnancy, childbirth, and the postnatal period – this may ensure positive maternal health and reduce morbidities and mortalities (Section 1.3).

The qualitative data also offered insights into changes between the time points and the areas. As seen, there were improvements in infrastructure and health practices (Section 6.4.1.1), maternal health awareness (Section 6.4.1.3), and women in the intervention area were more expressive/forthcoming in the interviews. There were also differences in responses between GTN group members, and non-members (Section 6.4.1.10). Overall, it appeared as though participants were more aware of their health and maternal health practices in the intervention area than in the control (Section 6.4.1.7). It was highlighted during the interviews that GTN worked with health workers in the area to improve maternal health and contraception practices (Sections 6.4.1.4 and 6.4.1.6.1).

In the qualitative findings, it came across that the process of diffusion of behavioural change within the population was through group members and health workers in the area. For example, both seemed to recall additional maternal health practices for the area. The added value of qualitative work was that it explained that the changes occurred via the positive spillover effect (Section 2.2), as GTN's groups worked with the whole community and not only the women in need of maternal health (Sections 6.4.1.1, 6.4.1.6 and 6.3.1.6.1). These community-based health promotion trials are more comprehensive as they are more holistic, i.e. not only concerned with health outcomes, as health promotion is concerned with salutogenesis (Section 2.2). Previous studies have found that a programme for improving birth preparedness in Nepal through women's groups increased their knowledge of obstetric danger signs but there was almost no change seen in the proportion of births involving an SBA (McPherson et al. 2006). As in the GTN intervention, despite an increase in awareness, other barriers to healthcare such as the cost of getting to a facility persisted. In health promotion, changing awareness versus changing behaviour is a common challenge in health promotion programmes; a mass media campaign may be beneficial in yielding positive changes in health uptake behaviour (Angus et al. 2013; Wakefield et al. 2010). The paper by Liu and colleagues (2010) using DiD on

evaluating a Safe Motherhood health system strengthening intervention found that despite an improvement in ANC uptake, there was no impact on delivery care. This is contrary to Ensor and colleagues' (2013) DiD evaluation of mothers' groups to improve both understanding of maternal health and of providing access (transport) to maternal healthcare services. They found improvements in delivery care due to provision of transport; yet not in the proportion of women who received antenatal and postnatal care.

In addition, there seem to exist sociocultural barriers to care. In the qualitative evaluation, shyness/timidity was referred to when considering accessing health services, although this changed in the last five years as women mentioned feeling empowered (Sections 6.4.1.7 and 6.4.1.8). Studies have found that women felt shy to be seen by an "unfamiliar" person i.e. not a relative or a male practitioner (Milne et al. 2014; Morrison, Thapa, et al. 2014). The interviews in the evaluation in this thesis also highlighted a number of key barriers to the first phase of delay of the *Three Delays* framework (Section 1.3.2), and they are common to both sample sites despite their different geographical locations and their capacities (equipment). The findings resonate with the literature predominantly collated from women's perspectives (Acharya et al. 2014).

By the final survey, women were one and a half times more likely to attend PNC by year 5 (Section 5.7). The qualitative findings suggested that in the postpartum period, women did not attend due to social constraints imposed on them, for example, being isolated or needing to rest (Section 6.4.1.9). Other literature had found similar reasons for PNC uptake being globally low for similar reasons to not attending ANC (Warren et al. 2006; Khanal et al. 2014; Sharma et al. 2016b). Yet in those studies where women went to their mothers' home, they were likely to receive a higher level of psychosocial care (Sections 1.3.1 and 2.3).

Finally, unintended consequences can occur during the delivery of intervention as highlighted by the mixed-methods evaluation (Sections 1.2 and 3.2.3.2). For instance, the qualitative study pointed to women's husbands being more supportive during pregnancy and childbirth (Section 6.4.1.6); this may mean they can play a supportive role in birth preparations

and PNC if they become a part of or involved in a health promotion intervention as discussed by Mullany and colleagues (2007) and Sharma and colleagues (2016b), and in Section 2.6.1. For maternal health, men (generally) have yet to be seen as part of the "solution" (Sternberg & Hubley 2004).

In the qualitative study it was suggested that disrespectful maternity care affects the enthusiasm of women to attend institutional deliveries. First, women are often treated as second-class citizen or are marginalised in LMICs, and as a result are prevented from attending health services (Sections 1.4.1, 2.3, and Section 6.4.1.9). Secondly, studies have ascertained that poor quality of care at health facilities may act as a barrier to pregnant women and their families accessing skilled care; as in LMICs the care women receive can be rude, disrespectful, and/or abusive. This leads to a violation of trust and poor quality care in the long-term (Rosen et al. 2015; Millar et al. 2016). The WHO (2015), among others such as the seven domains of disrespect and abuse (D&A) outlined in Bowser and Hill's Analysis and The White Ribbon Alliance Respectful Maternity Care Charter: The Universal Rights of Childbearing Women, stated that health systems must be responsible for the treatment of women during childbirth (Bowser & Hill 2010; WRA 2011). Health workers being overworked and underpaid (due to intractable health system problems) can lead to poor morale, compassion fatigue, and as an unintended consequence of disrespectful treatment of clients, and fellow providers. Therefore, there is a continued need for programmes designed to improve the quality of maternal healthcare, with a strong focus on respectful care. The latter should be an essential component of quality care for healthcare providers at all levels. Staff, therefore, require support and training to ensure that childbearing women are treated with compassion and dignity promoting evidence-based practices, and clientcentred and respectful maternity care services. Especially since educational interventions are an effective method of changing how providers communicate (Davis et al. 1995). Moreover, in the community, there should be an inclusive process that promotes the participation of women (Sharma et al. 2016b). Finally, strategies to track and continuously improve respectful care need to be measured and analysed - where disrespect and abuse is consistently identified and reported, and that locally appropriate preventative and therapeutic measures are implemented.

7.2.3 Groups

Community mobilisation through participatory women's groups to improve maternal and newborn health in rural settings have focused on health outcomes, such as neonatal mortality, stillbirth rate, pregnancy-related mortality ratio, and maternal mortality ratio (Houweling et al. 2007; Manandhar et al. 2004; Azad et al. 2010; Tripathy et al. 2010; More et al. 2012; Lewycka et al. 2013; Colbourn et al. 2013; Fottrell et al. 2013). Yet their effectiveness is debated on maternal mortality (WHO 2014b). However, one study found that community mobilisation and groups led to a reduction in neonatal mortality (Fottrell et al. 2013). Brody et al. (2016) did not find evidence for statistically significant effects of groups on women's psychological empowerment; however they found that women's groups (with a focus on economics) have positive effects on economic and political empowerment, women's mobility, and women's control over family planning.

Women's groups in Nepal have run the intervention over a shorter time period (<5 years): the MIRA trials the groups were run for 2.5 years (Manandhar 2004). As a result, the current recommendation is that groups should be run for no shorter than 3 years (WHO 2015). Furthermore, Prost and colleagues 2013 suggests that groups' studies ought to be placed into 'low coverage' (i.e. cut-off at <30% of pregnant women in the intervention area reached by the intervention) and 'high coverage'. Prost and colleagues' (2013) study, contrary to this evaluation, found no effect on use of antenatal care (receiving any/receiving recommended number of visits). Finally, similar to GTN, a study in Bangladesh found men rarely attend groups (Houweling et al. 2011).

7.2.4 Cost

For any intervention to improve maternal healthcare, it is important to know whether it is cost-effective, sustainable, and scalable (Ensor et al. 2009; Prost et al. 2013b). Although, a full cost-effectiveness analysis was outside the scope of this study as a far more sophisticated (health) outcome set was needed (Section 4.3.9.1.7). Key costs involved have been examined. As

seen in Section 3.4, if this type of community interventions is introduced into the health system, they need to take into account the importance of time and cost in delivering and evaluating health promotion. The costs of implementing and running intervention were described in Section 5.11. This latter analysis also included per percentage point increase, i.e. to increase ANC uptake by 1% in a population of VDC 8,569 (Table 19). The main startup costs were for training and group activities. The main recurrent costs were for salaries and transportation to the field. Here, the evaluation cost was close to 10% of the programme cost. The cost of evaluations has been debated (Section 3.4). If kept low as here, and in a study in Kenya with an evaluation cost of 17% of the programme cost, the majority of funds may be effectively used on programme activities; and be cheaper than when compared to the cost of an RCT (Dhaliwal et al. 2011; Larson and Wambua 2011). In the future, costs such as incremental cost per health promotion group, and impact on uptake outcomes would be of benefit to interventions like these. The economic case for investing in Safe Motherhood/maternal health promotion interventions is needed as little detailed evidence exists regarding the relative cost effectiveness of antenatal care, post-abortion care, and essential obstetric care (Jowett 2000).

7.2.5 Empowerment & decision-making

The DiD and qualitative analysis (Sections 7.2.1 and 7.2.2) revealed that there were other reasons for empowerment in addition to GTN (Sections 6.4.1.3 and 6.4.1.7). All the maternal indicators improved in time, but some of them were not directly caused by the intervention. For instance, women increased their level of empowerment in terms of autonomous decisionmaking within the family (Section 6.4.1.8), and may have potentially had an impact on maternal attendance outcomes. In this section, the nature of the intervention and its impact on empowerment is discussed. There was evidence that empowerment increased as seen by women deciding for themselves or with their family members when seeking ANC and delivery care (Table 7 and Section 6.4.1.8). However, decisionmaking/empowerment was not included in the estimated models as the trend was captured both by time and education (education, age, and parity were strongly correlated to the intervention). As highlighted in Section 1.4.4, both 'education level' and 'time' are associated/linked with empowerment. In other words, it was not straightforward to identify the impact of the intervention on empowerment in a context of general improvement in both women's conditions (e.g. education) and in healthcare attendance possibly caused by other factors, such as local literacy programmes and so on. During 2007-2010, the ANC increase appeared to not be caused by GTN intervention but was largely due to other determinant as witnessed by the significance of the variable time in both midline and overall regressions (Section 5.7). Either the progressive improvement in women's level of education (Section 5.3, Table 6) or the empowerment (Section 6.4.1.7) within the household may have played a role.

Although there are overarching reasons for non-attendance during pregnancy due to the patriarchal family structure in Nepal (Section 1.4), women have reduced decision-making power with regard to their reproductive health (Sections 6.4.1.7 and 6.4.1.8). Similarly, findings were identified by Puri et al. (2011). However, the qualitative findings indicated that patterns for decision-making had changed, with women deciding themselves or with their family members both when seeking ANC and delivery care (Section 6.4.1.8).

Therefore, complex relationships are likely to exist among education, empowerment, maternal outcomes, and the health promotion intervention as seen in studies looking at empowerment and health indicators (Varkey et al. 2010). As studies have found in maternal health programmes, women's participation in decision-making is essential whether they make decisions alone or jointly with their husbands or family (Acharya et al. 2010).

To summarise, this evaluation has largely achieved its aim, which was to compare the effectiveness of health promotion in a LMICs and to measure the uptake of ANC, DC, and PNC over the five years of the intervention. The objectives of the research are addressed by using DiD to assess the impact of the intervention on maternal health attendance, conducting a costing exercise, and detailing the cost of the evaluation as compared to the intervention. The research was also concerned with perceptions of change due to increased knowledge of maternal health, and any facilitators or barriers to uptake in relation to decision-making to attend care (Section

3.6.1). If more time had been available, further qualitative research could have been conducted at the same time of the midline survey. Although, the broad qualitative study design has helped to delve into areas that may not have been had it been narrower. The cost exercise in this thesis had limitations, as there are missing data for some of months. These were extrapolated from the data available; this estimation reduces reliability. In addition, the trainers may have underestimated some of the training costs. There are several aspects that may have been overlooked, such as controlling for group members or caste. In the former's case, it is argued that this study was concerned with the public health, and moreover the health promotion argument, and that a wider population benefits from interventions like these, not only those women in the groups in the evaluation. Finally, wealth as a potential confounding factor was more precise than caste. The study limitations and strengths are presented in further detail in the next sections.

7.3 Research limitations & strengths

In this section, the study limitations and strengths are presented. They include reflections on study design, sampling frame (total population), secondary analysis, reflections on social context, reflections on analytical approaches, validity (DiD, themes/pilot), and theoretical and researcher influence. As with all public health studies, the study has weaknesses. This section ends with the strengths of the research.

7.3.1 Limitations

The limitations of the evaluation include issues with regards to the methods and the time available to conduct the research. The principal limitation was time and money involved in conducting the interviews. This was a particular issue as the interviewing took place during June 2012 in Nepal, which is the monsoon season. This therefore posed further difficulties for the researcher in terms of reaching certain villages. Moreover, it proved arduous to speak to women: as they were busy prior to and during the start of the monsoon season, as this is the typical time to plant crops. Hence conducting fieldwork was more time-consuming due to "finding/gathering" participants as they might not attend groups/health-post clinics due to the heavy rains.

The aim of the evaluation was to capture a broader (spillover) effect of the intervention on the local population in that women surveyed were not necessarily the ones who received the intervention. The DiD method did not provide an analysis of those who were in the groups, although the qualitative analysis did.

Also, it was not a follow-up study with a control, which would have been more precise in determining effects over the five years on individual women. There is also a positive way of looking at this; there might have been positive/beneficial spillover effect occurring for those living in the intervention area, where those not part of the intervention still benefitted from the health promotion activities by word of mouth or simply communicating the information received (Section 2.2).

Another limitation is that some women interviewed might have been the same women during the baseline, midline, or final evaluation, which raises the issue of recall bias (Section 1.2).

Furthermore, the data are limited. The secondary data do not yield any financial data on healthcare uptake, nor did the survey query distance from home to health post/birth facility. Also, no questions were asked about the quality of care including: 1) the presence of male health workers at the facility; 2) other factors that might account for the changes found; and 3) lack thereof between the data collection points.

Another limitation is that it is difficult to ascertain what impact on longer-term maternal health the intervention had, as maternal mortality was not a primary outcome of the programme. As seen in Section 1.3, there exists the problem of linking long-term and relatively rare health outcomes to "real-world" community-based interventions (Gruber 1994; Bhutta et al. 2005; Reynolds et al. 2006; Waldinger 2010; Bouvier-Colle et al. 2012).

Language was also an issue, as the author did not speak Nepali. The research protocol detailed that the research translator briefed the

participants prior to the interview, and where possible, expanded on any terms and expressions particular to the Nepali, Newari, and Tamang language that could not be directly translated to English in order to be as comprehensive as possible (Sharma et al. 2015). In Section 5.3, 40.67% of the respondents were Tamang - a Tamang speaking moderator would have been beneficial. Also, during the process of the focus groups, there were three moderators and two transcribers/translators for this study from different ethnic backgrounds, each with their own range of experiences. The latter may mean that there are certain variances in the translation of the data despite the consensus aimed for.

As mentioned in Section 3.6, it is difficult to reconcile the stakeholders' diverse expectations of the purpose of an evaluation. For instance, during the interviews, the participants from the intervention area, the mothers-in-law groups and the GTN health promoters felt that this was an evaluation of the project, and thus they may have given "pleasing" (which is a type of bias) answers. This forms a threat to the study's validity. They felt a need to justify the continuation of the groups whereas the health promoters felt it was an evaluation of their job; some of their responses seemed to be a justification of their actions (van Teijlingen et al. 2013).

Issues in qualitative research also include the duration, interruptions, and audio-quality of the focus groups and interviews. Elevated background noise was often captured, as the interviews took place in the "open" field and near the road. These background noises were noticeable in the recordings and were noted during the transcription.

7.3.2 Strengths

One of the main strengths of this evaluation is the use of DiD analysis, as very few studies have used it in maternal health and LMICs. Furthermore, no other study has applied it in maternal health in Nepal. In addition, DiD is relatively low-cost (Section 7.2). However it is stressed that DiD could not capture all pre-existing differences in the control area compared to the intervention area. For example, there may have been interventions that the researcher was not aware of taking place in the control area (locality). Therefore, a qualitative study was needed to identify the "why and how"

behind what worked. This also helped determine what barriers existed, and if the intervention did address these, why and what the additional measures to improve the use of birth facilities and skilled birth attendants were. Thus using mixed methods is a strong point, as the combination of both quantitative and qualitative provides a deeper insight into the rationale/motivation underpinning the statistics/numbers (Section 4.2).

There is also strength in the outcomes selection - in this study there are counterfactual outcomes and a positive spillover effect, that is outcomes for participants who were not exposed to the programme and yet who benefit (Section 3.2.3.2). This thesis used proxy outcomes for health (Sections 1.3.1 and 4.3.9). In addition, these helped to look at improving maternal uptake, towards eventually reducing maternal morbidities and mortalities as they are linked.

Conducting a process, and impact, evaluation provides a strong way to address the complexity and flexibility of the GTN intervention (Section 2.5.4). For instance, as stated in Section 1.2 and in the limitations section (Section 7.3.1), there are challenges to measuring long-term health outcomes in health promotion interventions. Nonetheless, in an evaluation an intervention is "judged" beneficial if we see an increase in a timely and effective use of services and improved psychosocial state and outcomes. Furthermore, this evaluation shows that maternal health access is the vehicle for women in making their own decisions (empowerment) for reproductive health. Therefore, health promotion empowers women in the long-term to help them gain access to services they did not know about or could not attend due to power relations in the family.

Moreover, it requires knowledge and experience to conduct this type of evaluation, as was highlighted in the introduction (Section 1.2). The researcher knew a lot of key issues beforehand so could probe into maternal health and cultural issues, when the need arose in the qualitative study. In addition, the student has published a peer-reviewed article on Nepal (van Teijlingen et al. 2012; Joshi et al. 2013; Sharma et al. 2015; Sharma et al. 2016a; Sharma et al. 2016b and Sharma et al. 2017) and was able to ask questions on maternal health, cultural practices - being Hindu and general female issues. This study used primary data in addition to secondary data. Using secondary analysis was firstly a plus point, as a study of this scale of work as a primary study would not have been feasible due to the time and funding constraints of a PhD. Second, the secondary analysis data - frames were cleaned by the researcher prior to the analysis to obtain a fuller understanding in addition to the primary data. The third plus point was the high response rate of the survey that provided the secondary data, i.e. the near total coverage of the population (Section 4.3.4.2). An additional strength was that in the primary qualitative study, women, mothers-in-law, and men were also interviewed separately, which allowed them to speak about any issues anonymously.

The main translator had a health background and was trained prior to the research, as the interviewer spoke Hindi and a few Nepali words. This helped ensure the quality of the data. Also, the researcher was prepared for every eventuality: noise and interruptions and checked the recording for the transcription. In addition, the same translator was used throughout, thus providing consistency. A Newari-speaking translator had an added benefit (Sections 4.3.8 and 6.4.1.10). Two Nepalese translators who transcribed four of the interviews, independently of each other, ensured the accuracy of translation. The transcripts were then verified. Using a translator enabled access to the wider staff body, all of which have a role to play in providing childbirth services and thus influencing women's perceptions. One of the strengths of this study was the concurrent use of observations and semi-structured interviews (Section 4.3.8.2).

Richer data were provided by the frequent group members (women) who were more "open", i.e. willing to answer questions after the participants felt comfortable, as they were freely (and in confidence) able to express their views in the group discussions and in particular those mothers' groups with young children (<2 years old) (Section 6.4.1.10).

Finally, this thesis benefitted from a large and broad supervisory team with expertise in statistics, economics, qualitative evaluation, maternal health, LMICs experience, and mixed-methods research.

The chapters that follow outline the conclusion from the thesis and the recommendations from the research findings.

Chapter 8 Thesis conclusion

The quantitative research in this thesis leads to the conclusion that the GTN health promotion intervention appeared to have had a beneficial effect on selected maternal health-seeking behaviours. It improved in the maternal health services attendance outcomes in the intervention area relative to the control area for three of the six measured non-health (proxy) outcomes. Time is a factor in responding to the intervention. The improvement in PNC appeared to be subject to a "delay effect" (Section 7.2.1.1). This thesis also concludes that the GTN intervention had a greater spillover effect influence on the uptake of community-based ANC and PNC than on facility-based birth. ANC alone does not improve facility-based care in a health promotion intervention.

It can be concluded from the qualitative research that the intervention area's increase in awareness and empowerment is at least partly due to the GTN intervention. A more tentative conclusion is that the role and place of women in society probably has much more of a negative effect on postnatal women (Section 7.2.2).

The research contribution to new knowledge is that DiD is a suitable method to evaluate a complex community intervention in comparison to using expensive and cumbersome trials (Section 7.2). An analysis such as DiD provided a level of precision not available in simpler analysis, such as a before-after analysis of percentage change or Chi-square.

More generally, a mixed-method evaluation enables a more rounded understanding of potential causes of care-seeking behaviour in maternal health. Furthermore, one should not just do a quantitative analysis without a qualitative part, the statistics obtained may not imply anything if the context 'why' is not given, as solely percentages are inference without context (Section 7.2.2).

A control (area) is useful for comparison. The reader is drawn to the fact that health promotion interventions and their evaluations are complex yet a control provides the answer to the "what if" question, or what would happen if participants had not been exposed to the intervention.

8.1 Evaluation context

Evaluation is a key part of health promotion (Chapter 3) and the methods used in this thesis enable programmes or NGOs to add to their toolkit of evaluation design to provide accountability to the various stakeholders, assess if they have effectively met their programme aims, adapt their activities for future undertakings, and contribute to evidence in research (Sections 1.2, 2.2, and 3.2). As the main aim of the PhD was to evaluate what works in order to inform future implementations or upscaling (Sections 1.5 and 3.2), the key finding is that GTN was evidence-based and other projects aiming to be effective should be evidence-based too.

The apparent conclusion from the study results is that health promotion groups are effective in improving access to health. First, the increase in ANC uptake might lead to more women seeking delivery by SBA, and to the potential for postnatal care and rounded maternal care. Second, groups played a role in ensuring the continuation of care by increasing ANC and PNC, which are effective to target maternal and neonatal morbidity as well as infant mortality (Section 1.3).

There is a continuing need for conducting evaluations, as a review of quantitative methods and models of impact evaluation estimates how measured changes in wellbeing are attributable to a particular project or policy intervention (Sections 3.2.3.2 and 4.3.9.1.4). While this expanded range of methods for evaluation offers practical solutions to many of the problems facing health promotion evaluation, they are not a rapid, complete or easy answer. Scientific outcome methods for evaluating health promotion programmes have limits (Sections 3.2.3 and 3.6). Furthermore, the intervention was complex (several activities in the community) and unintended effects occurred. It is not completely possible to separate out the effect of the individual components, yet the qualitative analysis helped to a certain extent. As shown in this study, an effective evaluation should therefore be able to assess precisely the mechanisms by which beneficiaries are responding to the intervention (Manandhar et al., 2004; Osrin et al.,

2009). Evaluation, as seen in Section 1.2, is concerned with evidence and effectiveness for accountability and replicability/scalability. However, social science experiments are hard to replicate as often they are tailored to the existing population (Sections 1.2 and 3.2.3). Maternal access might be an issue in one area whilst in another the main issue might be sexual violence. The other query is whether there has been enough evidence/trials to warrant upscaling or to stop testing each new application of a development idea (Kremer 2003; Duflo 2004; Hobbes 2014). If an intervention is scaled, a rigorous evaluation of programmes' impacts can be a shared or public good and therefore scaled up (generalisable); the future application is that it offers reliable guidance to international organisations, governments, donors, and non-governmental organisations (NGOs) in their continual search for effective programmes (Section 1.2).

Therefore, it is beneficial to conduct integrated evaluations alongside complex community-based interventions in a transparent, measurable fashion (Judd et al. 2001; Duflo 2012; Datta and Petticrew 2013). Such rigorous scientific tests and controlled trials for social policy are needed to take the guesswork out of policy-making by knowing what works, what does not work, and why (Duflo 2012). In the hierarchy of evidence for effectiveness (Sections 1.2, 3.1, and 3.2.3), RCTs are ideal even in health promotion interventions, as seen in Section 2.6.1, where other costlier trials have used them. Yet as health promotion does emphasise working with the resources available, it lends to the idea/belief that appropriate less costly and more setting appropriate methods ought to be used, such as DiD.

One of the fallacies/misconceptions is that maternal health is (only) a "woman's issue". This evaluation has shown that in order to sustainably empower women to access health services, health promotion is needed and there is also the need to involve those who either make decisions for the women or with them: men, mothers-in-laws, their families or community 'leaders' (traditional healers, see Section 6.4.1.3.1).

To continue to improve awareness of maternal health and access, an investment can be made in health promotion. However, a concomitant investment must be made to strengthen the health system in general. It lends to that health promotion should be provided within the health system

and whichever model of care it suits whether primary, secondary or tertiary levels. This can be achieved by strengthening the health promotion capacity of health workers, skilled maternal, and newborn health workers (Section 1.4.1). If they are in short supply, community-based mobilisation may be a solution to empower individuals by improving access to knowledge and services (such as ANC and PNC), as seen in Section 2.4. It is cautioned that a similar point has been made with respect to conditional cash transfers, which increase demand for schooling but may not necessarily improve learning outcomes or even enrolments if there are supply-side constraints (White 2009). A programme such as this can improve uptake but not necessarily health outcomes over the long-term if the health system has constraints in providing quality care (Sections 2.4 and 2.6.1).

In summary, measuring the effects of a community-based intervention is not straightforward because of confounding factors in the wider social, ecological, environmental, and political arena. For instance, a large percentage of women may still prefer to be cared for at home by family members or other unskilled birth attendants. As mentioned in Chapter 1, this evaluation is important to the researcher. It was discussed that she wished to understand what works best for whom and to contribute to the body of accountability in programmes and inequities in health, in particular to the marginalised populations in LMICs. Ireland et al. (2015) have discussed that research in the field, such as Nepal, changes one's personal and professional knowledge. The researcher wished to understand the process of change or empowerment in marginalised women, and in that process, she herself changed and gained confidence in her own skills. As Chapter 1 detailed, health promotion is complex and intricate relationships exist. Thus the value of using mixed methods is that they address these issues. They measure and detail the complexity in order to interpret the findings. While Chapter 2 explored health promotion and behaviour change, here it is concluded that positive maternal health cannot be achieved solely through empowerment and women's groups. The effectiveness of women's groups has been discussed. These indeed are "means to an (public health) end" and can contribute significantly as seen here, particularly in community maternal health (Sections 2.5 and 2.6.1). However, public health needs to go the extra mile and improve the health system infrastructure and transport for these women to access delivery care. For instance, the SDGs call for

continued improvements in maternal health. If these evaluation results were to be placed in context, policy and planners now would need to address and strengthen the referral between the community and the facility using health promotion. Therefore various levels need to be included, such as grassroots, civil societies, and other social movements in health promotion activities. At a policy level, the next steps should be to build the capacities of grassroots organisations to work directly with communities. As this may yield progress towards capturing social indicators of change in attitudes, of empowerment, of support, and of those attributes of societies and relationships to strengthen (as GTN did) rather than to victimise. Towards these, the SDGs will also focus on countries that need to ensure that continued progress outcomes are recorded and analysed to ensure continuing progress to gender and development and women's rights with an additional need to focus not only on the economic but also on the social situation of women.

Chapter 9 Recommendations from the thesis

9.1 Overview

This thesis offers a set of different recommendations for Green Tara Nepal (GTN), other practitioners/implementers, policymakers, educators (health promoters), funders, researchers and those interested in implementing and evaluating community-based maternal health promotion interventions using groups.

9.2 Recommendations for GTN

As discussed in Chapter 1, GTN worked in is a particular social and geographical setting hence recommendations need to be seen in this light.

For (future) interventions' implementation:

- a) The two health promoters showed that it was possible to reach over two thousand individuals by being mobile (walking and scooter use). It is recommended that future programmes aiming to improve health access in poor, rural communities, particularly for those women who do not/cannot leave the home, consider using similar community mobilisation strategies complimented by the use of mobile clinics.
- b) The future design of the groups should be adapted to the level in question for example to account for preference for learning, such as oral versus visual methods, language, ethnic group mix or caste, etc.
- c) It is recommended to include a health promoter who can speak local languages while working with marginalised women (Sections 6.2 and 6.4.1.10).
- d) GTN had to deal with the (high) expectation(s) from the community, and often expectations from stakeholders are overwhelming. Therefore, health promoters should always explain what the direct benefits are (Sharma et al. 2017).
- e) Health promoters should limit the size of groups to deal with participation particularly if they feel overwhelmed, e.g. due to the large size of the groups; or if there is a loss of interest during meetings of the group and/or the time they have on offer to the groups.
- f) In the future GTN should grow "leaders of change", e.g. women who have been coming to the groups for 1-2 years as they may help to ensure health

promotion activities are continued. These leaders of change can perform a number of activities within the groups, such as training community members in safe birthing techniques, generation of community funds for maternal and infant care, stretcher provision schemes, distribution of clean delivery kits, home visits by women's group members to newly pregnant mothers, awareness raising with the help of video films in GTN's possession, social and psychological support, support for early initiation and maintenance of breastfeeding, etc.

- g) In the qualitative study, it was highlighted that men would like to join the GTN groups (Section 6.4.1.6). GTN should make further efforts to have male groups; perhaps this can be achieved with the appointment of a male health promoter.
- h) The qualitative study also showed that cultural beliefs and traditional/cultural practices around pregnancy were harmful, e.g. isolation or abuse of women in the postnatal period. GTN should address these ineffective practices in the future with evidence-based health promotion training to circumvent their continued practice. On the referral side, the GTN health promoters may consider exploring local institutions interested to collaborate with GTN, particularly in the areas of violence, often domestic and/or sexual. This latter exercise may also help manage the health promoter's time (and group size), as it will delegate certain tasks to these organisations.
- i) Health programmes and their evaluation need to be tailored for the needs of the community since a one-size-fits-all approach is not suitable. From the start, the local stakeholders should be involved in the needs assessment, this process improves the chances of empowerment occurring, of programme ownership and ensures sustainability in the long-term.

For the research and evaluation process:

- j) Continue to conduct research and evaluation around a humanitarian setting; and use methods such as DiD, qualitative, and costing methods where appropriate.
- k) GTN should consider continuing their endeavour of collecting and storing these rich data as it was at a relative low-cost (Section 5.11). Furthermore, it is recommended to consider the use of paperless technology (i.e. mobile phones) to collect field data.
- Health promotion interventions like GTN's often take place at a community level. The expected proportional benefits to individuals can be small, and

beneficial outcomes are delayed, particularly with regards to the health outcomes in these from health promotion exercises. Organisations, like GTN, who wish to continue or upscale their activities, will need to plan a rollout along a longer timeline and negotiate at the regional and national level for support.

- m) As in public health when there are changes in diseases patterns, for instance the growing burden of non-communicable diseases, GTN should seize the opportunity to test 'new' concepts on prevention/management of health promotion activities in these topic areas.
- n) There are growing numbers of open access journals, and interest from various organisations to set up a similar intervention. GTN participatory research activities ought to be published (better disseminated) so that others aiming to do this important work of improving health uptake via community mobilisation may also learn/benefit.

9.3 Recommendations for researchers

- a) In the first instance, DiD is recommended for evaluating quasiexperimental study designs to assess an NGO's or health promotion intervention's activities in LMICs.
- b) DiD can also help provide details of confounding factors, such as education.
- c) In addition, mixed methods are recommended as they use both qualitative and quantitative methods to answer what impact a complex evaluation has had in a community. The use of these methods also helps determine if a programme has had any unintended consequences.
- d) Costing an evaluation is difficult and should form part of evaluation. A rough costing of DiD would be useful in the future.
- e) As a result of being part of the intervention, women's empowerment may have increased. Complex relationships between education, empowerment, maternal outcomes, and health promotion may exist and further research will be devoted to shed a light on this.
- f) For the academic research community, the continuing way forward should ensure that more research is undertaken into social science in LMICs interventions. It is imperative that this research is applied to improve health services for Nepal's largely rural communities and that social policy can be shaped to be 'inclusive' of those marginalised.

9.4 Recommendations for practitioners

- a) Recommendations for practitioners include training/education in health promotion and community mobilisation. For example, health promotion officers employed in rural areas need to be trained in evidence-based health promotion to help them fulfil their role better in the community and thus assist them to determine the best way to improve maternal wellbeing and/via women empowerment.
- b) Since health promotion offers 'something for everyone', the groups may provide an impact into the wider community due to a spillover effect as health promoters may play a promising role in providing pregnancy and childbirth care, mobilising communities, and improving uptake outcomes in LMICs.
- c) In terms of service delivery in maternal health, other interventions, not just ANC, are needed to improve delivery care. Of all the outcomes, ANC improved, yet in the intervention area did not lead to an increase in uptake of SBA as suggested by a simpler analysis and the literature (Section 7.2). This means that ANC should not be targeted with the intended aim to increase institutional deliveries. Thus, additional measures are needed to improve the use of birth facilities and skilled birth attendants.
- In the future, the health promotion sessions may have to differ from those currently offered by GTN, for example specific training sessions for seeking a SBA at birth.
- e) In addition, one should not only do ANC. There are several components that are required to improve delivery care and PNC.
- f) In order to improve access to these services, an intervention in the health system should address both the supply of health workers and the transport to maternal health services.
- g) Also, it is not enough to 'get' women to care. The quality of care should be of importance.

 h) Part of addressing this lack of empowerment should involve training health workers in community mobilisation to include mothers-in-law and the men.

9.5 Recommendations for policy-makers

People who design interventions addressing the role and place of women in society and their access to health probably should take into account it is difficult to do so, especially over a short-term. This should also be taken into account if said interventions wish to consider evaluating their activities over the short-, mediumand long-term. Women's or mother-in-law groups may form part of a strategy, as they may be empowered to change or be part of a change with the information received during the groups. As seen in Chapter 1, government commitment was also a key ingredient in the success of mortality and fertility (Sections 1.3 and 1.4.4). This is similar to Bangladesh, (World Bank 2006); see Section 1.4.4 - Table 1. For policy-makers, recommendations from this evaluation include that maternal care interventions in Nepal and other LMICs should provide focused programmes for rural, uneducated, poor women so that they may delay childbearing; attend antenatal clinics and delivery care in case of pregnancy and postnatal care. These behavioural change health promotion programmes ought to include the accommodation of socio-cultural barriers; in order to achieve long-term change, with a slow build-up and not expect to see change overnight.

Appendix

Appendix I - Ethics Letter



Website: http://www.nhrc.org.np, Email : nhrc@nhrc.org.np

Appendix II - Table 2, Definitions & Sources

1. Population millions (2013) rounded off to closest million

Population, total: Total population is based on the de facto definition of population, which counts all residents regardless of legal status or citizenship--except for refugees not permanently settled in the country of asylum, who are generally considered part of the population of their country of origin. The values shown are midvear estimates. (1) United Nations Population Division. World Population Prospects, (2) United Nations Statistical Division. Population and Vital Statistics Report (various years), (3) Census reports and other statistical publications from national statistical offices, (4) Eurostat: Demographic Statistics. (5) Secretariat of the Pacific Community: Statistics and Demography Programme, and (6) U.S. Census Bureau: International Database. Catalog Sources World Development Indicators

http://data.worldbank.org/indicator/SP.POP.TOTL

2. Literacy rate, adult total (% of people ages 15 and above)

Adult (15+) literacy rate (%). Total is the percentage of the population age 15 and above who can, with understanding, read and write a short, simple statement on their everyday life. Generally, 'literacy' also encompasses 'numeracy', the ability to make simple arithmetic calculations. This indicator is calculated by dividing the number of literates aged 15 years and over by the corresponding age group population and multiplying the result by 100. **UNESCO** Institute for Statistics

Catalog Sources World Development Indicators http://data.worldbank.org/indicator/SE.ADT.LITR.ZS

3. Life expectancy at birth, total (years)

Life expectancy at birth indicates the number of years a newborn infant would live if prevailing patterns of mortality at the time of its birth were to stay the same throughout its life.

Derived from male and female life expectancy at birth from sources such as: (1) United Nations Population Division. World Population Prospects, (2) United Nations Statistical Division. Population and Vital Statistics Report (various years), (3) Census reports and other statistical publications from national statistical offices, (4) Eurostat: Demographic Statistics, (5) Secretariat of the Pacific Community: Statistics and Demography Programme, and (6)

U.S. Census Bureau: International Database.

Catalog Sources World Development Indicators

http://data.worldbank.org/indicator/SP.DYN.LE00.IN

4. Maternal mortality ratio (modelled estimate, per 100,000 live births)

Maternal mortality ratio is the number of women who die from pregnancy-related causes while pregnant or within 42 days of pregnancy termination per 100,000 live births. The data are estimated with a regression model using information on the proportion of maternal deaths among non-AIDS deaths in women ages 15-49, fertility, birth attendants, and GDP. WHO, UNICEF, UNFPA, The World Bank, and the United Nations Population Division. Trends in Maternal Mortality: 1990 to 2013. Geneva. World Health Organization. 2014 Catalog Sources World Development Indicators

http://data.worldbank.org/indicator/SH.STA.MMRT

5. Neonatal Mortality Ratio/1000 live births (2013)

Mortality rate, neonatal (per 1,000 live births)

Neonatal mortality rate is the number of neonates dying before reaching 28 days of age, per 1,000 live births in a given year.

Estimates developed by the UN Inter-Agency Group for Child Mortality Estimation (UNICEF, WHO, World Bank, UN DESA Population Division) at www.childmortality.org. Catalog Sources World Development Indicators http://data.worldbank.org/indicator/SH.DYN.NMRT

6. Births attended by skilled health staff (% of total)

Births attended by skilled health staff are the percentage of deliveries attended by personnel trained to give the necessary supervision, care, and advice to women during pregnancy, labor, and the postpartum period; to conduct deliveries on their own; and to care for

newborns.

UNICEF, State of the World's Children, Childinfo, and Demographic and Health Surveys by ICF International. Catalog Sources World Development Indicators 2011-2013

http://data.worldbank.org/indicator/SH.STA.BRTC.ZS

7. Birth rate, crude (per 1,000 people), 2011

Crude birth rate indicates the number of live births occurring during the year, per 1,000 population estimated at midyear. Subtracting the crude death rate from the crude birth rate provides the rate of natural increase, which is equal to the rate of population change in the absence of migration.

(1) United Nations Population Division. World Population Prospects, (2) United Nations Statistical Division. Population and Vital Statistics Report (various years), (3) Census reports and other statistical publications from national statistical offices, (4) Eurostat: Demographic Statistics, (5) Secretariat of the Pacific Community: Statistics and Demography Programme, and (6) U.S. Census Bureau: International Database.

Catalog Sources World Development Indicators 2013

http://data.worldbank.org/indicator/SP.DYN.CBRT.IN

8. Death rate, crude (per 1,000 people) (2013)

Crude death rate indicates the number of deaths occurring during the year, per 1,000 population estimated at midyear. Subtracting the crude death rate from the crude birth rate provides the rate of natural increase, which is equal to the rate of population change in the absence of migration.

(1) United Nations Population Division. World Population Prospects, (2) United Nations Statistical Division. Population and Vital Statistics Report (various years), (3) Census reports and other statistical publications from national statistical offices, (4) Eurostat: Demographic Statistics, (5) Secretariat of the Pacific Community: Statistics and Demography Programme, and (6) U.S. Census Bureau: International Database.

Catalog Sources World Development Indicators

http://data.worldbank.org/indicator/SP.DYN.CDRT.IN/countries

9. Fertility rate, total (births per woman)

Total fertility rate represents the number of children that would be born to a woman if she were to live to the end of her childbearing years and bear children in accordance with current age-specific fertility rates.

(1) United Nations Population Division. World Population Prospects, (2) United Nations Statistical Division. Population and Vital Statistics Repot (various years), (3) Census reports and other statistical publications from national statistical offices, (4) Eurostat: Demographic Statistics, (5) Secretariat of the Pacific Community: Statistics and Demography Programme, and (6) U.S. Census Bureau: International Database

Catalog Sources World Development Indicators 2011

http://data.worldbank.org/indicator/SP.DYN.TFRT.IN

10. Contraceptive prevalence (% of women ages 15-49)

Contraceptive prevalence rate is the percentage of women who are practicing, or whose sexual partners are practicing, any form of contraception. It is usually measured for married women ages 15-49 only.

Household surveys, including Demographic and Health Surveys by Macro International and Multiple Indicator Cluster Surveys by UNICEF.

Catalog Sources World Development Indicators

http://data.worldbank.org/indicator/SP.DYN.CONU.ZS

11. Pregnant women receiving prenatal care (%)

Pregnant women receiving prenatal care are the percentage of women attended at least once during pregnancy by skilled health personnel for reasons related to pregnancy. UNICEF, State of the World's Children, Childinfo, and Demographic and Health Surveys by ICF International.

Catalog Sources World Development Indicators 2011

http://data.worldbank.org/indicator/SH.STA.ANVC.ZS

12. Health expenditure per capita, PPP (constant 2011 international \$)

Total health expenditure is the sum of public and private health expenditures as a ratio of total population. It covers the provision of health services (preventive and curative), family planning activities, nutrition activities, and emergency aid designated for health but does not include provision of water and sanitation. Data are in international dollars converted using 2011 purchasing power parity (PPP) rates.

World Health Organization Global Health Expenditure database (see

http://apps.who.int/nha/database for the most recent updates).

Catalog Sources World Development Indicators

http://data.worldbank.org/indicator/SH.XPD.PCAP

World Health Organization Global Health Expenditure database (see

http://apps.who.int/nha/database for the most recent updates).

World Health Organization Global Health Expenditure database (see

http://apps.who.int/nha/database for the most recent updates).

Catalog Sources World Development Indicators

Source: http://data.worldbank.org/indicator/SH.XPD.OOPC.TO.ZS

13. Out-of-pocket health expenditure (% of total expenditure on health)

Out of pocket expenditure is any direct outlay by households, including gratuities and in-kind payments, to health practitioners and suppliers of pharmaceuticals, therapeutic appliances, and other goods and services whose primary intent is to contribute to the restoration or enhancement of the health status of individuals or population groups. It is a part of private health expenditure. World Health Organization National Health Account database (see http://apps.who.int/nha/database/DataExplorerRegime.aspx for the most recent updates). Catalog Sources World Development Indicators 2011

14. GNI per capita, PPP (current international \$)

GNI per capita based on purchasing power parity (PPP). PPP GNI is gross national income (GNI) converted to international dollars using purchasing power parity rates. An international dollar has the same purchasing power over GNI as a U.S. dollar has in the United States. GNI is the sum of value added by all resident producers plus any product taxes (less subsidies) not included in the valuation of output plus net receipts of primary income (compensation of employees and property income) from abroad. Data are in current international dollars based on the 2011 ICP round.

World Bank, International Comparison Program database.

Catalog Sources World Development Indicators

http://data.worldbank.org/indicator/NY.GNP.PCAP.PP.CD

15. GDP (purchasing power parity) rank/229 [1] 2014

GDP (purchasing power parity)

https://www.cia.gov/library/publications/the-world-factbook/rankorder/2001rank.html Country Comparison: GDP (Purchasing Power Parity)

GDP (purchasing power parity) compares the gross domestic product (GDP) or value of all final goods and services produced within a nation in a given year. A nation's GDP at purchasing power parity (PPP) exchange rates is the sum value of all goods and services produced in the country valued at prices prevailing in the United States.

<u>http://data.worldbank.org/indicator/NY.GNP.PCAP.PP.CD</u> and Source: The World Bank: World Development Indicators: Size of the economy (2013) The World Bank Group. http://wdi.worldbank.org/table/1.1

Appendix III - Questionnaire for Women in English and Nepali

Questionnaire for Women SN.:

VDC:

Ward no.:

Name of village:

Day:

Date: 2010/04/.....

SCREENING Q: DO YOU HAVE A CHILD UNDER 24 MONTHS (Not Completed)

1. Yes 2.No (if no, do not continue questionnaire)

1.0 If yes, how old is your youngest child?Months

1.0 A How many child(ren) do you have under 24 months?

SN	Questions	Coding categories	Skip
1.1	In what month and year were you	Month:	
	born?	Don't Know month	
		Year:	
		Don't Know year	
1.2	What is your age? (Compare and		
	correct 1.1 and/or 1.2 if inconsistent)	Years	
1.3	What is your caste /ethnicity?	1. Brahman	
		2. Chhetri	
		3. Tamang	
		4. Newar non Dalit	
		5. Newar Dalit	
		6. Balami	
		7. Dalit	
		8. Other (specify	
1.6	What is your religion?	1. Buddhist	
		2. Hindu	
		3. Christian	
		4. Other (specify)	
1.7	Can you read and write?	1. Yes	Go
		2. No	► to
			1.10
1.8	Have you ever attended school?	1. Yes	Go to
		2. No	1 .10
1.9	If yes, what is the highest grade you	1. Primary education	
	completed?	2. Secondary education (SLC)	
		3. Intermediate (PCL)	

Section 1: Household and Socio-demographic information

		4. Bachelor and above	
1.10	What is your current main	1. House wife	
	occupation?	2. Farmer	
		3. Service	
		4. Business	
		5. Other (Specify)	
1.11	What is your husband's level of	1. Illiterate	
	education?	2. Primary education	
		3. Secondary education (S.L.C)	
		4. Intermediate (PCL)	
		5. Bachelor and above	
1.12	What is your husband's main	1. Farmer	
	occupation?	2. Teacher	
		3. Business	
		4. Skilled labour	
		5. Unskilled labour	
		6. Other (Specify)	
1.13	How many people live in your	1. Total	
	house?	2. Young People & Adults (age	
		10 or above)	
		3. Children (below 10 yrs)	
1.14	How many rooms in your household	Rooms	
	are used for sleeping?		
1.15##	Do you and your family (household)	1. Land in Ropani-	
	have any property?		
		2. Number of Houses	
		3. Balance in cash (bank/ in	
		hand)	
		4. Yes, but do not know	
		amount	
		5. Cattle (Specify) A.	
		Cow	
		B. Buffalo C.	
		Goat	
		D. Chicken E.	
		Pig	
		F. Others	
		(Specify)	
1.16	Do you own any land or property?	1. Yes	
-------	---	--------------------------------	
		2. No	
		3. Don't Know	
1.16B	Do you have own Mobile Phone?	1. Yes	
		2. No	
1.17	Where are you currently living?	1. In own home	
		2. In rented property	
		3. Living with relative Go to	
		4. Other1.19	
		(specify)	
1.18	If it is your own home, What type of	1. Cemented	
	roof in your house? (Observation)	2. Tin	
		3. Tile	
		4. Hay	
		5. Other (specify):	
1.19	What is the main source of drinking	1. Piped water to own home	
##	water for members of your	2. Common/public piped water	
	household? (Max. 2 Answers)	3. Tube well or borehole	
		4. Surface water (river/dam/	
		lake/ pond/stream/canal/	
		irrigation canal	
		5. Stone tap/dhara	
		6. Jar/Bottled water	
		7. Others	
		(specify)	
1.20	Do you have your own toilet?	1. Yes Go to	
		2. N o 1 .23	
1.21	If yes, how many people use the	members	
	toilet?		
1.22	If Yes, what kind of toilet facility do	1. Flush to piped sewer system	
##	members of your household usually	2. Flush to septic tank	
	use? (max 2 answers)	3. Flush to somewhere else	
		4. Pit latrine with slab	
		5. Pit latrine without slab	
		6. Composting toilet	
1.23	Does your household have:	1. YES	
##		2.No	
		1. Electricity	

		2.	Radio		
		3.	Television		
		4.	Telephone		
		5.	Refrigerator		
		6.	Computer		
		7.	Wall clock		
		8.	Gas Geezer □		
		q	Solar papel	п	
		0.			
1.24	What type of fuel does your	1.	Electricity		
##	household mainly use for cooking?	2	LPG (Gas)		
	(max 2 answers)	2. 2	Biogas		
	· · · · · · · · · · · · · · · · · · ·	⊿	Korosono		
		т. 5	Wood		
		Э. С			
		ю. - 7	Animai dung		
4.00		1.	Other (specify):		O a ta
1.26	Do you have access to a motorable	1.	Yes		
4.07	road? (within five minutes waik)	Ζ.		NA:	1.28
1.27	If no, now long does it take to reach		Hours	Minutes	
	to motorable road? (TIME TAKEN				
4.00	WALKING)			4 1/100	
1.28 ##	Does any member of your nousehold	~		1. Yes	
##	own:	2. r	NU Diavala (Diakahaw	_	
		1.			
		2.	Motorcycle/scoote	r 🗆	
		3.	Tempo:		
		4.	Car/Truck:		
		5.	Other transport		
			(specify)		

1.30	How old were you when you got married?	yrs.	
1.31	How old were you when you had first		
	pregnancy?	yrs	
1.32	How many times have you been		
	pregnant?	times	
	How many live children do you		
	have?		
1.33	Have you had any	1. Yes	
	miscarriages/abortion/stillbirth?	2. No	
1.34	Is abortion legal in Nepal?	1. Yes	
		2. No	
		3. Don't know	

Section 2: Antenatal Care and seeking care – FOR ALL RESPONDENTS

Note: these questions relate to the woman's LAST pregnancy

2.12	How many dose of TT	1 Dose(s)	
	vaccine did you have in		
	your lifetime?		
2.1	Did you take iron/folic	1. Yes	
	acid (vitamin tabs) during	2. No	Go to 2.5
	pregnancy?	3. 3. Don't know	
2.2	If yes, for how long did	FromWeeks to weeks	
	you take them?	of pregnancy	
		Forweeks after delivery	
2.3	Where did you get these	1. HP/SHP	
##	tablets?	2. NGO/Manmohan Hospital	
		3. Private doctor or clinic	
		4. Pharmacy	
		5. Local health worker/FCHV	
		6. Outreach clinic	
		7. Hospital in Kathmandu	
		8. Others (Specify)	
2.4	Did you/anyone in your	1. Yes	
	family pay for the	2. No	
	tablets?	3. Don't know	

2.5	Did you see anyone for	1.	Yes	
	antenatal care this/ most	2.	No	Go to 2.22
	recent pregnancy?			
2.6	If yes: Whom did you see	1.	Doctor	
##	for your last visit/check-	2.	Nurse	
	up?	3.	HA/CMA/MCHW	
		4.	Health Worker (General)	
		5.	Other, (Specify)	
		6.	Don't know	
2.7	Where were the	1.	Hospital	
##	antenatal visits? (circle	2.	PHC/Manmohan Hospital	
	all that apply)	3.	HP/SHP	
		4.	Out Reach Clinic	
		5.	Private Clinic	
		6.	Other (specify)	
2.8	After how many months	1.	Month	
	of pregnancy did you first	2.	Don't know	
	have your antenatal visit			
	with above person?			
2.9	How many antenatal	1.	No. of visit	
	visits did you have during	2.	Don't know	
	your last pregnancy?			
2.10	How did you know about	1.	From Family members	
##	ANC check-ups?	2.	From Radio/TV	
		3.	Health workers	
		4.	Friends/relatives	
			/Neighbour/community people	
		5.	School/college/teacher	
		6.	Female community health	
			volunteer	
		7.	Green Tara Nepal's staff	
		8.	Green Tara Nepal's Group	
			Member	
		9.	Other	
			(specify)	

2.11	During antenatal visit,		
	was any of following		
##	done at least once during	A. 1.Yes 2. No 3. Don't know	
	your pregnancy?	B. 1.Yes 2. No 3. Don't know	
	A. Did you have weight	C. 1.Yes 2. No 3. Don't know	
	checked?	D. 1.Yes 2. No 3. Don't know	
	B. Was your height	E. 1.Yes 2. No 3. Don't know	
	measured?	F. 1.Yes 2. No 3. Don't know	
	C. Was blood pressure		
	measured?		
	D. Did you give urine		
	sample?		
	E. Did you give blood		
	sample?		
	F. They checked your		
	ankles for swelling?		
2.13	During pregnancy were	1. Yes	
	you given an injection in	2. No	Go to Q 2.15
	the arm to prevent baby	3. Don't know —	
	from getting tetanus?		
2.14	If yes, how many times?	times	
2.15	During the antenatal visit	1. Yes	
	did you get any advice	2. No	
	from health worker?	3. Don't know	
2.17	How long did it take to	Hours	
	travel from your home to	Minutes	
	place where you usually		
	went for antenatal check-		
	up?		
2.18	How did you get there?	1. Walking	
		2 Bus	
		2. 240	
Ì		3. Other (specify)	
2.19	Who decided that you	 Other (specify) Myself 	
2.19	Who decided that you would go for your	 Other (specify) Myself Husband 	
2.19	Who decided that you would go for your antenatal check-up?	 Other (specify) Myself Husband Mother-in-law 	
2.19	Who decided that you would go for your antenatal check-up?	 Other (specify) Myself Husband Mother-in-law Other (specify) 	
2.19	Who decided that you would go for your antenatal check-up? How much did you pay	 Other (specify) Myself Husband Mother-in-law Other (specify) Total cost Rs 	
2.19 2.20 ##	Who decided that you would go for your antenatal check-up? How much did you pay (including cash & kind)	 Other (specify) Myself Husband Mother-in-law Other (specify) Total cost Rs Kind: Labourhrs 	

	for each antenatal visit		
	during last pregnancy?		
2.21	How satisfied are you	1. Not at all	Go to 2.23
	with the antenatal care	2. Somewhat	
	you received from	3. Very Much	
	service providers during		
	pregnancy?		
2.22	If you did not have any	1. Shyness	
##	antenatal care visits, why	2. Health worker is a man	
	not?	3. Don't know about health services	
	(More than one answer	4. Too far to health facility	
	possible)	5. No money to pay for visit	
	Do not read out answers!	6. No time to go for visit	
		7. Family don't allow to go	
		8. No transportation	
		9. Other (specify)	
2.23	Did you have any health	1. Yes	
	problems during your	2. No	Go
	most recent pregnancy?	3. Don't know	Sec. 3
2.24	If yes, what problems did	1. Vaginal Bleeding	
##	you have?	2. Swelling body/ legs	
		3. High blood pressure	
		4. Dizziness	
		5. Abdomen pain	
		6. Vomiting in early pregnancy	
		7. Weakness	
		8. White discharge	
		9. Other (specify	
2.25	Where did you go to	1. Hospital	
##	solve these problems?	2. PHC/Manmohan Hospital	Skip 2.26
		3. HP/SHP	except
		4. Out Reach Clinic	8. Nowhere
		5. Private Clinic	
		6. Traditional Healers	
		7. Other (specify)	
		8. Nowhere	
2.26	If you did not seek care	1. No need perceived by woman	
##	from any one, why not?	2. No need perceived by family	Ask this Q if
	(Max. 3 Answers)	3. Not part of local tradition	2.25 is
		4. HW not in health facility	Nowhere

	5. HW is a man	
	6. Not aware of services	
	7. Too far to health facility	
	8. No money to pay for visit	
	9. No time to go for visit	
	10. The service is poor	
	11. Family don't allow to go	
	12. Any problem cured itself	
	13. Other (specify):	

FOR ALL RESPONDENTS, ASK ABOUT LAST DELIVERY:

Section 3: Delivery Care

3.1	When was your last antenatal	month of preg	nancy Only for
	visit before you gave birth?	Don't know	ANC
			Check up
3.2	Where did you deliver the baby?	1. Home	
##		2. Hospital	
		3. PHC/Manmohan Hosp	pital
		4. HP/SHP	
		5. Private clinic	
		6. Other (specify)	
3.3	Who decided where to deliver	1. Myself	
	your baby?	2. Husband	
		3. Mother-in-law/grandm	other
		4. Other (specify)	
3.4	Who assisted with the birth of	1. Doctor	
##	baby?	2. Nurse	
		3. Student Nurse/medica	d
		student	
		4. HA/CMA/MCHW	
		5. VHW	
		6. TBA	
		7. Family member/Relati	ves
		8. Health worker (genera	d)
		9. Other (specify)	
		10. No one	

3.5	Is there any local/ national	1.	Yes	
	financial help available for your	2.	No	Go to Q.
	delivery?	3.	Don't know	3.8
3.6	If yes, who are they?	1.	Government	
		2.	Private lender	
		3.	Local savings group	
		4.	Other:	
3.7	How much money did you	1.	Rs	
	receive?	2.	Don't know	
3.8	How much did you have to pay	1.	Total costrupees	
	drugs, registration procedures,	2.	Don't Know	
	travel, food etc.?			
3.9	What problems, if any, occurred	1.	Long labour (more than	
##	during the labour or delivery?		18hrs)	
	Specify?	2.	Retained placenta	
		3.	Excessive Vaginal Bleeding	If none,
		4.	Convulsion/fits	go to Q
		5.	Other	3.13
			(specify)	
		6.	None	
3.10	Who or Where did you visit to	1.	Hospital	
##	solve these problems?	2.	PHC/Manmohan Hospital	
		3.	HP/SHP	
		4.	Out Reach Clinic	
		5.	Private Clinic	
		6.	Traditional Healers	
		7.	Other (specify)	Go to
		8.	Nowhere	3.12
3.11	How soon did you seek help	1.	Immediately	Go to
	after the problem started?	2.	In less than 2 hours	3.13
		3.	Between 3-6 hours	
		4.	More than 6 hours	
3.12##	If you did not seek help	1.	No need perceived by	
	anywhere, why not?		woman	
		2.	No need perceived by family	
		3.	HW not in health facility	
		4.	HW is a man	
		5.	Not aware of services	
		6.	Too far to health facility	
		7.	No money to pay for visit	

			8. No time to go for visit
			9. The service is poor
			10. Family don't allow to go
			11. Too weak/sick to travel
			12. Other (specify)
3.13	How satisfied are you with t	he	1. Not at all
	care received during labour	and	2. Somewhat
	delivery? (Ask only to whom	n i	3. Very
	receive health services)		4. Not applicable
3.14	In your opinion, what are the	e	1. No trained health worker
##	main 3 problems with delive	ery	2. No transportation
	care in your community?	:	3. Too far health facility
			4. No health workers available
			at the time
			5. No money
			6. Not usual practice
			7. 7.Don't know where to get
			help
			8. Family do not perceive need
			9. Family refused to access
			care
			10. 10. Other
			(specify)
3.15##	What 3 things could improve	e	1. Health facilities in village
	delivery care for women in y	/our	2. Better trained staff in Health
	community?		facility
		:	3. More medicines
			4. More staff
		:	5. Inform women about
			available health services
			6. Increased awareness about
			delivery care
			7. More support from
			friends/family
			8. Other
			(specify)
Section	4: Postnatal Care of Woma	<u>in</u>	· · · · ·
4.1	After baby was born, did a	1. Ye	es
	health professional check	2. No	
	your own health?	3. Do	on't know Go to 4.5

4.2	How many days or weeks	1.	Same day	
	after the delivery did the first	2.	After 1 day	
	check take place?	3.	After 2 days	
		4.	Between 3-7 day	
		5.	Between 8- 14 day	
		6.	More than 14 days	
		7.	No Check	
4.5	Did you have any health	1.	Yes	
	problems within the first 42	2.	No	
	days after delivery?	3.	Don't know	Go to Q4.9
4.6	If yes, what problems did	1.	Vaginal Bleeding	
##	you have?	2.	Fever	
		3.	Weakness	
		4.	Convulsions/fits	
		5.	Breast infection	
		6.	Baby feeding problem	
		7.	Low mood/depression	
		8.	Offensive vaginal discharge	
		9.	Vaginal pain	
		10.	Faecal discharge from	
			vagina	
		11.	Other (specify)	
4.7	Where did you visit to solve	1.	Hospital	
##	these problems?	2.	PHC/Manmohan Memorial	Skip 4.8 except
			Hospital	8. Nowhere
		3.	HP/SHP	
		4.	Out Reach Clinic	
		5.	Traditional Healers	
		6.	Private Clinic	
		7.	Other (specify)	
		8.	Nowhere —	
4.8	If you did not seek help from	1.	No need perceived by	
##	anywhere, why not?		women	Ask this Q if 4.7
	(Max. 3 Answers)	2.	No need perceived by family	is Nowhere
		3.	HW not in health facility	
		4.	HW is a man	
		5.	Not aware of services	
		6.	Too far to health facility	
		7.	No money to pay for visit	
		8.	No time to go for visit	

		9. The service is poor	
		10. Family don't allow to go	
		11. Too weak/sick to travel	
		12. Other (specify)	
4.9	In your opinion, what are 3	1. No transportation facility	
##	main reasons that women do	2. Health facility too far	
	not check their health after	3. No health personnel in	
	delivery?	health centre	
		4. No money	
	(Max. 3 Answers)	5. No usual practice	
		6. No need perceived	
		7. Not allowed by family	
		8. Don't Know	
		9. Other (specify)	
4.10	In your opinion, what 3	1. Health facility in village	
##	things could help women	2. Better trained staff in health	
	access postnatal care more	facility	
	easily in your area?	3. More medicines facility	
	(Max. 3 Answers)	4. More staffs in health centre	
		5. Inform bout available health	
		services	
		6. Increase awareness on PNC	
		7. More support from	
		friends/family	
		8. Don't Know	
		9. Other (specify)	

Section 5: Neonatal care:

TELL WOMAN THIS RELATES TO HER MOST RECENTLY BORN CHILD

1.	Yes	ONLY
2.	No	for
3.	Don't know	НОМ
		Е
		Delive
		ry
1.	Clean blade	ONLY
2.	Unclean blade	for
2	Other (specify)	ном
3.	Other (Speerry)	110101
	1. 2. 3. 1. 2.	 Yes No Don't know Clean blade Unclean blade Other (specify)

				Delive
				ry
5.3	How far from the baby's body was the cord cut?	1.	(No. Of fingers)	Ask
		2.	Don't know	Home
				Delive
				ry
5.4	What was put on the cut cord?	1.	Nothing	ONLY
		2.	Antiseptic	for
		3.	Oil	HOM
		4.	Ghee/Butter	Е
		5.	Other (specify)	Delive
				ry
5.5	When was first time the baby was washed?	1.	Immediately after	ONLY
			birth	for
		2.	After	НОМ
			hrs	Е
		3.	Afterdays	Delive
		4.	Don't know	ry
5.6	How soon was the baby wrapped up after birth?	1.	Immediately	ONLY
		2.	Within one hour	for
		3.	More than one	НОМ
			hour	Е
		4.	Don't know	Delive
				ry
5.7	How old was the baby the first time they had	1.	1 month	
	anything other than breast milk? (E.g. animal milk,	2.	2 to 4 months	
	horlicks, medicine except vaccines, Jeevan jal, any	3.	5 to 6 months	
	foods)	4.	Over 7 months	
5.8	Was breast milk the first feed your baby was given?	1.	Yes	
		2.	No	
5.9	Did you give your baby the colostrum, the first yellow	1.	Yes	
	milk from the breast?	2.	No	
5.1	Did you breastfeed within the first hour after birth?	1.	Yes	
0		2.	No	
5.1	When did the baby have a first health check after		Hrs after	
1	delivery?	del	ivery	
			Days after	
		del	ivery	

5.1	Did the healthcare worker check your baby again in	1. Yes	
2	the first month after delivery?	2. No	
		3. Don't know	
5.1	Did your baby have any healthcare problems within	1. Yes	
3	the first month after delivery?	2. No Go to	0
		Q	
		5.17	
5.1	At what age did the baby have health problems?	days	
4		weeks	
5.1	What problems occurred with the baby after	1. Difficulty in	
5	delivery	breathing	
		2. Cold	
		3. Not feeding	
		4. Too sleepy	
		5. Diarrhoea	
		6. Other (specify)	
5.1	Where did you visit to solve these problems? Who	1. Hospital	
6	or where did u go to solve these problems?	2. PHC/Manmohan	
		Memorial	
		Hospital	
		3. HP/SHP	
		4. Out Reach Clinic	
		5. Private Clinic	
		6. Traditional	
		Healers	
		7. Other	
		(specify)	
		8. Nowhere	
5.1	Did you or anyone else register the birth of your	1. Yes	
7		2 No	
		2. No 3. Don't know	
		J. DUITEKHUW	

5.1	What was the baby wrapped in after delivery?	1.	Blanket
9		2.	Towel
		3.	Sari
		4.	Thin cloth
		5.	Other
			(state)

Section 6: Contraception and others FOR ALL RESPONDENTS

6.4	Was your last pregnancy planned?	1. Yes
		2. No
6.5	Were you using any kind of contraception, when you	1. Yes
	got pregnant?	2. No
6.6	Are you a member or ex member of any elected local	1. Yes (Your
	body? (E.g. VDC, ward etc.)	Post)
		2. No
6.7	Are you member of any voluntary organisation (e.g.	1. Yes
	NGO, User groups Cooperative etc.)	2. No
6.8	Who makes the decisions mainly about healthcare in	1. Myself
	the household?	2. My Husband
		3. Mother-in-Law
		4. Father-in-Law
		5. Other (Specify)
6.9	Sometimes a husband is annoyed or angered by	
	things that his wife does. In your opinion, is a husband	
	justified in hitting or beating his wife in the following	
	situations:	A. 1. YES 2.
		NO 3. Don't
	A. If she goes out without telling him?	Know
	B. If she neglects the children?	B. 1. YES 2.
	C. If she argues with him?	NO 3. Don't
	D. If she refuses to have sex with him?	Know
	E. If she burns the food?	C. 1. YES 2.
		NO 3. Don't
		Know
		D. 1. YES 2.
		NO 3. Don't
		Know
1		1

		E.	1. YE	S 2.	
			NO	3. Don't	
			Know	,	
1.4	Are you from Dalit Caste?	1.	Yes		
		2.	No		
		3.	Don'	t Know	

Section 7: EXTRA PAGE FOR INTERVENTION COMMUNITY ONLY

7.1	Have you heard of Green	1. Yes	
	Tara Nepal?	2. No	
7.2	Have you met any of the	1. Yes	
	Green Tara Nepal staff?	2. No	
7.3	If you know GTN, how?	1. Been to Group	lf Don't know
##		2. GTN Staff visited at home	about Green Tara
		3. Been to antenatal/	Don't ask this Q.
		postnatal gathering	
		4. Been to festival	
		5. Relative goes to group	
		6. Heard from others in	
		community.	
		7. Other	
		(Specify)	
7.4	Do you go to a GTN group?	1. Yes	
		2. No	Go to 7.7
7.5	If Yes, how many months	1. 0-3 months	
	ago was your 1 st meeting?	2. 4-6 months	
		3. 7-12 months	
		4. More than 12 months	
7.6	How many meetings have	meetings	
	you been to?		
7.7	Does anyone else in your	1. Yes	
	household go to a GTN	2. No	Go to 7.10
	group?		
7.8	If yes, who goes?	1. Mother-in-law	
		2. Sister-in-law	
		3. Husband	
		4. Father in law,	
		5. Brother-in-law	

		6.	Other (Specify)	
7.9	Did they share anything	1.	Yes	
	about health with you from	2.	No	
	their group meetings?			
7.10	Did you get given a blanket	1.	Yes	
	especially for the baby?	2.	No	

THANK YOU for taking part in this study

Please take this Nail-Clipper as our gift

SN.:

Household Survey Questionnaire for Women- 2012

VDC: Name of village:

Ward no.:
Day:
Date:2012/06/

SCREENING Q: DO YOU HAVE A CHILD UNDER 24 MONTHS (Not Completed)तपाइको २४ महिना भन्दा कम उमेरको बच्चा छ?1. Yes2.No (if no, do not continue questionnaire)/यदि छैन भने प्रश्न नसोध्ने)

1.0 If yes, how old is your youngest child?{यदि भएमा सबभन्दा सानो बच्चा कति उमेरको भयो?).......monthsDays

1.0a How many child(ren) do you have under 24 months?(तपाइको २४ महिना भन्दा कम उमेरको कतिजना बच्चा छन्?)....

	Section 1:Household and Socio-dem	ographic information	
SN	Questions	Coding categories	Skip
1.1	In what month and year were you born? Write in B.S. तपाइको जन्म कुन महिना र सालमााभएको थियो? वि. सं. मा लेख्नुहोस् ।	Month: Don't Know month Year: Don't Know year	
1.2	What is your age? (Write completed yrs) तपाइको उमेर कति भयो? पुरा भएको उमेर लेख्नुहोस् । (compare and correct 1.1 and/or 1.2 if inconsistent)प्रश्न1.1र 1.2को बीचमाएकरुपता छ बा छैन हेर्नुहोस्	Years	
1.3	What is your ethnicity? तपाईको थर के हो ? सबैको थर लेखने ।	 Yadav Tharu Muslim Gurung Other (specify) 	
1.4	What is your caste /ethnicity ? तपाई कुन जातमा पर्नुहुन्छ?	 Janjati Tharu Muslim Terai Dalit Pahadi Dalit Brahmin Chhetri Other (specify) 	
1.5	What is your religion? तपाइ कुन धर्म मान्नुहुन्छ:	 Buddhist Muslim Hindu Christian Other (specify) 	
1.6	Can you read and write? तपाइ पढ्न लेख्न सक्नुहुन्छ ?	1. Yes 2. No	Go to 1.9
1.7	Have you ever attended school? तपाइले स्कुल गएर पढ्नुभएको हो?	1. Yes 2. No	Go to 1.9
1.8	lf yes, what is the highest grade you completed? तपाईले कतिकक्षा सम्म पढ्नुभएको छ?	 Primary (completed class 5) Secondary (completedclass 10) Intermediate (PCL)-completed class 12 Bachelor and above 	
1.9	What is your current main occupation? तपाईको अहिलेको मुख्य पेशा (काम) के हो ?	 Student House wife Farmer Service Business Other (Specify) 	

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ))

1.10	What is your husband's level of education? तपाइको श्रीमान्ले कतिसम्म पद्नु भएको छ?	 Illiterate Primary education Secondary education (S.L.C) Intermediate (PCL) Bachelor and above 	
1.11	What is your husband's main occupation? तपाइको श्रीमान्को मुख्य पेशा के हो ?	 Farmer Teacher Business Skilled labour Unskilled labour Other (Specify) 	
1.12	How many people live in your house? तपाइको घरमा कति जना हुनुहुन्छ ?	 Total Young People & Adults (age 10 or above) Children (below 10 yrs) 	
1.13 ##	Do you and your family(household) have any property? तपाइको आफ्नै वा परिवारको कति सम्पत्ती छ?	1. Land in Bigha/kaththa	
1.14	Do you have any property in your name? के तपाईको आफ्नै नाममा जमिन वा सम्पत्ती छ?	1. Yes 2. No 3. Don't Know	
1.15	Do you have own Mobile Phone? तपाइको आफ्नै मोवाइल फोन छ?	1. Yes 2. No	
1.16	Where are you currently living? तपाई अहिले कहाँ बसि रहनु भएको छ ?	 In own home In rented property Living with relative Other (specify) 	
1.17	What type of roof in your house? (Observation) छाना कस्तो प्रकारको छानो भएको घरमा वस्नुभएको छ? (अवलोकन)	1. Cemented 2. Tin 3. Tile 4. Hay 5. Other (specify):	
1.18 ##	Does your household have: तपाइको घरमा निम्न सुविधाहरु छन् ?	1. YES 2.No 1. Electricity 2 2. Radio 2 3. Television 2 4. Telephone (landline) 2 5. Refrigerator 2 6. Computer 2 7. Wall clock 2 8. Gas geyser 2 9. Solar panel 2 10. Fan 2	
1.19 ##	What type of fuel does your household mainly use for cooking?(max 2 answers) तपाइको घरमा खाना पकाउन कुन इन्धन प्रयोग गर्नुहुन्छ? (वढीमा २ वटा उत्तरहरु) (max 2 answers)	 Electricity LPG (Gas) Biogas Kerosene Wood Animal dung Other (specify): 	
1.20 ##	Does any member of your household own: तपाईको परिवार सदस्यहरुको तलका मध्ये कनै सवारी साधन छ?	1. Yes 2.No 1. Bicycle /Rickshaw 2. 2. Motorcycle/scooter 2. 3. Tempo: 2. 4. Car/Truck: 2. 5. Cart (bull cart) 2. 6. Other transport (specify)	
1.21	ls anyone in your family currently working abroad? के तपाईको परिवार सदस्यहरुमध्ये कोहि विदेशमा काम गरि रहनु भएको छ⁄	 Yes No-skip to section 2 3. 	

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

1	.22	lf yes, in which country/countries are they working? यदि हो भने कुन देशमा काम गरि रहनु भएको छ, देशको नाम लेख्नुहोस्? यदि दूईजना भएमा दुवै लेख्ने । worth giving option of more than 1?	1 2	
1	.23	Are they sending money to your household? के बहाले नियमित रुपमा घरमा पैसा पठाउने गर्नु भएको छ:	1. Yes 2. No	

Section 2: Water and sanitation

2.1	What is the main source of drinking water fo	r members	 Piped water to own home 	
##	of your household? (Max. 2 Answers)		Common/public piped water	
	तपाइको घरमा खानेपानी कहाँबाट ल्याइन्छ? (बढीमा २ वटा उत्त	तरहरु)	3. Tube well or borehole	
			4. Surface water (river/dam/ lake/	
			pond/stream/canal/irrigation canal	
			5. Stone tap/dhara	
			6 Jar/Bottled water	
			7 Other (specify)	
2.2	De yey heye teilet in yeyr heyes?			Cata
2.2	Do you have tollet in your houser dylsalet and	મ્ન ખપાછ⊰:	1. fes	3.4
			2. No	2.4
2.3	If Yes, What kind of toilet facility do member	s of your	 Flush to piped sewer system 	
##	household usually use? (max 2 answers)		Flush to septic tank	
	यदि भएमा, कस्तो प्रकारको छ? (बढीमा २ वटा उत्तरहरु)		3. Flush to somewhere else	
			4. Pit latrine with slab	
			5. Pit latrine without slab	
			6. Composting toilet	
2.4	If you have no toilet at your house. what is the	ne reason?	1. Lack of financial resources	
##	तपाईको घरमा चर्पी नभएको भए चर्पी नबनाउनको कारण के व	ते? ग्रे?	2. Lack of time	
			3. No land	
			4 Do not like defecating in toilet	
			5 Do not know importance of toilet	
			6 New house/still to make toilet	
			7 Other (specify)	
25	\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\		7. Other (specify)	
2.5	where do you defecate if you have no tollet	at your	1. Use neighbours toilet	
	home? तपाईको घरमा चपी नभएको भए कहा दिशा गर्ने ग	नुभएको छ?	2. Use public toilet	
			3. Open defecation (bush, field)	
			4. Other (specify)	
2.6	Is there enough water in the toilet, where do	you go for	1. Yes	
	defecation? के तपाईले प्रयोग गर्ने चर्पीमा प्रशस्त पानीक	व्यवस्था छ?	2. No	
2.7	Have you ever beard about open defecation	free	1 Yes	
	compaign in your VDC3 & auto ben derecation	गर्द जन्म	2 No	
	Campaign in your VDC? के तपाइल यस गा. 14. स. र	ମାହ खुल्लା	2. 10	
	दिशामुक्त क्षत्र बनाउन लागका सुन्तु भएका छ?	1		
2.8	Is there toilet facility in the following places		1. YES 2.No 3. Don't know	
	within your VDC? के यस गा. वि. स. का निम्न	1. School		
	स्थानहरुमा चर्पी छ?	2. VDC Offi	ce 🛛 🗖 🗖	
		3. HP/SHP		
		4. Temple/	religious place 🛛 🗖 🗖	
		5. Other (s	pecify)	
2.9	When is it necessary, if at all, to wash your	wash your 1. After defecation		
##	hands? हात कतिबेला धनपर्क?	2. Before e	ating meal	
1	านกษระ พ.ศ. พ.ศ.พ.ศ. ษูกูรณะ	3. After cle	aning faces of the child	
1		4. After wo	rking at the dust/ field/animal sheds	
1		5 Other (s	necify	
		J. Other (s		
2.10	In the last one month, has your child had dia	rrhoea?	1. Yes	
	(Loose stool more than 3 times in 24 hours)		2. No	Go to
1	के गत १ महिनाभित्रमा तपाईको बच्चालाई पखाला लागेको थिय	ो? (२४ घण्टामा		next sec
1	तीन पटकभन्दा बढि पातलो दिशा लागेको)			

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

2.11	During the time (your child had diarrhea), how much	1.	Less than usual
	food was he given to eat? तपाईको बच्चालाई पखाला लागेको बेलामा	2.	More than usual
	साबिक भन्दा कति मात्रामा खान दिईएको थियो?	3.	About same
		4.	Other ()
2.12	Was anything given to treat diarrhoea? पखालाको उपचार गर्न	1.	Yes
	केहि दिईएको थियो?	2.	No
		3.	Don't know/ don't remember
2.13	What was given to treat diarrhoea?	1.	Pills/ syrup antibiotic
##	पखालाको उपचार गर्न केहि दिईएको थियो?	2.	Oral rehydration solution
		3.	Zink tablets
		4.	Injection
		5.	Intra-venous fluid
		6.	Home remedy/ herbal medicine
		7.	Other (specify

Section 3: Antenatal Care and seeking care – FOR ALL RESPONDENTS Note: these questions relate to the woman's LAST pregnancy

NOL	a mese questions relate to the woman's LAST pregnan	
3.1	Did you take iron/folic acid (vitamin tabs) during	1. Yes
	pregnancy? तपाइले पछिल्लो पटक गर्भवती हुँदा आइरन र भिटामिनका चक्की	2. No Go to
	खानुभएको थियो?	3. Don't know 3.4
3.2	Did you take all of the tables OR did you complete the	1. Yes
	course? के तपाईले सबै चक्की खानुभयो वा सबै औषधि खानुभयो?	2.No
	Complete course means 4 th month of pregnancy to birth	
3.3	Where did you get these tablets?	1. HP/SHP
##	तपाइले यि चक्की कहाँबाट लिनुभयो?	2. NGO clinic
		3. Private doctor or clinic/
		Pharmacy
		Iocal health worker/FCHV/Out
		reach clinic
		5. Hospital in Kathmandu
		6. Other (Specify)
3.4	Did you see anyone for antenatal care during your most	1. Yes
	recent pregnancy? तपाइले पछिल्लो पटक गर्भवती हुँदा कसैसँग जाँच	2. No Go to
	गराउनु भएको थियो?	3.16
3.5	If yes: Whom did you see for your last visit/check-up?	1. Doctor
##	यदि जँचाउनुभएको भए को सँग जँचाउनु भयो?	2. Nurse (SN, ANM)
		3. HA/CMA/MCHW
		4. Other (Specify)
		5. Don't know
3.6	Where were the antenatal visits? (circle all that apply)	1. Hospital
##	तपाइले पछिल्लो पटक गर्भवती हुँदा गर्भवती जाँच कहाँ गराउनुभयो?	2. PHC
		3. HP/SHP
		4. Out Reach Clinic
		5. Private Clinic
		6. Other (Specify)
3.7	After how many months of pregnancy did you have your	1. Month
	first antenatal visit with above person? तपाइ गर्भवती भएको कती	2. Don't know
	महिनामा पहिला जाच गराउनु भयो?	
3.8	How many antenatal visits did you have during your last	1. No. of visit
	pregnancy?तपाइले पछिल्लो पटक गर्भवती हुँदा जम्मा कति पटक जँचाउनु भयो?	2. Don't know
3.9	How did you know about ANC checkups? तपाइले गर्भवती	1. From Family members
##	जाँचबारे कसबाट थाहापाउनु भयो ?	2. From Radio/TV
		3. Health workers
		4. Friends/relatives
		/Neighbour/community people
		5. School/college/teacher
		Female community health volunteer
		7. Other (specify)

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

	3.10	During antenatal visit, was any of following done at least		
	##	पटक तलका प्रक्रियाहरु गरीयो?		
		A. Did you have weight checked?तौल लिइयो?	A. 1.Yes 2.No 3.Don't know	
		B. Was your height measured?उचाइ लिइयो?	B. 1.Yes 2.No 3.Don't know	
		C.Was blood pressure measured?रक्तचाप जॉचियो?	C. 1.Yes 2.No 3.Don't know	
		D. Did you give urine sample ?पिसाब जांचियो?	D. 1.Yes 2.No 3.Don't know	
		E. Did you give blood sample?रगत जॉचियो?	E. 1.Yes 2.No 3.Don't know	
		F. Did they check your ankles for swelling?सुन्निएको छ छन पिडौलामा औला गाडेर जॉचियो?	F. 1.Yes 2.No 3.Don't know	
	3.11	During pregnancy were you given an injection in the arm	1. Yes (if yes, times)	
		to prevent the baby from getting tetanus? तपाइ गभेवती हुँदा	2. No	
ļ		ाट.ाट.खाप लगाउनुभया?	3. Don't know	
	3.12	During the antenatal visit did you get any advice from	1. Yes	
		health worker regarding ANC, use of iron, deliver at HF	2. No	
		and PNC? गर्भवती जांचका समयमा तपाइल स्वास्थ्यकमीबाट गर्भवती जाच, आईरन चक्की खाने, स्वासथ्य संस्थामा प्रसुती गर्नेवारे कुनै सल्लाह पाउनुभयो?	3. Don't know	
	3.13	How long did it take to travel from your home to place	Hours	
		where you usually went for antenatal check-up? तपाइको घरबाट प्रायः गर्भवती जाँच गराएको स्थानमा पुग्न कति समय लाग्यो?	Minutes	
ſ	3.14	Who decided that you would go for your antenatal check-	1. Myself	
		up? गर्भवती जाँच गराउने निर्णय कसले गऱ्यो?	2. Husband	
			3. Mother-in-law	
-	2.45	The sector of the design of th	4. Other (specify)	C - +-
	3.15	How satisfied are you with the antenatal care you received	1. NOL dL dll 2. Somewhat	3 17
		rom service providers during pregnancy (स्वास्थ्यकमाक) व्यवहार	3. Very Much	5.17
ŀ	3.16	If you did not have any antenatal care visits, why not?यदि	1. Shyness	
	##	तपाइले गर्भवती जाँच नगराउन् भएको भए, किन जाँच गराउनुभएन?	2. Health worker is a man	
		(more then one answer possible)	3. Don't know about health services	
		Do not read out answers!	4. Too far to health facility	
			5. No money to pay for visit	
			6. No time to go for visit	
			7. Family don't allow to go	
			9 Other (specify)	
ŀ	3.17	Did you have any health problems during your most recent	1. Yes	Go to
		pregnancy? के तपाइलाइ गर्भवती अवस्थामा कनै समस्या भयो?	2. No	next
		F8,	3. Don't know 🔶	Section 4
ŀ	3.18	If ves, what problems did you have?	1. Vaginal Bleeding	
	##	यदि समस्या आएको भए के के भयो?	2. Swelling body/ legs	
			3. High blood pressure	
			4. Dizziness	
			5. Abdominal pain	
I			6. Vomiting in early pregnancy	
			7. Weakness	
I			o. white discharge	
ŀ	3.19	Where did you go to solve these problems????????????????????????????????????	1 Hospital	
I	##	समाधानको लागि कहा जानभयो ?	2. PHC	
I			3. HP/SHP	
I			4. Out Reach Clinic	
I			5. Traditional Birth Attendants	
I			6. FCHV	
I			7. Private Clinic	
I			8. Traditional Healers	
Т			9. Other (specify)	
I			10 Nowhoro	

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

 3.20 If you did not seek care from any one, why not? (Max. 3 ## Answers) Only if nowhere to previous question. यदि सहयोग लिनु नमएको भए, किन लिनु भएन ?(बढीमा ३ वटा उत्तरहरु) अधिल्लो प्रश्नमा कहिपनि नगएको वताएमा मात्र) 	 No need perceived by woman No need perceived by family Not part of local tradition HW not in health facility HW is a man Not aware of services Too far to health facility No money to pay for visit No time to go for visit The service is poor Family don't allow to go The problem cured itself Other (specify)
--	---

<u>Section 4: Delivery Care</u> FOR ALL RESPONDENTS, ASK ABOUT LAST DELIVERY:

4.1	Where did you deliver the baby? तपाइले बच्चा कहाँ जन्माउनुभयो?	1. Home 2. Hospital 3. PHC 4. HP/SHP 5. Private clinic 6. Other(specify)
4.2 ##	Who decided where to deliver your baby?बच्चा कहाँ जन्माउनेवारे कसले निर्णय लियो?	 Myself Husband Mother-in-law/grandmother Other (specify)
4.3 ##	Who assisted with the birth of baby? बच्चा जन्माउन कसले मद्दत गऱ्यो?	 Doctor Nurse Student Nurse/medical student HA/CMA/MCHW VHW TBA Family member/Relatives Health worker (general) Other (specify) No one
4.4	Did you receive money from GON as transportation allowance for delivering at health facility? तपाइ सुत्केरी हुँदा कतैबाट (सरकारी वा गैरसरकारी) आर्थिक सहयोग पाउनुभयो?	1. Yes
4.5	How much money did you receive? तपाइले सुत्केरी हुँदा कति रकम सहयोग स्वरुप पाउनुभयो?	 Did not get any rupees Don't know
4.6	How much did you have to pay drugs, registration procedures, for staff, for delivery, travel, food etc? तपाई सुत्केरी हुँदा कति खर्च लाग्यो? (दर्ता शुल्क, औषधि उपचार, कर्मचारी, यातायात, खाना, आदि सबैमा)	 Total costrupees Don't Know
4.7	How satisfied are you with the care received during labour and delivery? तपाई प्रसव सेवावाट कत्तिको सन्तुष्ट हुनुहुन्छ?	 Not at all Somewhat Very Not applicable

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

5.1	After baby was born, did a health professional check your	1. Yes	
	own health? बच्चा जन्मेपछि, स्वास्थ्यकर्मीले तपाईलाई जाँच गऱ्यो?	2. No	Go to
		3. Don't know	5.3
5.2	How many days or weeks after the delivery did the first	1. Same day	
	check take place?बच्चा जन्मेको कति समयपछि जाँच गरियो?	2. next day	
		3. After 2 days	
		4. No Check	
5.3	Did you have any health problems within the first 42 days	1. Yes	
	after delivery?तपाईलाई प्रसव भएको पहिलो४२ दिन भित्रमा कुनै	2. No	
	स्वास्थ्य समस्या आयो?	3. Don't know	

Section 6: Neonatal Care : TELL WOMAN THIS RELATES TO HER MOST RECENTLY BORN CHILD

6.1	If you had your baby at home, was a Home Delivery Kit	1.	Yes	ONLY for
	box (safe delivery kit box) used?यदि तपाइले घरमै बच्चा जन्माउनु	2.	No	HOME
	भएको भए, सुत्केरी सामग्रीको बट्टा प्रयोग गर्नुभयो?	3.	Don't know	Delivery
6.2	With what was the cord-cut? साल के ले काटियो वा काट्नुभयो ?	1.	Clean blade	ONLY for
		2.	Unclean blade	HOME
		3.	Other(specify)	Delivery
		4.	Don't know	
6.3	How far from the baby's body was the cord cut?	1.	(no. of fingers) अंगुल	
	कति लामो नाभि छाडेर साल काट्नुभयो?	2.	Don't know	
6.4	What was put on the cut cord?	1.	Nothing	
	साल काटेपछि नाभिमा के राख्नुभयो?	2.	Antiseptic	
		3.	Oil	
		4.	Ghee/Butter	
		5.	Sindur	
		6.	Turmeric (Besar)	
		7.	Other (specify)	
6.5	When was first time the baby was washed?	1.	Immediately after birth	
	बच्चालाइ कति समय पछि नुहाइदिनु भयो?	2.	Afterhrs	
		3.	Afterdays	
		4.	Don't know	
6.6	How soon was the baby wrapped up after birth?बच्चालाइ	1.	Immediately	
	कति समय पछि न्यानो कपडाले वेर्नुभयो?	2.	Within one hour	
		3.	More than one hour	
		4.	Don't know	
6.7	What was the baby wrapped in after delivery?	1.	Blanket	
	बच्चा जन्मेपछि बच्चालाइ के ले बेनुंभयो?	2.	Towel	
		3.	Sari	
		4.	Thin cloth	
		5.	Nothing	
		6.	Don't know	
		7.	Other (state)	
6.7	How old was the baby the first time they had anything	1.	1 month	
	other than breast milk? (eg animal milk, horlicks, medicine	2.	2 to 4 months	
	except vaccines, Jeevan jal, any foods, alcohol or holy	3.	5 to 6 months	
	water from priest/ trad healer) तपाइको बच्चा कति महिनाको हुँदा	4.	Over / months	
	आफ्नो दूध बाहेकका खानेकुरा ख्वाउन शुरु गर्नुभयो ?	5.	Still breast feeding	
6.8	Was breast milk the first feed your baby was given?	1.	Yes	
	तपाइले बच्चालाई पहिलो पटक तपाइको आफ्नै दूध खुवाउनुभयो?	2.	No	
6.9	Did you give your baby the colostrum, the first yellow milk from	1.	Yes	
	the breast? तपाइको बच्चालाइ तपाइले पहिलोपल्ट आउने बाक्लो पहेँलो (2.	No	
	विधाती) दुध खुवाउनुभयो?	<u> </u>		
6.10	Did you breast feed within the first hour after birth?	1.	Yes	
	तपाइका बच्चालाइ जन्मका एक घण्टा भित्रमा दूध खुवाउनुभया?	2.	No	
6.11	Was your baby weighed soon after delivery? के तपाइको बच्चालाई	1.	Yes	
	जन्मन बित्तिक तालिएका थिया?	2.	No	Go to Q
		3.	Don't know	6.13

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

6.12	lf yes, what was his/her weight? यदि थियो भने उसको तौल कति थियो?	 1 gram 2. Don't remember 	
6.13	When did the baby have a first health check after delivery? बच्चा जन्मेको कति समयपछि पहिलो पटक स्वास्थ्य जाँच गराउनुभयो?	Hrs after delivery Days after delivery Had none	
6.14	Did the healthcare worker check your baby a second time in the first month after delivery? तपाइको बच्चालाइ स्वास्थ्यकर्मीले एक महिनाभित्र फेरी जॉच गराउन लानुभयो?	 Yes No Don't know 	
6.15	Did your baby have any healthcare problems within the first month after delivery? तपाइको वच्चालाइ जन्मेको एक महिना भित्र कुनै स्वास्थ्य समस्या आयो?	1. Yes 2. No →	Go to Q 6.19
6.16	At what age did the baby have health problems? तपाइको बच्चा कति दिन वा हप्ताको हुँदा स्वास्थ्य समस्या आयो?	days weeks	
6.17	What problems occurred with the baby after delivery यदि स्वास्थ्य समस्या आएको भए के के भयो?	 Difficulty in breathing Cold/ hypothermia Fever Not feeding Too sleepy Diarrhoea Other (specify) 	
6.18	Where did you visit to solve these problems? Who or where did u go to solve these problems? यी समस्याहरु समाधान गर्न कहां जानुभयो?	Hospital PHC/ Hospital HP/SHP Out Reach Clinic Frivate Clinic Traditional Healers Family members Other (specify) Nowhere	
6.19	Did you or anyone else register the birth of your baby? तपाइको बच्चाको जन्म दर्ता भयो?	1. Yes 2. No 3. Don't know	

Section7: Women empowerment

7.1	Are you a member or ex member of elected local body? (eg VDC, ward etc) तपाइ कुने संस्थाको निर्वाचित सदस्य हुनुहुन्छः वा हुनुहुन्थो? (गा.वि.स., वडा)	1. Yes (your post) 2. No
7.2	Are you member of any voluntary organisation (NGO, User groups etc) तपाइ कुनै समुह वा गैरसरकारी संस्थाको सदस्य हुनुहुन्छ?	1. Yes 2. No
7.3	Who makes the decisions mainly about health care in the household? तपाइको घरमा स्वास्थ्य सेवा लिनमा कसले निर्णय गर्छ?	1. Myself 2. My husband 3. Mother in law 4. Other (specify who)
74	Sometimes a husband is annoyed or angered by things that his wife does. In your opinion, is a husband justified in hitting or beating his wife in the following situations: कहिलेकाहि श्रीमान् श्रीमतीसँग केहि कुरा पर्दा रिसाउछत्। तपाइका विचारमा निम्न अवस्थामा श्रीमतीलाइ गालि गर्नु वा कुटपिट गर्नु जायज हो ? A. If she goes out without telling him? यदि श्रीमान्लाइ नसोधि वाहिर गएमा B. If she neglects the children? यदि उनले वच्चाहरूलाइ वेवास्ता गरेमा C. If she argues with him? यदि उनले श्रीमान्सँग वादविवाद वा वहस गरेमा D. If she refuses to have sex with him? यदि उनले यौन सम्पर्क गर्न नमानेमा E. If she burns the food? यदि उनले खाना डढाएमा	 A. 1. YES 2. NO 3. Don't Know B. 1. YES 2. NO 3. Don't Know C. 1. YES 2. NO 3. Don't Know D. 1. YES 2. NO 3. Don't Know E. 1. YES 2. NO 3. Don't Know

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

7.5	How old were you when you (Ask only for married woman) तपाई कतिवर्षको हुनुहुन्थ्यो? (विवाहितमहिलालाई मात्र सोध्ने) 1. promised to marry विवाह गर्ना निर्णय हुराको उमेर 2. got married विवाह गर्दाको उमेर 3. Living with husband श्रीमानसित सगै वस्वाको उमेर Who selected your husband? (Ask only for married woman) तपार्वको श्रीमानकमले क्लीर गर्नमएको थियों/विवाहित महिलालाई	1 yrs. Promise marriage 2 yrs- age at marriage 3 yrs -started to live with husband or husband house 1. Myself 2. My family
	मात्र सोधने)	 Myself and my family jointly Other (specify)
7.7	In your opinion, who should have right to choose the husband/partner? तपाईको विचारमा श्रीमान छनौट गर्नमा कसको अधिकार हुन्छ होला? सबैलाई सोध्ने ।	 Own Family member (parents, own brother and sisters) Relatives Jointly Other (Specify)
7.8	How old were you when you had first pregnancy? तपाइ पहिलो पटक गर्भवती हुँदा कती वर्षकी हुनुहन्थ्यो?	yrs
7.9	How many times have you been pregnant? तपाइ अहिले सम्म कति पटक गर्भवती हुनुभयो?	times
7.10	How many live children do you have? तपाईको कति जना जीवित बच्चा छन्?	
7.11	Have you had any miscarriages/abortion/stillbirth? तपाईको बच्चा खेर गएको वा गर्भपतनभएको वा मरेको बच्चाजन्मेको छ ?	1. Yes 2. No
7.12	If yes, what has happen? यदि थियो भने के भएको थियो?	1. Miscarriages (28 wks before) 2. Abortion (after 28 weeks) 3. Stillbirth (मरेको बच्चा जन्मेको)
7.13	Have you ever discussed, with your partner, his attitudes towards family planning? परिवार नियोजनका साधनको प्रयोगवारे आफ्नो श्रीमान वा साथीसित कहिल्यै सल्लाह गर्नुभएको छ:	 Yes No No partner/unmarried
7.14	ls abortion legal in Nepal? के नेपालमागर्भपतनले कानूनीमान्यतापाएको छ ?	1. Yes 2. No 3. Don't know
7.15	Where do women go to have abortion in this village? यदि गर्भपतन गराउन परेमा सो सेवा कहा पाईन्द्र?	India Butwal Parasi Bhairahawa Others (specify)

Section 8: Contraception and others FOR ALL RESPONDENTS				
8.1	Were you using contraception when you became pregnant before your recent/last pregnancy? तपाइले अहिले वा गत गर्भावस्था भन्दा पहिले कुनै परिवार नियोजनको साधन प्रयोग गर्नुभएको थियो?	1. Yes 2. No	Go to 8.5	
8.2 ##	If yes, what did you use? यदि गर्नु भएको भए के प्रयोग गर्नुभएको थियो?	1. Oral pills 2. IUD(Copper T)/ IUI (Mirena) 3. Injections (DEPO) 4. Implants (Norplant) 5. Condom 6. Diaphragm/cap 7. Foam/jelly 8. female sterilization (minilap) 9.male sterilization (vasestomy) 8. Other (specify)		
8.3	How long had you been using that method of contraception? ती साधनहरु कति समयदेखि प्रयोग गरि रहनु भएको छ?	 Less than one week Less than 1 month Between 1-6 months\ Less than one year More than one year 		
8.4 ##	Where did you obtain this contraception last time? यि साधनहरु कहाँवाट प्राप्त गनुभयो?	1. Gov hospital/clinic 2. PHC 3. Health post/sub health post 4. mobile clinic 5. FCHV		

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

		6. FPAN
		7. Marie Stopes
		8. Private hospital/clinic
		9. pharmacy
		10. Others (specify)
8.5	Was your last pregnancy planned? तपाइको गत गर्भ योजनाअनसार	1. Yes
	भएको हो?	2. No
8.6	Have you ever discussed with your partner, his attitudes	1. Yes
	towards family planning? परिवार नियोजनका साधनको पर्यागवारे आफनो	2. No
	श्रीमान वा साथीसित कहिल्यै सल्लाह गर्नुभएको छ?	
8.7	Who makes the decisions about family planning for you? परिवार	1. No partner/unmarried
	नियोजनका साधनको प्रयोगबारे कसले निर्णय गर्छ? परिवार नियोजनका साधन प्रयोग	2. Don't use FP
	गर्नेलाई मात्र सोध्ने?	3. You
		4. Partner
		5. Mutual (you & partner)
		6. Relatives
		7. Health workers
		8. Other (specify)
8.8	Have you ever used emergency contraceptives? के तपाईले कहिल्पै	1. Yes
	आकस्मिक गर्भ निरोधको साधनहरु प्रयोग गर्नुभएको छ?	2. No
		Never heard of it

Section 9: Knowledge on HIIV and its prevention

9.1	 Can people reduce their chance of getting the AIDS virus by having just one uninfected sex partner who has sexual 	1. YES	2. NO	3. Don't Know □	
	intercourse with no other partners? के मानिसले एकजना विश्वासिलो व्यक्तिसंग मात्र यौन सम्पर्क गरेमा (जसले अन्यसग यौन सम्पर्क गरेको छैन) एड्स लाग्ने संभावना घट्छ?				
	 Can people get the AIDS virus from mosquito bites? के लामखुट्टेको टोकाइबाट एड्स सर्छः 				
	 Can people reduce their chance of getting the AIDS virus by using a condom every time they have sex? हरेक पटकको यौन सम्पर्कमा जाउंग प्रोण गरेण प्रदेश प्रवेत प्रवेत प्रवेत 				
	 Can people get the AIDS virus by sharing food with a person who 				
	has AIDS? एड्स लागेको व्यक्तिको जुठो खाादा एचआइमि स्छं/ Is it possible for a healthy-looking person to have the AIDS virus? हेर्दा स्वस्थ देखिने व्यक्तिलाइ पनि एड्स लागेको हन सक्छ/				
	 If a member of your family became sick with AIDS, would you be willing to care for her or him in your own household? ব্যাহুকা প্ৰিয়াক কাঁটি ব্যবহাৰা তুত্তন ময়া মন ব্যাহ ত্ৰমলাহ ঘৰ্মী হ্যালাৰে গাঁৱলক কি চলা? 				

THANK YOU for taking part in this study (धन्यवाद **!)** Please accept this small gift

Multiple Answers (यस्तो चिन्ह ## भएमा एक भन्दा धेरै उत्तर हुन सक्छ 1)

Appendix IV - Variables Description

	-		
Table S1: Variables	description and c	codification, overal	l evaluation.

Variable	Description	Values
Regressions	• •	1
Intervention	Dichotomical variable	0=control group;
		1= intervention group
After (midline regression)	Dichotomical variables	0= time at baseline;
		1= time at midline
After*intervention (midline	Dichotomical variables	0= control group at midline;
regression)		d intervention means at
		T= Intervention group at midling
Afterafter (final regression)	Categorical variables	0 = time at baseline:
		· · · · · · · · · · · · · · · · · · ·
		1= time at midline
		2= time at final
Afterafter*intervention (final	Dichotomical variables	0=control group at final;
regression)		
		1= intervention groups at
		final
		2= intervention group at
Age	Continuous variable	110ai 15 - 49
Education	Categorical variables	0 = none
		· ·····,
		1= primary;
		2= Secondary and
		higher/tertiary
Components for wealth inde	x construction	
1. Materials used for roofing	Dicnotomical variables	0= root made of tin, hay,
rooning		stone,
		1- roof made of cement
		tile.
		,
2. Area of land owned	Dichotomical variables	0= own land less than 3
		Ropani (0.38 acres in the
		hills);
		1 own lond greater than 2
		Ropani (0.38 acres in the
		hills)
3. Goat	Dichotomical variables	0=none;
		1= owns goats
4. Motorised vehicle	Dichotomical variables	0=none;
		1= owns a motorcycle
5. Car	Dichotomical variables	0=none;

		1- owns a car
6. Source of drinking water	Dichotomical variables	0= non-piped source of water to the home (common or public piped water, well, borehole, rain water, surface water such as rain, dam, lake, pond, stream, stone tap or <i>dhara</i>); 1= piped source of water to the home
7. Type of toilet	Dichotomical variables	0= pit latrine (with or without slab) or composting toilet; 1= owns a flush toilet (flush to piped sewer system, septic tank or pit latrine)
8. Number of rooms in dwelling	Ratio of room: person	Total household member/rooms in dwelling
9. Type of energy used to cook (natural)	Dichotomous variables	0=none; 1= uses natural source directly (kerosene, wood, animal dung, coal, straw, shrubs, grass)
10. Type of energy used to cook (biogas)	Dichotomous variables	0=none; 1= uses biogas (made from raw materials converted to gas: agricultural waste, manure, municipal waste, plant material, sewage, green waste or food waste)
11. Type of energy used to cook (LP gas)	Dichotomous variables	0=none; 1= uses liquid petroleum
12. Type of energy used to cook (electricity)	Dichotomous variables	0=none; 1= uses electricity to cook
13. Bicycle	Dichotomical variables	0=none; 1= owns a bicvcle
14. Mobile phone	Dichotomical variables	0=none; 1= owns a mobile phone
15. Fridge	Dichotomical variables	0=none; 1= owns a fridge
16. Computer	Dichotomical variables	0=none; 1= owns a computer

Appendix V - Annex-1: Qualitative topic guide, Intervention Area Intervention area: Very young, young, and older mothers

- 1. What has changed in this area since 2007?
- 2. How has the population changed distribution of the communities in the last 5 years?
- 3. What government programmes (financial schemes or interventions) for health have been introduced since 2007?
- 4. In your opinion, what has changed in healthcare in the last five years?
- 5. Tell me how you use mobile phones to enquire about healthcare and advice in the community? How?
- 6. What would you like to change in the community to help improve the health of babies?
- 7. What would you like to change in the community to help improve the health of babies?
- 8. What NGO's work in this area?
- 9. Who make the decision to attend ANC? Tell me about it?
- 10. Who makes the decision to attend delivery care? Tell me about it?
- 11. Who makes the decision to attend PNC? Tell me about it?
- 12. Do you attend GT health promotion groups?
- What do you discuss at GTN? (Hand washing, hygiene sanitation, problems related to ANC, delivery and PNC; Infection/exclusive breast feeding/skin-to-skin; and Contraception)
- 14. What barriers are there to women like you to attend sessions? (Time, cost for transportation, had to work in the field, house and permission from the family)
- 15. Are you able to communicate freely and honestly with GTN staff? What is good, what is bad? How can GTN want to improve?
- 16. Since you started going to GTN, do you feel different?
- 17. Does anyone else in your household go to a GTN group? (Mother-in-law, sister-inlaw, husband, father-in-law? Did they share anything about health with you from their group meetings?)
- 18. Are there any other members of the community who don't attend GTN, who are benefitting in any other way?
- 19. Do you go to GTN groups? Do you go to ANC, if you don't why?
- 20. Would you recommend a new mother (to be) in the community to attend?
- 21. Tell me what it is like having a new baby or being a new mother? What is the tradition in your family? (When can the new mother go out, go to her parent's house/in-laws house? When you start working?)
- 22. What kind of cultural practices exist around pre-pregnancy?

- 23. What kind of cultural practices exist around pregnancy?
- 24. What kind of cultural practices exist around post-pregnancy care?
- 25. Tell what is commonly practiced in the community during antenatal, pregnancy and postnatal care? Do mothers give ayurvedic/herbal medicines to babies? From where? Presents? What can you do? What can't you do? Intervention area: GTN Staff Interview
- 1. What has changed in this area since 2007?
- 2. How has the population changed distribution of the communities in the last 5 years?
- 3. What government programmes (financial schemes or interventions) for health have been introduced since 2007?
- 4. In your opinion, what has changed in healthcare in the last five years?
- 5. Tell me how you use mobile phones to enquire about healthcare and advice in the community? How?
- 6. What would you like to change in the community to help improve the health of babies?
- 7. What would you like to change in the community to help improve the health of babies?
- 8. What NGO's work in this area?
- 9. Who make the decision to attend ANC? Tell me about it?
- 10. Who makes the decision to attend delivery care? Tell me about it?
- 11. Who makes the decision to attend PNC? Tell me about it?

GTN:

- 12. How do you get new mothers to attend?
- 13. Do you go to house visits? Why? What do you do? How different is it from the group meeting?
- 14. Has the home visit made any difference or any changes to the women?
- 15. Are there any other members of the community who don't attend GTN, who are benefitting in any other way?
- 16. Can you tell me why some people might not attend? How do you cope with nonattendance?
- 17. How do you keep people motivated with during the group meeting?
- 18. How do you keep people motivated after they have been to several group meetings?
- 19. Do you give any incentives/gifts at the meeting or at the house visits?
- 20. What exercises do you do? In problem solving?
- 21. Who do you refer to?

- 22. Which other agency do you work with? What partnerships do you have in the community?
- 23. Tell me what it is like having a new baby or being a new mother? What is the tradition in your family? (When can the new mother go out, go to her parent's house/in-laws house? When you start working?)
- 24. What kind of cultural practices exist around pre-pregnancy?
- 25. What kind of cultural practices exist around pregnancy?
- 26. What kind of cultural practices exist around post-pregnancy care? Do mothers give ayurvedic/herbal medicines to babies? From where? Presents? What can you do? What can't you do?

Intervention area: FCHVs

- 1. What has changed in this area since 2007?
- 2. How has the population changed distribution of the communities in the last 5 years?
- 3. What government programmes (financial schemes or interventions) for health have been introduced since 2007?
- 4. In your opinion, what has changed in healthcare in the last five years?
- 5. Tell me how you use mobile phones to enquire about healthcare and advice in the community? How?
- 6. What would you like to change in the community to help improve the health of babies?
- 7. What would you like to change in the community to help improve the health of mothers?
- 8. What NGO's work in this area?
- 9. Who make the decision to attend ANC? Tell me about it?
- 10. Who makes the decision to attend delivery care? Tell me about it?
- 11. Who makes the decision to attend PNC? Tell me about it?
- 12. Are you able to communicate freely and honestly with GTN staff? What is good, what is bad? How can GTN want to improve?
- 13. Can you tell me why some people don't attend?
- 14. Would you recommend a new mother (to be) in the community to attend?
- 15. Can you tell me why some people might not attend?
- 16. What care advice in your role as FCHV do you give to the new baby?
- 17. What care advice in your role as FCHV do you give to the new mother?
- 18. What is the tradition in the family?
- 19. When can the new mother go out, go to her parent's house/in-laws house?
- 20. What kind of cultural practices exist around pre-pregnancy?
- 21. What kind of cultural practices exist around pregnancy?
- 22. What kind of cultural practices exist around post-pregnancy care?

Do mothers give ayurvedic/herbal medicines to babies? From where? Presents? What can you do? What can't you do?

Appendix V - Annex-2: qualitative topic guide, Control area

Controlled area: Very young, young, and older mothers

- 1. What has changed in this area since 2007?
- 2. How has the population changed distribution of the communities in the last 5 years?
- 3. What government programmes (financial schemes or interventions) for health have been introduced since 2007?
- 4. In your opinion, what has changed in healthcare in the last five years?
- 5. Tell me how you use mobile phones to enquire about healthcare and advice in the community? How?
- 6. What would you like to change in the community to help improve the health of babies?
- 7. What would you like to change in the community to help improve the health of babies?
- 8. What NGOs work in this area?
- 9. Who make the decision to attend ANC? Tell me about it?
- 10. Who makes the decision to attend delivery care? Tell me about it?
- 11. Who makes the decision to attend PNC? Tell me about it?
- 12. Would you recommend a new mother (to be) in the community to attend?
- 13. Tell me what it is like having a new baby or being a new mother? What is the tradition in your family? (When can the new mother go out, go to her parent's house/in-laws house? When you start working?)
- 14. What kind of cultural practices exist around pre-pregnancy?
- 15. What kind of cultural practices exist around pregnancy?
- 16. What kind of cultural practices exist around post-pregnancy care?
- 17. Tell what is commonly practiced in the community during antenatal, pregnancy and postnatal care? Do mothers give ayurvedic/herbal medicines to babies? From where? Presents? What can you do? What can't you do?

Intervention area: GTN Staff Interview

- 27. What has changed in this area since 2007?
- 28. How has the population changed distribution of the communities in the last 5 years?
- 29. What government programmes (financial schemes or interventions) for health have been introduced since 2007?
- 30. In your opinion, what has changed in healthcare in the last five years?

- 31. Tell me how you use mobile phones to enquire about healthcare and advice in the community? How?
- 32. What would you like to change in the community to help improve the health of babies?
- 33. What would you like to change in the community to help improve the health of babies?
- 34. What NGO's work in this area?
- 35. Who make the decision to attend ANC? Tell me about it?
- 36. Who makes the decision to attend delivery care? Tell me about it?
- 37. Who makes the decision to attend PNC? Tell me about it?
- 38. Tell me what it is like having a new baby or being a new mother? What is the tradition in your family? (When can the new mother go out, go to her parent's house/in-laws house? When you start working?)
- 39. What kind of cultural practices exist around pre-pregnancy?
- 40. What kind of cultural practices exist around pregnancy?
- 41. What kind of cultural practices exist around post-pregnancy care?Do mothers give ayurvedic/herbal medicines to babies? From where?Presents? What can you do? What can't you do?

Intervention area: FCHVs

- 1. What has changed in this area since 2007?
- 2. How has the population changed distribution of the communities in the last 5 years?
- 3. What government programmes (financial schemes or interventions) for health have been introduced since 2007?
- 4. In your opinion, what has changed in healthcare in the last five years?
- 5. Tell me how you use mobile phones to enquire about healthcare and advice in the community? How?
- 6. What would you like to change in the community to help improve the health of babies?
- 7. What would you like to change in the community to help improve the health of mothers?
- 8. What NGOs work in this area?
- 9. Who make the decision to attend ANC? Tell me about it?
- 10. Who makes the decision to attend delivery care? Tell me about it?
- 11. Who makes the decision to attend PNC? Tell me about it?
- 12. What care advice in your role as FCHV do you give to the new baby?
- 13. What care advice in your role as FCHV do you give to the new mother?
- 14. What is the tradition in the family?
- 15. When can the new mother go out, go to her parent's house/in-laws house?
- 16. What kind of cultural practices exist around pre-pregnancy?
- 17. What kind of cultural practices exist around pregnancy?
- 18. What kind of cultural practices exist around post-pregnancy care?
 - a. Do mothers give ayurvedic/herbal medicines to babies? From where? Presents? What can you do? What can't you do?

Appendix VI – Definitions

Basic emergency obstetric and newborn care (BEmONC) is critical to reducing maternal and neonatal death. This care, which can be provided with skilled staff in health centres, includes the capabilities for:

- Administering antibiotics, uterotonic drugs (oxytocin) and anticonvulsants (magnesium sulphate);
- Manual removal of the placenta;
- Removal of retained products following miscarriage or abortion;
- Assisted vaginal delivery, preferably with vacuum extractor;
- Basic neonatal resuscitation care.

Comprehensive emergency obstetric and newborn care, typically delivered in hospitals, includes all the basic functions above, plus capabilities for:

- Performing Caesarean sections;
- Safe blood transfusion;
- Provision of care to sick and low-birth weight newborns, including resuscitation.

Timing is critical in preventing maternal death and disability: Although post-partum haemorrhage can kill a woman in less than two hours, for most other complications, a woman has between six and 12 hours or more to get life-saving emergency care. Similarly, most perinatal deaths occur around delivery or in the first 48 hours afterward.

Source: UNFPA 2014, Setting standards for emergency obstetric and newborn care. <u>http://www.unfpa.org/resources/setting-standards-emergency-obstetric-and-newborn-care#sthash.WStPGEUc.dpuf</u>
Appendix VII – Published papers

Published paper from PhD thesis:

 Sharma, S., van Teijlingen, E., Belizán, J.M., Hundley, V., Simkhada, P., Sicuri, E. 2016a. Measuring What Works: Impact evaluation of women's groups on maternal health uptake in rural Nepal, *PLOS One* 11(5): e0155144. Available from: http://iournals.plos.org/plosope/article2id=10.1371%/2Eiournal.pope.0155144

http://journals.plos.org/plosone/article?id=10.1371%2Fjournal.pone.0155144 [Accessed 25 January 2017].

 Sharma, S., van Teijlingen, E., Hundley, V., Angell, C. and Simkhada, P., 2016b. Dirty and 40 days in the wilderness: Eliciting childbirth and postnatal cultural practices and beliefs in Nepal. *BMC Pregnancy and Childbirth*, 16(1), 147. Available from:

https://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/s12884-016-0938-4 [Accessed 25 January 2017].



OPEN ACCESS

Citation: Sharma S, Teijingen Ev, Belizán JM, Hundley V, Simkhada P, Sicuri E (2016) Measuring What Works: An Impact Evaluation of Women's Groups on Maternal Health Uptake in Rural Nepal. PLoS ONE 11(5): e0155144. doi:10.1371/journal. pone.0155144

Editor: C. Mary Schooling, Hunter College, UNITED STATES

Received: March 2, 2015

Accepted: April 25, 2016

Published: May 23, 2016

Copyright: © 2016 Sharma et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data are thirdparty data. The data sets belong to Green Tara Negal who gave us permission to use the data for this paper. As a charity working in the development field Green Tara Nepal is the ethical and legal guardian of the data which they are willing to provide to researchers on request. We confirm that other interested researchers can obtain the data the same way the authors obtained the data by contacting by email the Programme Manager. greentaranegal@gmail.com.

RESEARCH ARTICLE

Measuring What Works: An Impact Evaluation of Women's Groups on Maternal Health Uptake in Rural Nepal

Sheetal Sharma¹*, Edwin van Teijlingen¹, José M. Belizán², Vanora Hundley¹, Padam Simkhada³, Elisa Sicuri^{4,5}

1 Faculty of Health and Social Sciences, Bournemouth University, Dorset, United Kingdom, 2 Institute for Clinical Effectiveness and Health Policy, Buenos Aires, Argentina, 3 Centre for Public Health, Liverpool John Moores University, Liverpool, United Kingdom, 4 I SGlobal, Barcelona Ctr. Int. Health Res. (CRESIB), Hospital Clinic—Universitat de Barcelona, Barcelona, Spain, 5 Health Economics Group, Department of Infectious Disease Epidemiology, School of Public Health, Imperial College London, London, United Kingdom

* ssharma@bournemouth.ac.uk

Abstract

Background

There is a need for studies evaluating maternal health interventions in low-income countries. This paper evaluates one such intervention designed to promote maternal health among rural women in Nepal.

Methods and Results

This was a five-year controlled, non-randomised, repeated cross-sectional study (2007, 2010, 2012) of a participatory community-based maternal health promotion intervention focusing on women's groups to improve maternal health services uptake. In total 1,236 women of childbearing age, who had their last child \leq two years ago, were interviewed. Difference-in-Difference estimation assessed the effects of the intervention on selected outcome variables while controlling for a constructed wealth index and women's characteristics. In the first three years (from 2007 to the 2010), the intervention increased women's likelihood of attending for antenatal care at least once during pregnancy by seven times [OR = 7.0, 95%CI (2.3; 21.4)], of taking iron and folic acid by three times, although not significantly [OR = 2.2, 95%CI (1.0; 4.7)]. Over five years, women were more likely to seek antenatal care at least once [OR = 3.0, 95%CI (1.5; 5.2)], to take iron/folic acid [OR = 1.9, [95% CI (1.1; 3.2)], and to attending antenatal care [OR = 1.5, [95% CI (1.1; 2.2)]. No improvement was found on attending antenatal care in the first trimester, birthing at an institution or with a skilled birth attendant.

Conclusion

Community-based health promotion has a much stronger effect on the uptake of antenatal care and less on delivery care. Other factors not easily resolved through health promotion

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

PLOS ONE

Funding: This work was supported by Green Tara Trust (UK) and Bournemouth University (UK). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

interventions may influence these outcomes, such as costs or geographical constraints. The evaluation has implications for policy and practice in public health, especially maternal health promotion.

Introduction

Health and development interventions in low-income countries (LICs) need high-quality evaluation [1–3]. Although evidence-based practice is making inroads into public health, there are still too many programmes that lack impact evaluation [4–6].

The main question of impact evaluation is one of attribution: isolating the effect of the programme from other intervening factors and potential selection bias of participants [7–10]. Impact evaluation assesses the medium- to longer-term changes of selected outcomes that can be attributed to a particular intervention, such as a project, programme or policy. These may be both the intended and, unintended outcomes [9,11].

In 2007, Green Tara Nepal (GTN), a non-governmental organisation (NGO), designed and implemented a five-year intervention to improve maternal health uptake in rural Nepal. The intervention included participatory action approaches, health promotion and small incentives. In Nepal, there is inequitable access to maternal health services due to factors such as geography, caste, ethnicity, religion and corruption [12–15].

The GTN intervention was designed to fit national policies and incorporated maternal health promotion focusing on women's groups [16–18]. Such health promotion empowers women to seek care when it is needed, which is particularly important in low-income settings [4,19].

This paper reports on the impact evaluation of GTN, selecting as surrogate maternal mortality outcomes factors such as antenatal care (ANC), skilled birth attendance (SBA) and institutional delivery. These outcomes were chosen because they are indirectly correlated with maternal mortality [20–22]. Difference-in-Difference (DiD), analysis was used to identify and measure the effect of the intervention on the selected outcomes [23–28].

Methods

Study area and population

The programme selected two districts (one intervention and one control) with similar socioeconomic characteristics. The districts were close to Kathmandu, each with a total population of just under 9,000. Baseline population characteristics (wealth, age, and education) and health system characteristics were not statistically different between intervention and control districts [29–30]. For instance, the intervention area has a community hospital providing basic emergency obstetric care and two health posts. The control area has two health posts and a primary health care centre nearby (similar to the community hospital in intervention area).

The intervention villages were chosen from a few pre-selected districts not far from Kathmandu that were (a) safe to work at the time of Maoist rebellion (1996–2006), which was still ongoing at the time the intervention was designed (2005/2006); (b) with the local maternal health needs identified by the community and (c) with political commitment towards change [31].

Intervention design

Two health promoters, an auxiliary nurse midwife and a community medical assistant, established and supported women's groups with enrolment running between 2006 and 2012.

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

Groups comprised (a) women age 15 to 49 with at least one child under the age of two; (b) mothers-in-law; and (c) husbands, as the latter two are particularly influential in terms of women's ability to access health services [19, 29, 32–33]. Family decision-making affects the first of the "3 delays" in the maternal morbidity and mortality conceptual framework, namely the delay in making the decision to seek care [34–35].

The intervention took advantage of existing regular monthly women's group meetings, which were originally savings or literacy-based. The rationale was that women's groups have the potential for scaling-up within existing social systems, where community and health institutions are working in synergy to improve access to quality maternal and newborn health services [36]. Once the GTN intervention was implemented, its groups met regularly (monthly) for health promotion activities. The evidence-based maternal health promotion was designed by GTN and maternal health researchers and delivered by GTN health promotion. For evidence signal converting and postnatal care (PNC). The implementation, danger signs of pregnancy, safe delivery and postnatal care (PNC). The implementation consisted of 24 sessions of health promotion, each one lasting one hour. In 2010, there were 37 groups (of 12 daughters-in-law, two men and 23 mixed mothers-in-law/ daughters-in-law) reaching over 1,100 people.

By 2012, there were 46 groups (11 daughter-in-law groups, two male only groups, and 33 mother-in-law and daughter-in-law groups) reaching over 733 people.

GTN's prerogative was to include all castes and empower lower castes to attend. The groups were generally mixed-caste, however, in certain areas, (where not all castes are represented) groups consisted of only one or two castes (S2 Fig).

The GTN health promoters supported the existing health system of sub-health posts by health communication training with mother-child health workers, traditional healers and hospital staff (e.g. neonatal training); and mobile clinics visits to outlying areas.

The intervention included small individual incentives consisting of a baby blanket awarded on completion of four ANC visits, safe delivery kits available at subsidised prices and other goods of the value of less than ten US cents. These incentives aimed to encourage group attendance and health- care seeking behaviour.

Finally, the intervention was flexible; at each stage, barriers were identified; solutions incorporated and the intervention was reapplied to meet the local needs [31].

Evaluation design

This was a five-year controlled non-randomised intervention. Surveys for its evaluation took place at three points in time (repeated cross-sections): before the intervention started in 2007 (baseline), after 2½ years, in 2010, and after five years in 2012. Trained fieldworkers conducted survey interviews. Surveys were conducted in four villages, two in the intervention and two in control districts. The two villages of the intervention district were the ones where the intervention took place; the two villages of the control area were very similar to the intervention villages in terms of health system characteristics and socio-economic conditions. Within the district, the intervention was carried out in two villages selected based on (a) the presence of an adequate health system infrastructure, specifically a community hospital basic emergency obstetric care and two health posts; (b) having obtained approval from the village head; (c) the willingness to collaborate of the local health staff; (d) being neither the richest nor poorest in the district. The list of all women in the four villages (N = 474, 484 and 463, in the first, second and third survey, respectively) meeting inclusion criteria (having at least one child under the age of two years at the moment of survey) was compiled for both the intervention and the control areas. Hence all women meeting the inclusion criteria in the area determined the sample size,

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

and the sampling frame covered the total sample. Women were then approached and interviewed at home. If women were not available on the first home visit, interviewers returned on several occasions, and after the third negative attempt women were dropped from the list. Health and socio-economic data were collected on individual, household and village level predicators/indicators using a structured questionnaire.

All women who met the inclusion criteria, not only those who participated in the groups, were eligible for this evaluation: the evaluation aimed to capture positive 'spillover' effects as any community-based programme can have an impact not only on the women directly receiving the intervention but also on the overall community [37,38].

The Nepal Health Research Council (NHRC) granted ethical approval for the study. The study aims were verbally explained to participants, as well as the consequences of their involvement and the right to withdraw at any point. After verbal consent all data were recorded anonymously.

Data analysis

Descriptive analysis included demographic, cultural and socio-economic characteristics as well as maternal outcomes data and decision-making. DiD assessed the individual probability of engaging in each of the outcome variables and measured the difference in each outcome between intervention and control groups and before and after treatment while controlling for potential explanatory variables.

Logistic multivariate regressions were applied. Control variables, in addition to the ones representing the impact of the intervention, were chosen based on the literature and from previously published Demographic Health Studies data based studies [12,39].

With the aim of evaluating the intervention at two points in time, two different types of regressions were estimated: (a) regressions on the sample of women interviewed at baseline (year 2007) and at midline (year 2010); (b) regressions on the sample of all women in the study, including baseline, midline and final evaluation (years 2007, 2010 and 2012). The former (a) permitted the research team to assess the effects of the intervention after 2 years and the latter (b) to evaluate the overall impact of the intervention after 5 years from start.

Specifically, the regressions from baseline to midline were:

$$Yni = \beta 0 + \beta 1 \text{ interventioni} + \beta 2 \text{ after } + \beta 3 \text{ after } * \text{ interventioni} + \beta 4 \text{ age} \\ + \beta 5 \text{ wealth indexi} + \beta 6 \text{ educationi} + \beta 7 \text{ parityi} \qquad \text{Eq[1]}$$

The regressions from baseline to final evaluation were:

 $\begin{array}{l} Yni = \beta 8 + \beta 9 \ interventioni + \beta 10 \ afterafteri + \beta 11 afterafter * interventioni + \beta 12 \ age \\ + \beta 13 \ wealth \ indexi + \beta 14 \ educationi + \beta 15 \ parityi \end{array} \\ \begin{array}{l} Eq[2] \end{array}$

Where i = 1, ..., N is indicator of each women participating in surveys. Y_n are the binary response variables and n indicates:

- (*n* = 1) ANC attendance at least once during whole pregnancy;
- (*n* = 2) ANC attendance at least once during first trimester;
- (n = 3) at least four ANC visits. The WHO recommends a minimum of four ANC visits and that the first one should be within the first trimester of pregnancy [40];
- (n = 4) presence of a SBA during delivery; SBA in Nepal was defined as nurse-midwives, auxiliary nurse-midwives and obstetricians. The following groups were, excluded: traditional birth attendants, health attendants, medical students as they are not classified by WHO as SBA [41];

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

- (n = 5) institutional delivery (ID, including at hospital, primary health centre, private hospital or clinic. This was chosen as an outcome because it is a recognised strategy to improve maternal child health [42–44];
- (n = 6) attending PNC defined as mother and newborn being seen 24 hours after birth, as 60% of maternal deaths in the low and middle-income countries occur postpartum [45–47]; and
- (n = 7) taking iron and folic acid; in Nepal, supplementation is provided at government health facilities [39] to prevent anaemia and neural tube defects [48].

The control variables were:

- Intervention (treat) {Eqs [1] and [2]} denotes the observations of the two groups: intervention and control;
- After {Eq [1]} denotes time, before (baseline) and after intervention started;
- After* intervention (treat-after) {Eq [1]} identifies the group of observations belonging to the intervention group after the intervention started as compared to the remaining observations (namely all observations belonging to control group and intervention group before the programme started), its estimated coefficient, $\hat{\beta}_3$ represents the impact of the intervention [49].
- Afterafter {Eq [2]} identifies the observations collected both in the midline and in the final evaluation as compared to baseline. As for the variable after in Eq [1], this variable also represent time;
- Afterafter*intervention (treat-afterafter) {Eq [2]} identifies observations belonging to the intervention group at the midline and at the final evaluation as opposed to baseline and to all the observations in the control group at any time. Its estimated coefficient, $\hat{\beta}_{11}$ represents the impact of the overall intervention [49].
- Age = {Eqs [1] and [2]} represents the age of the individual.
- Wealth index {Eqs [1] and [2]} is extracted from a series of assets owned (see details below). This variable tests the hypothesis of maternal attendance/compliance depending on women's socio-economic status [50–52]. This variable was included instead of caste due to the statistically significant relationship between the two variables both in the midline and in the overall evaluation (Respectively: Pearson chi²(14) = 326.14; Pr = 0.00 and Pearson chi²(14) = 424.75 Pr = 0.00).
- Education = {Eqs [1] and [2]} indicates women's level of education.
- Parity = {Eqs [1] and [2]}.
- S1 Table further describes control variables.

Wealth index construction

The wealth index was constructed using Principal Component Analysis (PCA) among a number of assets owned by women's families [53–54]. PCA is commonly used when household expenditure or income data are not available [55]. Assets included in the PCA were: owning a bicycle, motorcycle, goat or car, type of access to hygienic facilities (source of drinking water, type of toilet), number of rooms in house, and materials used in the dwelling (materials used

for flooring, walls, roofing). A description of variables included in the PCA is provided in S1 Table. In S1 Fig we present the distribution of wealth according to high, middle and low castes, according to the published definition of caste [39,56]. High caste (= 1) included: Brahmin, Chhetri, Newar; middle caste (= 2) included: Tamang middle, or low (= 3): Newar Dalit, Balami, Dalit and others (Christian or Muslim), S1. In S1 Fig the zero represents the median of the wealth index ranking. The distribution is skewed to the left due to the poverty level of this population.

Data were analysed with STATA" version 11.0 (Stata/SE 11.0 Stata corporation, College Station, TX, USA).

Results

A total of 1,236 women (611 in the control and 625 in intervention area) completed the surveys, with an overall average response rate of 87%. Specifically, N = 412 women participated in the first, N = 421 in the second and N = 403 in the third survey (S2 Fig). Overall, the mean age of respondents was 25.4 ± 5.1 years, the mean age of marriage 19.6 ± 3.3 years and the mean age at their first pregnancy 20.9 ± 3.2 years. The main occupation of respondents was either housewife or farmer (89.5%). Most women were multigravida (56.1%) and 43.9% were primigravida (Table 1).

In the five-year period, the characteristics of the sample changed, the number of Buddhists decreased, while Hindus remained the majority. At baseline, most women were married before the age of 20; this proportion fell to 40.4% in the control group and to 48.5% in the intervention group in last survey. Literacy rates increased steadily (Table 1).

In terms of decision-making, over time a larger proportion of women reported to have planned pregnancy and to have more autonomy over maternity care, for instance ANC and place of delivery self-decisions increased over the duration of the study (Table 2). These changes were statistically significant in the intervention area.

At baseline 80.4% of women in the control area attended ANC compared to 84.6% in the intervention area. In the intervention area, from baseline to final evaluation, the proportion of women who sought ANC at least once significantly increased from 84.6% to 98.0%. The proportion seeking ANC in the first trimester significantly increased from 47.7% to 62.4%; those seeking ANC four or more times significantly increased from 67.3% to 81.0%. In addition, a greater proportion of women reported taking iron/folic acid (from 86.5% to 96.0%) and seeing a SBA (from 60.6% to 82.0%). Significant increases were also seen in seeking an institutional delivery (from 60.6% to 76.0%) and PNC (from 52.2% to 85.9%). Use of safe delivery kit significantly increases from 5.0% to 34.3%. Improvements were also registered in the control group but not all were significant (Table 3).

Impact of the intervention

Tables 4 and 5 show the estimated odds ratios (OR) for the mid and overall evaluations, respectively. The effect of the intervention is represented by the estimated values of the OR of the variables treat-after (midline evaluation) and treat-afterafter (final evaluation). From baseline to the midline there was an increase in women's likelihood of attending ANC at least once during whole pregnancy by 7.0 times [OR = 7.0, 95%CI (2.3; 21.4)]. A significant increase was also seen in the probability of taking iron/folic acid [OR = 3.0, 95%CI (1.2; 7.9)]. The probability of seeking four or more ANC check-ups had doubled, but this increase was not statistically significant (OR = 2.2, 95%CI (1.0; 4.7)] (Table 4). Over the five years (from baseline to final term), women were three times more likely of seeking ANC at least once [OR = 3.0, 95%CI (1.5; 5.8)].

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

PLOS ONE

Table 1. Characteristics of respondents (intervention/control).

		Control			ļ	Intervention		
Characteristic used as denominator	Baseline2007	Midline 2010	Final 2012	p-va l ue*	Baseline2007	Midline2010	Final 2012	p-value*
N	204	204	203		208	217	200	
Religion		%		0.0014		%		0.0042
Buddhist	34.8	18.1	31.0		31.7	22.6	18.5	
Hindu	62.3	80.4	68.0		66.4	74.7	77.5	
Other (Christian, Muslim)	2.9	1.5	1.0		2.0	2.8	3.0	
Caste/Ethnicity				0.2048				0.0358
Brahmin	10.3	10.8	6.9		19.2	13.8	13.5	
Chhetri	20.6	18.1	17.2		14.9	7.8	12.5	
Tamang	38.2	35.8	38.4		39.9	51.2	40.5	
Newar non Dalit	25.5	27.5	26.1		14.4	13.4	19.0	
Newar Da l it	1.5	2.5	2.5		1.4	0.5	1.5	
Dalit	1.5	2.0	2.5		1.9	3.2	3.0	
Balami	0	0	0		6.3	5.5	7.0	
Other (Gurung etc.)	2.5	3.4	6.4		1.9	4.6	3.0	
Age of marriage (yrs ¹)				0.0173				0.0165
15–19	50.0	52.9	41.4		60.6	60.4	48.5	
20–24	38.4	40.7	45.3		37.0	35.0	43.5	
25–29	10.8	5.4	10.3		1.9	4.2	7.5	
30 and above	2.0	2.0	3.0		0.5	0.5	0.5	
Age of 1 st pregnancy				0.0135				0.0001
14–19	37.7	37.3	27.6		51.4	37.8	31.5	
20–24	47.1	52.0	52.7		42.8	52.5	53.5	
25–29	13.7	8.8	15.3		4.8	8.7	13.0	
30 and above	1.5	2.0	4.43		2.0	2.0	2.0	
Literacy	64.2	77.0	70.4	0.0187	66.4	73.3	81.0	0.0037
Education				0.1269				0.0025
None	43.6	31.4	33.0		39.9	33.2	24.5	
Primary	28.4	41.7	34.5		31.7	42.9	38.0	
Secondary and higher	27.9	27.0	32.5		28.4	24.0	37.5	

¹ Yrs-Years

*p-values are based on Kruskal Wallis test to compare each categorical variable across time.

doi:10.1371/journal.pone.0155144.t001

Women were nearly twice as likely [OR = 1.9, [95% CI (1.1; 3.2)] to take iron/folic acid, and once and a half times as likely to attend PNC [OR = 1.5, [95% CI (1.1; 2.2)].

No effect was seen in the midline, or in the overall evaluation, on attending ANC in the first trimester, seeking an institutional delivery and having a SBA (Tables 4 and 5). A high OR was found for four ANC visits (Table 4), but not in the overall evaluation (Table 5).

Results on remaining covariates

Wealth was a significant factor explaining a high proportion of the variation in all the outcomes both in the midline and in the overall evaluation. In particular, being richer (3^{rd} tertile) compared to being poorer (1^{st} tertile) increased substantially the probability of having a SBA at birth by the midline [(OR = 11.0, 95% CI (6.3; 19.4)] and by the overall evaluation [OR = 9.3, 95% CI (5.9; 14.7)].

PLOS ONE DOI:10.1371/journal.pone.0155144 May 23, 2016

Impact Evaluations or How to Measure What Works' in Health

Table 1. Characteristics of respondents (intervention/control).

		Control				Intervention		
Characteristic used as denominator	Baseline2007	Midline 2010	Final 2012	p-value*	Baseline2007	Midline2010	Final 2012	p-value*
N	204	204	203		208	217	200	
Religion		%		0.0014		%		0.0042
Buddhist	34.8	18.1	31.0		31.7	22.6	18.5	
Hindu	62.3	80.4	68.0		66.4	74.7	77.5	
Other (Christian, Muslim)	2.9	1.5	1.0		2.0	2.8	3.0	
Caste/Ethnicity				0.2048				0.0358
Brahmin	10.3	10.8	6.9		19.2	13.8	13.5	
Chhetri	20.6	18.1	17.2		14.9	7.8	12.5	
Tamang	38.2	35.8	38.4		39.9	51.2	40.5	
Newar non Da l it	25.5	27.5	26.1		14.4	13.4	19.0	
Newar Dalit	1.5	2.5	2.5		1.4	0.5	1.5	
Dalit	1.5	2.0	2.5		1.9	3.2	3.0	
Balami	0	0	0		6.3	5.5	7.0	
Other (Gurung etc.)	2.5	3.4	6.4		1.9	4.6	3.0	
Age of marriage (yrs ¹)				0.0173				0.0165
15-19	50.0	52.9	41.4		60.6	60.4	48.5	
20–24	38.4	40.7	45.3		37.0	35.0	43.5	
25-29	10.8	5.4	10.3		1.9	4.2	7.5	
30 and above	2.0	2.0	3.0		0.5	0.5	0.5	
Age of 1 st pregnancy				0.0135				0.0001
14–19	37.7	37.3	27.6		51.4	37.8	31.5	
20-24	47.1	52.0	52.7		42.8	52.5	53.5	
25-29	13.7	8.8	15.3		4.8	8.7	13.0	
30 and above	1.5	2.0	4.43		2.0	2.0	2.0	
Literacy	64.2	77.0	70.4	0.0187	66.4	73.3	81.0	0.0037
Education				0.1269				0.0025
None	43.6	31.4	33.0		39.9	33.2	24.5	
Primary	28.4	41.7	34.5		31.7	42.9	38.0	
Secondary and higher	27.9	27.0	32.5		28.4	24.0	37.5	

1 Yrs-Years

*p-values are based on Kruskal Wallis test to compare each categorical variable across time.

doi:10.1371/journal.pone.0155144.t001

Women were nearly twice as likely [OR = 1.9, [95% CI (1.1; 3.2)] to take iron/folic acid, and once and a half times as likely to attend PNC [OR = 1.5, [95% CI (1.1; 2.2)].

No effect was seen in the midline, or in the overall evaluation, on attending ANC in the first trimester, seeking an institutional delivery and having a SBA (Tables 4 and 5). A high OR was found for four ANC visits (Table 4), but not in the overall evaluation (Table 5).

Results on remaining covariates

Wealth was a significant factor explaining a high proportion of the variation in all the outcomes both in the midline and in the overall evaluation. In particular, being richer (3^{rd} tertile) compared to being poorer (1^{st} tertile) increased substantially the probability of having a SBA at birth by the midline [(OR = 11.0, 95% CI (6.3; 19.4)] and by the overall evaluation [OR = 9.3, 95% CI (5.9; 14.7)].

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

Impact Evaluations or How to 'Measure What Works' in Health

	Seeking ANC ¹ at least once	Seeking ANC in the 1st Trimester	Seeking ANC 4 or more times	Taking Iron/Folic Acid during pregnancy	SBA ²	ID ³	Seeking PNC ⁴
Observations	832	714	832	831	832	830	832
Treat	1.3 (0.7; 2.4)	0.7 (0.4; 1.1)	1.2 (0.7; 2.0)	2.3 (1.2; 4.1)**	1.2 (0.7; 1.9)	1.2 (0.8; 2.0)	1.5 (0.9; 2.3)
After	1.2 (0.6; 2.2)	1.6 (1.0; 2.6)	0.9 (0.5; 1.5)	1.0 (0.6; 1.7)	1.3 (0.8; 2.0)	0.8 (0.5; 1.3)	2.3 (1.5; 3.6)**
Treat-after	7.0 (2.3; 21.4)**	1.2 (0.6; 2.4)	2.2 (1.0; 4.7)*	3.0 (1.2; 7.9)**	1.5 (0.7; 2.8)	1.7 (0.9; 3.3)	1.6 (0.9; 3.1)
Wealth							
Wealth 2	3.4 (1.9; 6.0)**	1.7 (1.1; 2.5)**	1.2 (0.8; 1.9)	2.6 (1.5; 4.4)**	2.7 (1.9; 3.9)**	2.2 (1.5; 3.2)**	1.7 (1.2; 2.5)**
Wealth 3	6.0 (2.3; 15.7)**	3.1 (1.9; 4.9)**	5.2 (2.6; 10.8)**	2.8 (1.3; 5.8)**	11.0 (6.3; 19.4)**	7.6 (4.6; 12.7)**	4.0 (2.5; 6.4)**
Age	0.9 (0.9; 1.0)**	1.0 (1.0; 1.1)	1.0 (1.0; 1.0)	0.9 (0.9; 1.0)**	1.0 (1.0; 1.0)**	1.0 (1.0; 1.0)	1.0 (1.0; 1.0)
Education							
Education 2	5.2 (2.7; 10.1)**	1.5 (1.0; 2.3)**	1.0 (0.6; 1.5)	4.0 (2.3; 7.0)**	2.0 (1.3; 2.9)**	2.0 (1.3; 2.9)**	2.1 (1.4; 3.0)**
Education 3	9.3 (3.1; 28.0)**	2.6 (1.6; 4.2)	2.0 (1.0; 3.9)**	10.1 (3.9; 25.8)**	4.7 (2.8; 8.1)**	4.3 (2.6; 7.1)**	4.5 (2.8; 7.3)**
Parity							
Parity 2	0.7 (0.4; 1.2)	0.6 (0.4; 0.9)**	0.6 (0.4; 0.9)**	0.5 (0.3; 1.0)**	0.5 (0.4; 0.8)**	0.53 (0.4; 0.8)**	0.7 (0.5; 1.0)**
Parity 3	0.5 (0.3; 0.9)**	0.6 (0.4; 0.9)**	0.6 (0.4; 1.1)	0.3 (0.2; 0.6)**	0.6 (0.4; 1.0)*	0.6 (0.4; 1.0)**	0.75 (0.5; 1.2)

Table 4. Difference in difference analysis of maternal health uptake (intervention and control) at the midline evaluation.

¹ ANC—Antenatal care

² SBA–Skilled birth attendant

³ ID-Institutional delivery

⁴ PNC-Postnatal care

**pvalue<0.05

doi:10.1371/journal.pone.0155144.t004

Relative to the construction of the wealth index, the first component extracted explained 20% of total variability in the population. The scores based on the first component were grouped into tertiles in S1 Fig, with the lowest (group 1) representing the poorest and the higher (group 3) representing the richest women. Age was a significant factor in determining whether women sought one antenatal visit and took iron/folic acid both at the midline and final evaluation. In both cases, being older lowered the probability of a positive outcome. Having higher education compared to no education increased the probability of all the outcomes considered. In particular, having secondary school or higher-level education increased the probability of attending ANC at least once in the midline [OR = 9.3, 95% CI (3.1; 28.0)] and overall evaluation [OR = 11.0, 95% CI (4.2; 29.0)].

In the intervention area (variable treat), women were more predisposed to seek an institutional delivery [OR = 1.5, 95% CI (1.0; 2.3)] and PNC [OR = 1.5, 95% CI (1.5; 2.3)] at any time. Women in the intervention area were 2.3 [OR = 2.3, 95% CI (1.2; 4.1) times more likely at midline to take iron/folic and 2.4 [OR = 2.4, 95% CI (1.4; 4.2)] by year 5. Over time (variable afterafter), women become increasingly more likely to have a SBA at birth [OR = 1.3, 95% CI (1.4; 2.2)], institutional delivery [OR = 1.3, 95% CI (1.0; 1.6)] and PNC [OR = 1.8, 95% CI (1.4; 2.2)], reflecting background changes. With increasing parity, the ORs for all outcomes remain significantly below 1, indicating a negative relationship between having more than one child and the

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

aaa in diffa

Table 5 Differe

Impact Evaluations or How to Measure What Works' in Health

Table 5. Differe	nee in amerence and	arysis of maternal neuro	aptake in the overall	evaluation.			
	Seeking ANC ¹ at least once	Seeking ANC in the 1st Trimester	Seeking ANC 4 or more times	Taking Iron/Folic Acid during pregnancy	SBA ²	ID ³	Seeking PNC ⁴
Observations	1235	1086	1235	1233	1235	1233	1235
Treat	1.5 (0.8; 2.6)	0.7 (0.5; 1.0)	1.5 (0.9; 2.4)	2.4 (1.4; 4.2)**	1.3 (0.9; 2.0)	1.5 (1.0; 2.3)**	1.5 (1.0; 2.3)**
Afterafter	1.2 (0.9; 1.6)	1.0 (0.8; 1.3)	1.0 (0.8; 1.3)	0.9 (0.7; 1.2)	1.3 (1.1; 1.7)**	1.3 (1.0; 1.6)**	1.8 (1.4; 2.2)**
Treat- afterafter	3.0 (1.5; 5.8)**	1.2 (0.9; 1.7)	1.1 (0.7; 1.6)	1.9 (1.1; 3.2)**	1.0 (0.7; 1.5)	0.9 (0.7; 1.3)	1.5 (1.1; 2.2)**
Wealth							
Wealth 2	2.5 (1.6; 4.0)**	1.8 (1.3; 2.4)**	1.4 (1.0; 2.0)	2.6 (1.7; 4.0)**	2.6 (1.9; 3.5)**	2.2 (1.6; 3.0)**	1.7 (1.2; 2.3)**
Wealth 3	4.7 (2.2; 10.4)**	2.8 (1.9; 4.1)**	3.3 (2.0; 5.6)**	2.4 (1.4; 4.2)**	9.3 (5.9; 14.7)**	7.0 (4.6; 10.6)**	3.8 (2.6; 5.7)**
Age	0.9 (0.9; 1.0)**	1.0 (1.0; 1.0)	1.0 (1.0; 1.0)	1.0 (1.0; 1.0)**	1.0 (1.0; 1.0)	1.0 (1.0; 1.0)	1.0 (1.0; 1.0)
Education							
Education 2	4.7 (2.7; 8.1)**	1.4 (1.0; 1.9)	1.0 (0.7; 1.5)	3.3 (2.1; 5.1)**	1.9 (1.4; 2.6)**	1.9 (1.4; 3.0)**	2.2 (1.6; 3.1)**
Education 3	11.0 (4.2; 29.0)**	2.2 (1.5; 3.3)**	1.8 (1.1; 3.0)**	9.3 (4.6; 18.9)**	4.6 (2.9; 7.1)**	4.0 (2.6; 6.0)**	4.7 (3.1; 7.1)**
Parity							
Parity 2	0.6 (0.4; 1.0)*	0.7 (0.5; 1.0)**	0.7 (0.5; 1.04)	0.6 (0.4; 0.9)**	0.5 (0.3; 0.6)**	0.5 (0.4; 0.7)**	0.6 (0.4; 0.7)**
Parity 3	0.4 (0.3; 0.8)	0.6 (0.4; 1.0)**	0.7 (0.4; 1.09)	0.3 (0.2; 0.5)**	0.5 (0.3; 0.8)**	0.5 (0.4; 0.8)**	0.7 (0.4; 1.0)**

nce analysis of maternal health untake in the overall evaluation

¹ ANC—Antenatal care

² SBA–Skilled birth attendant

³ ID-Institutional delivery

⁴ PNC-Postnatal care

**pvalue<0.05

doi:10.1371/journal.pone.0155144.t005

attendance to any type of maternal health service (either ANC or PNC) or the adherence to any type of pregnancy intervention, such as folic acid and iron intake.

Discussion

This evaluation showed that the health promotion intervention had a positive effect on the uptake of ANC (attending at least once), iron/folic acid intake and PNC, but not on institutional delivery. While there was a positive effect on ANC attendance at least once during pregnancy, no effect was seen on ANC attendance in the first trimester. This may be because women become aware of the pregnancy status later or there are cultural reasons for the pregnancy to be kept a 'secret' [33–34, 42, 57]. Often first-time mothers need to ask permission and money from her family to attend ANC [58].

ANC attendance reduced from the midline to the overall evaluation, suggesting that the intervention may have diminishing returns: it could be argued that the effect was stronger when moving from low coverage to a medium level beyond which marginal improvements started decreasing [59]. The same trend as for ANC at least once was seen for iron/folic acid supplementation [60].

The barriers to accessing institutional delivery may be due to distance and socio-cultural factors (e.g. not part of a family's birth preparedness plans) that cannot be overcome by a

PLOS ONE DOI:10.1371/journal.pone.0155144 May 23, 2016

community-based intervention. Other studies in Nepal have suggested socio-economic, financial and geographical obstacles to seeking delivery care [45,61]. Studies with a focus in similar interventions have found that women's groups in LICs increase the knowledge of obstetric 'danger' signs but have little impact on SBA due to other barriers, such as the cost of reaching a facility [62–64].

Although ANC attendance and provision of iron/folic acid during pregnancy are targets likely to be achieved in a community-based intervention such interventions without additional resources (such as ambulances) may not be influential enough to increase SBA or institutional delivery. Whilst PNC uptake is generally low for similar reasons to ANC non-attendance [65– 66]. According to descriptive statistics, coverage of PNC greatly increased over the five years, yet the intervention only had a positive and significant effect at the very end of the intervention.

However, during 2007–2010, improvements were not associated with the GTN intervention but were due to other determinants, as witnessed by the significance of the variable "time" in both midline and overall regressions. The progressive improvement in women's level of education and empowerment within the household may have played a role. Empowerment increased for women in particular with regard to decision-making power for ANC and delivery care. However, empowerment was not included in the estimated models as the trend was captured both by time and education (education, age and parity were strongly correlated). Interestingly, the increased women's empowerment could be attributed to the health promotion intervention itself. Complex relationships are likely to exist among education, empowerment, maternal outcomes and the health promotion intervention [67,68].

In the overall evaluation, the factor "time" (represented by variable "afterafter") was significant for PNC, delivery care (SBA and institutional delivery) outcomes, highlighting that other factors, not the GTN intervention, played a role. For instance, women were more likely to attend care if they have high household wealth, higher levels of education and lower parity [57,69]. A wealth index was preferred to caste as a more refined measurement for its greater inclusivity as a socio-economic indicator.

These results suggest that health promotion groups improve access to maternal health when individual, socio-economic and environment (health system) conditions are addressed [70]. Moreover, for this rural LICs setting, a health promotion intervention facilitating behavioural change may be more suitable as opposed to a cash transfer scheme alone. Cash transfer schemes, such as the 'Safe Mother Program' (*Aama*-Suraksha-Karyakram) maternity incentive scheme, are often not financially sustainable [71–72].

Measuring the effect of a community-based intervention is not straightforward because of confounding environmental factors. Therefore, a (quasi-) experimental design is needed to ascertain whether the changes or improvements are due to the intervention or to external factors [5,28,73]. Previous studies have evaluated community-based interventions through randomised controlled trials (RCTs), the gold standard methodology for measuring effectiveness [74]. However, RCTs are difficult to organise, expensive and often setting-inappropriate for up-scaling or replicating and difficult to set up and run in LICs [25,75].

There is a need for community-based interventions to be accompanied by proper evaluations. Such evaluations can take the guesswork out of policy-making by knowing what works, what does not work and why [76–79]. Given the quasi-experimental study design, DiD permitted an approximation as close to an assessment of the effect of the programme on the outcomes of interest. Previous studies have assessed the effectiveness of programmes whereby a DiD approach was applied to community-based interventions [24–25]. In terms of data analysis, the criteria for inclusion of variables in a multivariate model vary between problems and disciplines. We followed methodologists suggesting the inclusion of all relevant variables in the

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

model regardless of their significance in order to control for as many as possible covariates [80]. This approach, however, led in some cases to wide standard errors in estimated parameters, but as control variables were based on the literature, the impact of irrelevant variables was limited.

For this evaluation, maternal health behaviour before and after the intervention was not independently measured but based on self-reporting, which may have led to issues of validity and to recall bias. However, it is unlikely that the quality of self-reporting was different between intervention and control groups.

A weakness of this study is that the data collected did not identify any social barriers (e.g. gender of health worker), financial or geographical barriers (e.g. travel distance) to healthcare uptake that might account for the changes found, or lack thereof, between the data collection points. Moreover, for any intervention to improve maternal care and leverage commitment that maternal health should be a human right, it is important to know whether it is sustainable, scalable and cost-effective [81].

Conclusion

This impact evaluation suggested that the community-based health promotion intervention had a greater effect on the uptake of ANC than on delivery care. Other factors, not easily resolved through health promotion interventions, may influence the latter outcomes, such as costs or geographical constraints. However, all the selected maternal outcomes improved in time while, in parallel, women's level of education and empowerment also increased, which may either have facilitated or be the consequence of the intervention. Interventions should prioritise impact evaluations to inform future health policies.

Supporting Information

S1 Fig. Wealth index distribution of participants. (DOCX)

S2 Fig. GTN health promotion intervention in Nepal and its evaluation. (DOCX)

S1 Table. Variables description and codification, overall evaluation. (DOCX)

Acknowledgments

We thank all research participants, also the staff of Green Tara Nepal, the health posts and the Female Community Health Volunteers. We also thank the anonymous reviewers for PLOS One for their helpful comments. We are grateful to the rural communities in Nepal who participated in the interventions and gave us advice.

Author Contributions

Conceived and designed the experiments: SS $\rm EvT$ PS. Analyzed the data: SS ES. Wrote the paper: SS $\rm EvT$ JMB VH PS ES.

References

 Bhutta ZA, Darmstadt GL, Hasan BS, Haws RA. Community-based interventions for improving perinatal and neonatal health outcomes in developing countries: a review of the evidence. Pediatrics. 2005; 115(Suppl 2):519–617.

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

- Godin G, Gagnon H, Alary M, Levy J, Otis J. The degree of planning: an indicator of the potential success of health education programs. Promot Educ. 2007; 14(3):138–42. PMID: 18154222
- Adam T, Hsu J, de Savigny D, Lavis JN, Røttingen J-A, Bennett S. Evaluating health systems strengthening interventions in low-income and middle-income countries: are we asking the right questions? Health Policy Plan. 2012; 27(Suppl 4): iv9–19. PMID: 23014156
- World Health Organization (WHO) Regional Publications. European Series. Evaluation in health promotion: principles and perspectives. 92 ed. Rootman I, Goodstadt M, Hyndman B, McQueen D, Potvin L, Springet J, et al., editors. Geneva: WHO; 2001. Available from: http://www.euro.who.int/_data/ assets/pdf_file/0007/108934/E73455.pdf
- Subba N, Subba S. Modification of delivery practice in Rajbanshi mothers of Nepal. J Nobel Med Coll. 2014; 3(1):10–5.
- Malenga G, Molyneux M. The Millennium Villages project. Lancet. 2012; 379(9832):2131–33. doi: 10. 1016/S0140-6736(12)60369-9 PMID: 22572601
- Christiansen G. Evaluation as a quality assurance tool in health promotion. Research and practice for health promotion. Research and Practice of Health Promotion. Cologne, Germany: Federal Centre for Health Education; 1999.
- Fitz-Gibbon C, Morris L. How to design a program evaluation. Newbury Park, CA: Sage Publications; 1987.
- Khandker SR, Koolwal GB, Samad HA. Handbook on impact evaluation: quantitative methods and practices. Washington, D.C.: The World Bank; 2010. Available from: http://books.google.es/books? id=vVOQUUZmNqMC
- J-PAL. Impact Evaluation. 2015 [cited 2015 Feb]. Available from: http://www.povertyactionlab.org/ methodology/what-evaluation/impact-evaluation
- The World Bank. Impact evaluation—The experience of the independent evaluation group of the World bank. Washington, D.C.: The independent evaluation group of the World Bank. 2006 [cited 2016 March]. Available from: http://inweb90.worldbank.org/oed/oeddoclib.nsf/ DocUNIDViewForJavaSearch/35BC42095BF58F8852571 E00068C6BD/Sfile/impact_evaluation.pdf
- Ministry of Health and Population (MOHP) New Era and ICF International Inc. Nepal Demographic and Health Survey 2006. Kathmandu, Nepal: Ministry of Health and Population (MOHP) New Era and ICF International Inc. 2007.
- 13. Transparency International. Transparency International: the global coalition against corruption. Nepal. 2014. Available from: http://www.transparency.org/country#NPL
- Oxford Poverty and Human Development Initiative. "Nepal Country Briefing", Multidimensional Poverty Index Data Bank. Oxford: Oxford University: 2013. Available from: www.ophi.org.uk/multidimensionalpoverty-index/mpi-country-briefings/
- Muldoon KA, Galway LP, Nakajima M, Kanters S, Hogg RS, Bendavid E, et al. Health system determinants of infant, child and maternal mortality: A cross-sectional study of UN member countries. Global Health. 2011; 7(1):42.
- Rosato M, Malamba F, Kunyenge B, Phiri T, Mwansambo C, Kazembe P, et al. Strategies developed and implemented by women's groups to improve mother and infant health and reduce mortality in rural Malawi. Int Heal. 2012; 4 (3): 176–84.
- Manandhar DS, Osrin D, Shrestha BP, Mesko N, Morrison J, Tumbahangphe KM, et al. Effect of a participatory intervention with women's groups on birth outcomes in Nepal: cluster-randomised controlled trial. Lancet. 2004; 364(9438):970–79. PMID: 15364188
- Shrestha B, Bhandari B, Manandhar D, Osrin D, Costello A, Saville N. Community interventions to reduce child mortality in Dhanusha, Nepal: Study protocol for a cluster randomized controlled trial. Trials. 2011; 12(1):136.
- Simkhada B, van Teijlingen E, Porter M, Simkhada P. Major problems and key issues in maternal health in Nepal. Kathmandu Univ Med J. 2006; 4(2):258–63.
- Say L, Raine R. A systematic review of inequalities in the use of maternal health care in developing countries: examining the scale of the problem and the importance of context. Bull World Health Organ. 2007; 85(10):812–9. PMID: 18038064
- Filippi V, Ronsmans C, Campbell OMR, Graham WJ, Mills A, Borghi J, et al. Maternal health in poor countries: the broader context and a call for action. Lancet. 2006; 368(9546):1535–41. PMID: 17071287
- 22. Donnay F. Maternal survival in developing countries: what has been done, what can be achieved in the next decade. Int J Gynecol Obstet. 2000; 70(1):89–97.

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

- Bonell CP, Hargreaves J, Cousens S, Ross D, Hayes R, Petticrew M, et al. Alternatives to randomisation in the evaluation of public health interventions: Design challenges and solutions. J Epidemiol Community Health. 2011; 65 (7): 582–7. doi: 10.1136/jech.2008.082602 PMID: 19213758
- 24. Liu X, Yan H, Wang D. The evaluation of "Safe Motherhood" program on maternal care utilization in rural western China: a difference in difference approach. BMC Public Health. 2010; 10(1):566.
- Alderman H, Ndiaye B, Linnemayr S, Ka A, Rokx C, Dieng K, et al. Effectiveness of a community-based intervention to improve nutrition in young children in Senegal: A difference in difference analysis. Public Health Nutr. 2009; 12(05):667–73.
- 26. Waldinger F. Difference-in-Differences. Lecture 3 [Internet]. 2010. Available from: http://www2. warwick.ac.uk/fac/soc/economics/staff/academic/waldinger/teaching/ec9a8/slides/lecture_3_-_did.pdf
- 27. Reynolds HW, Wong EL, Tucker H. Adolescents use of maternal and child health services in developing countries. Int Fam Plan Perspect. 2006; 32(1):6–16. PMID: 16723297
- Rauniar G, Das B, Manandhar T, Bhattacharya S. Effectiveness of an educational feedback intervention on drug prescribing in dental practice. Kathmandu Univ Med J. 2012; 10(40):30–5.
- Simkhada P, van Teijlingen E, Kadel S, Stephens J, Sharma S, Sharma M. Reliability of national data sets: evidence from a detailed small area study in rural Kathmandu Valley, Nepal. Asian J Epidemiol. 2009; 2:44–8.
- Central Bureau of Statistics. Statistical yearbook of Nepal. Kathmandu: Central Bureau of Statistics; 2001. Available from: http://cbs.gov.np/?page_id=1233
- van Teijlingen E, Simkhada P, Stephens J, Simkhada B, Rogers S, Sharma S. Making the best use of all resources: Developing a health promotion intervention in rural Nepal. Health Renaissance. 2012; 10 (3):229–35.
- Milne L, van Teijlingen E, Hundley V, Simkhada P, Ireland J. Staff perspectives of barriers to women accessing birthing services in Nepai: A qualitative study. BMC Pregnancy Childbirth. 2015; 15: 142 doi:10.1186/s1284-015-0664-0 PMID: 26133977
- Simkhada B, Porter M, van Teijlingen E. The role of mothers-in-law in antenatal care decision-making in Nepal: A qualitative study. BMC Pregnancy Childbirth. 2010; 10(1):34. Available from: http://www. biomedcentral.com/1471-2393/10/34
- Thaddeus S, Maine D. Too far to walk: Maternal mortality in context. Soc Sci Med. 1994; 38(8):1091– 110. PMID: 8042057
- Nour NM. An introduction to maternal mortality. Rev Obstet Gynecol. 2008; 1(2):77–81. PMID: 18769668
- World Health Organization (WHO). Technical Brief No.3 Scaling Up Health Services: Challenges and Choices [Internet]. 2008. Available from: http://www.who.int/healthsystems/topics/delivery/technical_ brief_scale-up_june12.pdf
- de Heer HD, Koehly L, Pederson R, Morera O. Effectiveness and spillover of an after-school health promotion program for Hispanic elementary school children. Am J Public Health. 2011; 101: 1907–13. doi: 10.2105/AJPH.2011.300177 PMID: 21852659
- Vanderweele TJ, Hong G, Jones SM, Brown JL. Mediation and spillover effects in group-randomized trials: A case study of the 4Rs educational intervention. J Am Stat Assoc. 2013; 108: 469–482. doi: 10. 1080/01621459.2013.779832 PMID: 23997375
- Ministry of Health and Population. Nepal annual report 2009/2010 [Internet]. Kathmandu: Ministry of Health and Population, Nepal; 2012. Available from: http://dohs.gov.np/wp-content/uploads/2014/04/ Annual_Report_2066_67.pdf
- 40. World Health Organization (WHO), United Nations Children's Fund (UNICEF). Antenatal Care in Developing Countries: Promises, Achievements and Missed Opportunities: An Analysis of Trends, Levels, and Differentials: 1990–2001., Geneva: WHO; 2003. Available from: http://www.childinfo.org/files/antenatal_care.pdf
- World Health Organization (WHO). Making pregnancy safer: the critical role of the skilled attendant: A joint statement by WHO, ICM and FIGO [Internet]. 2004. Available from: http://www.who.int/maternal_ child_adolescent/documents/9241591692/en/
- 42. Asefa A, Teshome W, Mesele T, Letamo Y. Use of institutional delivery services in the Southern Nations, Nationalities, and People's Region, Ethiopia: A cross-sectional comparative mixed methods study. Lancet. 2013; 382:9. Available from: http://www.sciencedirect.com/science/article/pii/ S0140673613621704 doi: 10.1016/S0140-6736(13)60225-1
- Kesterton A, Cleland J, Sloggett A, Ronsmans C. Institutional delivery in rural India: the relative importance of accessibility and economic status. BMC Pregnancy Childbirth. 2010; 10(1):30.
- 44. Kestler E, Walker D, Bonvecchio A, de Tejada SS, Donner A. A matched pair cluster randomized implementation trail to measure the effectiveness of an intervention package aiming to decrease perinatal

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

mortality and increase institution-based obstetric care among indigenous women in Guatemala: Study protocol. BMC Pregnancy Childbirth. 2013; 13(1):73.

- World Health Organization (WHO). WHO recommendations on postnatal care of the mother and newborn. Geneva: WHO; 2013. Available from: http://apps.who.int/iris/bitstream/10665/97603/1/ 9789241506649_eng.pdf
- Middleberg MI. Promoting reproductive security in developing countries. New York: Kluwer Academic/ Plenum Publishers; 2003.
- Li XF, Fortney JA, Kotelchuck M, Glover LH. The postpartum period: The key to maternal mortality. Int J Gynecol Obstet. 2014; 54(1):1–10.
- World Health Organization (WHO). Daily iron and folic acid supplementation in pregnant women: guidelines. Geneva: WHO; 2012. Available from: http://apps.who.int/iris/bitstream/10665/77770/1/ 9789241501996_eng.pdf
- Schlotter M, Schwerdt G, Woessmann L, Econometric Methods for Causal Evaluation of Education Policies and Practices: A Non-Technical Guide. Ifo Institute, University of Munich and IZA Discussion Paper No. 4725, 2010. Available from: http://repec.iza.org/dp4725.pdf
- Ahmed S, Creanga AA, Gillespie DG, Tsui AO. Economic status, education and empowerment: Implications for maternal health service utilization in developing countries. PLoS One. 2010; 5(6):e11190. Available from: http://www.plosone.org/article/info:doi/10.1371/journal.pone.0011190 doi: 10.1371/ journal.pone.0011190 PMID: 20585646
- Simkhada B, van Teijlingen ER, Porter M, Simkhada P. Factors affecting the utilization of antenatal care in developing countries: Systematic review of the literature. J Adv Nurs. 2008; 61(3):244–60. doi: 10.1111/j.1365-2648.2007.04532.x PMID: 18197860
- Tuntiseranee P, Olsen J, Chongsuvivatwong V, Limbutara S. Socioeconomic and work related determinants of pregnancy outcome in southern Thailand. J Epidemiol Community Health; 1999; 53(10):624– 9. PMID: 10616674
- Howe L, Hargreaves J, Huttly S. Issues in the construction of wealth indices for the measurement of socio-economic position in low-income countries. Emerg Themes Epidemiol. 2008; 5(1):1–14. http://dx. doi.org/10.1186/1742-7622-5-3
- Vyas S, Kumaranayake L. Constructing socio-economic status indices: how to use principal components analysis. Heal Policy Plan. 2006; 21 (6): 459–68.
- 55. Filmer D, Pritchett L. Estimating wealth effects without expenditure data—or tears: an application to educational enrollments in states of India. Demography. 2001; 38(1):115–32. PMID: 11227840
- Bennett L, Dahal DR, Govindasamy PCaste, Ethnic and Regional Identity in Nepal: Further Analysis of the 2006 Nepal Demographic and Health Survey. Calverton, Maryland, USA: Macro International Inc., 2008. Available at: http://pdf.usaid.gov/pdf_docs/Pnadm638.pdf
- Finlayson K, Downe S. Why do women not use antenatal services in low- and middle-income countries? A meta-synthesis of qualitative studies. PLoS Med. 2013; 10(1):e1001373. http://dx.doi.org/10. 1371/journal.pmed.1001373 doi: 10.1371/journal.pmed.1001373 PMID: 23349622
- Puri M, Tamang J, Shah I. Suffering in silence: consequences of sexual violence within marriage among young women in Nepal. BMC Public Health. 2011; 11(1):29. Available from: http://www. biomedcentral.com/1471-2458/11/29
- 59. Davies M, Macdowall W. Health promotion theory. Maidenhead: Open University Press; 2005.
- 60. Khanal V, Zhao Y, Sauer K. Role of antenatal care and iron supplementation during pregnancy in preventing low birth weight in Nepal: comparison of national surveys 2006 and 2011. Arch Public Heal. 2014; 72(1):4.
- McPherson RA, Khadka N, Moore JM, Sharma M. Are birth-preparedness programmes effective? Results from a field trial in Siraha district, Nepal. J Health Popul Nutr. 2006; 24(4):479–88. PMID: 17591345
- Borghi J, Ensor T, Neupane BD, Tiwari S. Implicaciones financieras de la atención cualificada en el momento del parto en Nepal. Trop Med Int Health. 2006; 11(2):228–37. http://dx.doi.org/10.1111/j. 1365-3156.2005.01546.x PMID: 16451348
- Choulagai B, Onta S, Subedi N, Mehata S, Bhandari G, Poudyal A, et al. Barriers to using skilled birth attendants' services in mid- and far-western Nepal: a cross-sectional study. BMC Int Health Hum Rights. 2013; 13(1):49.
- Furuta M, Salway S. Women's position within the household as a determinant of maternal health care use in Nepal. Int Fam Plan Perspect. 2006; 32(1):17–27. PMID: 16723298
- 65. Warren C, Daly P, Toure L, Mongi P. Postnatal care. In: Lawn J and Kerber K, ed. Opportunities for Africa's newborns: Practical data, policy and programmatic support for newborn care in Africa. Cape Town, South Africa: WHO on behalf of The Partnership for Maternal Newborn and Child Health; 2006.

PLOS ONE | DOI:10.1371/journal.pone.0155144 May 23, 2016

- 66. Khanal V, Adhikari M, Karkee R, Gavidia T. Factors associated with the utilisation of postnatal care services among the mothers of Nepal: Analysis of Nepal Demographic and Health Survey 2011. BMC Womens Health. 2014; 14(1):19.
- 67. United Nations Population Fund (UNFPA). International Conference on Population and Development. Gender equality, equity and empowerment of women. Cairo, Egypt, 1995.
- National Institute for Health and Care Excellence (NICE). Behaviour change: the principles for effective interventions. NICE public health guidance 6. NICE; 2007. Available from: http://www.nice.org.uk/ guidance/ph6/resources/guidance-behaviour-change-the-principles-for-effective-interventions-pdf
- Lee RB, Abellera M, Triunfante CC, Mirandilla MEL. Utilization of maternal health care services among low income Filipino women. Asia-Pacific E-Journal Heal Soc Sci. 2012; 1(1).
- Irwin A, Valentine N, Brown C, Loewenson R, Solar O, Brown H, et al. The commission on social determinants of health: Tackling the social roots of health inequities. PLoS Med 2006; 3(6): e106. PMID: 16681414
- Witter S, Khadka S, Nath H, Tiwari S. The national free delivery policy in Nepal: early evidence of its effects on health facilities. Heal Policy Plan. 2011; 26 (suppl 2): ii84–91.
- 72. Ensor T, Clapham S, Prasai DP. What drives health policy formulation: Insights from the Nepal maternity incentive scheme? Health Pol. 2014; 90(2):247–53.
- Coast E, Jones E, Portela A, Lattof SR. Maternity care services and culture: a systematic global mapping of interventions. PLoS One. 2014; 9(9):e108130. doi: 10.1371/journal.pone.0108130 PMID: 25268940
- 74. Osrin D, Azad K, Fernandez A, Manandhar DS, Mwansambo CW, Tripathy P, et al. Ethical challenges in cluster randomized controlled trials: Experiences from public health interventions in Africa & Asia. Bull World Health Organ. 2009; 87(10):772–9. PMID: 19876544
- Bhutta ZA, Ali S, Cousens S, Ali TM, Haider BA, Rizvi A, et al. Interventions to address maternal, newborn, and child survival: What difference can integrated primary health care strategies make? Lancet. 2008; 372(9642):972–89. doi: 10.1016/S0140-6736(08)61407-5 PMID: 18790320
- 76. Datta J, Petticrew M. Challenges to evaluating complex interventions: A content analysis of published papers. BMC Public Health. 2013; 13(1):568.
- Ensor T, Green C, Quigley P, Badru AR, Kaluba D, Kureya T. Mobilizing communities to improve maternal health: Results of an intervention in rural Zambia. Bull World Health Organ. 2014; 92(1):51–9. doi: 10.2471/BLT.13.122721 PMID: 24391300
- Judd J, Frankish CJ, Moulton G. Setting standards in the evaluation of community-based health promotion programmes—a unifying approach. Heal Promot Int. 2001; 16 (4):367–80.
- 79. Duflo E. Rigorous evaluation of human behavior. Science. 2012; 336(6087):1398. Available from: http://www.sciencemag.org/content/336/6087/1398.2.short
- Bursac Z, Gauss CH, Williams DK, Hosmer DW. Purposeful selection of variables in logistic regression. Source Code Biol Med. 2008; 3:17. doi: 10.1186/1751-0473-3-17 PMID: 19087314
- Prost A, Colbourn T, Seward N, Azad K, Coomarasamy A, Copas A, et al. Women's groups practising participatory learning and action to improve maternal and newborn health in low-resource settings: a systematic review and meta-analysis. Lancet. 2013; 381(9879):1736–46. doi: 10.1016/S0140-6736 (13)60685-6 PMID: 23683640

ossMark

RESEARCH ARTICLE	Open Acc
Dirty and 40 days in the wilderness:	Cr

Sheetal Sharma^{1,2*}, Edwin van Teijlingen², Vanora Hundley², Catherine Angell² and Padam Simkhada³

cultural practices and beliefs in Nepal

Abstract

Background: Pregnancy and childbirth are socio-cultural events that carry varying meanings across different societies and cultures. These are often translated into social expectations of what a particular society expects women to do (or not to do) during pregnancy, birth and/or the postnatal period. This paper reports a study exploring beliefs around childbirth in Nepal, a low-income country with a largely Hindu population. The paper then sets these findings in the context of the wider global literature around issues such as periods where women are viewed as polluted (or dirty even) after childbirth.

Methods: A qualitative study comprising five in-depth face-to-face interviews and 14 focus group discussions with mainly women, but also men and health service providers. The qualitative findings in Nepal were compared and contrasted with the literature on practices and cultural beliefs related to the pregnancy and childbirth period across the globe and at different times in history.

Results: The themes that emerged from the analysis included: (a) cord cutting & placenta rituals; (b) rest & seclusion; (c) purification, naming & weaning ceremonies and (d) nutrition and breastfeeding. Physiological changes in mother and baby may underpin the various beliefs, ritual and practices in the postnatal period. These practices often mean women do not access postnatal health services.

Conclusions: The cultural practices, taboos and beliefs during pregnancy and around childbirth found in Nepal largely resonate with those reported across the globe. This paper stresses that local people's beliefs and practices offer both opportunities and barriers to health service providers.

Maternity care providers need to be aware of local values, beliefs and traditions to anticipate and meet the needs of women, gain their trust and work with them.

Keywords: Traditional practices, Nepal, Postnatal period, Women, Babies newborns postnatal care

Background

Pregnancy and childbirth encompass many physiological changes which social and cultural norms influence (Fig. 1). Every society has cultural practices, beliefs, superstitions or taboos concerning pregnancy and childbirth. These can translate into restrictions governed by the family, for instance what women can eat, with many cultures making distinctions between 'hot' and 'cold' foods, a distinction not necessarily related to temperature or how spicy a

perceived as not being hungry and is therefore not fed immediately after the birth [3]. Other examples include limitations on women's mobility, such as being prevented from crossing a river during pregnancy, which can restrict access to antenatal care [4]. Therefore, traditions or cultural practices may restrict what new mothers can do. The culture or traditions remain very strong even among relatively highly educated women, as will be discussed in this paper.

particular food is [1, 2]. Moreover, foods are deemed in-

appropriate (taboo) for consumption in pregnancy or during lactation; in some instances the new mother is

Correspondence: sheetale@gmail.com ¹KenBerry Solutions Ltd., Nairobi, Kenya ²Faculty of Health and Social Sciences, Bournemouth University, Bournemouth, UK Full list of author information is available at the end of the article



© 2016 The Author(s). Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creative.commons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creative.commons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.



Fig. 1 Photograph of a mural of a pregnant woman in Nepal

In countries like Nepal, socio-cultural practices around childbirth include maternal seclusion after labour and birth. Moreover, cultural beliefs in a community play a vital role in non-utilisation of postnatal care (PNC) [5, 6]. Coverage of PNC in Nepal is inadequate, especially among the poorest women and those living in remote rural areas; a recent study suggested that only 21 % of new mothers receive any PNC whether they birth in a facility or at home [6]. Both the neonatal and infant mortality have declined over the past two decades, levelling off at 33/1,000 for neonatal mortality and 46/1,000 for infant mortality [7], while the maternal mortality is estimated at 258/ 100,000 [8]. It has been suggested that simply providing a postnatal check-up on the first day before women are discharged from the health facility could prevent up to 38.7 % of maternal deaths during that period [4].

This paper explores social and cultural practices that have health implications in the childbirth and postnatal periods of rural Nepali women; and places it in the context of the global literature on such practices. The study explores the extent to which such practices, especially in low-income countries, are part of the cultural adaptation of becoming a mother and identifies why women might not access services, particularly PNC.

Methods

Qualitative approach

This study consisted of primary qualitative research on traditional practices around pregnancy, childbirth and the puerperium in rural Nepal. A qualitative approach was considered appropriate for exploring the views of women and health care providers [9].

The study setting was both PNC clinics in a community hospital and participants' homes, open fields and the village health post in two villages in rural Nepal (identified as A and B for the purpose of this paper). The new mothers went to these clinics for check-ups and vaccinations. The health service characteristics were similar between village A and B. In village A, there were two health posts and a primary health care centre nearby. In village B, the hospital was a community hospital with maternity services (Basic Emergency Obstetric Care Centre) and there were two health posts. Data collection took place in July 2012.

As qualitative methods are most appropriate for exploring complex phenomena [9], interviews and focus groups (FG) were used to explore behaviour and practice [10, 11]. Face-to-face interviews and FGs were conducted with: (a) women with a recent pregnancy and/or with a child under the age of two; (b) their mothers-in-law; (c) their husbands; and (d) healthcare workers in the area.

The sampling was purposive and potential participants were recruited through a network of health centres and women's groups. Purposive sampling was conducted in order to obtain a broad view of perspectives from a range of participants that included health workers and health service users of diverse social classes. As various ethnicities exist in Nepal (Gurung, Newari, Tamang, etc.) each with their own practices, any cultural issues raised by participants around childbirth were explored in-depth and the all-female interview team enabled postnatal practices to be probed [12]. The FGs and interviews lasted some 45 min each and were recorded (with permission), translated and transcribed [12]. First, five semi-structured interviews involving eight health workers were conducted in English by the first author (SS), as the participants spoke English, typical of higher caste/educated/health professionals in Nepal [13]. The researcher did not have a dual (clinical) role. The interviews were conducted in offices to explore the responses of the participants and gather more and deeper information by probing their answers. Secondly, fourteen FGs (each with 3-9 participants to keep the group manageable but yield good discussion) were conducted in participants' homes, open fields or the village health post with the aid of a Nepali translator. The latter was a maternal health researcher, as recommended in a study by Pitchforth and van Teijlingen [10]. The qualitative data were analysed independently by two of the researchers (SS and EvT) using a thematic approach. They then compared and contrasted the findings to minimise bias and to ensure the reliability of the data [13].

The Nepal Health Research Council granted ethical approval for the qualitative study (Reg. No. 37/2011 on 1/08/2011). Consent was obtained from each individual participant, and participants were assured that they were able to withdraw, if they so wanted.

Literature review methods

The literature review on global PNC provision and utilisation was undertaken alongside the qualitative data collection to offer suggestions for areas to explore during or

after the interviews/focus groups. We searched the following electronic databases: PubMed (or Ovid MEDLINE), EMBASE, Cochrane Library, PsycINFO, Scopus, Wed of Science, WHO (World Health Organization), CINAHL and Popline. Databases were searched from the start of the database until May 2013 for cultural issues, practices and beliefs. Inclusion criteria were: English language; qualitative and quantitative research; primary studies; all health care settings, including general practice, midwifery, outpatient, clinics, hospitals; all participants; with no time limit. Exclusion criteria were non-English language, papers that did not contain primary research or did not focus on maternity and childbirth. The search terms were 'cultural practices', 'cultural practices AND beliefs related to postpartum/natal period', and finally '40 AND days AND postnatal AND belief'. The term '40 davs' was included as a search term as it was mentioned in the interviews and the initial literature search revealed that these 40 days are considered the postpartum period [14].

The qualitative findings are presented first and the captured key study themes are then put into context through an analysis of the global literature.

Results

The five interviews included eight health workers who were aged between 18 and 37 years old. The inter-viewees included primary healthcare workers (Table 1). The 14 FG included: 9 groups of women separated into distinct groups of women who had recently given birth and mothers-in-law, aged 17-38; two groups involving men aged 35 to 61 years (some were partners of women who had recently given birth) and two FG with female community health volunteers (FCHV) and one FG with maternal and child health workers (MCHW) aged 32 to 52 (Table 2). Nearly all participants were Hindu. All of the men and two-thirds of the women were literate, at least to primary school level. All 70 participants were married, and many were initially reticent to speak of 'traditional practices' for fear of being judged, however as the interview went on they became more confident and open in their disclosures.

Table 1	Characteristics	of interviewed	health workers

Tuble T characteristics of interviewed he	and it wonters	
Health worker interviews	Age	Village
Hospital staff, Primary Health Nurses (2)	25 and 28	В
Sub-Health Post In-charge, Community Medical Assistant	40	В
Health promoters, Auxiliary Nurse Midwives (2)	30 and 40	В
Maternal & Child Health worker (Outreach clinic)	36	A
Primary Health Nurses/Maternal & Child Health worker (2)	42 and 52	A

Table 2 Characteristics of focus groups

Group Interviewed	Age	Village
Mothers with children ≤24 months	25-35	В
Mothers with children ≤24 months	21-28	В
Mothers with children ≤24 months	17-23	В
Mothers with children ≤24 months	26-48	В
Mother-in-law groups	37-55	В
Mother-in-law groups	40-62	В
Female Community Health Volunteer	26-48	В
Maternal & Child Health workers (MCH)	24 and 31	В
Extended family (Husband/Father-in-law) Focus Group Discussions	29-47	В
Female Community Health Volunteer	32-36	А
Mother-in-law groups	55-60	А
Mothers with children ≤24 months	22-28	А
Mothers with children ≤24 months	28-34	А
Extended family groups (Husband/Father-in-law) Group Interview	46-66	А

The key findings will be presented within four themes that permeated from a thematic analysis of the research, starting with: (a) cord cutting & placenta rituals; (b) rest & seclusion; (c) purification, naming & weaning ceremonies and (d) nutrition & breastfeeding. Under each heading the findings from the qualitative interviews are presented, followed by the findings of the literature analysis to put the former into context.

a) Cord cutting & placenta rituals

Participants discussed how the cord was cut, and how the placenta was expelled and then buried. No participant mentioned eating the placenta. Cord-cutting practices could involve a variety of substances and diverse types of tools to cut the cord, for example one health worker said:

(Mothers) say to us they have used sickle to cut the cord or razor blade...and then apply sometimes antiseptic, cooking oil, ghee (butter), toothpaste, ash or nothing to the umbilical cord. Health promoter/ Auxiliary Nurse Midwife, Interview

A few postnatal Tamang women mentioned the cutting of the umbilical cord at home:

The cord was cut by hasiya' (scythe)... (and then) cleaned with plain water. Tamang mother, FG

Gurung women commonly bury the placenta at the foot of a tree, while the Newari and Tamang bury it at a junction or under the road. For example, a Tamang mother referred to burying the placenta. Then placenta was buried in the road where there is a junction (laughing...), Tamang mother, FG

It is believed that the child's future health is linked to the method of disposal of the placenta:

The placenta here is buried. It should not be eaten by a crow and should not be eaten by insects. If it happens, it won't be good for child, they will be sick. Tamang mother, FG

There are "low-resource" interventions (i.e. practices) used to expel a retained placenta:

If placenta is not expelled, a piece of cloth is inserted/packed in woman's mouth so that she has nausea. That helps to expel the placenta. Tamang mother, FG

b) Rest & seclusion

Generally, Nepalese women rest after labour and delivery, with duration of rest differing between the various castes. Dates of significance in Nepal include the date when the new mother can leave her in-laws' home (after 30 days) to go to her parents' home for a period of rest that lasts from a few days up to a month. At this time her nutrition becomes a priority. One Tamang respondent describes this period:

... Then the postnatal mother is allowed to go in kitchen on the 9th day. The postnatal mother is sent to her mother's home at around 10-15 days and she can stay in mother's home as she wishes, sometimes up to a month. Tamang woman, FG

In this time a massage is given to both baby and mother, it typically involves mustard oil to relax the muscles and help the child grow with the smoothing of joints. Nepalese women have a postnatal caregiver who comes to massage the abdomen. In her maternal home, the new mother is cared for and is fed a specific diet of lentils and spices, such as cumin, believed to stimulate her breast milk production.

...from this day (nwaran or baptism) onwards, mother and baby are given mustard oil massage and stay in the sun (laughing). **Brahmin woman, FG**

Most women reported that, unlike in the past (in the last decade) where they would return to household duties soon after birth, they were now allowed to rest in their in-laws' home. Perhaps, the underlying issue is that there existed a lack of awareness of the physiology of pregnancy and childbirth; that in the postnatal period women need 'time to heal' and were therefore 'isolated'. Isolating pregnant women in the shed was also mentioned during the men's FG, typically in a cowshed. In Nepali the word for such a shed is 'chhaupadi', which is

There is not such an influence or effect in this village, but in some place there is still the practice of isolating women during menstruation/pregnancy (chhaupadi), sometimes the women have to stay in the stable (shed) also. In my thinking in this place (cough) there are not so bad practices. Everyone is doing equity behaviour (treating women equally to men). Tamang and Newari Men FG

c) Purification, naming & weaning ceremonies

also the word for this excluding practice:

The qualitative data indicate that in rural areas of Nepal, a purification ceremony is performed as birth is believed to be unclean. The naming and purification ceremony *nwaran* involves a day of cleaning the home, bathing the mother and baby, and choosing a name. The ceremony takes place anywhere between the 3rd and 12th day after birth. A priest performs the 'baptism' ceremony of *"nwaran"* at home, which includes the horoscope from the child's birth details, and the mother and child are 'purified' (from their 'past birth', in the religious sense).

In the days before the *nwaran* ceremony, the new mother cannot go out. She should not visit the (Hindu) temple and no one will touch her or take the child directly from her, as a vaginal birth is deemed 'dirty'. The mother-in-law will use her old sari or old clothes to pick up the child. In the past women would warm/cleanse their hands over the fire before holding the baby. Finally, fire is also used to burn the woven mat on which the *Bahun* women lie during and after childbirth.

Thus, birth is considered 'dirty' and the *nwaran* ceremony comprises the cleansing of the mother and house as well as naming of the child.

Delivery is considered dirty and untouchable. Usually, nwaran ... here takes place on 3rd day of delivery. Some (families) invite the lama (local Tamang priest) to pray and purify home and to name the newly born baby. Tamang woman FG

After the *nwaran* ceremony the new mother is included back into the household activities. The sleeping child is placed in a sari, as the used cloth is believed to offer protection from the 'evil eye':

The evil eye is averted when you have the baby sleep in the mother-in-law's sari; even mothers who are

teachers (or who are educated) follow these practices. Brahmin new mother and schoolteacher, Interview

Until they are 'cleansed', women have other restrictions imposed on them, including not being allowed near a deity's statue or a temple (Nepal is a strongly Hindu country) during late pregnancy:

She (mother) is not allowed to worship God from around the 6th month of pregnancy until nwaran. Bahun and Chhetri mothers FG

Furthermore, a secret *nwaran* name based on their '*rashi*' (astrological sign) is given to the baby in Nepal. This rashi is astrological and dependent on the time of day of the baby's birth.

Nwaran celebrated on 9th day for girl and 11th day for boy, which at the same time they do the purification ceremony. The name is given based on time, day and date of the birth. The Jyotishi (astrologer) is responsible to extract the name for the baby. **High caste (Brahmin)** health worker, Interview

The new mother in Nepal also has dietary restrictions.

Until the nwaran purification day, (the) a mother is not given salt, green vegetables, and is not exposed to the sun. Nwaran is done on 9th day of delivery both in Bahun and Chhetri communities. Local hospital staff. Interview

Among some Tamang there are two stages of purification *chokyaune* usually on 7th day of delivery, and *nwaran* usually on the 9th day and on that day the mother and baby are taken outside for the first time. Then they are considered 'clean' and can be 'touched' by others.

We cannot go to the kitchen, we stay (in) one corner of the ground floor; we cannot go up the stairs (laughing). Tamang mothers FG

The laughter seemed to be more about the perceived oddness of the situation rather than embarrassment.

The father is part of the cleansing ritual *nwaran*, after the child's mother is allowed outside the house.

Then only we are taken out of the home in sun. Usually the child's father does hom (special worshiping) on the third day also. Tamang mothers FG

These ceremonies vary between different ethnic groups. *Nwaran* takes place on the 3rd day among in Tamangs, the

7th day among the Bahun and Chhetri; and in Newar communities between the 9th and 12th day. It seems the higher the caste the later the ceremony. Similar to the rice 'weaning' ceremony *pasni, nwaran* takes place later for boys than girls.

Keep her (the mother) warm, give hot food, oil massage, and keep in the sun, burn lamp on 6th day and rice feeding on 5th month for daughter and on 6th month for son. **Tamang and Newar FG**

The rituals carry on until the 10th day:

On the 10th day lanterns surround the baby's mother. Nwaran is done usually on 12^{th} day for daughter and 5^{th} or 7^{th} day for a son in the Newar community. Baby and mother are given a bath, and then the priest comes home and organizes the puja (prayers). Then, (they are) considered purified. Afterwards, mother and baby (are exposed to the) sun every day. Newar hospital staff, Interview

Alongside *nwaran*, cow urine is used to purify the walls of the house. Family members of the newly born baby consume a drop of cow urine and cow dung is used with red soil to clean the house; as the cow is considered the national and holy symbol of Nepal.

The weaning *Pasni* ceremony occurs at 5 months for girls and at 6 months for boys. In this ceremony solid silver anklets called *khalis*, carved with dragons at both the ends, are given to keep the bad omens away from baby, and they are believed to help the baby's legs grow stronger. Gifts are given to the child and the mother may observe a fast. The fasting then coincides with weaning of the baby and the introduction of solid food.

d) Nutrition & breastfeeding

Food practices mentioned by women in the interviews included special attention given to nutritious food and the use of aryuvedic medicine. The new mothers are given a special diet in the postnatal period like "kwati', a special soup prepared from various beans with some meat. Several Tamang women said that "dahi-chyura" (curd and beaten rice accompanied by meat curry) is postnatal food. Mothers are also given "gudpakh", a special sweet in the form of a cake, rich in calories, made from flour, clarified butter, cashew nuts and coconut.

Mothers take gudpakh; mother becomes strong during pregnancy time. They prepare it at home. Those who visit the mother, they get it from outside. At home they prepare with milk... till baby is 2 years old we put a chain around their neck made of the umbilical cord. Newari mothers-in-law FG

Additionally, *janma ghuti* (a commercial aryuvedic medicine for digestion), *balmrita* (herbal aryuvedic medicine) and *jaiphal* (nutmeg) and herbs and spices are given to the newborn baby. Traditional healers use these as medicine.

Some of the aryuvedic foods include quati quati or ghuti aushodi (a soup of mixed pulses); in it is supari (crushed betel-nut), jaiphal, pepper and cashew nut... So that baby's heart become strong and even for strong bones. Newari mothers-in-law FG

Discussion

The qualitative research conducted in Nepal highlighted that (a) birth was perceived as 'polluting'; (b) postnatal women were perceived as being 'polluted'; and therefore isolated and (c) cleansing rituals were required for mothers after the resting/isolation period. Consent was obtained from each participant, this was particularly important within a culture where most women have to ask their husbands. Although, the husband's permission is needed, during the interviews and focus groups the discussions were 'organic', i.e., women openly spoke of their beliefs and practices.

There was a major overlap with existing literature, with the themes of the qualitative study (cord cutting and placenta, purification, naming and weaning ceremonies, rest and seclusion, nutrition and breastfeeding) echoing what researchers had found in other countries. This sets the study in a wider global perspective.

(a) Cord cutting & placenta rituals;

In Nepal, it is considered lucky to cut the umbilical cord on a coin [15]. The treatment of umbilical cords is very ritualistic, and various household tools are used to cut and tie the cord. The qualitative research suggests that in rural areas the cord is often cut with a sickle or an unsterilized knife, a practice noted in similar communities in Bangladesh [3, 16].

Poor cord hygiene is a common issue in many lowincome countries and particularly in births taking place outside of health facilities. For instance, in India the tool used is related to the trade among the caste; for example the use of a scythe by farmers, however tetanus is reported in new-borns [17, 18]. In Bangladesh, the cord is only cut after the placenta is delivered; the 'cord cutter' remains 'unholy' and cannot go for prayer for 41 days. The mother is already considered unclean due to having recently given birth, so she can cut the cord, as can a child that has not begun to pray as (s)he is also considered to be unclean [3]. In Nepal the placenta is generally buried, to protect the baby. If the placenta is retained the practice is to try to make the woman vomit to help expel it; while in other low-income countries accounts exist of massaging and sitting on the abdomen [19]. The practice in Mayan Yucatan is to treat a retained placenta with abdominal massage, applying hot water and alcohol, and then covering the woman with blankets [20]. In Malaysia, the midwife massages the mother's abdomen after the birth to facilitate the expulsion of the placenta [21].

All of these are low-cost, but not necessarily low-risk, interventions to address the problem of a retained placenta. However, the placenta's "low-resource" practices mentioned are not without risk, indeed as the literature shows sepsis remains the major cause of neonatal mortality. WHO recommendations for achieving a clean birth include a clean surface for delivery, clean hands of the birth attendant, clean cutting of the umbilical cord, clean perineum, clean cord tying, and clean cord care, since use of household tool and substances may lead to sepsis [22, 23]. It has been estimated that these clean birth practices can avert 20-30 % of newborn deaths due to sepsis and tetanus [24].

In the literature, many cultures link the baby's demeanour and future with the placenta. Placenta, the Latin word for cake, is referred to in France as a baked good; the 'other' bun in the oven [17]. Furthermore, a placental recipe from 1983 published in the magazine *Mothering* mentions the oxytocin contained within the placenta might prevent postpartum haemorrhage; placentaphagy benefits are known [17, 25]. However, we found no evidence of this in our study in Nepal.

There are also rituals associated with placental burial. For example, placentae are buried at a junction in Mexico, similar to the Newari community in Nepal [17]. One possible explanation can be identified from Indian, Semific myths; old Jewish texts tell pregnant women not to stand alone at the crossroads as they may "see the foetus taken away by evil powers". It seems a crossroad is the place where spirits dwell [26]. Perhaps burying the placenta at a crossroad diverts evil spirits away from the new baby towards the 'useless'/less important placenta. Similar to Nepal, in Lao the placenta is considered a dirty object to be buried and a fire is lit over the buried area in order to prevent spirits and animals from reaching it. If any part of the woman touches the placenta, it is believed that the lochia might dry up, causing harm to her baby and even neonatal death [27]. Lao and Burmese ethnic women still practise traditional childbirth rituals during birth preparedness, umbilical cord cutting, where they 'roast' or provide heat to mothers to stimulate healing [2, 27]; a practice also seen in traditional medicine in Laos [28].

(b)Resting and Seclusion

From the data it seems that women were housebound for a number of days after the birth and the length of this period of seclusion varied by caste or ethnic group. This is a phenomenon found across the globe, including in high-income countries in the recent past. The length of time a woman is secluded or rested varied across different countries and the principles underpinning this isolation (to heal vs. being unclean) also seem to differ greatly. After the period of seclusion there is often a ceremony to purify women to publically accept them back into daily life. The literature supports the concept of a resting - a lengthy lie-in or lying-in period, a period of seclusion, as women need to rest in order to heal, yet it may mean that they are neglected. In Greece, birth customs include women and babies resting and being isolated for 40 days after birth, a period that is still observed [29]. The 40-day period is called the lochial period, from 'lochia' the normal vaginal discharge of cell debris and blood after birth. The Bible says "40 days" for the vaginal discharge resulting from involution and can also be described as the red lochia, lasting 4-6 weeks [29]. The lochial period is a time when the woman can be cherished and pampered without feeling inadequate or shamed", noted Mead and Wolfenstein, some 60 years ago [30]. As mentioned in the interviews, in remote rural parts of Nepal women are isolated made to birth in the cowshed 'chhaupadi'; women menstruating or in labour are thought to be ritually polluted and must be kept at a distance from the family in these sheds [31]. Women in Zaire and India are also secluded in a hut [32]. For Muslims the period of postnatal seclusion traditionally lasts 40 days. The religious rituals are performed on the 40th day and these include shaving the child's head, as a vaginal birth is considered unclean. This act permits, what is considered, the growth of 'new' and 'clean' hair [33]. This 'seclusion' around the time of birth also occurs in Burma and in Turkey where it is believed that postnatal women are more vulnerable to evil forces and "the grave of women who have just given birth is open for 40 days": postnatal women are at risk and can easily die in 40 days and in that period, mother and baby are not left alone, lactating women do not go out, they and their children are not bathed [2, 34]. This is in contrast to women in Nepal; where they are left alone. Purdah (female seclusion) is observed in Bangladesh lasting 5 to 9 days and there exist dietary restrictions that last up to 6 months [3]. Similarly, among the Negev Bedouin in Israel, a 40-day postnatal period includes seclusion (homestay), followed by congratulatory visiting, the reciprocal exchange of gifts and money, and observance of a special diet [35].

A number of cultures have beliefs, taboos and behaviours relating to women and newborns in the postnatal period, a period lasting up to 40 days. Among Mayans the period lasted 20 days and Japanese mothers remained in a birth chamber for 3 weeks [17]. In Chinese the postnatal period of rest is called the 'sitting month' or 'doing the month' and lasts for 30 or 40 days. This exists, according to Chinese traditional medicine, as postpartum women are considered to be in a 'weak' state, and the practice is still observed with primiparous women [1]. Keeping mothers together with their babies is medically important but also culturally: in southwest Nepal new mothers stay with their babies continuously for 6 days [36]. Higginbottom refers to a 40-day period after the birth in which particular foods are eaten [37]. Cassidy also refers to "the upsitting" where bed linens would be changed and on the 10th day the mother was allowed to perform housework, and that hard labour ought to be avoided in the weeks after birth for the risk of uterine prolapse [17]. Burmese women also observed rest in the postpartum period [2]. The 40-day period has often been put into practice as the 'quarantine' period for women, a period of rest and purification [17]. The word "quarantine" originates from the Venetian dialect quaranta giorni, meaning 'forty days' for the length of isolation of ships for detection of plague symptoms [38]. This separation of infected people was used to prevent the spread of disease, and is recorded as far back as the Old Testament [39]. Culturally and historically, birthing women are considered 'unclean' [2, 19]. In many cultures postnatal women are believed to be dirty and weak [16, 40, 41]. Moreover, the pollution of birth is detailed; for example in Nepal, Maori (Aetoroa/New Zealand), Japan, China, Inuits in Canada, Turkey, and Bangladesh [1, 3, 16, 17, 34, 40].

Evidence of isolating practices can also be seen in western countries. In Europe in the recent past women were considered 'polluted' and dangerous to men, so new mothers were not allowed to prepare or cook food for 40 days [17]. The immediate period after childbirth is referred to historically as the 'lying-in period' in English and "Wochenbett" in German or "week bed". Browne and Browne refer to the lying-in period as 8–10 days after labour and birth; similar to the time it takes for the stump of the umbilical cord to fall off naturally [42].

Historically women in the British Isles were unclean after birth [43]. Purification as a ritual is likely to have at least some physical foundation, such as notion of infection control in modern medicine. The 40-day period presents vulnerability in mother and child which can be targeted in this time; as a frequently described postpartum problem is infection [3].

In the USA, self-help books on childbirth inform new mothers and their partners that the postnatal period lasts 6 weeks [44]. Six weeks is, of course, a different way of expressing 'the 40-day period'. The explanation such self-help books give is that "the uterus has returned to a non-pregnant size and bleeding has abated" [44]. Similarly, one of the first UK guides for new mothers recommended that women visit their doctor at 6-weeks postpartum for a range of physiological check-ups (a period of 6–8 weeks for uterus and other pelvic structures to 'heal' the puerperium), [45, 46]. Eastman and Russell also suggested that energy demanding activities such as tennis, cycling, jogging and heavy housework/lifting be postponed until the "lochia has ceased" [29]. In the 1960s, Browne and Browne claimed that red lochia lasted 24 days and only after that time should women resume household duties, start going out again or drive a car, whereas the shampooing of hair could be done as desired [42].

Caring or nourishing of women during this period is seen in the literature. In Nepal, women can have a postnatal massage to the abdomen in order to promote blood circulation and therefore healing in the first weeks post-childbirth [36]. Mayan women get "one or more massages" from their midwives 28 days post-partum [47]. In Nepal, our findings were that traditional postnatal care includes baby massage with mustard oil, massaging the mother, and an emphasis on nutrition. In higher castes (Brahmin, Chhetri, Newar and Bahun; in Nepal Tamangs are lower caste) these tasks are performed by a birth assistant, who will stay in the house for a month to wash the child's clothes and cook for the mother.

Mothers need rest and seclusion, thus there are advantages for new mothers of having a lying-in period with its associated rituals and taboos. The historian Cressy (1993) uses the term postnatal privileges to reflect this positive notion [48]. Women need rest after childbirth but should not be treated as 'infected' or 'dirty' during their seclusion period; research from China has found that the 40-day seclusion custom can adversely affects women's mental health with reports of postnatal depression occuring due to the feeling of isolation [49]. Also adversely affecting women's mental health are folk beliefs or traditional attitudes around stillbirth, which are slightly different in Nepal. This might reflect a lack of research on the impact of stillbirth on maternal mental health. Another concern is the issue of alcohol consumption as mentioned by some ethnic groups in Nepal, although this appeared to be a less common issue globally. Another issue of concern is sexual violence as the prevalence of sexual violence within marriage ranged from 12 to 50 % in Nepal [50]. Lastly, three studies have reported post-partum depression among women in Nepal to be between 4.9 and 12 % [51-53]. However, postpartum depression, it seems, is not discussed with women from low-income countries [54].

(c) Purification, naming & weaning ceremonies

The Hindu caste system and its associated behaviours have an impact on birth customs. Similar to Nepal, in India the naming ceremony takes place on the 10th or 12^{th} day after birth after which the mother is considered 'clean' and can carry out normal household chores (e.g., cooking): furthermore male visitors can visit the nursing mother. A weaning ceremony at 6 months (Annaprassana) is believed to be necessary for the baby to become more mobile; gifts here too are given to the child and the mother may observe a fast. Glass bangles worn during pregnancy are gifted to the midwives. Mothers in India also return to their parental home for 40 days after the birth. These customs are also practised by the Hindu diaspora and can lead to antenatal and postnatal nonattendance [40]. The literature also demonstrates the religious importance of ritual cleansing. Traditionally the Church of England had a thanksgiving ritual welcoming new mothers back in the church after childbirth, which was also a ritual cleansing ceremonial. The ritual referred to as 'churching' lasted well into the twentieth century [43]. Similarly, historically 'kirking' was found in the Highlands of Scotland. Associated with the Church of Scotland, it referred to the cleansing ritual to allow the women polluted in childbirth to come back into the kirk (church) [55]. The Greek Orthodox Archdiocese in the U.S.A. states that women may stay home for a period of 6 weeks after giving birth [56]. The Holy Bible in Leviticus XII: 2 notes that where the woman

"born a man child: then she shall be unclean seven days; according to the days of the separation for her infirmity shall she be unclean." ... "And she shall then continue in the blood of her purifying three and thirty days; she shall touch no hallowed thing, nor come into the sanctuary, until the days of her purifying be fulfilled" (*The Holy Bible*, Leviticus XII:4) [57].

Similarly, Jewish women were allowed back into the temple 33 days after the birth of a son and 66 days after the birth of a daughter [17]. The notion of purification in the 40 days also denotes the temptation of Christ when Jesus was in the wilderness "And when he had fasted forty days forty and forty nights..." (*The Holy Bible*, Matthew, IV: 2) [57]. We must bear in mind that in *The Bible* 40 days may refer to a long period of time rather than exactly 40 days [58].

Ceremonies frequently involve burning as part of the cleansing. In Indochina fire in the postpartum period plays a central role in ritual cleansing. In Lao PDR the confinement period of rest and "lie by the fire" is perceived as positive, the 'bad blood' bleeds out as women lie on the floor [28]. Furthermore, women in Laos refer to being 'roasted' and that advance preparation of, for example, baby clothes would lead to death of the newborn [27]. Interestingly, allopathic practitioners have now incorporated some of these traditional practices in Laos [28].

(d)Nutrition & breastfeeding

The literature discussed the role of food in the postpartum period. In Bangladesh on the first day after birth, to continue the healing of the birth passage, no food is given, and in the following days meals are nutritionally deficient consisting of rice only, as polluted women are not perceived to be hungry. Burmese women and women in Turkey who adhered to traditions of food restrictions and prescriptions during the postnatal period were traditionally not given any water to drink for 2–3 days after the birth [2, 34].

There are many references in the literature to hot and cold foods [40, 59]; and it is worth noting that hot in one country is not necessarily hot in another [60, 61]. For instance, China has the notion of 'Qi' deficiency and blood loss, 'heat' or 'cold', which may cause health problems like dizziness; thus 'cold' foods should be avoided 'hot' should be encouraged [1]. This notion of hot and cold also exists in Laos, whilst taboos include not bathing, no hair washing or teeth brushing and staying in bed between 18 h to 2 days [1, 2]. The notion of 'hot' and 'cold' is not only related to the food, but can also relate to the stage of pregnancy and birth. In Malaysia pregnancy is 'hot' [59], in Cantonese China the pregnant mother is 'cold' and the foetus 'hot' [62], whilst in Vietnam both the mother and foetus change from 'cold' in the first trimester to 'hot' in the last [41]. The notion of 'hot' and 'cold' with regards to pregnancy also exists in Laos [28]. In the literature however there was no indication of herbal aryuvedic medicine/food being harmful, suggesting a gap in existing evidence.

Dietary and breastfeeding restrictions exist; some offer women poor diets for a variety of days in Laos [28]. In Nepal, India and elsewhere in South Asia, colostrum is not given until a priest approves it, as it is considered to be pus [3, 17]. This is not unlike seventeenth century England when medical texts recommended against the feeding of colostrum [63]. In common with Nepal, the initial breastfeeding practice in Bangladesh is poor, as colostrum is not given, as it is deemed to be 'dirty milk' due to its pus-like appearance. Taboos are also evident in relation to the baby. In Bangladesh in the first 40 days breast milk is given; as is sweet water "misri pani". The latter is thought to have benefits. While, in richer households goat or cow milk is given after 40 days, yet in poorer houses misri pani often leads to a high incidence of diarrhoea [3]. The breastfeeding diet is observed for 40 days. In Cairo, infants are breastfed exclusively for the first 40 days after birth [64].

The majority of women in the qualitative interviews reported that they discarded their colostrum, which they felt was inadequate in nutritional value. Whilst this is contrary to the World Health Organization (WHO) recommendation that breastfeeding should commence in the first hour after birth [65]; it is significant because it indicates that women were actively expressing colostrum. This is vital in terms of stimulating their breast-milk supply [66]. This may be why the practice of discarding colostrum as done in South-East Asia is not as detrimental as previously thought [67]. Although, this is less than ideal in terms of the beneficial constituents of colostrum and definitely harmful if other substances are given, such as honey, butter or unclean water. Breastfeeding rates are high in Nepal (although not necessarily exclusively breastfeeding) when compared with UK rates at 6 weeks and 6 months, but not on the first day or two, as is common in South Asia. Some authors have suggested that women in South Asia generally do not breastfeed on the first or second day; but they do stimulate their breasts for the milk supply [3, 33]. Breastfeeding statistics for the UK show reasonably high initiation rates (81 %) but at 6 to 8 weeks the prevalence is down to only 47.2 % [68, 69]. Breastfeeding is associated with reduced risk of infection (colostrum contains elevated concentrations of multiple antimicrobial proteins), prevention of dehydration and hypoglycaemia in babies and reduced risk of breast and ovarian cancer in mothers and increased mother-baby bonding. Breastfeeding has short and long-term health benefits for both baby and mother [70]. Potential long-term health benefits in children include reduced blood pressure, cholesterol concentrations, and obesity [67, 70].

The literature illustrates that alcohol plays an important role in both birth and the postpartum period. In Nepal the Tamang mothers drink *jad* during their pregnancy and post-pregnancy, similar to the 'god-sips'. The term 'god-sips' is thought to have arisen because when a woman went into labour, her 'gossips' were sent forth to gather for merriment and to partake in a drink at the labour [71]. Indeed the drinking midwife is mentioned in Shakespeare's Twelfth Night: "like aqua vitae with a midwife" [72]. In the Tamang community *jad*, an alcohol, is taken during pregnancy and post-pregnancy to celebrate the birth, and Gurung women may drink it to put the child to sleep during breastfeeding as alcohol will certainly pass into the breast-milk [68].

Alcohol and pregnancy are linked culturally, for instance in Africa rum was given to the Akan and Igbo child. Furthermore, the birth was celebrated with alcohol and at the naming ceremony [73]. Drugs passing into milk as cathartics were identified by Greek physicians and Gurung women use it to put the child to sleep [29]. Alcohol is also used in the protection of children; Malaysians bathe children in stout as they believe it protects babies and to help new-borns suffering from jaundice [32, 74].

This article has discussed that there are logical reasons for practices that are linked to the physiology of pregnancy, birth and the postpartum period. The umbilical cord if left alone typically falls off after six to ten days. whilst the lochia heal in about 40 days/6 weeks [75]. The latter ties in with the International Classification of Diseases' definition of maternal mortality [76]. However, the origin of the 42 days limit is historical, i.e., in the old Anglican Church and the Jewish faith where purified women resumed attending prayers 40 days after childbirth, rather than medical. Clinically, the relevance lies in the first menstrual bleeding in non-lactating women occurring 6 to 8 weeks after parturition. The literature finds that the 42 day limit was not based on a study of the timing of maternally related deaths [14]. Some practices, however, clearly put women at risk; isolation may mean rest but if the woman is alone and suffers a postpartum haemorrhage, this may result in a preventable maternal death. Also unclean tools such as a scythe used to cut the cord may lead to infection, or fasting can lead to malnutrition, and most commonly the discarding colostrum may reduce the protective effects of early breastfeeding.

It is easy to forget that childbirth is a hazard for mother and child in many low-income countries; some traditional practices reduce and others increase the chance of dying. This article highlights that cultural practices exist universally between days 3–10 and 40, and that many of these can be linked to physiology. The timing of these important events means cultural influences play a role in postnatal practices [77]. In society, rituals develop over time to deal with the physiological and social aspects of birth and are internally consistent.

The study and literature within this article have shown that reproductive health is shaped by culture and women's position may be influenced by social and cultural aspects rather than biological factors: The role and place of women in society is 'lowered' in a patriarchal society where historical social norms are maintained. Several studies refer to cultural sensitivity when dealing with women, focusing attention on improving the maternity services rather than on women and their cultural differences [12, 78, 79]. As social cultural practices are passed down from senior females to younger generations, postpartum home visits may play an important role in helping women to change behaviours [1]. Nepalese maternity care should focus on the rural population to be more sustainable and maternity nurses/midwives can use health promotion interactions during home visits [37].

Strengths and limitations of the study

This is one of the first studies of its kind in Nepal. Women were interviewed individually, which allowed them to

speak about the issues anonymously. However, due to accessibility, time and resource constraints men had to be interviewed in groups. This is a limitation as male participants in the focus group stated that the topic was women's 'business' and they felt that they could not comment in any depth. A minority of the interviews were conducted in English, which may have influenced the way Nepali professionals expressed themselves. Most interviews and all focus groups relied on a translator, which also may have affected the data. The translator had a health background and was trained prior to the research and the interviewer spoke Hindi and a few words of Nepali which helped ensure the quality of the data. A limitation to the search strategy is that it did not include 'stillbirth' and 'nutrition'.

Policy relevance

A gap in knowledge surrounding social cultural conditions may explain the failure of some health policies and programmes to address such issues [80]. Therefore, it is important that, even in the postnatal period, childbearing women feel they can discuss non-health worries that relate to superstition, myths and taboos. Culture and traditions are fraught with ambiguity, especially as many health programmes aim to integrate 'evolving modernities' with the influence of globalisation [37]. Furthermore, social cultural practices can affect women's health status, and therefore a westernised model of care is not advocated, rather informed decisions should be taken regarding locally appropriate illness prevention. In addition, health policymakers and international development advisers need to take social and cultural conditions into consideration to formulate evidence-based policies to reduce morbidity and mortality in mothers and their babies, and reduce gender inequalities [80-82].

Implication for practice

Understanding childbirth values and beliefs of specific cultural groups can promote culturally appropriate evidence-based care. Cultural postnatal practices can be harmful or ineffective, but changing deep-rooted practices, often with religious origins, is challenging even among educated women. Understanding the social cultural environments should be part of health providers training to change these behaviours or incorporate them into the care. The clearly detrimental behaviour will require culturally sensitive re-educative programmes that create new understandings in both practitioners as well as women of childbearing age and their family and local communities. Some of the interventions should address the physiology of childbirth, which is often poorly understood by rural women and/or those with low education levels. If local people know how their traditional behaviour fits with the physiology of childbirth it might be slightly easier to change some of the undesirable or

risky behaviours. This understanding is also of importance when designing culturally appropriate interventions: such as birth kits in low-income countries [83].

Finally, practitioners in high-income countries can learn from those in low-income countries to help provide culturally appropriate care that is accessible. This will be especially beneficial to high-income countries with large ethnic minorities to help avoid discriminatory policy and practices.

Conclusion

Social cultural practices may prevent women from accessing postnatal care. Although there are physiological explanations that underpin some of the beliefs and practices around, for example, a new mother needing rest for a 40day period, it is important to stress that not all practices necessarily have a physiological origin. The role and place of women in society probably have a much greater negative effect on postnatal women. Some of these influences are negative as they can prevent women accessing postnatal care in low-income countries and the more positive consequences of having a 40-day period come from a 'socially enforced' rest and seclusion.

Abbreviations

FCHV, female community health volunteers; FG, focus groups; I, interviews; MCHW, maternal and child health workers; PNC, postnatal care; WHO, World Health Organization

Acknowledgments

We thank all research participants, also the staff of Green Tara Nepal, the health posts and the Female Community Health Volunteers. This study had financial support from Green Tara Trust, UK and Bournemouth University, UK We are grateful to Sofia Castro Lopes, Lindsay Menard-Freeman and Dr. Andrea Nove for their advice.

Authors' contributions

EVT and SS, VH, CA and PS have contributed to design the original study including analysis plan and initial interpretation. SS collected the data. All assisted in the interpretation of data, commented on drafts and approved the final submission.

Competing interests

The author(s) declare that they have no competing interests.

Author details

¹/KenBerry Solutions Ltd., Nairobi, Kenya. ²Faculty of Health and Social Sciences, Bournemouth University, Bournemouth, UK. ³Centre for Public Health, Liverpool John Moores University, Liverpool, UK.

Received: 9 March 2015 Accepted: 24 June 2016 Published online: 05 July 2016

- References
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, Liu L, Chen B, Ding Q. Postpartum practices of puerperal
 1. Liu N, Mao L, Sun X, L Public Health: 2006;6:274.
- Sein KK. Beliefs and practices surrounding postpartum period among Myanmar women. Midwifery. 2013;29:1257–63. Darmstadt GL, Syed U, Patel Z, Kabir N. Review of domiciliary newborn-care
- 3 practices in Bangladesh, J Health Pop Nutr. 2006;24(4):380-93
- Pradhan A, Suvedi B, Sharma S, Puri M, Poudel P, Chitrakar S, KC N, Hulton L. Nepal Maternal Morbidity and Mortality Study 2008/2009. Kathmandu:

Family Health Division, Department of Health Services, Ministry of Health and Population. 2010. Mesko N, Osrin D, Tamang S, Shrestha BP, Manandhar DS, Manandhar M,

- Standing H, Costello AM de L. Care for perinatal illness in rural Nepal: a descriptive study with cross-sectional and qualitative components. BMC Int Health Hum Rights: 20033-3. Dhakal S, Chapman GN, Simkhada PP, van Teijlingen ER, Stephens J, Raja AE.
- Utilisation of postnatal care among rural women in Nepal. BMC Pregnancy Childbirth. 2007;7:19.
- Paudel D, Shrestha IB, Siebeck M, Rehfuess EA. Neonatal health in Nepa
- Fouce: O, Janesona ID, Janesona ID, Jacobs KM, Remuess EA, Neonatai Realth III Nepat: analysis of absolute and relative inequalities and impact of current efforts to reduce neonatal mortality. BMC Public Health. 2013;13:1239.
 WHO, UNICEF, UNIFPA, Bank TW, UN Population Division. Trends in Maternal Mortality: 1990 to 2015. 2015.
 Tashakkori A, Teddile C, Mixed Methodology. Combining Qualitative and Competence Anonecomposition. (In Constructions) 2000. 9.
- Quantitative Approaches, vol. 46. London: Sage; 1998. Pitchforth E, van Teijlingen E. International public health research involving interpreters: a case study from Bangladesh. BMC Public Health. 2005;5:71. 10.
- 11.
- van Teijlingen E, Pitchforth E. Focus group research in family planning and reproductive health care. J Fam Plann Reprod Health Care. 2006;32:30–2. Thwala SBP, Holroyd E, Jones LK, Heatth belief dualism in the postnatal practices of rural Swazi women: An ethnographic account. Women Birth. 2012;25. 12.
- 13.
- Forrest Keenam K, van Teijlingen E, Pitchforth E. The analysis of qualitative research data in family planning and reproductive health care. J Fam Plann Reprod Health Care. 2005;31:40–3. Høj L, Da Silva D, Hedegaard K, Sandström A, Aaby P. Maternal mortality: 14.
- Only 42 days? BJOG. 2003;110;995–1000. PATH. Maternal and Child Health: Simple Kits Save Lives. 2013. Goodburn EA, Gazi R, Chowdhury M. Bellefs and practices regarding delivery and postpartum maternal morbidity in rural Bangladesh. Stud Fam Plann. 16. 1995:26:22-32
- Cassidy. The Surprising History of How We Are Born. New York: Grove/Atlantic, Incorporated; 2007.
- Hassan H, Jokhio AH, Winter H, MacArthur C. Safe delivery and newborn 18. care practices in Sindh, Pakistan: A community-based investigation of mothers and health workers. Midwifery. 2012;28:406–11. Jeffery R, Jeffery PM, Traditional Birth Attendants in Rural North India: The
- 19. Social organisation of childbearing. In: Midwifey and Medicalization of Childbirth: Comparative Perspectives. New York: Nova; 1999. p. 269. Anderson BA, Anderson EN, Franklin T, Dzib-Xihum de Cen A. Pathways
- of decision making among Yucatan Mayan traditional birth attendants. J Midwifery Womens Health. 2004;49:312–9.
- 21.
- J MidWrety Womens Health. 2004;99:312-9; Laderman C. A baby is born in Merchang. In: van Teijlingen ER, Lowis R, McCaffery P, Porter M, editors. Midwifery and Medicalization of Childbirth: Comparative Perspectives. New York Nova; 1999. p. 235–44. Karumbi J, Mulaku M, Aluvaala J, English M, Opiyo N. Topical umbilical cord care for prevention of infection and neonatal mortality. Pediatr Infect Dis J. 2013;32:78–83. 22.
- Blencowe H, Cousens S, Mullany LC, Lee ACC, Kerber K, Wall S, Darmstadt 23 Gic, Lawn JE, Clean birth and postnatal care practices to reduce neonatal deaths from sepsis and tetanus: a systematic review and Delphi estimation of mortality effect. BMC Public Health. 2011;11 Suppl 3511.
 Friberg IK, Bhutta ZA, Darmstadt GL, Bang A, Cousens S, Baqui AH, Kumar V,
- 24 Walker N, Lawn JE. Comparing modelled predictions of neonatal mortality impacts using list with observed results of community-based intervention trials in South Asia. Int J Epidemiol. 2010;39.
- Donnison J. Midwives and Medical Men: A History of Inter-Professional 25.
- Rivalites and Women's Rights. New York: Schocken Books; 1977. Chawla J. Understanding "Narak": Rethinking Pollution. An Interpretation of Data from Dais in North India. In "Exploring Pollution and Safety—The Ditry Side of Women's Health". Sheffield: Sheffield University's School of
- Nursing & Midwifery; 2004. Sychareun V, Phengsavanh A, Hansana V, Somphet V, Menorah S, Cultural Beliefs and Traditional Rituals about Child Birth Practices in Lao PDR. Kuala Lumpur: The 27 Asian-Pacific Resource & Research Centre for Women (ARROW); 2009.
- Davia S. Beliefs, Taboos, Practices and Behaviors around Birth in Lao. PDR. Presented to the World Health Organization. 2003. Eastman NJ, Russell KP. Eastman's Expectant Motherhood. 6th ed. New York: 28. 29.
- Little: 1977. 30.
- Mead M, Wolfenstein M. Childhood in Contemporary Cultures. Chicago University of Chicago Press; 1955

Sharma et al. BMC Pregnancy and Childbirth (2016) 16:147

Page 12 of 12

- 31 Ghimire LV Unclean & unseen Student Br Med 1 2005/13/177-220
- Ginning LY, Orlectan & different student of mild J. 2003 (3) 77–220.
 Dunham C. Mamatoto: A Celebration of Birth. New York: Penguin; 1993.
 Bandyopadhyay M. Impact of ritual pollution on lactation and breastfeeding
- practices in rural West Bengal India. Int Breastfeed J. 2009;4:2. 34
- Geckil E, Şahin T, Ege E. Traditional postpartum practices of women and infants and the factors influencing such practices in South Eastern Turkey. Midwifery. 2009;25:62-71. 35.
- Hundric L, Beckerleg S, Kassem F, Abu Jafar AM, Belmaker I, Abu Saad K, Shoham-Yardi I. Women's health custom made: building on the 40 days postpartum for Arab women. Health Care Women Int. 2000;21:529–42. Wullany LC, Darmstadt GL, Khatry SK, Telsch JM. Traditional massage of
- 36. newborns in Nepal: Implications for trials of improved practice. J Trop Pediatr. 2005;51:82–6. Higginbottom GM, Vallianatos H, Forgeron J, Gibbons D, Malhi R, Marnede F
- 37.
- Food choices and practices during pregnancy of immigrant and Aboriginal women in Canada: a study protocol. BMC Pregnancy Childbirth. 2011;11:100. Sehdev PS. The origin of quarantine. Clin Infect Dis. 2002;35:1071–2. Eccles R. Understanding the symptoms of the common cold and influenza. Lancet Infect Dis. 2005;5:718–25. 39.
- Gatrad AR, Ray M, Sheikh A, Hindu birth customs, Arch Dis Child, 2004;89:1094-7. 40 Manderson L, Mathews M. Vietnamese attitudes towards maternal and infant health. Med J Aust. 1981;1:59–72. 41.
- 42. Browne FJ, Browne JCMC. Antenatal and Postnatal Care. UK.: J. & A Churchill 1960
- Hogan S. Breasts & the Beestings: Rethinking Breast-Feeding Practices, Maternity 43. Rituals, & Maternal Attachment in Britain & Ireland. J Int Womens Stud.
- 2008:10:21-30. Simonds W. Watching the clock: Keeping time during pregnancy, birth, and postpartum experiences. Soc Sci Med. 2002;55:559–70. 44
- 45. Phillips A, Lean N, Jacobs B, Jacobs B. Your Body, Your Baby, Your Life. London: Sphere: 1985.
- Leonard M. Your Baby's Early Days. Risborough: Shire Publications, Limited; 1977. Jordan B. Buscando La Forma: An Ethnography of Contemporary Maya 47.
- Childbirth, 4th ed. Prospect height: Waveland Press; 1993. 48.
- Cressy D, Purification, Thanksylving and the Churching of Women in Post-Reformation England. Past Present. 1993;141:106–46. Wong J, Fisher J. The role of traditional confinement practices in determining 49. postpartum depression in women in Chinese cultures: a systematic review of
- 50.
- posporter ocposition of the moment of the control o 51. Regmi S. Sligl W. Carter D. Grut W. Seear M. A controlled study of postpartum
- negrin 5, and 7, and 0, she is a second state of the among postnatal women - A population based guestionnaire study upplemented by sleep diary and actigraphy. Arch Womens Meni lealth. 2011;14:S17–8.
- Sabba NR. Post Partum Depression among Rajbansi Women in Nepal. 53 Res A Res J Cult Soc. 2013:1.
- Griffith LB. Practitioners, postnatal depression, and translation: An investigation into the representation of Bangladeshi mothers in the East End. Anthropol 54. Med. 2009;16:267-78.
- 55. Brvers HM, Midwifery & Maternity Services in the Gaidhealtachd and the North of Scotland 1912-2005, Aberdeen: University of Aberdeen; 2010. PhD thesis. Pregnancy, Birth, Naming and 40 day Churching. [http://www.stgeorgegoc
- 56. org/baptism.htm]
- The Holy Bilds, Containing the Old and New Testaments: Translated out of the Original Tongues; and with the Former Translations Diligently Compared and Revised, by His Majesty's Special Command. Glasgow: William Collins, 57 5ons and Co.: 1870.
- Warfield B. Inspiration and Authority of the Bible. University of Virginia: P & R Publishing, Presbyterian and Reformed Pub. Co., University of Virginia; 1970. Manderson L. Hot-cold food and medical theories: overview and introduction. 58
- 59. Soc Sci Med. 1987:25:329-30.
- Molony CH. Systematic valence coding of Mexican "hot"—"cold" food. Ecol Food Nutr. 1975;4:67–74. Logan MH. Humoral Medicine in Guatemala and Peasant Acceptance of 60.
- 61.
- Modern Medicine, Hum Organ 1973;22:385–96. Topley M. Chinese and Western medicine in Hongkong: some social and cultural determinants of variation, interaction and change. In: Medicine in 62.

- Chinese Cultures: Comparative Studies of Health Care in Chinese and Other Societies: Papers and Discussions from a Conference Held in Settle, USA Bethesda; 1974. Washington: Dept. of Health, Education, and Welfare, Public Health Service, National Institutes of Health, 1976. p. 241-271.
- 63.
- Wickes IG. A history of infant feeding. I. Primitive peoples; ancient works; Renaissance writers. Arch Dis Child. 1953;28:151–8. Pak-Gorstein S, Haq A, Graham EA. Cultural influences on infant feeding practices. Pediatr Rev. 2009;30:e11–21. 64.
- World Health Organization (WHO). World Health Organization, Nutrition: Exclusive Breastfeeding. 2014. Riordan J, Hoover K. Perinatal and intrapartum care. In: Riordan J, editor. 65.
- Breastfeeding & Human Lactation, 3rd ed. Sudbury: Jones and Bartlett
- Dublishers; 2005, p. 185–216.
 Odent PLG, Early infant feeding and neonatal survival in Nepal: breastfeeding, colostrum and discarding of the first milk. Doctral thesis, UCL (University College London, 2011, Available on http://discovery.uclac.uk/1310430/. 67
- 68.
- McAndrew F, Thompson J, Fellows L, Large A, Speed M, Renfrew MJ. Infant Feeding Survey 2010. London: Health and Social Care Information Centre; 2012. Department of Health. Breastleeding Stats. Indicators on Breastfeeding: Quarter 4 2012/13. London: Department of Health; 2013. 69.
- Quarter 4 2012/13. London: Department of Health; 2013. Hoddinott P. Tappin D, Winght C. Breast feeding, BMJ. 2008;336:881–7. Donnison J. The Decline of the Midwife. In: van Teijlingen E, Lowis G, McCaffery PPM, editors. Midwifery and Medicalization of Childbirth: Comparative Perspectives. New York: Nova; 2004. p. 101–15. Shakespeare. London: Chancellor Press; 1982. Smith FH. Spints And Spirituality: Alcohol In Caribbean Slave Societies. 2012. [http://www.kislahfoundation.org/prize/200102.html]. 71
- 72.
- 73.
- 74 New Born Babies and Beers. [http://look-amazing.blogspot.com.es/2012/07/
- In Byrom S, Edwards G, Bick D, editors. Essential Midwifery Practice: Postnatal Byrom S, Edwards G, Bick D, editors. Essential Midwifery Practice: Postnatal
- Int bytom 5, cowards 5, bick J, eators, essential MidWiney Plattice Positiata Gare, Chichester Wiey-Blackweil (2010, p. 1–26. International Classification of Diseases, International Classification of Diseases, Geneva: World Health Organization; 2004. Lozoff B, Jordan B, Malone S, Childbirth in Cross-Cultural Perspective. Marriage Fam Rev. 1988;12:35–60. 76
- 77.
- 78
- 79. 80.
- Mariage Fam Rev. 1988;12:35–60. Rocherson Y. The Asian Mother and Baby Campaign: The Construction of Ethnic Minorities' Health Needs. Crit Soc Policy. 1988;8:4–23. Stein HF. The Culture of the Patient as a Red Herring in Clinical Decision Making: A Case Study. Med Anthropol Q. 1989;172–5. Smith SL, Neupane S. Factors in health initiative success Learning from Nepal's newborn survival initiative. Soc Sci Med. 2011;72:268–75. World Health Organization (WHO). World Health Organization, Global Health Osteratory Data Repository (GHO). Antrality: Neonatal Deaths and Mortality Bata by Reviol. 1907;012. 81. Rate by Region, 1990-2012, 2013.
- Evans EC. A review of cultural influence on maternal mortality in the developing world. Midwifery. 2013;29:490–6. Hundley VA, Avan BJ, Braunholtz D, Fitzmaurice AE, Graham WJ. Lessons 82.
- 83. regarding the use of birth kits in low resource countries. Midwifery. 2011;27.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit

BioMed Central

Bibliography

- AbouZahr, C., 2003. Global burden of maternal death and disability. *British Medical Bulletin*, 67 (1), 1–11. Available from: http://bmb.oxfordjournals.org/content/67/1/1.full [Accessed 22 December 2015].
- Abou-Zahr, C.L. and Wardlaw, T.M., 2003. Antenatal care in developing countries: promises, achievements and missed opportunities: an analysis of trends, levels and differentials, 1990-2001 (No. 618.2: 316). Geneva: World Health Organization. Available from: http://www.who.int/maternal_child_adolescent/documents/9241590947/en/ [Accessed 13 December 2016].
- Acharya, D. R., Bell, J. S., Simkhada, P., van Teijlingen, E. R., and Regmi, P. R., 2010. Women's autonomy in household decision-making: a demographic study in Nepal. *Reproductive Health*, 7 (1),15. Available from: http://www.reproductive-health-journal.com/content/7/1/15 [Accessed 22 December 2015].
- Ahmed, S., Creanga, A. A., Gillespie, D. G., and Tsui, A. O., 2010. Economic status, education and empowerment: implications for maternal health service utilization in developing countries. *PLoS ONE*, 5 (6), e11190. Available from: http://www.plosone.org/article/info:doi/10.1371/journal.pone.0011190.
- Akhund, S. and Yousafzai, A. K., 2011. How successful are women's groups in health promotion and disease prevention? A synthesis of the literature and recommendations for developing countries. *Eastern Mediterranean Health Journal*, 17 (5), 446–52.
- Al-Busaidi, Z. Q., 2008. Qualitative research and its uses in health care. *Sultan Qaboos University Medical Journal*, 8 (1), 11–19.
- Alderman, H., Ndiaye, B., Linnemayr, S., Ka, A., Rokx, C., Dieng, K., and Mulder-Sibanda, M., 2009. Effectiveness of a community-based intervention to improve nutrition in young children in Senegal: a difference in difference analysis. *Public Health Nutrition*, 12 (5), 667–73.
- Altman, D. G. and Bland, J. M., 1998. Generalisation and extrapolation. *BMJ*, 317 (7155), 409–410. Available from: http://www.bmj.com/content/317/7155/409.abstract [Accessed 1 April 2014].
- Ameh, C. and van den Broek, N., 2014. Strategies to improve global maternal and neonatal health. *In: Magowan, Brain A., Owen, Philip and Thomson, A., ed. Clinical Obstetrics & Gynecology*. Edinburgh, UK: Elsevier Saunders, 229–238.
 Anon., 2012. With transparency comes trust. *Nature*, 485 (7397), 147.
- Angus, K., Cairns, G., Purves, R., Bryce, S., MacDonald, L. and Gordon, R., 2013. Systematic literature review to examine the evidence for the effectiveness of interventions that use theories and models of behaviour change: towards the prevention and control of communicable diseases. Stockholm: European Centre for Disease Prevention and Control. Available from: http://ecdc.europa.eu/en/publications/Publications/health-communicationbehaviour-change-literature-review.pdf [Accessed 24 January 2017].

- Antonovsky, A., 1996. The salutogenic model as a theory to guide health promotion. *Health Promotion International*, 11 (1), 11–18. Available from: http://heapro.oxfordjournals.org/content/11/1/11.abstract [Accessed 29 September 2015].
- Arnold, R., van Teijlingen, E., Ryan, K. and Holloway, I., 2015. Understanding Afghan healthcare providers: a qualitative study of the culture of care in a Kabul maternity hospital. *BJOG: An International Journal of Obstetrics & Gynaecology*, 122(2), 260-267.

Asefa, A., Teshome, W., Mesele, T., and Letamo, Y., 2013. Use of institutional delivery services in the Southern Nations, Nationalities, and People's Region, Ethiopia: a cross-sectional comparative mixed methods study. *The Lancet*, 382, 9. Available from: http://www.sciencedirect.com/science/article/pii/S0140673613621704 [Accessed 1 December 2014].

- Asian Development Bank (ADB), 2001. *Handbook on stakeholder consultation and participation in ADB operations*. Abidjan, Côte d'Ivoire: Environment and Sustainable Development Unit. Available from: http://www.afdb.org/fileadmin/uploads/afdb/Documents/Policy-Documents/Handbook%20on%20Stakeholder%20Consultaion.pdf [Accessed 15 December 2015].
- Asthana, S., 1996. Women's health and women's empowerment: a locality perspective. *Health & Place*, 2 (1), 1–13. Available from: http://www.sciencedirect.com/science/article/pii/1353829295000232.[Accessed 15 February 2017].
- Azad, K., Barnett, S., Banerjee, B., Shaha, S., Khan, K., Rego, A. R., Barua, S., Flatman, D., Pagel, C., Prost, A., Ellis, M., and Costello, A., 2015. Effect of scaling up women's groups on birth outcomes in three rural districts in Bangladesh: a cluster-randomised controlled trial. *The Lancet*, 375 (9721), 1193–1202.
- Badshah, S., Mason, L., McKelvie, K., Payne, R., and Lisboa, P. J., 2008. Risk factors for low birthweight in the public-hospitals at Peshawar, NWFP-Pakistan. *BMC Public Health*, 8, 197.
- Baez, J. E., 2007. *Guidelines for designing evaluations of community-based nutrition promotion programs*. Washington D.C: World Bank. Available from: http://siteresources.worldbank.org/EXTLACREGTOPNUT/Resources/Guideline sForDesigningEvaluations.pdf [Accessed 2 September 2015].
- Baker, E., 1999. Principles of practice for academic/practice/community research partnerships. *American Journal of Preventive Medicine*, 16 (3), S1, 86–93.
- Bamberger, M., Rao, V., and Woolcock, M., 2010. Using Mixed Methods in Monitoring and Evaluation: Experiences from International Development. World Bank Development Research Group, 30. Available from: wwwwds.worldbank.org/servlet/WDSContentServer/WDSP/IB/2010/03/23/0001583 49_20100323100628/Rendered/PDF/WPS5245.pdf [Accessed 2 September 2015].

- Banerjee, A. V., Cole, S., Duflo, E., and Linden, L., 2007. Remedying Education: Evidence from Two Randomized Experiments in India. *The Quarterly Journal* of *Economics*, 122 (3), 1235–1264. Available from: http://qje.oxfordjournals.org/content/122/3/1235.abstract [Accessed 12 November 2015].
- Banerjee, A. and Duflo, E., 2012. *Poor economics: A radical rethinking of the way to fight global poverty*. New York City: PublicAffairs. Available from: http://economics.mit.edu/files/5993 [Accessed on 12 March 2015].
- Bang, A. T., Reddy, H. M., Deshmukh, M. D., Baitule, S. B., and Bang, R. A., 2005. Neonatal and infant mortality in the ten years (1993 to 2003) of the gadchiroli field trial: effect of home-based neonatal care. *Journal of Perinatology*, 25 (S1), S92–S107.
- Baqui, A. H., Rosecrans, A. M., Williams, E. K., Agrawal, P. K., Ahmed, S.,
 Darmstadt, G. L., Kumar, V., Kiran, U., Panwar, D., Ahuja, R. C., Srivastava, V. K., Black, R. E., and Santosham, M., 2008. NGO facilitation of a government community-based maternal and neonatal health programme in rural India: Improvements in equity. *Health Policy and Planning*, 23, 234–243.
- Baqui, A. H., El-Arifeen, S., Darmstadt, G. L., Ahmed, S., Williams, E. K., Seraji, H. R., Mannan, I., Rahman, S. M., Shah, R., Saha, S. K., Syed, U., Winch, P. J., Lefevre, A., Santosham, M., and Black, R. E., 2015. Effect of community-based newborn-care intervention package implemented through two service-delivery strategies in Sylhet district, Bangladesh: a cluster-randomised controlled trial. *The Lancet*, 371 (9628), 1936–1944.
- Baral Y.R., Lyons, K., Skinner, J., van Teijlingen, E., 2010. Determinants of skilled birth attendants for delivery in Nepal. *Kathmandu University Medical Journal*, 8 (31), 325–32.
- Barnett, K., 2012. Best Practices for Community Health Needs Assessment and Implementation Strategy Development: A review of scientific methods, current practices, and future potential. In: public forum convened by centers for disease control and prevention. Atlanta, GA, USA: Public Health Institute.
- Basevi, V. and Lavender, T., 2014. Routine perineal shaving on admission in labour. *The Cochrane Database of Systematic Reviews*, 11, CD001236. Available from: http://onlinelibrary.wiley.com/doi/10.1002/14651858.CD001236/full [Accessed 16 December 2015].
- Baskerville, N. B., Hogg, W., and Lemelin, J., 2001. Process evaluation of a tailored multifaceted approach to changing family physician practice patterns improving preventive care. *The Journal of Family Practice*, 50 (3), W242–9.
- Bassani, D., Surkan, P., and Olinto, M., 2009. Inadequate use of prenatal services among Brazilian women: the role of maternal characteristics. *International Perspectives on Sexual and Reproductive Health*, 35 (1), 15-20. Available from: http://www.jstor.org/stable/25472411 [Accessed 21 December 2015].
- Baum, F., MacDougall, C., and Smith, D., 2006. Participatory action research. Journal of Epidemiology and Community Health, 60 (10), 854–857.

Bayer, A., 2001. Executive summary: maternal mortality and morbidity. *Population*

Resource Center, 2001. Available from: http://www.prcdc.org/files/Maternal_Mortality.pdf [Cited 9 September 2015].

- Beattie, A., 1995. Evaluation in community development for health: an opportunity for dialogue. *Health Education Journal*, 54 (4), 465–472.
- Berg, M., 2005. A midwifery model of care for childbearing women at high risk: genuine caring in caring for the genuine. *The Journal of Perinatal Education*, 14 (1), 9–21.
- Bertrand, M., Duflo, E., and Mullainathan, S., 2004. How much should we trust differences-in-differences estimates? *The Quarterly Journal of Economics*, 119 (1), 249–275. Available from: http://qje.oxfordjournals.org/content/119/1/249.abstract [Accessed 21 September 2015].
- Bhandari, N., Bahl, R., Mazumdar, S., Martines, J., Black, R. E., and Bhan, M. K., 2003. Effect of community-based promotion of exclusive breastfeeding on diarrhoeal illness and growth: a cluster randomised controlled trial. *The Lancet*, 361 (9367), 1418–23.
- Bhattacharya, J., Tu, P., and Hyde, T., 2013. *Health Economics.* Basingstoke, UK: Palgrave Macmillan.
- Bhuiya, A. and Chowdhury, M., 2002. Beneficial effects of a woman-focused development programme on child survival: evidence from rural Bangladesh. Social Science & Medicine, 55 (9), 1553–1560. Available from: http://www.sciencedirect.com/science/article/pii/S0277953601002878 [Accessed 1 September 2015].
- Bhutta, Z. A., Darmstadt, G. L., Hasan, B. S., and Haws, R. A., 2005. Community-based interventions for improving perinatal and neonatal health outcomes in developing countries: a review of the evidence. *Pediatrics*, 115 (Supplement 2), 519–617. Available from: http://pediatrics.aappublications.org/content/115/Supplement_2/519.abstract [Accessed 2 September 2015].
- Bhutta, Z. A., Soofi, S., Cousens, S., Mohammad, S., Memon, Z. A., Ali, I., Feroze, A., Raza, F., Khan, A., Wall, S., and Martines, J., 2011. Improvement of perinatal and newborn care in rural Pakistan through community-based strategies: a cluster-randomised effectiveness trial. *The Lancet*, 377 (9763), 403–412.
- Bishai, D., Niessen, L.W. and Shrestha, M., 2002. Local governance and community financing of primary care: evidence from Nepal. *Health Policy and Planning*, *17*(2), 202-206. Available from: https://oup.silverchaircdn.com/oup/backfile/Content_public/Journal/heapol/17/2/10.1093_heapol_17. 2.202/1/170202.pdf?Expires=1487003986&Signature=VeEIrOqYQjYMCC4caN QfEajnyHfd6DcTYO6E6r7fnJVWv9EfDhmlfITzVzqOZ7JTOeeVpHohwc5jJ8gYYeORRB69gxZ2hI12CoHYOUF1embo~Oa3cnhfsRpYCaXVqG7QaMo ch5IYwBExCISDsJGEPeEBXOTrcbvH9unwctJrF3ONrEIBNbKbRosNLAhMAc DH37X55IW9iTd7A9hgx7wJscZeAqiAMZD6IFxZrggYu2~PBLWRGclcFT5mXH hP~DQxEt0-sKoTqeSmJtf~-v9WzyvtLQDwZB6o5hPp-5yx2Q~UWaTZoW0BYSLJPnANfLFjoKgH8GyhKofCYRtAFfRw__&Key-Pair-Id=APKAIUCZBIA4LVPAVW3Q [Accessed on 20 January 2017].

- Bogren, M. U., van Teijlingen, E., and Berg, M., 2013. Where midwives are not yet recognised: a feasibility study of professional midwives in Nepal. *Midwifery*, 29 (10), 1103–9.
- Bolam, A., Manandhar, D. S., Shrestha, P., Ellis, M., Malla, K., and Costello, A. M., 1998. Factors affecting home delivery in the Kathmandu Valley, Nepal. *Health Policy and Planning*, 13 (2), 152–8.
- Bonell, C. P., Hargreaves, J., Cousens, S., Ross, D., Hayes, R., Petticrew, M., and Kirkwood, B. R., 2011. Alternatives to randomisation in the evaluation of public health interventions: design challenges and solutions. *Journal of Epidemiology and Community Health*, 65 (7), 582–587. Available from:

http://jech.bmj.com/content/65/7/582.abstract [Accessed 21 September 2015]. Boone, P., Mann, V., Eble, A., Mendiratta, T., Mukherjee, R., Figueiredo, R.,

- Jayanty, C., Frost, C., Padmanabh, M. R., and Elbourne, D., 2007. Community health and medical provision: impact on neonates (The CHAMPION trial). *BMC Pediatrics*, 7 (1), 1–11. Available from: http://dx.doi.org/10.1186/1471-2431-7-26 [Accessed 21 October 2015].
- Booth, M. L. and Samdal, O., 1997. Health–promoting schools in Australia: models and measurement. *Australian and New Zealand Journal of Public Health*, 21 (4), 365–370. Available from: http://doi.wiley.com/10.1111/j.1467-842X.1997.tb01716.x [Accessed 18 November 2015].
- Borghi, J., Ensor, T., Neupane, B. D., and Tiwari, S., 2006. Financial implications of skilled attendance at delivery in Nepal. *Tropical Medicine & International Health*, 11 (2), 228–237. Available from: http://dx.doi.org/10.1111/j.1365-3156.2005.01546.x [Accessed 21 October 2015].
- Bothwell, L.E., Greene, J.A., Podolsky, S.H. and Jones, D.S., 2016. Assessing the gold standard—lessons from the history of RCTs. Available from: http://www.nejm.org/doi/full/10.1056/NEJMms1604593#t=article [Accessed 27 January 2017].
- Bouvier-Colle, M.-H., Mohangoo, A. D., Gissler, M., Novak-Antolic, Z., Vutuc, C., Szamotulska, K., Zeitlin, J., and Committee, E.-P. S., 2012. What about the mothers? An analysis of maternal mortality and morbidity in perinatal health surveillance systems in Europe. *BJOG: An International Journal of Obstetrics & Gynaecology*, 119 (7), 880-890.Available from: http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=medl&NEWS= N&AN=22571748 [Accessed 21 October 2015].
- Bowling, A., 2014. *Research Methods in Health: Investigating Health and Health Services*. London, UK: McGraw-Hill Education (UK).
- Bowser D., and Hill K., 2010. *Exploring evidence for disrespect and abuse in facilitybased childbirth: report of a landscape analysis.* 2010. Boston, MA: URC-Traction Project, Harvard School of Public Health. Available form: http://www.tractionproject.org/sites/default/files/Respectful_Care_at_Birth_9-20-101_Final.pdf [Accessed 13 December 2016].
- Brannen, J., 1992. *Mixing methods: Quantitative and qualitative research.* Aldershot, UK: Avebury (UK).

- Brauman, K., Suchinfran, C., and Murray, D., 1999. The paucity of effects in community trials: is secular trend the culprit? *Preventive Medicine*, 28, 426–429.
- British Medical Journal (BMJ), 2015. The *Chi square tests*. London, UK: BMJ Publishing Group Ltd. Available from: http://www.bmj.com/aboutbmj/resources-readers/publications/statistics-square-one/8-chi-squared-tests [Accessed 8 Aug 2015].
- Britten, N., 1995. Qualitative Research: Qualitative interviews in medical research. *BMJ*, 311 (6999), 251-253.
- Britton, A., Thorogood, M., Coombes, Y., and Lewando-Hundt, G., 1998. Search for evidence of effective health promotion. Quantitative outcome evaluation with qualitative process evaluation is best. *BMJ (Clinical research ed.)*, 316 (7132), 703–4.
- Brody, C., Hoop, T.D., Vojtkova, M., Warnock, R., Dunbar, M., Murthy, P. and Dworkin, S.L. 2016. Can self-help group programs improve women's empowerment? A systematic review. *Journal of Development Effectiveness*, 1-26.
- Brown, S. and Felbab-Brown, V., 2012. *Nepal, on the Brink of Collapse*. New York Times, 5. Available from: *www.nytimes.com/2012/06/06/opinion/nepal-on-thebrink-of-collapse.html* [Accessed 24 January 2014].
- Brown, C. A., Sohani, S. B., Khan, K., Lilford, R., and Mukhwana, W., 2008.
 Antenatal care and perinatal outcomes in Kwale district, Kenya. *BMC Pregnancy and Childbirth*, 8 (1), 2. Available from: http://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/1471-2393-8-2 [Accessed 21 December 2015].
- Bryce, J., el Arifeen, S., Pariyo, G., Lanata, C. F., Gwatkin, D., and Habicht, J.-P., 2003. Reducing child mortality: can public health deliver? *The Lancet*, 362 (9378), 159–164.
- Bryman, A., 2007. Barriers to Integrating Quantitative and Qualitative Research. *Journal of Mixed Methods Research*, 1 (1), 8–22. Available from: http://mmr.sagepub.com/content/1/1/8.abstract [Accessed 10 May 2015].
- Campbell, M., Fitzpatrick, R., Haines, A., Kinmonth, A. L., Sandercock, P., Spiegelhalter, D., and Tyrer, P., 2000. Framework for design and evaluation of complex interventions to improve health. *BMJ*, 321 (7262), 694–696. Available from: http://www.bmj.com/content/321/7262/694.abstract [Accessed 10 May 2015].
- Campbell, O. M. R. and Graham, W. J., 2006. Strategies for reducing maternal mortality: getting on with what works. *The Lancet*, 368 (9543), 1284–99.
- Carr, L.T., 1994. The strengths and weaknesses of quantitative and qualitative research: what method for nursing? *Journal of Advanced Nursing*, *20*(4), 716-721.
- Carson, J., 2005. The strengths and weaknesses of research designs involving quantitative measures. *Journal of Research in Nursing*, *10*(5), 583-583.

- Central Bureau of Statistics (CBS), 2001. *Statistical Yearbook of Nepal.* Kathmandu, Nepal: National Planning Commission Secretariat, Government of Nepal. Available from: http://cbs.gov.np/?page_id=1233 [Accessed 12 March 2014].
- Central Bureau of Statistics (CBS), 2012. *Nepal Nepal Living Standards Survey* 2010-2011, NLSS Third. Kathmandu: CBS - National Planning Commission Secretariat, Government of Nepal. Available from: http://cbs.gov.np/nada/index.php/catalog/37 [Accessed 24 January 2017].
- Central Bureau of Statistics (CBS), 2015. National Planning Commission Secretariat, Government of Nepal (GON). *Nepal - Annual Household Survey* 2013-2014. Kathmandu: GoN and UNDP. Available from: http://cbs.gov.np/nada/index.php/catalog/52 [Accessed 24 January 2017].
- Central Intelligence Agency (2013). *The World Factbook: Nepal.* Washington, DC: Central Intelligence Agency. Available from: https://www.cia.gov/library/publications/resources/the-worldfactbook/geos/np.html [Accessed 24 January 2017].
- *Centres* for Disease Control and Prevention (CDC), 2013. *A sustainability planning guide for healthy communities*. Atlanta, GA: CDC. Available from: http://www.cdc.gov/nccdphp/dch/programs/healthycommunitiesprogram/pdf/su stainability_guide.pdf [Accessed 2 March 2014].
- Choulagai, B., Onta, S., Subedi, N., Mehata, S., Bhandari, G., Poudyal, A., Shrestha, B., Mathai, M., Petzold, M., and Krettek, A., 2013. Barriers to using skilled birth attendants' services in mid- and far-western Nepal: A crosssectional study. *BMC International Health and Human Rights*, 13 (1), 49. Available from: http://www.biomedcentral.com/1472-698X/13/49 [Accessed 1 May 2015].
- Chowdhury, A. M., Karim, F., and Ahmed, J., 1988. Teaching ORT to women: individually or in groups? *The Journal of Tropical Medicine and Hygiene*, 91 (6), 283–287.
- Christian, P., Khatry, S. K., Katz, J., Pradhan, E. K., LeClerq, S. C., Shrestha, S. R., Adhikari, R. K., Sommer, A., and Keith, P. W., 2003. Effects of alternative maternal micronutrient supplements on low birth weight in rural Nepal: Double blind randomised community trial. *BMJ*, 326 (7389), 571.
- Christiansen, G., 1999. *Evaluation as a quality assurance tool in health promotion. Research and practice for health promotion an expert report.* Research and Practice of Health. Cologne, Germany: Federal Centre for Health Education.

Choy, L.T., 2014. The strengths and weaknesses of research methodology: Comparison and complimentary between qualitative and quantitative approaches. *IOSR Journal of Humanities and Social Science*, *19*(4), 99-104. Available from: http://s3.amazonaws.com/academia.edu.documents/37208325/N0194399104. pdf?AWSAccessKeyId=AKIAIWOWYYGZ2Y53UL3A&Expires=1486207489&S ignature=j3DpolmILARIPfjNkh5foJOIDfU%3D&response-contentdisposition=inline%3B%20filename%3DThe_Strengths_and_Weaknesses_of_ Research.pdf [Accessed 29 January 2017].
- Clancy, M. J., 2002. Overview of research designs. *Emergency Medicine Journal*, 19 (6), 546–549. Available from: http://emj.bmj.com/content/19/6/546.abstract [Accessed 10 May 2015].
- Clarke, M. and Oxman, A., 2003. Formulating the problem. In: Clarke M., Oxman A.D., eds. *Cochrane reviewers' handbook*. Available from: http://www.cochrane.dk/cochrane/handbook/handbook.htm [Accessed 3 April 2015].
- Cleland, J., Conde-Agudelo, A., Peterson, H., Ross, J. and Tsui, A., 2012. Contraception and health. *The Lancet*, *380*(9837), 149-156.
- Clinton, M., 2010. Ethical conduct of research in developing countries. *International Journal of Evidence-based Healthcare*, 8 (2), 43-44. Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1279905/ [Accessed 22 December 2016].
- Clore, G.L. & Schnall, S., 2005. The influence of affect on attitude. *In: Albarracin, D.* & *Zanna, M. P., ed. Handbook of attitudes*. Mahwah: Erlbaum, 437–489. Available from: https://www.repository.cam.ac.uk/bitstream/handle/1810/239306/Clore & Schnall %282005%29.pdf;jsessionid=48BBE98CEC3BF9FCE84E14050EA06098?seq uence=1 [Accessed 3 April 2015].
- Colbourn, T., Pulkki-Brännström, A.-M., Nambiar, B., Kim, S., Bondo, A., Banda, L., Makwenda, C., Batura, N., Haghparast-Bidgoli, H., Hunter, R., Costello, A., Baio, G., and Skordis-Worrall, J., 2015. Cost-effectiveness and affordability of community mobilisation through women's groups and quality improvement in health facilities (MaiKhanda trial) in Malawi. *Cost Effectiveness and Resource Allocation : C/E*, 13 (1), 1. Available from: http://www.resourceallocation.com/content/13/1/1 [Accessed 12 February 2015].
- Corner, J., 1991. In search of more complete answers to research questions Quantitative versus qualitative research methods is there a war forward? *Journal of Advanced Nursing* 16(3), 718-727.
- Corti, L., Day, A., and Backhouse, G., 2000. Confidentiality and Informed Consent: Issues for Consideration in the Preservation of and Provision of Access to Qualitative Data Archives. *Forum Qualitative Sozialforschung/Forum: Qualitative Social Research*. Available from: http://www.qualitativeresearch.net/index.php/fqs/article/view/1024/2207 [Accessed 10 February 2015].
- Costello, A., Azad, K., and Barnett, S., 2006. An alternative strategy to reduce maternal mortality. *The Lancet*, 368 (9546), 1477–9. Available from: http://www.ncbi.nlm.nih.gov/pubmed/17071268 [Accessed 29 October 2015].
- Coyne, I. T., 1997. Sampling in qualitative research. Purposeful and theoretical sampling; merging or clear boundaries? *Journal of Advanced Nursing*, 26, 623–630.
- Craig, P., Dieppe, P., Macintyre, S., Michie, S., Nazareth, I. and Petticrew, M., 2011. Developing and evaluating complex interventions: new guidance.

London, UK: *Medical Research Council*, UK. Available from: https://www.mrc.ac.uk/documents/pdf/complex-interventions-guidance/ [Accessed 22 December 2016].

- Craig, P. 2013 *Health Inequalities Action Framework*. Edinburgh: NHS Health Scotland Available from: https://elearning.healthscotland.com/pluginfile.php/30245/mod_resource/conte nt/0/NHS_Health_Scotland_Corporate_resources/HealthInequalitiesActionFra mework.pdf [Accessed 25 February 2016].
- Creswell, J. W., 2003. *Research design: qualitative, quantitative, and mixed method approaches.* California, USA; Thousand Oaks.
- Creswell, J. W., Plano-Clark, V. L., Gutmann, M. L., and Hanson, W. E., 2003. Advanced Mixed Methods Research Designs. In: *Handbook of Mixed Methods in Social and Behavioral Research*, 209–240. Available from: http://www.sagepub.com/upm-data/19291_Chapter_7.pdf [Accessed 29 October 2015].
- Creswell, J.W. and Plano-Clark, V.L., 2007. Choosing a mixed methods design. Designing and conducting mixed methods research, 58-88. London, UK: SAGE Publications.
- Creswell, J. W. and Plano-Clark, V. L., 2011. *Designing and Conducting Mixed Methods Research*. London, UK: SAGE Publications.

Cryder, C. E., London, A. J., Volpp, K. G., and Loewenstein, G., 2010. Informative inducement: Study payment as a signal of risk. *Social Science & Medicine*, 70 (3), 455–464. Available from: http://www.sciencedirect.com/science/article/pii/S0277953609007370 [Accessed 10 May 2015].

- Dane, A. V and Schneider, B. H., 1998. Program integrity in primary and early secondary prevention: Are implementation effects out of control? *Clinical Psychology Review*, 18 (1), 23–45. Available from: http://www.sciencedirect.com/science/article/pii/S0272735897000433 [Accessed 10 May 2015].
- Darmstadt, G. L., Lee, A. C. C., Cousens, S., Sibley, L., Bhutta, Z. A., Donnay, F., Osrin, D., Bang, A., Kumar, V., Wall, S. N., Baqui, A., and Lawn, J. E., 2009.
 60 Million non-facility births: Who can deliver in community settings to reduce intrapartum-related deaths? *International Journal of Gynaecology and Obstetrics: The Official Organ of the International Federation of Gynaecology and Obstetrics*, 107 Suppl., S89–112.
- Datta, J. and Petticrew, M., 2013. Challenges to evaluating complex interventions: a content analysis of published papers. *BMC Public Health*, 13 (1), 568. Available from: http://www.biomedcentral.com/1471-2458/13/568 [Accessed 10 June 2015].
- Dawson, C., 2009. Introduction to research methods: a practical guide for anyone undertaking a research project. Hachette, UK: Little Brown Book Group.
- Dawson, A. and Grill, K., 2012. Health Promotion: Conceptual and Ethical Issues. *Public Health Ethics*, 5 (2), 101–103. Available from:

http://phe.oxfordjournals.org/content/5/2/101.full [Accessed 17 November 2015].

- Davis D.A, Thomson M.A, Oxman A.D., Haynes R.B. 1995. Changing physician performance. A systematic review of the effect of continuing medical education strategies. *JAMA* 274(9):700–5. Available from: http://jamanetwork.com/journals/jama/article-abstract/389532 [Accessed 26 January 2017].
- Dearden, K., Altaye, M., De Maza, I., De Oliva, M., Stone-Jimenez, M., Burkhalter, B. R., and Morrow, A. L., 2002. The impact of mother-to-mother support on optimal breast-feeding: a controlled community intervention trial in peri-urban Guatemala City, Guatemala. *Pan American Journal of Public Health*, 12 (3), 193–201.
- Deeks, J., Dinnes, J., D'Amico, R., Sowden, A., and Sakarovitch, C., 2003. Evaluating non-randomised intervention studies. *Health Technology Assessment*, 7 (27), 186. Available from: http//journalslibrary.nihr.ac.uk/hta/hta7270 [Accessed 17 November 2015].
- Defo, B. K., 1997. Effects of socioeconomic disadvantage and women's status on women's health in Cameroon. Social Science & Medicine, 44 (7), 1023–1042. Available from: http://www.sciencedirect.com/science/article/pii/S0277953696002286 [Accessed 17 November 2015].
- Deniston, O. L., 1980. Whither evaluation-whither utilisation. *Evaluation and Program Planning*, (3), 91–94.
- Dennis, C.-L., Hodnett, E., Kenton, L., Weston, J., Zupancic, J., Stewart, D. E., and Kiss, A., 2009. Effect of peer support on prevention of postnatal depression among high risk women: Multisite randomised controlled trial. *BMJ*, 338. Available from: http://www.bmj.com/content/338/bmj.a3064.abstract [Accessed 17 March 2015].
- Denzin, N. K. and Lincoln, Y. S., 1994. *Handbook of qualitative research*. London, UK: SAGE Publications: (page vii).
- Department for International Development (DFID), 2010. Department for International Development in 2009-10: response to the International Development (Reporting and Transparency) Act 2006. Available from: www.legislation.gov.uk/ukpga/2006/31/pdfs/ukpga_20060031_en.pdf [Accessed 22 December 2016].
- Dhaliwal, I., Duflo, E., Glennerster, R., and Caitlin, T., 2011. Comparative costeffectiveness analysis to inform policy in developing countries: A general framework with applications for education. Cambridge, MA: Abdul Latif Jameel Poverty Action Lab (J-PAL), MIT. Available from: http://economics.mit.edu/files/6959 [Accessed 22 December 2015].
- Dharmalingam, A., Navaneetham, K., and Krishnakumar, C. S., 2010. Nutritional status of mothers and low birth weight in India. *Maternal and Child Health Journal*, 14, 290–298.

Dixon, J., Biehal, N., Green, J., Sinclair, I., Kay, C., and Parry, E., 2013. Trials and

tribulations: challenges and prospects for randomised controlled trials of social work with children. *British Journal of Social Work*, 44 (6), 1563–1581. Available from:

http://bjsw.oxfordjournals.org/content/early/2013/03/04/bjsw.bct035.abstract [Accessed 21 December 2015].

- Donabedian, A., 1988. The quality of care. How can it be assessed? *JAMA*, 260 (12), 1743–1748.
- Downe, S., Finlayson, K., Walsh, D., and Lavender, T., 2009. 'Weighing up and balancing out': a meta-synthesis of barriers to antenatal care for marginalised women in high-income countries. *BJOG: An International Journal of Obstetrics* & *Gynaecology*, 116 (4), 518–529.
- Dowswell, T., Carroli, G., Duley, L., Gates, S., Gülmezoglu, A. M., Khan-Neelofur, D., and Piaggio, G. G., 2010. Alternative versus standard packages of antenatal care for low-risk pregnancy. *The Cochrane Database of Systematic Reviews*, (10), CD000934.
- Duflo, E., 2004. Scaling up and Evaluation. Annual World Bank Conference on Development Economics. Available from: http://economics.mit.edu/files/766 [Accessed 17 April 2015].
- Duflo, E., 2012. Rigorous evaluation of human behavior. *Science*, 336 (6087), 1398. Available from: http://www.sciencemag.org/content/336/6087/1398.2.short. [Accessed 29 April 2015].
- Duffy, M.E. 1985. Quantitative and qualitative research antagonistic or complementary? *Nursing and Health Care* 8(6), 356-357.
- Ensor, T., Clapham, S., and Prasai, D. P., 2009. What drives health policy formulation: insights from the Nepal maternity incentive scheme? *Health Policy*, 90 (2-3), 247–53.
- Ensor, T., Green, C., Quigley, P., Badru, A. R., Kaluba, D., and Kureya, T., 2014. Mobilizing communities to improve maternal health: results of an intervention in rural Zambia. *Bulletin of the World Health Organization*, 92 (1), 51–59. Available from: www.who.int/bulletin/volumes/92/1/13-122721/en/ [Accessed 21 December 2016].
- EPOC, 2005. Effective Practice and Organisation of Care Group EPOC methods paper: including interrupted time series (ITS) designs in a EPOC Review. Available from: http://epoc.cochrane.org/sites/epoc.cochrane.org/files/public/uploads/inttime.pd f [Accessed 21 December 2016].
- Eriksson, M. and Lindström, B., 2008. A salutogenic interpretation of the Ottawa Charter. *Health Promotion International*, 23 (2), 190–199.
- Ezzy, D., 2013. Qualitative analysis. London, UK: Routledge.
- Evans, D., and Killoran, A. 2000. Tackling health inequalities through partnership working: learning from a realistic evaluation. *Critical Public Health*,10(2), 125-140. Available from: http://www.tandfonline.com/doi/pdf/10.1080/09581590050075899 [Accessed

27 January 2017].

- Every Woman Every Child (EWEC), 2015. Saving lives, protecting futures. Progress Report on the Global Strategy for Women's and Children's Health 2010-2015. Washington DC: UN Foundation. Available from: http://www.everywomaneverychild.org/images/EWEC_Progress_Report_FINA L 3.pdf [Accessed 12 May 2015].
- Falle, T.Y., Mullany, L.C., Thatte, N., Khatry, S.K., LeClerq, S.C., Darmstadt, G.L., Katz, J. and Tielsch, J.M., 2009. Potential role of traditional birth attendants in neonatal healthcare in rural southern Nepal. *Journal of Health, Population and Nutrition*, pp.53-61. Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2650835/ [Accessed 27 January 2017].
- Filmer, D. and Pritchett, L., 2001. Estimating wealth effects without expenditure data--or tears: an application to educational enrollments in states of India. *Demography*, 38 (1), 115–32.
- Finlayson, K. and Downe, S., 2013. Why Do Women Not Use Antenatal Services in Low- and Middle-Income Countries? A Meta-Synthesis of Qualitative Studies. *PLoS Medicine*, 10 (1), e1001373. Available from: http://dx.plos.org/10.1371/journal.pmed.1001373 [Accessed 24 January 2013].
- Fitz-Gibbon, C.T. and Morris, L.L., 1987. *How to design a program evaluation* (Vol. 3). Newbury Park, California: London, UK: SAGE Publications.
- Flynn, B. B., Schroeder, R. G., and Sakakibara, S., 1994. A framework for quality management research and an associated measurement instrument. *Journal of Operations Management*, 11 (4), 339–366.
- Forrest Keenan, K., van Teijlingen, E., and Pitchforth, E., 2005. The analysis of qualitative research data in family planning and reproductive health care. *The Journal of Family Planning and Reproductive Health Care*, 31 (1), 40–3. Available from: http://jfprhc.bmj.com/content/31/1/40.citation [Accessed 3 Mar 2015].
- Fottrell, E., Azad, K., Kuddus, A., Younes, L., Shaha, S., Nahar, T., Aumon, B.H., Hossen, M., Beard, J., Hossain, T. and Pulkki-Brannstrom, A.M., 2013. The effect of increased coverage of participatory women's groups on neonatal mortality in Bangladesh: a cluster randomized trial. *JAMA Pediatrics*, 167(9), 816-825.
- Fujita, N., Matsui, M., Srey, S., Po, C. S., Uong, S., and Koum, K., 2005. Antenatal care in the capital city of Cambodia: current situation and impact on obstetric outcome. *Journal of Obstetrics and Gynaecology Research*, 31 (2), 133–139. Available from: http://dx.doi.org/10.1111/j.1341-8076.2005.00259.x [Accessed 3 March 2015].
- Furuta, M. and Salway, S., 2006. Women's position within the household as a determinant of maternal health care use in Nepal. *International Family Planning Perspectives*, 32 (1), 17–27.
- Gartlehner, G., Hansen, R. A., Nissman, D., Lohr, K. N., and Carey, T. S., 2006. Criteria for distinguishing effectiveness from efficacy trials in systematic

reviews. *Technical Reviews, No. 12.* (Prepared by the RTI-International– University of North Carolina Evidence-based Practice Center under Contract No. 290-02-0016) AHRQ Publication No.06-0046. Rockville, MD: Agency for Healthcare Research and Quality. Available from: http://www.ncbi.nlm.nih.gov/books/NBK44029ILLLL/ [Accessed 11 November 2015].

- Gertler, P. J., Martinez, S., Premand, P., Rawlings, L. B., and Vermeersch, C. M., 2011. *Impact evaluation in practice*. Washington DC, USA: World Bank Publications. Available from: siteresources.worldbank.org/EXTHDOFFICE/.../Impact_Evaluation_in_Practice .pdf [Accessed 21 December 2016].
- Gill, P., Stewart, K., Treasure, E., and Chadwick, B., 2008. Methods of data collection in qualitative research: interviews and focus groups. *British Dental Journal*, 204 (6), 291–5. Available from: http://dx.doi.org/10.1038/bdj.2008.192 [Accessed 11 Jul 2014].
- Glanz, K. and Rimer, B.K., 1997. *Theory at a glance: A guide for health promotion practice* (No. 97). US Dept. of Health and Human Services, Public Health Service, National Institutes of Health, National Cancer Institute.
- Glasgow, R. E., Vogt, T. M., and Boles, S. M., 1999. Evaluating the public health impact of health promotion interventions: the RE-AIM framework. *American Journal of Public Health*, 89 (9), 1322–1327.
- Godin, G., Gagnon, H., Alary, M., Levy, J. J., and Otis, J., 2007. The degree of planning: an indicator of the potential success of health education programs. *Promotion & Education*, 14 (3), 138–42.
- Goodman, R. M., 1998. Principals and tools for evaluating community-based prevention and health promotion programs. *Journal of Public Health Management and Practice*, 4 (2), 37–47.
- Government of Nepal. National Planning Commission Secretariat (GON), 2014. Kathmandu, Nepal: *Central Bureau of Statistics*. Available from: http://cbs.gov.np/ [Accessed 15 November 2015].
- Government of Nepal, MInistry of Health, Department of Health Services, (GoN, MoH and DHS), 2014. *Organization Structure*. Available from: http://dohs.gov.np/about-us/organization-structure/ [Accessed 25 Ferbruary 2017].
- Graham, W., Brass, W. and Snow, R.W., 1989. Estimating maternal mortality: the sisterhood method. *Studies in Family Planning*, 20 (3), 125-135. Available from: http://www.jstor.org/stable/1966567?seq=1#page_scan_tab_contents [Accessed 5 December 2015].
- Graham, W. J., Fitzmaurice, A. E., Bell, J. S., and Cairns, J. A., 2004. The familial technique for linking maternal death with poverty. *The Lancet*, 363 (9402), 23–27.
- Graham, H. and Kelly, M.P., 2004. *Health inequalities: concepts, frameworks and policy*. London: Health Development Agency. Available from: http://publicaciones.ops.org.ar/publicaciones/piezas%20comunicacionales/curs oDDS/cursoesp/Textos%20Completos/health%20inequalities_concepts,%20fr

amework%20and%20policy.pdf [Accessed 24 February 2017].

- Grant, R. W. and Sugarman, J., 2004. Ethics in Human Subjects Research: Do Incentives Matter? *Journal of Medicine and Philosophy*, 29 (6), 717–738.
- Green, C., 1998. *Mother support groups: a review of experience in developing countries*. Arlington, Virginia: Partnership for Child Health Care, Basic Support for Institutionalizing Child Survival [BASICS]. Available from: http://www.basics.org/documents/Mother-Support-Groups_Cynthia-Green.pdf [Accessed 21 December 2016].
- Green, J., 2000. The role of theory in evidence-based health promotion practice. *Health Education Research*, 15 (2), 125–129. Available from: http://her.oxfordjournals.org/content/15/2/125.long [Accessed 8 September 2015].
- Green, J., South, J., 2006. *Evaluation*. New York, USA: McGraw-Hill Companies, Incorporated.
- Green, J. and Tones, K., 1999. For debate. Towards a secure evidence base for health promotion. *Journal of Public Health*, 21 (2), 133–139. Available from: http://jpubhealth.oxfordjournals.org/content/21/2/133.abstract?ijkey=271ec4372 62e5da9b544a4101aca001562b34be1&keytype2=tf_ipsecsha [Accessed 22 December 2015].
- Green, J., Willis, K., Hughes, E., Small, R., Welch, N., Gibbs, L., and Daly, J., 2007. Generating best evidence from qualitative research: the role of data analysis. *Australian and New Zealand Journal of Public Health*, 31 (6), 545–550.
- Green Tara Trust Nepal (GTN), 2008. *Baseline Report 2008* (Comparison of Intervention and Control community), Kathmandu, Nepal: GTN.
- Gruber, J., 1994. The Incidence of Mandated Maternity Benefits. *The American Economic Review*, 84 (3), 622–641.
- Hadi, A., 2001. Promoting health knowledge through micro-credit programmes: experience of BRAC in Bangladesh. *Health Promotion International*, 16 (3), 219–227.
- Hart, E., 1996. Action research as a professionalizing strategy: issues and dilemmas. *Journal of Advanced Nursing*, 23 (3), 454–461.
- Hawe, P., Degeling, D., and Hall, J., 1990. *Evaluating Health Promotion: A Health Worker's Guide*. Sydney Australia: MacLennan & Petty.
- Hawkes, N., 2013. Deworming debunked. *BMJ* 346. p.e8558. Available from: http://www.bmj.com/content/346/bmj.e8558.short [Accessed 16 December 2016].
- de Heer, H.D., Koehly, L., Pederson, R., Morera, O. 2011. Effectiveness and spillover of an after-school health promotion program for Hispanic elementary school children. *The American Journal of Public Health*, 101, 1907–13.
- Higginbottom, G. M. A., 2004. Sampling issues in qualitative research. *Nurse Researcher*, 12 (1), 7-19. Available from: http://journals.rcni.com/doi/abs/10.7748/nr2004.07.12.1.7.c5927?url_ver=Z39.8

8-2003&rfr_id=ori%3Arid%3Acrossref.org&rfr_dat=cr_pub%3Dpubmed& [Accessed 12 January 2017].

- Higgins, K. L., 2014. *Economic Growth and Sustainability: Systems Thinking for a Complex World*. Edinburgh, UK: Elsevier Science.
- Hobbes, W., 2014. Stop trying to save the world. New Republic. New York, USA: New Republic. Available from: http://www.newrepublic.com/article/120178/problem-internationaldevelopment-and-plan-fix-it [Accessed 17 November 2014].
- Hogan, M. C., Foreman, K. J., Naghavi, M., Ahn, S. Y., Wang, M., Makela, S. M., Lopez, A. D., Lozano, R., and Murray, C. J., 2010. Maternal mortality for 181 countries, 1980-2008: a systematic analysis of progress towards Millennium Development Goal 5. *The Lancet*, 375, 1609–1623.
- Houweling, T.A., Azad, K., Younes, L., Kuddus, A., Shaha, S., Haq, B., Nahar, T., Beard, J., Fottrell, E.F., Prost, A. and Costello, A., 2011. The effect of participatory women's groups on birth outcomes in Bangladesh: does coverage matter? Study protocol for a randomized controlled trial. *Trials*, *12*(1), 208. Available from: https://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-12-208 [Accessed 21 December 2015].
- Houweling, T. A., Ronsmans, C., Campbell, O. M., and Kunst, A. E., 2007. Huge poor–rich inequalities in maternity care: an international comparative study of maternity and child care in developing countries. *Bulletin of the World Health Organization*, 85 (10), 745–754. Available from: /pmc/articles/PMC2636501/?report=abstract [Accessed 21 December 2015].
- Howe, K. R., 1988. Against the Quantitative-Qualitative Incompatibility Thesis or Dogmas Die Hard. *Educational Researcher*, 17 (8), 10–16.
- Howe, L., Hargreaves, J., and Huttly, S., 2008. Issues in the construction of wealth indices for the measurement of socio-economic position in low-income countries. *Emerging Themes in Epidemiology*, 5 (1), 1–14.
- Housing Recovery and Reconstruction Platform Nepal (HRRP), 2015. *Nepal: Kathmandu District Administrative Map.* Kathmandu: HRRP, Nepal. Available from:

http://hrrpnepal.org/media/38582/160226_kathmandu_admin_district_v2.pdf [Accessed 25 January 2017].

- Houweling, T.A., Azad, K., Younes, L., Kuddus, A., Shaha, S., Haq, B., Nahar, T., Beard, J., Fottrell, E.F., Prost, A. and Costello, A., 2011. The effect of participatory women's groups on birth outcomes in Bangladesh: does coverage matter? Study protocol for a randomized controlled trial. *Trials*, 12 (1), 1.
- Hulton, L. A., Matthews, Z., and Stones, R. W., 2007. Applying a framework for assessing the quality of maternal health services in urban India. *Social Science & Medicine (1982)*, 64 (10), 2083–95.
- Hultsch, D. F., MacDonald, S. W. S., Hunter, M. A., Maitland, S. B., and Dixon, R. A., 2002. Sampling and generalisability in developmental research: comparison of random and convenience samples of older adults. *International Journal of Behavioral Development*, 26 (4), 345–359.

- Hundley, V. A., 2013. Early labour should we be telling women to stay at home? *MIDIRS Midwifery Digest*, 3 (23), 337–340.
- Hundley, V. A., 2014. Early labour: Should we be telling women to stay at home? *In: Gold Perinatal Care Online Conference.* https://www.goldlearning.com/lecture/240 [Accessed 20 November 2016].
- Hussein, J., Bell, J., Iang, M.D., Mesko, N., Amery, J. and Graham, W., 2011. An appraisal of the maternal mortality decline in Nepal. *PloS one*, *6*(5), p.e19898. Available from: http://journals.plos.org/plosone/article?id=10.1371/journal.pone.0019898 [Accessed 3 January 2016].
- The International Fund for Agricultural Development (IFAD) and Tango International, 2009. *Sustaibability of rural development projects*. Available from: http://www.ifad.org/operations/projects/regions/pi/paper/8.pdf [Accessed 20 March 2015].
- Imbens, G. and Wooldridge, J., 2007. Lecture 1: Estimation of average treatment effects under unconfoundedness. *In: What's New in Econometrics" NBER Summer 2007 Seminar*. Available from: http://www.nber.org/WNE/WNEnotes.pdf [Accessed 2 January 2015].
- Index Mundi, 2012. *Nepal Factbook.* Kathmandu, Nepal: CIA. Available from: http://www.indexmundi.com/nepal/ [Accessed 5 March 2016].
- Institute of Medicine (IMS, US) Committee on Health and Behavior: Research, Practice, and Policy, 2001. *Health and behavior: the interplay of biological, behavioral and societal influences*. Washington D.C.: National Academies Press (US). Available from: http://www.ncbi.nlm.nih.gov/books/NBK43742/ [Accessed 2 March 2016].
- Institute for Work & Health (IWH), 2015. *What researchers mean by... primary data and secondary data.* At Work, 82 Fall 2015. Toronto: Institute for Work & Health. Available from: http://www.iwh.on.ca/wrmb/primary-data-andsecondary-data [Accessed 27 January 2017].
- Ireland, J., van Teijlingen, E., and Kemp, J., 2015. Twinning in Nepal: the Royal College of Midwives UK and the Midwifery Society of Nepal working in partnership Journal of Asian Midwives. *Journal of Asian Midwives*, 2 (1), 26– 33.
- Irwig, L., Zwarenstein, M., Zwi, A., and Chalmers, I., 1998. A flow diagram to facilitate selection of interventions and research for health care. *Bulletin of the World Health Organization*, 76 (1), 17–24.
- Israel, B. A., Schulz, A. J., Parker, E. A., and Becker, A. B., 2001. Communitybased participatory research: policy recommendations for promoting a partnership approach in health research. *Education for Health*, 14 (2), 182– 197.

The Abdul Latif Jameel Poverty Action Lab (J-PAL), 2015. *Impact Evaluation*. Cambridge, MA: J-PAL. Available from: http://www.povertyactionlab.org/methodology/what-evaluation/impactevaluation [Accessed 5 February 2015].

Jackson, S. F., Edward, R., Kahan, B., and Al, E., 2001. An assessment of the methods and concepts used to synthesize the evidence of effectiveness in health promotion: a review of 17 initiatives. Toronto, Canada: Canadian Consortium for Health Promotion Research. Available from:

http://s3.amazonaws.com/academia.edu.documents/44700993/An_assessmen t_of_the_methods_and_concept20160413-17077-

1jvamzg.pdf?AWSAccessKeyId=AKIAJ56TQJRTWSMTNPEA&Expires=14819 15558&Signature=1mWL1TaReFe%2FNdhksF4b8NoerNE%3D&responsecontent-

disposition=inline%3B%20filename%3DAn_assessment_of_the_methods_and _concept.pdf [Accessed 12 December 2016].

- Jacobs, B., Ir, P., Bigdeli, M., Annear, P. L., and Van Damme, W., 2012. Addressing access barriers to health services: an analytical framework for selecting appropriate interventions in low-income Asian countries. *Health Policy and Planning*, 27 (4), 288–300. Available from: http://heapol.oxfordjournals.org/content/27/4/288.full [Accessed 29 April 2015].
- John, A., 2015. Towards midwifery education and regulation in Nepal. *The Practising Midwife*, 18 (8), pp.24-26. Available from: http://www.ingentaconnect.com/content/mesl/tpm/2015/00000018/0000008/ar t00009?crawler=true [Accessed 21 December 2015].
- Johnson, R.B. and Onwuegbuzie, A.J., 2004. Mixed methods research: A research paradigm whose time has come. *Educational Researcher*, *33*(7), 14-26. Available from: http://journals.sagepub.com/doi/abs/10.3102/0013189X033007014 [Accessed on 29 January 2017].
- Jokhio, A. H., Winter, H. R., and Cheng, K. K., 2005. An intervention involving traditional birth attendants and perinatal and maternal mortality in Pakistan. *The New England Journal of Medicine*, 352 (20), 2091–9.
- Joshi, R., Sharma, S., and van Teijlingen, E., 2013. Improving neonatal health in Nepal: Major challenges to achieving millennium development goal 4. *Health Science Journal*, 7 (3), 247 - 257. Available from: http://eprints.bournemouth.ac.uk/20916/1/Joshi et al 2013.pdf [Accessed 22 December 2015].
- Joshi, C., Torvaldsen, S., Hodgson, R., and Hayen, A., 2014. Factors associated with the use and quality of antenatal care in Nepal: a population-based study using the demographic and health survey data. *BMC Pregnancy and Childbirth*, 14, 94.
- Jowett, M., 2000. Safe motherhood interventions in low-income countries: an economic justification and evidence of cost effectiveness. *Health Policy*, 53 (3), 201-228.
- Judd, J., Frankish, C. J., and Moulton, G., 2001. Setting standards in the evaluation of community-based health promotion programmes- a unifying approach. *Health Promotion International*, 16 (4), 367–380.

Kabir, M., Iliyasu, Z., Abubakar, I. S., and Asani, A., 2005. Determinants of

utilization of antenatal care services in Kumbotso Village, northern Nigeria. *Tropical Doctor*, 35 (2), 110–1.

- Khan, A.A. and Bhardwaj, S.M., 1994. Access to health care a conceptual framework and its relevance to health care planning. *Evaluation & the Health Professions*, *17*(1), 60-76. Available from: http://journals.sagepub.com/doi/abs/10.1177/016327879401700104 [Accessed 24 January 2017].
- Kesterton, A., Cleland, J., Sloggett, A., and Ronsmans, C., 2010. Institutional delivery in rural India: the relative importance of accessibility and economic status. *BMC Pregnancy and Childbirth*, 10 (1), 30. Available from: http://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/1471-2393-10-30 [Accessed 22 July 2015].
- Kestler, E., Walker, D., Bonvecchio, A., de Tejada, S. S., and Donner, A., 2013. A matched pair cluster randomized implementation trail to measure the effectiveness of an intervention package aiming to decrease perinatal mortality and increase institution-based obstetric care among indigenous women in Guatemala: study protocol. *BMC Pregnancy and Childbirth*, 13 (1), 73. Available from: http://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/1471-2393-13-73 [Accessed 2 July 2015].
- Khanal, V., Adhikari, M., Sauer, K., and Zhao, Y., 2013. Factors associated with the introduction of prelacteal feeds in Nepal: findings from the Nepal Demographic and Health Survey 2011. *International Breastfeeding Journal*, 8 (1), 9.
- Khanal, V., Sauer, K., and Zhao, Y., 2013a. Determinants of complementary feeding practices among Nepalese children aged 6-23 months: findings from Demographic and Health Survey 2011. *BMC Pediatrics*, 13 (131). Available from: http://bmcpediatr.biomedcentral.com/articles/10.1186/1471-2431-13-131 [Accessed 19 January 2015].
- Khanal, V., Sauer, K., and Zhao, Y., 2013b. Exclusive breastfeeding practices in relation to social and health determinants: a comparison of the 2006 and 2011 Nepal Demographic and Health Surveys. *BMC Public Health*, 13 (958). Available from: http://bmcpublichealth.biomedcentral.com/articles/10.1186/1471-2458-13-958 [Accessed 19 May 2015].
- Khanal, V., Adhikari, M., Karkee, R., and Gavidia, T., 2014. Factors associated with the utilisation of postnatal care services among the mothers of Nepal: analysis of Nepal Demographic and Health Survey 2011. *BMC Women's Health*, 14 (1), 19. http://bmcwomenshealth.biomedcentral.com/articles/10.1186/1472-6874-14-19 [Accessed 2 July 2015].
- Khandker, S. R., Koolwal, G. B., and Samad, H. A., 2010. Handbook of impact evaluation. Quantitative methods and practices. Available from: http://wwwwds.worldbank.org/external/default/WDSContentServer/WDSP/IB/2009/12/10/ 000333037_20091210014322/Rendered/PDF/520990PUB0EPI1101Official0U se0Only1.pdf [Accessed 19 January 2015].
- Kinney, M.V., Boldosser-Boesch, A. and McCallon, B., 2016. Quality, equity, and dignity for women and babies. *The Lancet*, 388 (10056), 2066-2068.

- Kitzinger, J., 1995. Qualitative research. Introducing focus groups. *BMJ (Clinical research ed.)*, 311, 299–302.
- Koblinsky, M.A. ed., 2003. Reducing Maternal Mortality: Learning from Bolivia, China, Egypt, Honduras, Indonesia, Jamaica, and Zimbabwe. Washington D.C.: World Bank Publications. Available from: http://publications.worldbank.org/ecommerce/catalog/product?item_id=192997 9 [Accessed 25 February 2017].
- Koblinsky, M., Matthews, Z., Hussein, J., Mavalankar, D., Mridha, M. K., Anwar, I., Achadi, E., Adjei, S., Padmanabhan, P., and van Lerberghe, W., 2015. Going to scale with professional skilled care. *The Lancet*, 368 (9544), 1377–1386.
- Koirala, K. R. (2012). Volunteers help bring down Nepal's maternal mortality rate. Available at: Khabar South Asia. Available at: http://astanar.blogspot.com.es/2014/03/criket-became-identity-of-nepal.html [Accessed 27 January 2017].
- Koné, A., Sullivan, M., Senturia, K. D., Chrisman, N. J., Ciske, S. J., and Krieger, J.
 W., 2000. Improving collaboration between researchers and communities.
 Public Health Reports, 115 (2-3), 243–248.
- Kollmair, M., Manandhar, S., Subedi, B. and Thieme, S., 2006. New figures for old stories: Migration and remittances in Nepal. *Migration Letters*, *3*(2), 151.
- Kouyate, B., Some, F., Jahn, A., Coulibaly, B., Eriksen, J., Sauerborn, R., Gustafsson, L., Tomson, G., Becher, H., and Mueller, O., 2008. Process and effects of a community intervention on malaria in rural Burkina Faso: randomized controlled trial. *Malaria Journal*, 7 (1), 50. Available from: http://www.malariajournal.com/content/7/1/50 [Accessed 12 January 2015].
- Kremer, M., 2003. Randomized Evaluations of Educational Programs in Developing Countries: Some Lessons. *The American Economic Review*, 93 (2), 102–106. Available from: http://www.jstor.org/stable/3132208 [Accessed 19 January 2015].
- Kumar, V., Mohanty, S., Kumar, A., Misra, R. P., Santosham, M., Awasthi, S., Baqui, A. H., Singh, P., Singh, V., Ahuja, R. C., Singh, J. V., Malik, G. K., Ahmed, S., Black, R. E., Bhandari, M., and Darmstadt, G. L., 2015. Effect of community-based behaviour change management on neonatal mortality in Shivgarh, Uttar Pradesh, India: a cluster-randomised controlled trial. *The Lancet*, 372 (9644), 1151–1162. Available from: http://dx.doi.org/10.1016/S0140-6736(08)61483-X [Accessed 9 March 2015].
- Kwambai, T. K., Dellicour, S., Desai, M., Ameh, C. A., Person, B., Achieng, F., Mason, L., Laserson, K. F., and Ter Kuile, F. O., 2013. Perspectives of men on antenatal and delivery care service utilisation in rural western Kenya: a qualitative study. *BMC Pregnancy and Childbirth*, 13 (1), 134. Available from: http://www.biomedcentral.com/1471-2393/13/134 [Accessed 14 December 2015].
- Langer, A., Campero, L., Garcia, C., and Reynoso, S., 1998. Effects of psychosocial support during labour and childbirth on breastfeeding, medical interventions, and mothers' wellbeing in a Mexican public hospital: a randomised clinical trial. *British Journal of Obstetrics and Gynaecology*, 105 (10), 1056–1063.

- Larson, B.A. and Wambua, N., 2011. How to calculate the annual costs of NGOimplemented programmes to support orphans and vulnerable children: a sixstep approach. *Journal of the International AIDS Society*, *14*(1), 1. Available from: https://jiasociety.biomedcentral.com/articles/10.1186/1758-2652-14-59 [Accessed 21 December 2015].
- van Lerberghe, W. and De Brouwere, V., 2001. Reducing maternal mortality in a context of poverty. *In: Safe Motherhood Strategies: A Review of the Evidence. Studies in Health Services Organization and Policy,* 17, 1–5.
- Levin, L. and Ziglio, E., 1996. Health Promotion as an investment strategy: a perspective for the 21st Century. *In: Debates and dilemmas in promoting health: a reader.* London: Open University Press, 363–69.
- Lewin, S., Glenton, C., and Oxman, A. D., 2009. Use of qualitative methods alongside randomised controlled trials of complex healthcare interventions: methodological study. *BMJ (Clinical research ed.)*, 339, b3496. Available from: http://www.bmj.com/content/339/bmj.b3496 [Accessed 21 October 2015].
- Lewis, G., 2011. Saving Mothers' Lives: reviewing maternal deaths to make motherhood safer: 2006–08. The Eighth Report on Confidential Enquiries into Maternal Deaths. *BJOG: an International Journal of Obstetrics and Gynaecology,* 118 (s1), 1–203 Available from: http://onlinelibrary.wiley.com/doi/10.1111/j.1471-0528.2010.02847.x/pdf [Accessed 21 December 2015].
- Lewis, S., Lee, A., and Simkhada, P., 2015. The role of husbands in maternal health and safe childbirth in rural Nepal: a qualitative study. *BMC Pregnancy and Childbirth*, 15 (162).
- Lewycka, S., Mwansambo, C., Rosato, M., Kazembe, P., Phiri, T., Mganga, A., Chapota, H., Kainja, E., Newell, M.L., Greco, G. and Pulkki-Brännström, A.M., 2013. Effect of women's groups and volunteer peer counselling on rates of mortality, morbidity, and health behaviours in mothers and children in rural Malawi (MaiMwana): a factorial, cluster-randomised controlled trial. *The Lancet*, 381(9879), 1721-1735.
- Li, X. F., Fortney, J. A., Kotelchuck, M., and Glover, L. H., 2014. The postpartum period: the key to maternal mortality. *International Journal of Gynecology and Obstetrics*, 54 (1), 1–10. Available from: http://www.ijgo.org/article/0020-7292(96)02667-7/abstract [Accessed 14 December 2015].
- Lincetto, O., Mothebesoane-Anoh, S., Gomez, P. and Munjanja, S., 2006. Antenatal care. *Opportunities for Africa's newborns: Practical data, policy and programmatic support for newborn care in Africa*, 80-90. Geneva: WHO. Available from: http://www.who.int/pmnch/media/publications/aonsectionIII_2.pdf [Accessed 24 January 2017].
- Liu, X., Yan, H., and Wang, D., 2010. The evaluation of 'Safe Motherhood' program on maternal care utilization in rural western China: a difference in difference approach. *BMC Public Health*, 10 (1), 566. Available from: http://bmcpublichealth.biomedcentral.com/articles/10.1186/1471-2458-10-566 [Accessed 19 December 2016].

- Lopez, G.I., Figueroa, M., Connor, S.E. and Maliski, S.L., 2008. Translation barriers in conducting qualitative research with Spanish speakers. *Qualitative Health Research*, *18*(12), 1729-1737. Available from: http://journals.sagepub.com/doi/abs/10.1177/1049732308325857 [Accessed 19 December 2016].
- Lozano, R., Wang, H., Foreman, K. J., Rajaratnam, J. K., Naghavi, M., Marcus, J. R., Dwyer-Lindgren, L., Lofgren, K. T., Phillips, D., Atkinson, C., Lopez, A. D., and Murray, C. J. L., 2011. Progress towards Millennium Development Goals 4 and 5 on maternal and child mortality: an updated systematic analysis. *The Lancet*, 378 (9797), 1139–65.
- Macdonald, G., Veen, C., and Tones, K., 1996. Evidence for success in health promotion: suggestions for improvement. *Health Education Research*, 11 (3), 367–376. Available from: her.oxfordjournals.org/content/11/3/367.full.pdf [Accessed 12 December 2016].
- Malenga, G. and Molyneux, M., 2012. The Millennium Villages project. *The Lancet*, 379 (9832), 2131–3. Available from: http://www.thelancet.com/article/S0140673612603699/fulltext [Accessed 5 Nov 2015].
- Manandhar, D. S., Osrin, D., Shrestha, B. P., Mesko, N., Morrison, J., Tumbahangphe, K. M., Tamang, S., Thapa, S., Shrestha, D., Thapa, B., Shrestha, J. R., Wade, A., Borghi, J., Standing, H., Manandhar, M., and Costello, A. M. de L., 2004. Effect of a participatory intervention with women's groups on birth outcomes in Nepal: cluster-randomised controlled trial. *The Lancet*, 364 (9438), 970–9.
- Manktelow, B., Smith, L., Evans, T., Hyman-Taylor, P., Kurinczuk, J., Field, D., Smith, P., Mielewczyk, F., Draper, E., and MBRRACE-UK, 2015. *MBRRACE-UK Perinatal Mortality Surveillance Report UK Perinatal Death for Births from January to December 2013 – Supplementary Report.* Leicester, UK: Trusts and Health Boards. Available from: https://www.npeu.ox.ac.uk/downloads/files/mbrrace-uk/reports/MBRRACE-UK%20Perinatal%20Surveillance%20Report%202013.pdf [Accessed 2 January 2016].
- Mason, J., 2002. Qualitative Interviewing. In: *Qualitative Researching*, 236. London, UK: SAGE Publications.
- Mays, N. and Pope, C., 1995. Rigour and qualitative research. *BMJ*, 311 (6997), 109.
- Mays, N. and Pope, C., 2000. Qualitative research in health care: Assessing quality in qualitative research. *BMJ*, 320, 50–52.
- McColl, E., Jacoby, A., Thomas, L., Soutter, J., Bamford, C., Steen, N., Thomas, R., Harvey, E., Garratt, A., and Bond, J., 2001. Design and use of questionnaires: a review of best practice applicable to surveys of health service staff and patients. *Health Technology Assessment*, 5 (31), 1–256.
- McLeroy K, D, B., A, S., and K, G., 1988. An ecological perspective on health promotion programs. *Health Education Quaterly*, (15), 351–377.

McPherson, R. A., Khadka, N., Moore, J. M., and Sharma, M., 2006. Are birthpreparedness programmes effective? Results from a field trial in Siraha district, Nepal. *Journal of Health, Population, and Nutrition*, 24 (4), 479–488.

- McQueen, D. V., 2001. Strengthening the evidence base for health promotion. *Health Promotion International*, 16 (3), 261–268. Available from: http://heapro.oxfordjournals.org/content/16/3/261.full [Accessed 23 December 2015].
- McQueen, D.V. and Jones, C.M., 2007. Global perspectives on health promotion effectiveness. In: *Global perspectives on health promotion effectiveness*, 3-11. New York, USA: Springer New York.
- McQueen, D. V and De Salazar, L., 2011. Health promotion, the Ottawa Charter and 'developing personal skills': a compact history of 25 years. *Health Promotion International*, 26 Suppl 2 (suppl_2), ii194–201. Available from: http://heapro.oxfordjournals.org/content/26/suppl_2/ii194.full [Accessed 19 October 2015].
- Meissner, J. O. and Sprenger, M., 2010. Mixing Methods in Innovation Research: Studying the Process-Culture-Link in Innovation Management. *Forum Qualitative Sozialforschung / Forum: Qualitative Social Research*. Available from: http://www.qualitative-research.net/index.php/fqs/article/view/1560/3073 [Accessed 22 December 2015].

Meissner, H., Creswell, J., Klassen, A. C., Plano, V., and Smith, K. C., 2011. Best Practices for Mixed Methods Research in the Health Sciences. *Methods*, 29, 1–39. Accessed: https://www2.jabsom.hawaii.edu/native/docs/tsudocs/Best_Practices_for_Mixe d Methods Research Aug2011.pdf [Accessed 12 December 2016].

Merton, R., 1936. The unanticipated consequences of purposive social action. *American Sociological Review*, 1(6), 894-904. Available from: http://www.jstor.org/stable/2084615 [Accessed 21 December 2015].

Merzel, C. and D'Afflitti, J., 2003. Reconsidering community-based health promotion: promise, performance, and potential. *American Journal of Public Health*, 93 (4), 557–74. Available from: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=1447790&tool=pmce ntrez&rendertype=abstract [Accessed 27 April 2015].

- Miaskowski, C., Dodd, M., West, C., Schumacher, K., Paul, S. M., Tripathy, D., and Koo, P., 2004. Randomized clinical trial of the effectiveness of a self-care intervention to improve cancer pain management. *Journal of Clinical Oncology : Official Journal of the American Society of Clinical Oncology*, 22 (9), 1713–20. Available from: http://www.ncbi.nlm.nih.gov/pubmed/15117994 [Accessed 22 December 2015].
- Middleberg, M. I., 2003. *Promoting reproductive security in developing countries*. London, UK: Springer Science & Business Media.
- Milburn, K., Fraser, E., Secker, J., and Pavis, S., 1995. Combining methods in health promotion research: some considerations about appropriate use. *Health Education Journal*, 54 (3), 347–356.

Milne, L., van Teijlingen, E., Hundley, V., Simkhada, P. and Ireland, J., 2015. Staff

perspectives of barriers to women accessing birthing services in Nepal: a qualitative study. *BMC Pregnancy and Childbirth*, 15 (1), 142. Available from: http://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/s12884-015-0564-6 [Accessed 20 December 2016].

- Miller, S., Abalos, E., Chamillard, M., Ciapponi, A., Colaci, D., Comandé, D., Diaz, V., Geller, S., Hanson, C., Langer, A. and Manuelli, V. 2016. Beyond too little, too late and too much, too soon: a pathway towards evidence-based, respectful maternity care worldwide. *The Lancet*, 388 (10056), 2176-2192.
- Ministry of Health (MOH) [Nepal], New ERA, and ORC Macro. 2002. *Nepal Demographic and Health Survey 2001.* Calverton, Maryland, USA: Family Health Division, Ministry of Health; New ERA; and ORC Macro. Available from: dhsprogram.com/pubs/pdf/fr132/fr132.pdf [Accessed 18 Dec 2016].

Ministry of Health and Population (MOHP) [Nepal], New ERA, and Macro International Inc. 2007. *Nepal Demographic and Health Survey 2006.* Kathmandu, Nepal: Ministry of Health and Population, New ERA, and Macro International Inc. Available from: http://dhsprogram.com/pubs/pdf/fr191/fr191.pdf [Accessed 18 December 2016].

Ministry of Health (MOH) [Nepal], New ERA, and USAID, 2008. An analytical report on national survey of female community health volunteers of Nepal. Kathmandu: MOH and USAID. Available from: http://www.dhsprogram.com/pubs/pdf/FR181/FCHV_Nepal2007.pdf [Accessed 24 January 2017].

- Ministry of Health and Population (MOHP), 2010. *Nepal Health Sector Programme-II (NHSP-II) 2010 - 2015*. Kathmandu: Government of Nepal. Available from: http://nhsp.org.np/wp-content/uploads/2014/11/NHSP-II-Final.pdf [Accessed 24 January 2017].
- Ministry of Health and Population, Government of Nepal (MOHP), 2012. *Nepal annual report 2066/2067 (2009/2010)*. Kathmandu, Nepal: Department of Health Services, Ministry of Health and Population, Nepal. Available from: http://dohs.gov.np/wp-content/uploads/2014/04/Annual_Report_2066_67.pdf [Accessed 18 December 2016].

Ministry of Health and Population, Government of Nepal (MOHP), New ERA and ICF International, 2012. Kathmandu, Nepal: *Nepal Demographic and Health Survey 2011*. Available from: http://www.measuredhs.com/pubs/pdf/FR257/FR257 [13April2012].pdf [Accessed 22 December 2015].

- Ministry of Health and Population Nepal, Partnership for Maternal, Newborn & Child Health, WHO, World Bank and Alliance for Health Policy and Systems Research (MOHP), 2014. *Success factors for women's and children's health: Nepal.* Geneva: World Health Organization. Available from: http://www.who.int/pmnch/knowledge/publications/nepal_country_report.pdf [Accessed 25 January 2017].
- Minkler, M., 1997. *Community Organizing and Community Building for Health.* London, UK: Rutgers University Press.

- Mittelmark, M., 2002. Health Promotion: Effectiveness, Efficiency and Equity, 3rd edn Keith Tones and Sylvia Tilford Nelson Thornes, Cheltenham, 2001 524 pp, ISBN 0-7487-4527-0 (pb). *Health Education Research*, 17 (3), 376-377. Available from: http://her.oxfordjournals.org/content/17/3/376.short [Accessed 18 December 2016].
- Moore, G., Audrey, S., Barker, M., Bond, L., Bonell, C., Hardeman, W., Moore, L., O'Cathain, A., Tinati, T., Wight, D. and Baird, J., 2015. *Process evaluation of complex interventions: a summary of Medical Research Council guidance.* Glasgow: MRC Population Health Sciences Research Network, Available from: https://www.mrc.ac.uk/documents/pdf/mrc-phsrn-process-evaluation-summaryguidance/ [Accessed 25 January 2017].
- More, N. S., Bapat, U., Das, S., Patil, S., Porel, M., Vaidya, L., Koriya, B., Barnett, S., Costello, A., Fernandez, A., and Osrin, D., 2008. Cluster-randomised controlled trial of community mobilisation in Mumbai slums to improve care during pregnancy, delivery, postpartum and for the newborn. *Trials*, 9 (1), 1. Available from: http://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-9-7 [Accessed 18 December 2016].
- More, N.S., Bapat, U., Das, S., Alcock, G., Patil, S., Porel, M., Vaidya, L., Fernandez, A., Joshi, W. and Osrin, D. 2012. Community mobilization in Mumbai slums to improve perinatal care and outcomes: a cluster randomized controlled trial. *PLoS Med*, 9 (7), e1001257. Available from: http://journals.plos.org/plosmedicine/article?id=10.1371/journal.pmed.1001257 [Accessed 2 May 2015].
- Morgan, D. L., 2007. Paradigms Lost and Pragmatism Regained: Methodological Implications of Combining Qualitative and Quantitative Methods. *Journal of Mixed Methods Research*, 1 (1), 48–76. Available from: http://mmr.sagepub.com/content/1/1/48.abstract. [Accessed 8 March 2015].
- Morrison, J., Basnet, M., Budhathoki, B., Adhikari, D., Tumbahangphe, K., Manandhar, D., Costello, A., and Groce, N., 2014. Disabled women's maternal and newborn health care in rural Nepal: a qualitative study. *Midwifery*, 30 (11), 1132–9.
- Morrison, J., Osrin, D., Shrestha, B., Tumbahangphe, K. M., Tamang, S., Shrestha, D., Thapa, S., Mesko, N., Manandhar, D. S., and Costello, A., 2008. How did formative research inform the development of a women's group intervention in rural Nepal? *Journal of Perinatology*, 28 (Suppl 2), S14–S22. Available from: http://www.nature.com/jp/journal/v28/n2s/full/jp2008171a.html [Accessed 2 May 2015].
- Morrison, J., Thapa, R., Basnet, M., Budhathoki, B., Tumbahangphe, K., Manandhar, D., Costello, A., and Osrin, D., 2014. Exploring the first delay: a qualitative study of home deliveries in Makwanpur district Nepal. *BMC Pregnancy and Childbirth*, 14 (1), 89. Available from: http://www.biomedcentral.com/1471-2393/14/89 [Accessed 2 May 2015].
- Morrow, A. L., Guerrero, M. L., Shults, J., Calva, J. J., Lutter, C., Bravo, J., Ruiz-Palacios, G., Morrow, R. C., and Butterfoss, F. D., 1999. Efficacy of homebased peer counselling to promote exclusive breastfeeding: a randomised controlled trial. *The Lancet*, 353 (9160), 1226–1231.

Morse, J.M. and Niehaus, L., 2009. *Mixed method design: Principles and procedures* (Vol. 4). Oakland: USA Left Coast Press.

Mseu, D., Nyasulu, B. M., and Muheriwa, S. R., 2014. Evaluation of a Safe Motherhood project in Ntcheu district, Malawi. *International Journal of Women's Health*, 6, 1045–55. Available from: http://s3.amazonaws.com/academia.edu.documents/39569556/f_IJWH-40109evaluation-of-maternal-and-child-health-project-that-useswo_121214_22816.pdf?AWSAccessKeyId=AKIAJ56TQJRTWSMTNPEA&Expi res=1482129099&Signature=%2F3uX23IN0wpzD1fo9Dyk6gIAcjU%3D&respo nse-contentdisposition=inline%3B%20filename%3DEvaluation_of_Safe_Motherhood_in_N tcheu.pdf [Accessed 22 November 2015].

Mullany, L. C., Darmstadt, G. L., Khatry, S. K., and Tielsch, J. M., 2005. Traditional massage of newborns in Nepal: Implications for trials of improved practice. *Journal of Tropical Pediatrics*, 51, 82–86.

Mullany, B. C., Becker, S., and Hindin, M. J., 2007. The impact of including husbands in antenatal health education services on maternal health practices in urban Nepal: results from a randomized controlled trial. *Health Education Research*, 22 (2), 166–76. Available from: http://her.oxfordjournals.org/content/22/2/166.abstract [Accessed 30 January 2015].

- Mushamiri, I., Luo, C., liams-Hauser, C., and Ben Amor, Y., 2015. Evaluation of the impact of a mobile health system on adherence to antenatal and postnatal care and prevention of mother-to-child transmission of HIV programs in Kenya. *BMC Public Health*, 15 (1), 102. Available from:
- http://www.biomedcentral.com/1471-2458/15/102 [Accessed 9 February 2015]. Naidoo, J. and Wills, J., 2000. *Health promotion: foundations for practice*. Edinburgh, UK: Elsevier Health Sciences.
- Nair, N. S., Rao, R. S., Chandrashekar, S., Acharya, D., and Bhat, H. V, 2000. Socio-demographic and maternal determinants of low birth weight: a multivariate approach. *Indian Journal of Pediatrics*, 67 (1), 9–14. Available from: http://link.springer.com/article/10.1007/BF02802625 [Accessed 12 February 2015].
- Nepal Law Commission, 2007. *The Interim Constitution of Nepal (2063) 2007.* Kathmandu: Nepal Law Commission, Government of Nepal. Available from: http://www.wipo.int/edocs/lexdocs/laws/en/np/np006en.pdf [Accessed 25 February 2017].
- Neupane, S. P., 2015. Immediate lessons from the Nepal earthquake. *The Lancet*, 385 (9982), 2041–2042.
- Nour, N. M., 2008. An introduction to maternal mortality. *Reviews in Obstetrics and Gynecology*, 1 (2), 77–81.
- Nutbeam, D., 1998. Health Promotion Glossary. *Health Promotion International*, 13 (4), 349–364. Available from: http://heapro.oxfordjournals.org/content/13/4/349.short [Accessed 11 February 2015].

- Nutbeam, D., 2006. Using theory to guide changing individual behaviour. *In*: Davies, M. and McDowall, W., eds. *Health Promotion Theory*. Open University Press. Available from: http://www.sanjeshp.ir/phd/phd_91/Pages/Refrences/health education and promotion/[Maggie_Davies,_Wendy_Macdowall]_Health_Promotion_(BookFi.o r.pdf [Accessed 12 February 2015].
- Nutbeam, D. and Harris, E., 2004. *Theory in a Nutshell: A Practical Guide to Health Promotion Theories.* London, UK: McGraw-Hill.
- O'Connor-Fleming, M. L., Parker, E., Higgins, H., and Gould, T., 2006. A framework for evaluating health promotion programs. *Health Promotion Journal of Australia : Official Journal of Australian Association of Health Promotion Professionals*, 17 (1), 61–66.
- O'Rourke, K., Howard-Grabman, L., and Seoane, G., 1998. Impact of community organization of women on perinatal outcomes in rural Bolivia. *Revista panamericana de salud publica = Pan American Journal of Public Health*, 3 (1), 9–14. Available from: http://www.scielosp.org/scielo.php?pid=S1020-49891998000100002&script=sci_arttext [Accessed 12 February 2015].
- Oanda, 2015. Oanda Exchange rates. Oanda: London, UK. Available from: https://www.oanda.com/ [Accessed 12 October 2015].
- Orb, A., Eisenhauer, L. and Wynaden, D., 2001. Ethics in qualitative research. *Journal of Nursing Scholarship*, 33 (1), 93-96. Available from: http://onlinelibrary.wiley.com/doi/10.1111/j.1547-5069.2001.00093.x/epdf [Accessed 20 December 2016].
- Ory, M. G., Lee Smith, M., Mier, N., and Wernicke, M. M., 2010. The science of sustaining health behavior change: the health maintenance consortium. *American Journal of Health Behavior*, 34 (6), 647–59. Available from: http://www.ingentaconnect.com/content/png/ajhb/2010/00000034/0000006/art 00002 [Accessed 8 December 2015].
- Osrin, D., Mesko, N., Shrestha, B.P., Shrestha, D., Tamang, S., Thapa, S., Tumbahangphe, K.M., Shrestha, J.R., Manandhar, M.K., Manandhar, D.S. and Standing, H., 2003. Implementing a community-based participatory intervention to improve essential newborn care in rural Nepal. *Transactions of the Royal Society of Tropical Medicine and Hygiene*, *97* (1), 18-21. Available from: http://trstmh.oxfordjournals.org/content/97/1/18.short [Accessed 8 December 2015].
- Osrin, D., Azad, K., Fernandez, A., Manandhar, D. S., Mwansambo, C. W., Tripathy, P., and Costello, A. M., 2009. Ethical challenges in cluster randomized controlled trials: experiences from public health interventions in Africa and Asia. *Bulletin of the World Health Organization*, 87 (10), 772–779. Available from: http://www.scielosp.org/scielo.php?pid=S0042-96862009001000013&script=sci_arttext&tlng=pt [Accessed 8 December 2015].
- Parker, E. A., Eng, E., Laraia, B., Ammerman, A., Dodds, J., Margolis, L., and Cross, A., 1998. Coalition building for prevention: lessons learned from the North Carolina Community-Based Public Health Initiative. *Journal of Public Health Management and Practice*, 4 (2), 25–36.

- Pasick, R. J., Burke, N. J., Barker, J. C., Joseph, G., Bird, J. A., Otero-Sabogal, R., Tuason, N., Stewart, S. L., Rakowski, W., Clark, M. A., Washington, P. K., and Guerra, C., 2009. Behavioral Theory in a Diverse Society: Like a Compass on Mars. *Health Education & Behavior : the Official Publication of the Society for Public Health Education*, 36 (5 Suppl), 11S–35S.
- Patton, M. Q., 1996. *Utilization-Focused Evaluation: The New Century Text.* London, UK: SAGE Publications.
- Patton, M. Q., 2002. *Qualitative Evaluation and Research Methods*. London, UK: SAGE Publications.
- Pawson, R. and Tilley, N., 2004. *Realist evaluation*. London: Monograph prepared for British Cabinet Office. Available from: http://www.communitymatters.com.au/RE_chapter.pdf [Accessed 22 January 2017].
- Pell, C., Meñaca, A., Were, F., Afrah, N. A., Chatio, S., Manda-Taylor, L., Hamel, M. J., Hodgson, A., Tagbor, H., Kalilani, L., Ouma, P., and Pool, R., 2013. Factors affecting antenatal care attendance: results from qualitative studies in Ghana, Kenya and Malawi. *PloS One*, 8 (1), e53747. Available from: http://journals.plos.org/plosone/article?id=10.1371/journal.pone.0053747 [Accessed 27 October 2015].
- Perkins, E. R., Simnett, I., Wright, L., and Eds., 1999. Evidence based health promotion. *Health Education*, 99 (4), 46. Available from: http://dx.doi.org/10.1108/he.1999.99.4.46.2 [Accessed 8 June 2015].
- Pitchforth, E. and van Teijlingen, E., 2005. International public health research involving interpreters: a case study from Bangladesh. *BMC Public Health*, 5, 71 Available from: http://bmcpublichealth.biomedcentral.com/articles/10.1186/1471-2458-5-71 [Accessed 15 June 2015].
- Pitchforth, E., van Teijlingen, E., Graham, W., and Fitzmaurice, A., 2007. Development of a proxy wealth index for women utilizing emergency obstetric care in Bangladesh. *Health Policy and Planning*, 22 (5), 311–319. Available from: http://heapol.oxfordjournals.org/content/22/5/311.abstract [Accessed 28 June 2015].
- Pitchforth, E., van Teijlingen, E., Watson, V., Tucker, J., Kiger, A., Ireland, J., Farmer, J., Rennie, A.-M., Gibb, S., Thomson, E., and Ryan, M., 2009. 'Choice' and place of delivery: a qualitative study of women in remote and rural Scotland. Quality & Safety in Health Care, 18 (1), 42–8. Available from: http://qualitysafety.bmj.com/content/18/1/42.abstract [Accessed 21 December 2015].
- Pitchforth, E., Watson, V., Tucker, J., Ryan, M., van Teijlingen, E., Farmer, J., Ireland, J., Thomson, E., Kiger, A., and Bryers, H., 2008. Models of intrapartum care and women's trade-offs in remote and rural Scotland: A mixed-methods study. *BJOG: An International Journal of Obstetrics and Gynaecology*, 115, 560–569.
- Plano-Clark, V. L. and Creswell, J. W., 2010. *Understanding research: a consumer's guide*. Boston, MA. Available from: https://csuglobal.blackboard.com/bbcswebdav/institution/FCC

Content/csfiles/home_dir/externalFiles_20130401041211/library__xid-1005_5/Textbook Reserve__xid-13309_5/OTL__xid-14610_5/OTL568__xid-21568_5/OTL568_Mod1Reading__xid-21569_5-3.pdf [Accessed 9 June 2015].

- Pope, C., Ziebland, S., and Mays, N., 2000. Qualitative research in health care. Analysing qualitative data. *BMJ*, 320, 114–116.
- Povlsen, L. and Borup, I. K., 2011. Holism in nursing and health promotion: distinct or related perspectives?--A literature review. *Scandinavian Journal of Caring Sciences*, 25 (4), 798–805. Available from: http://www.ncbi.nlm.nih.gov/pubmed/21443544 [Accessed 17 Nov 2015].
- Poudyal, A.K., Jimba, M., Murakami, I., Silwal, R.C., Wakai, S. and Kuratsuji, T., 2003. A traditional healers' training model in rural Nepal: strengthening their roles in community health. *Tropical Medicine & International Health*, 8(10), 956-960. Available from: http://onlinelibrary.wiley.com/doi/10.1046/j.1365-3156.2003.01094.x/full [Accessed 24 January 2013].
- Powell-Jackson, T. and Hoque, M. E., 2012. Economic consequences of maternal illness in rural Bangladesh. *Health Economics*, 21 (7), 796–810. Available from: http://dx.doi.org/10.1002/hec.1749 [Accessed 4 April 2015].
- Pradhan, A., Aryal, R.H., Regmi, G., Ban, B. and Govindasamy, P., 1997. *Nepal Family Health Survey 1996*. Kathmandu: Department of Health Services, Family Health Division, Available from: http://dhsprogram.com/pubs/pdf/fr78/fr78.pdf [Accessed 24 January 2017].
- Pronyk, P. M., Muniz, M., Nemser, B., Somers, M.-A., McClellan, L., Palm, C. A., Huynh, U. K., Amor, Y. Ben, Begashaw, B., McArthur, J. W., Niang, A., Sachs, S. E., Singh, P., Teklehaimanot, A., and Sachs, J. D., 2012. The effect of an integrated multisector model for achieving the Millennium Development Goals and improving child survival in rural sub-Saharan Africa: a non-randomised controlled assessment. *The Lancet*, 379 (9832), 2179–2188.
- Prost, A., Colbourn, T., Seward, N., Azad, K., Coomarasamy, A., Copas, A., Houweling, T. A. J., Fottrell, E., Kuddus, A., Lewycka, S., MacArthur, C., Manandhar, D., Morrison, J., Mwansambo, C., Nair, N., Nambiar, B., Osrin, D., Pagel, C., Phiri, T., Pulkki-Brännström, A.-M., Rosato, M., Skordis-Worrall, J., Saville, N., More, N. S., Shrestha, B., Tripathy, P., Wilson, A., and Costello, A., 2013. Women's groups practising participatory learning and action to improve maternal and newborn health in low-resource settings: a systematic review and meta-analysis. *The Lancet*, 381 (9879), 1736–46. Available from: http://www.ncbi.nlm.nih.gov/pubmed/23683640 [Accessed 14 November 2015].
- Puri, M., Tamang, J., and Shah, I., 2011. Suffering in silence: consequences of sexual violence within marriage among young women in Nepal. *BMC Public Health*, 11 (1), 29. Available from: http://www.biomedcentral.com/1471-2458/11/29 [Accessed 8 May 2015].
- Rauniar, G., Das, B., Manandhar, T., and Bhattacharya, S., 2012. Effectiveness of an educational feedback intervention on drug prescribing in dental practice. *Kathmandu University Medical Journal*, 10 (40), 30 – 5. Available from: http://www.nepjol.info/index.php/KUMJ/article/view/10991 [Accessed 2 June 2016].

Reeve, J. and Peerbhoy, D., 2007. Evaluating the evaluation: Understanding the

utility and limitations of evaluation as a tool for organizational learning. *Health Education Journal*, 66 (2), 120–131. Available from: http://hej.sagepub.com/content/66/2/120.abstract [Accessed 4 June 2016].

Reveiz, L., Gaitán, H. G., and Cuervo, L. G., 2007. Enemas during labour. *The Cochrane Database of Systematic Reviews*, (4), CD000330. Available from: http://onlinelibrary.wiley.com/doi/10.1002/14651858.CD000330.pub2/full [Accessed 5 November 2015].

Reynolds, H. W., Wong, E. L., and Tucker, H., 2006. Adolescents use of maternal and child health services in developing countries. *International Family Planning Perspectives*, 32 (1), 6 – 16. Available from: http://www.guttmacher.org/pubs/journals/3200606.html [Accessed 8 July 2015].

Rogers, E. M., 2002. Diffusion of preventive innovations. *Addictive Behaviors*, 27 (6), 989–993. Available from: http://www.sciencedirect.com/science/article/pii/S0306460302003003 [Accessed 4 March 2015].

- Rogo, K., Oucho, J., and Mwalali, P., 2006. Maternal Mortality. *In*: Jamison, D., Feachem, R., Makgoba, M., and et al., eds. *Disease and Mortality in Sub-Saharan Africa*. Washington DC: World Bank. Available from: https://openknowledge.worldbank.org/bitstream/handle/10986/7050/363050Dis ease01010FFICIAL0USE0ONLY1.pdf?sequence=1 [Accessed 12 July 2015].
- Ronsmans, C., 2009. Severe acute maternal morbidity in low-income countries. Best practice & research. Clinical Obstetrics & Gynaecology, 23 (3), 305–16. Available from: http://www.ncbi.nlm.nih.gov/pubmed/19201657 [Accessed 22 December 2015].
- Ronsmans, C., and Graham, W. J., 2006. Maternal mortality: who, when, where, and why. *The Lancet*, 368(9542), 1189–1200.

Rootman, I., Goodstadt, M., Hyndman, R., McQueen, D., Potvin, L., Springett, J., and E Ziglio, 2001. *Evaluation in health promotion: principles and perspectives*. 92nd ed. World Health Organization Regional Publications. European Series. Available from: http://www.euro.who.int/__data/assets/pdf_file/0007/108934/E73455.pdf [Accessed 13 January 2015].

- Rosato, M., Laverack, G., Grabman, L. H., Tripathy, P., Nair, N., Mwansambo, C., Azad, K., Morrison, J., Bhutta, Z., Perry, H., Rifkin, S., and Costello, A., 2008. Community participation: lessons for maternal, newborn, and child health. *The Lancet*, 372, 962–971. Available from: http://www.ncbi.nlm.nih.gov/pubmed/18790319\nhttp://journals.ohiolink.edu/ejc /pdf.cgi/Rosato_Mikey.pdf?issn=01406736&issue=v372i9642&article=962_cplf mnach.
- Rosato, M., Mwansambo, C. W., Kazembe, P. N., Phiri, T., Soko, Q. S., Lewycka, S., Kunyenge, B. E., Vergnano, S., Osrin, D., Newell, M. L., and Costello, A. M., 2006. Women's groups' perceptions of maternal health issues in rural Malawi. *The Lancet*, 368, 1180–1188.
- Rosen, L., Manor, O., Engelhard, D., and Zucker, D., 2006. In defense of the randomized controlled trial for health promotion research. *American Journal of*

Public Health, 96 (7), 1181–6. Available from:

http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=1483860&tool=pmce ntrez&rendertype=abstract [Accessed 21 December 2015].

- Rosen, H.E., Lynam, P.F., Carr, C., Reis, V., Ricca, J., Bazant, E.S. and Bartlett, L.A. 2015. Direct observation of respectful maternity care in five countries: a cross-sectional study of health facilities in East and Southern Africa. *BMC Pregnancy and Childbirth*, 15(1), 1.
- Rowe, R. E. and Garcia, J., 2003. Social class, ethnicity and attendance for antenatal care in the United Kingdom: a systematic review. *Journal of Public Health Medicine*, 25 (2), 113–9. Available from: http://www.ncbi.nlm.nih.gov/pubmed/12848399 [Accessed 21 December 2015].
- Ryan, P., 2009. Integrated Theory of Health Behavior Change: background and intervention development. *Clinical Nurse Specialist* (CNS), 23 (3), 161–70.. Available from: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2778019&tool=pmce ntrez&rendertype=abstract [Accessed 5 August 2015].
- Ryan, M., Scott, D. A., Reeves, C., Bate, A., van Teijlingen, E. R., Russell, E. M., Napper, M., and Robb, C. M., 2001. Eliciting public preferences for healthcare: a systematic review of techniques. *Health Technology Assessment*, 5 (5), 1– 186.
- Rychetnik, L., 2002. Criteria for evaluating evidence on public health interventions. Journal of Epidemiology & Community Health, 56 (2), 119–127. Available from: http://jech.bmj.com/content/56/2/119.abstract [Accessed 10 January 2015].
- Sandall, J., 2012. Every Woman Needs a Midwife, and Some Women Need a Doctor Too. *Birth*, 39 (4), 323–326. Available from: http://dx.doi.org/10.1111/birt.12010 [Accessed 2 November 2015].
- Sandelowski, M., 1995. Sample size in qualitative research. *Research in Nursing & Health*, 18, 179–183. Available from: http://dx.doi.org/10.1002/nur.4770180211 [Accessed 27 August 2015].
- Sandelowski, M., 2000. Combining Qualitative and Quantitative Sampling, Data Collection, and Analysis Techniques in Mixed-Method Studies. *Research in Nursing & Health*, 23 (3), 246–255. Available from: http://dx.doi.org/10.1002/1098-240X(200006)23:3<246::AID-NUR9>3.0.CO;2-H [Accessed 27 August 2015].
- Sanford, N., 1981. A model for action research. *In: Human inquiry: a sourcebook of new paradigm research*. London, UK: J. Wiley.
- Saunders, R. P., Evans, M. H., and Joshi, P., 2005. Developing a Process-Evaluation Plan for Assessing Health Promotion Program Implementation: A How-To Guide. *Health Promotion Practice*, 6 (2), 134–147. Available from: http://hpp.sagepub.com/content/6/2/134.abstract [Accessed 4 November 2015].
- Scriven, M., 2008. A Summative Evaluation of RCT Methodology: & An Alternative Approach to Causal Research. *Journal of MultiDisciplinary Evaluation*, 5 (9), 11–24. Available from: http://eric.ed.gov/?id=EJ800127 [Accessed 21

December 2015].

- Sein, K. K., 2013. Beliefs and practices surrounding postpartum period among Myanmar women. *Midwifery*, 29, 1257–1263.
- Selby, J. V., 2010. Health systems, patients factors, and quality of care for diabetes: A synthesis of findings from the TRIAD study. *Diabetes Care*, 33 (4), 940–947.
- Sharma, D. C., 2015. Nepal earthquake exposes gaps in disaster preparedness. *The Lancet*, 385 (9980), 1819–1820.
- Sharma, S., Joshi, P., and van Teijlingen, E., 2015. 'Nepenglish' or 'Nepali English': A New Version of English? *Asian Journal of Social Sciences & Humanities*, 4(2), 188-193. Available from: http://eprints.bournemouth.ac.uk/21967/1/AJSSH2015 Nepenglish 2015.pdf [Accessed 22 Dec 2015].
- Sharma, S., van Teijlingen, E., Belizán, J.M., Hundley, V., Simkhada, P., Sicuri, E. 2016a. Measuring What Works: Impact evaluation of women's groups on maternal health uptake in rural Nepal, *PLOS One* 11(5): e0155144. Available from:

http://journals.plos.org/plosone/article?id=10.1371%2Fjournal.pone.0155144 [Accessed 25 January 2017].

- Sharma, S., van Teijlingen, E., Hundley, V., Angell, C. and Simkhada, P., 2016b. Dirty and 40 days in the wilderness: Eliciting childbirth and postnatal cultural practices and beliefs in Nepal. *BMC Pregnancy and Childbirth*, 16(1), 147. Available from: https://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/s12884-016-0938-4 [Accessed 25 January 2017].
- Sharma, S., Simkhada, P., van Teijlingen, E., Stephens J, Hundley, V., Angell, C., 2017. Evaluation of a Maternity Care Intervention in Rural Nepal: Lessons learnt, *Journal of Asian Midwives* [accepted March 2017].
- Shaw, I.F., 2003. Ethics in qualitative research and evaluation. *Journal of Social Work*, 3(1), 9-29.
- Shrestha, S., Bell, J.S. and Marais, D., 2014. An analysis of factors linked to the decline in maternal mortality in Nepal. *PloS one*, *9*(4), p.e93029. Available from: http://journals.plos.org/plosone/article?id=10.1371/journal.pone.0093029 [Accessed 24 January 2017].
- Shrestha, B., Bhandari, B., Manandhar, D., Osrin, D., Costello, A., and Saville, N., 2011. Community interventions to reduce child mortality in Dhanusha, Nepal: study protocol for a cluster randomized controlled trial. *Trials*, 12 (1), 136. Available from: http://www.trialsjournal.com/content/12/1/136 [Accessed 8 April 2015].
- Simkhada, B., Porter, M., and van Teijlingen, E., 2010. The role of mothers-in-law in antenatal care decision-making in Nepal: a qualitative study. *BMC Pregnancy and Childbirth*, 10 (1), 34. Available from: http://www.biomedcentral.com/1471-2393/10/34 [Accessed 7 May 2015].

Simkhada, B., van Teijlingen, E., Porter, M., and Simkhada, P., 2006. Major

problems and key issues in maternal health in Nepal. *Kathmandu University Medical Journal*, 4 (2), 258–63. Available from: http://eprints.bournemouth.ac.uk/10148/1/Bibha_Simkhada_review_KUMJ_200 6.pdf [Accessed 7 May 2015].

- Simkhada, P., van Teijlingen, E., Kadel, S., Stephens, J., Sharma, S., and Sharma, M., 2009. Reliability of national data sets: evidence from a detailed small area study in rural Kathmandu Valley, Nepal. Asian Journal of Epidemiology, 2, 44 – 48. Available from: http://scialert.net/qredirect.php?doi=aje.2009.44.48&linkid=cite [Accessed 9 May 2015].
- Simkhada, B., Teijlingen, E. R. van, Porter, M., and Simkhada, P., 2008. Factors affecting the utilization of antenatal care in developing countries: systematic review of the literature. *Journal of Advanced Nursing*, 61 (3), 244–260. Available from: http://dx.doi.org/10.1111/j.1365-2648.2007.04532.x [Accessed 8 May 2015].
- Smith, R. D. and Petticrew, M., 2010. Public health evaluation in the twenty-first century: time to see the wood as well as the trees. *Journal of Public Health*, 32 (1), 2–7. Available from: http://jpubhealth.oxfordjournals.org/content/32/1/2.abstract [Accessed 29 April 2015].
- van Spall, H.G., Toren, A., Kiss, A. and Fowler, R.A., 2007. Eligibility criteria of randomized controlled trials published in high-impact general medical journals: a systematic sampling review. *JAMA*, 297(11), 1233-1240.
- Speller, V., Learmonth, A., and Harrison, D., 1998. Evaluating health promotion is complex. *BMJ*, 316 (7142), 1463–1463. Available from: http://www.bmj.com/content/316/7142/1463.1?hwshib2=authn%3A1447353592 %3A20151111%253A89c7606b-5835-4b8b-91cf-98a99a3ec0ad%3A0%3A0%3A0%3AKvTjlo3kEhdRfwA38g0MUg%3D%3D [Accessed 11 November 2015].
- Sreeramareddy, C.T., Hari S.J., Sreekumaran B.V., Giri S, and Chuni N., 2011. Home delivery and newborn care practices among urban women in western Nepal: a questionnaire survey. *BMC Pregnancy Childbirth*, (6), 27. Available from: http://bmcpregnancychildbirth.biomedcentral.com/articles/10.1186/1471-2393-6-27 [Accessed 11 February 2015].
- Sreeramareddy, C. T., Shidhaye, R. R., and Sathiakumar, N., 2011. Association between biomass fuel use and maternal report of child size at birth--an analysis of 2005-06 India Demographic Health Survey data. *BMC Public Health*, 11, 403. Available from: http://bmcpublichealth.biomedcentral.com/articles/10.1186/1471-2458-11-403 [Accessed 13 February 2015].
- Stange, K. C., Crabtree, B. F., and Miller, W. L., 2006. Publishing Multimethod Research. *Annals of Family Medicine*, 4 (4), 292–294.
- Starrs, A. M., 2006. Safe motherhood initiative: 20 years and counting. *The Lancet*, 368 (9542), 1130–1132. Available from: http://linkinghub.elsevier.com/retrieve/pii/S0140673606693859 [Accessed 11 May 2015].

- Steckler, A. and Linnan, L., 2002. *Process Evaluation for Public Health Interventions and Research*. London, UK: Wiley.
- Steckler, A. and McLeroy, K. R., 2008. The importance of external validity. *American Journal of Public Health*, 98 (1), 9–10. Available from: http://ajph.aphapublications.org/doi/abs/10.2105/AJPH.2007.126847 [Accessed 22 November 2015].
- Sternberg, P. and Hubley, J., 2004. Evaluating men's involvement as a strategy in sexual and reproductive health promotion. *Health Promotion International*, 19 (3), 389–96. Available from: http://heapro.oxfordjournals.org/content/19/3/389.abstract [Accessed 25 February 2015].
- Stevens, P., 2004. *Diseases of Poverty of poverty and the 10/90 Gap.* Geneva: World Health Organization. Available from: http://www.who.int/intellectualproperty/submissions/InternationalPolicyNetwork. pdf [Accessed 13 December 2015].
- Stokols D, 1996. Translating social ecological theory into guidelines for community. *American Journal of Health Promotion*, (10), 282–298.
- Suchman, E., 1968. *Evaluative Research: Principles and Practice in Public Service and Social Action Programs.* USA: Russell Sage Foundation.
- Suvedi, Krishna, B., Pradhan, A., Barnett, S., Puri, M., Chitrakar, S. R., Poudel, P., Sharma, S., and Hulton, L., 2009. Nepal Maternal Mortality and Morbidity Study 2008/2009: Summary of Preliminary Findings. Kathmandu, Nepal. Available from: http://www.dpiap.org/resources/pdf/nepal_maternal_mortality_2011_04_22.pdf [Accessed 3 July 2015].
- Syed, U., Khadka, N., Khan, A. and Wall, S., 2008. Care-seeking practices in South Asia: using formative research to design program interventions to save newborn lives. *Journal of Perinatology*, *28*, S9-S13. Available from: http://www.nature.com/jp/journal/v28/n2s/full/jp2008165a.html [Accessed 25 January 2017].
- STATA 2017. *Reading and Using STATA Output.* MIT: STATA. Available from: http://web.mit.edu/course/17/17.846/OldFiles/www/Readout.html [Accessed 1 February 2017.
- Tashakkori, A. and Creswell, J. W., 2007. Editorial: The New Era of Mixed Methods. *Journal of Mixed Methods Research*,1(1), 3-7.
- Tashakkori, A. and Teddlie, C., 1998. *Mixed methodology: Combining qualitative and quantitative approaches (Vol. 46).* London, UK: SAGE Publications.
- Tashakkori, A. and Teddlie, C., 2010. Sage handbook of mixed methods in social & behavioral research. London, UK: SAGE Publications.
- Teddlie, C. and Yu, F., 2007. Mixed methods sampling a typology with examples. *Journal of Mixed Methods Research*, 1(1), 77-100.

van Teijlingen, E., 2005. A Critical Analysis of the Medical Model as used in the

Study of Pregnancy and Childbirth. *Sociological Research Online*, *10*(2). Available from: http://www.socresonline.org.uk/10/2/teijlingen.html [Accessed 5 December 2015].

- van Teijlingen, E. R. and Hundley, V., 2001. The importance of pilot studies. *Social Research Update*, 35. Available from: http://aura.abdn.ac.uk/bitstream/2164/157/1/SRU35 pilot studies.pdf [Accessed 19 July 2015].
- van Teijlingen, E. and Pitchforth, E., 2006. Focus group research in family planning and reproductive health care. *Journal of Family Planning and Reproductive Health Care*, 32 (1), 30–2.
- van Teijlingen, E., Simkhada, P., and Stephens, J., 2013. Doing Focus Groups in the Health Field: Some Lessons from Nepal. *Health Prospect.* Available from: http://www.nepjol.info/index.php/HPROSPECT/article/view/8722 [Accessed 20 December 2015].
- van Teijlingen, E., Simkhada, P., Stephens, J., Simkhada, B., Rogers, S., and Sharma, S., 2012. Making the best use of all resources: developing a health promotion intervention in rural Nepal. *Health Renaissance*, 10 (3), 229–235. Available from: http://www.nepjol.info/index.php/HREN/article/view/7141 [Accessed 9 January 2015].
- van Teijlingen, E, Simkhada, P., Wasti, P.P. 2015. Nepal is Changing: Modernisation and Diversity in Healthcare. In: *Wasti, S.P., Simkhada, P.P.* & *van Teijlingen, E. (Eds.) The Dynamics of Health in Nepal*, Kathmandu, Nepal: Social Science Baha & Himal Books: 1-15.
- Tengland, P., 2012. Behaviour Change or Empowerment: On the Ethics of Health-Promotion Strategies. *Public Health Ethics*, (5), 140–153.
- Thaddeus, S. and Mainde, D., 1994. Too far to walk: maternal mortality in context. Social Science & Medicine, 38 (8), 1091 – 110.
- Thomas, J. and Harden, A. 2008. Methods for the thematic synthesis of qualitative research in systematic reviews. *BMC Medical Research Methodology*, 8 (1), 1. Available from: http://eprints.ncrm.ac.uk/468/1/1007_JTAHthematic_synthesis.pdf [Accessed 6 December 2016].
- Thompson, B.T. and Schoenfeld, D., 2007. Usual care as the control group in clinical trials of nonpharmacologic interventions. *Proceedings of the American Thoracic Society*, *4*(7), 577-582. Available from: http://www.atsjournals.org/doi/abs/10.1513/pats.200706-072JK [Accessed 27 January 2017].
- Thomson, G., Dykes, F., Singh, G., Cawley, L., and Dey, P., 2013. A public health perspective of women's experiences of antenatal care: an exploration of insights from a community consultation. *Midwifery*, 29 (3), 211–6. Available from: http://www.midwiferyjournal.com/article/S0266613812000034/fulltext [Accessed 21 December 2015].
- Thrall, J. H., 2011. Unintended consequences of health care legislation. *Journal of the American College of Radiology*, 8 (10), 687–91. Available from:

http://www.jacr.org/article/S1546-1440(11)00332-2/abstract [Accessed 5 December 2015].

- Thwala, S. B. P., Holroyd, E., and Jones, L. K., 2012. Health belief dualism in the postnatal practices of rural Swazi women: An ethnographic account. *Women and Birth*, 25.
- Tollefson, J., 2015. Can randomized trials eliminate global poverty? *Nature*, 524 (7564), 150–3. Available from: http://www.nature.com/news/can-randomized-trials-eliminate-global-poverty-1.18176 [Accessed 21 December 2015].
- Tones, K., 2000. Evaluating health promotion: a tale of three errors. *Patient Education and Counseling*, 39 (2–3), 227–236. Available from: http://www.sciencedirect.com/science/article/pii/S073839919900035X. [Accessed 2 March 2017].
- Tones, K. and Tilford, S., 2001. *Health Promotion: Effectiveness, Efficiency and Equity*. Cheltenham, UK: Nelson Thornes.
- Tracy, S. K. and Tracy, M. B., 2003. Costing the cascade: estimating the cost of increased obstetric intervention in childbirth using population data. *BJOG : an international journal of Obstetrics and Gynaecology*, 110 (8), 717–24. Available from: http://onlinelibrary.wiley.com/doi/10.1111/j.1471-0528.2003.02045.x/full [Accessed 17 November 2015].
- Trinh, L. T. T., Dibley, M. J., and Byles, J., 2007. Determinants of Antenatal Care Utilization in Three Rural Areas of Vietnam. *Public Health Nursing*, 24 (4), 300– 310. Available from: http://onlinelibrary.wiley.com/doi/10.1111/j.1525-1446.2007.00638.x/full [Accessed 21 Dec 2015].
- Tripathy, P., Nair, N., Barnett, S., Mahapatra, R., Borghi, J., Rath, S., Rath, S., Gope, R., Mahto, D., Sinha, R., Lakshminarayana, R., Patel, V., Pagel, C., Prost, A., and Costello, A., 2010. Effect of a participatory intervention with women's groups on birth outcomes and maternal depression in Jharkhand and Orissa, India: a cluster-randomised controlled trial. *The Lancet*, 375 (9721), 1182–92.
- Tuladhar, H. and Dhakal, N., 2012. Impact of Antenatal Care on Maternal and Perinatal utcome: A Study at Nepal Medical College Teaching Hospital. Nepal Journal of Obstetrics and Gynaecology; Vol 6, No 2 (2011). Available from: http://www.nepjol.info/index.php/NJOG/article/view/6755 [Accessed 12 April 2015].
- Tuntiseranee, P., Olsen, J., Chongsuvivatwong, V., and Limbutara, S., 1999. Socioeconomic and work related determinants of pregnancy outcome in southern Thailand. *Journal of Epidemiology and Community Health*, 53 (10), 624–629.
- Ulin, P., Robinson, E., and Tolley, E., 2012. Qualitative methods in public health: a field guide for applied research. United Nations (UN), 2015. The Millennium Development Goals Report 2015. Available from: http://www.un.org/millenniumgoals/2015_MDG_Report/pdf/MDG 2015 rev %28July 1%29.pdf [Accessed 7 April 2014].

United Nations Development Fund (UNDP), 2014. Nepal Millennium Development

Goals Progress Report 2013. Kathmandu: Government of Nepal, National Planning Commission / United Nations Country Team of Nepal. Available from: http://www.np.undp.org/content/dam/nepal/docs/reports/millennium%20develo pment%20goals/UNDP_NP_MDG_Report_2013.pdf [Accessed 27 January 2017].

- United Nations Development Fund (UNDP), 2014. World Urbanization Prospects. New York: United Nations. Available from: https://esa.un.org/unpd/wup/publications/files/wup2014-highlights.Pdf [Accessed 24 January 2017].
- United Nations Development Fund (UNDP), 2015. Table 1: Human Development Index and its components. *Human Development Report 2015: Work for Human Development.* New York: UNDP. Available from: http://hdr.undp.org/en/composite/HDI [Accessed 24 January 2017].
- United Nations Population Fund (UNFPA), 2006. *Maternal and neonatal health in east and south-east asia*. Bangkok, Thailand: UNFPA. Available from: http://www.unfpa.org/sites/default/files/pub-pdf/MaternalHealth_Asia.pdf [Accessed 1 May 2014].
- United Nations Population Fund (UNFPA), 2014. The state of the world's midwifery 2014: A universal pathway. A women's right to health. New York, USA: UNFPA. Available from: www.unfpa.org/sowmy [Accessed 19 December 2016].
- United Nations (UN), 2011a. *United Nations Millennium Development Goals*. UN Millennium Development Goals. New York, USA: UN. Available from: http://www.un.org/millenniumgoals/poverty.shtml [Accessed 11 April 2014].
- United Nations (UN), 2011b. *The Millennium Development Goals Report 2011*. New York, USA: UN. Available from: http://www.un.org/millenniumgoals/pdf/(2011_E) MDG Report 2011_Book LR.pdf [Accessed 11 April 2014].
- United Nations (UN), 2016. Provisional proposed tiers for global SDG indicators. (March 24, 2016). New York, U.S.A: United Nations Statistics Division. Available from: http://unstats.un.org/sdgs/files/meetings/iaeg-sdgs-meeting-03/Provisional-Proposed-Tiers-for-SDG-Indicators-24-03-16.pdf [Accessed 24 January 2017].
- United Nations General Assembly (UNGA), 2000. *Resolution 55/2 United Nations Millennium Declaration*. New York, USA: UN. Available from: http://www.un.org/millennium/declaration/ares552e.pdf [Accessed 11 April 2014].
- United Nations University World Institute for Development (UNU-WIDER), 2005. World Income Inequality Database (V 2.0a). World Institute for Development Economics Research, United Nations University. Available from: https://www.wider.unu.edu/project/wiid-world-income-inequality-database [Accessed 24 January 2017].
- US Department of Health and Human services, 2002. *Physical activity evaluation handbook.* Atlanta, GA: US Department of Health and Human Services, Centers for Disease Control and Prevention; 2002. Available from:

https://www.cdc.gov/nccdphp/dnpa/physical/handbook/pdf/handbook.pdf [Accessed 19 December 2016].

- Vanderweele, T.J., Hong, G., Jones, S.M., Brown, J.L., 2013. Mediation and spillover effects in group-randomized trials: A case study of the 4Rs educational intervention. *Journal of the American Statistical Association.* 108, 469–482.
- Varkey, P., Mbbs, Kureshi, S., and Lesnick, T., 2010. Empowerment of women and its association with the health of the community. *Journal of Women's Health* (2002), 19 (1), 71–6.
- Verheijen, E. C., Raven, J. H., and Hofmeyr, G. J., 2009. Fundal pressure during the second stage of labour. *The Cochrane Database of Systematic Reviews*, (4), CD006067. Available from: http://onlinelibrary.wiley.com/doi/10.1002/14651858.CD006067.pub2/full [Accessed 19 December 2016].
- Victora, C. G., Aquino, E. M. L., do Carmo Leal, M., Monteiro, C. A., Barros, F. C., and Szwarcwald, C. L., 2011. Maternal and child health in Brazil: progress and challenges. *The Lancet*, 377 (9780), 1863–76.
- Vijeyarasa, R., 2009. Putting Reproductive Rights on the Transitional Justice Agenda: The Need to Redress Violations and Incorporate Reproductive Health Reforms in Post-Conflict Development. New England Journal of International and Comparative Law, 15, 41. Available from: http://heinonline.org/HOL/LandingPage?handle=hein.journals/newenjic15&div= 4&id=&page= [Accessed 24 February 2017].
- Vogel, I., 2012. Review of the use of 'Theory of Change'in international development. London, UK: Department for International Development (DFID). Available from: https://assets.publishing.service.gov.uk/media/57a08a5ded915d3cfd00071a/D FID_ToC_Review_VogelV7.pdf [Accessed 19 December 2016].
- Vyas, S. and Kumaranayake, L., 2006. Constructing socio-economic status indices: how to use principal components analysis. *Health Policy and Planning*, 21 (6), 459–468. Available from: http://heapol.oxfordjournals.org/content/21/6/459.abstract [Accessed 19 June 2015].
- Wakefield, M. A., Loken, B., and Hornik, R. C., 2010. Use of mass media campaigns to change health behaviour. *The Lancet*, 376 (9748), 1261–71.
- Waldinger, F., 2010. *Difference-in-Differences. Lecture 3.* Available from: http://www2.warwick.ac.uk/fac/soc/economics/staff/academic/waldinger/teachin g/ec9a8/slides/lecture_3_-_did.pdf [Accessed 28 June 2015].
- Walker, W., 2005. The strengths and weaknesses of research designs involving quantitative measures. *Journal of Research in Nursing*, *10*(5), 571-582.
 Available from: http://journals.sagepub.com/doi/pdf/10.1177/136140960501000505 [Accessed on 28 January 2017].
- Wallerstein, N., 1992. Powerlessness, Empowerment, and Health: Implications for Health Promotion Programs. *American Journal of Health Promotion*, 6 (3), 197–205. Available from: http://dx.doi.org/10.4278/0890-1171-6.3.197

[Accessed 21 April 2015].

- Warren C., Daly P., Toure L., Mongi P. Postnatal care. In: Lawn J. and Kerber K., (Eds.). 2006. Opportunities for Africa's Newborns. Cape Town, South Africa: Partnership for Maternal, Newborn and Child Health, 79–90. Available from: http://www.who.int/pmnch/media/publications/oanfullreport.pdf [Accessed 19 December 2016].
- Wasti, S.P., Simkhada, P.P. & van Teijlingen, E. (Eds.). 2015. *The Dynamics of Health in Nepal*. Kathmandu, Nepal: Social Science Baha & Himal Books. Available from: http://himalbooks.com/shop/products/The-Dynamics-of-Health-in-Nepal.html [Accessed 20 December 2016].
- Waters, E., Doyle, J., Jackson, N., Howes, F., Brunton, G., and Oakley, A., 2006.
 Evaluating the effectiveness of public health interventions: the role and activities of the Cochrane Collaboration. *Journal of Epidemiology and Community health*, 60 (4), 285–9. Available from: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2566164&tool=pmce ntrez&rendertype=abstract [Accessed 11 November 2015].
- Watkins, L. and Weeks, A. D., 2009. Providing information to pregnant women: how, what and where? *BJOG : an International Journal of Obstetrics and Gynaecology*, 116 (7), 877–9. Available from: http://www.ncbi.nlm.nih.gov/pubmed/19522792 [Accessed 5 December 2015].
- Webb, P. and Harinarayan, A., 1999. A Measure of Uncertainty: The Nature of Vulnerability and Its Relationship to Malnutrition. *Disasters*, 23 (4), 292–305.
- Wee, D.W., Mullany, L.C., Katz, J., Khatry, S.K., LeClerq, S.C. and Tielsch, J.M., 2010. Pregnancy-related mortality in southern Nepal between 2001 and 2006: independent estimates from a prospective, population-based cohort and a direct sisterhood survey. *American Journal of Epidemiology*, 172:855-60.
- West, K. P., Katz, J., Khatry, S. K., LeClerq, S. C., Pradhan, E. K., Shrestha, S. R., Connor, P. B., Dali, S. M., Christian, P., Pokhrel, R. P., and Sommer, A., 1999.
 Double blind, cluster randomised trial of low dose supplementation with vitamin A or beta carotene on mortality related to pregnancy in Nepal. The NNIPS-2 Study Group. *BMJ (Clinical research ed.)*, 318 (7183), 570–5. Available from: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=27760&tool=pmcent rez&rendertype=abstract [Accessed 3 December 2015].
- Westhrop, G., 2014. Realist impact evaluation: an introduction. *Working and discussion papers*. London, UK: Overseas Development Institute (ODI). Available from: https://www.odi.org/publications/8716-methods-lab-realist-impact-evaluation-introduction [Accessed 16 November 2015].
- White, H., 2006. Impact evaluation: the experience of the independent evaluation group of the World Bank, Washington D.C: World Bank. Available from: http://Inweb90.worldbank.org/oed/oeddoclib.nsf/DocUNIDViewForJavaSearch/ 35BC420995BF58F8852571E00068C6BD/\$file/impact_evaluation.pdf [Accessed 1 November 2015].
- White, H., 2009. Theory-based impact evaluation: principles and practice. *Journal of Development Effectiveness*, 1 (3), 271–284. Available from: http://dx.doi.org/10.1080/19439340903114628 [Accessed 2 November 2015].

- White, H., 2013. An introduction to the use of randomised control trials to evaluate development interventions. *Journal of Development Effectiveness*, 5 (1), 30–49. Available from: http://dx.doi.org/10.1080/19439342.2013.764652 [Accessed 18 September 2015].
- Whitehead, M., 1991. The concepts and principles of equity and health. *Health Promotion International*, *6*(3), 217-228. Available from: https://academic.oup.com/heapro/article/6/3/217/742216/The-concepts-and-principles-of-equity-and-health [Accessed 27 January 2017].
- Whitehead, D., 2003. Health promotion and health education viewed as symbiotic paradigms: bridging the theory and practice gap between them. *Journal of Clinical Nursing*, 12 (6), 796–805. Available from: http://dx.doi.org/10.1046/j.1365-2702.2003.00804.x [Accessed 16 October 2015].
- Whitehead, D., 2004. Health promotion and health education: advancing the concepts. *Journal of Advanced Nursing*, 47 (3), 311–320. Available from: http://www.ncbi.nlm.nih.gov/pubmed/15238126 [Accessed 18 Oct 2015].

White Ribbon Alliance (WRA)., 2011. *Respectful maternity care charter.* Washington, DC: White Ribbon Alliance. Available from: http://whiteribbonalliance.org/wpcontent/uploads/2013/10/Final_RMC_Charter.pdf [Accessed 13 December 2016].

Witter, S., Khadka, S., Nath, H., and Tiwari, S., 2011. The national free delivery policy in Nepal: early evidence of its effects on health facilities. *Health Policy* and Planning, 26 (Suppl 2), ii84–ii91. Available from: http://heapol.oxfordjournals.org/content/26/suppl_2/ii84.abstract [Accessed 5 May 2015].

The World Bank 2006. *The World Bank annual report 2006*. Washington D.C., U.S.A. Available from: http://siteresources.worldbank.org/INTANNREP2K6/Resources/2838485-1158333614345/AR06_final_LO_RES.pdf [Accessed 24 January 2017].

- The World Bank, 2010. *The World Bank: Nepal.* Washington D.C.: The World Bank. Available from: http://www.worldbank.org/en/country/nepal [Accessed 24 January 2017].
- The World Bank, 2011. Adult literacy rate, population 15+ years, both sexes (%). Washington D.C.: The World Bank. Available from: http://data.worldbank.org/indicator/SE.ADT.LITR.ZS [Accessed 24 January 2017].
- The World Bank, 2013. *Population total.* Catalog Sources World Development Indicators. Washington D.C.: The World Bank. Available form: http://data.worldbank.org/indicator/SP.POP.TOTL [Accessed 2 January 2016].
- World Health Organization (WHO), 1978. Declaration of Alma-Ata: International Conference on Primary Health Care, Alma-Ata. USSR, 6–12 September 1978. Retrieved February, 14, p.2006. Geneva: WHO. Available from: www.who.int/publications/almaata_declaration_en.pdf?ua=1 [Accessed 16 November 2015].

- World Health Organization (WHO), 1981. Health programme evaluation, Guiding principles for its application in the managerial process for national health development. Health for all series, No. 6. Geneva: WHO. Available from: http://apps.who.int/iris/bitstream/10665/40674/1/9241800062.pdf [Accessed 25 January 2017].
- World Health Organization (WHO), 1986. The Ottawa charter for health promotion: first international conference on health promotion, Ottawa, 21 November 1986.
 Geneva: WHO. Available from: www.who.int/healthpromotion/conferences/previous/ottawa/en/ [Accessed 16 November 2015].
- World Health Organization (WHO), 1992. International Classification of Diseases and Related Health Problems, Tenth Revision. Vol. 1. Geneva: WHO. Available from: www.who.int/classifications/icd/en/ [Accessed 16 November 2015].
- World Health Organization (WHO), 1998. *Health promotion evaluation: recommendations to policymakers.* Copenhagen: WHO Regional Office for Europe. Available from: http://www.euro.who.int/__data/assets/pdf_file/0004/159871/E60706.pdf [Accessed 15 September 2013].
- World Health Organization (WHO), 2000. *The World Health Report 2000: Health systems: improving performance.* Geneva: WHO. Available from: *www.who.int/whr/2000/en/* [Accessed 16 November 2015].
- World Health Organization (WHO), 2004a. Chapter 3 Community participation: advocacy and action. In: *Changing history. The World Health Report: 2004.* Geneva: WHO. Available from: *www.who.int/whr/2004/en/05_chap3_en.pdf* [Accessed 16 November 2015].
- World Health Organization (WHO), 2004b. *Making pregnancy safer: the critical role of the skilled attendant: a joint statement by WHO, ICM and FIGO.* Available from:

http://www.who.int/maternal_child_adolescent/documents/9241591692/en/ [Accessed 16 November 2015].

- World Health Organization (WHO), 2005a. *The Jakarta Declaration on health promotion in the 21st century*, 1997. Available from: http://www.who.int/healthpromotion/conferences/previous/jakarta/declaration/e n/ [Accessed 21 December 2016].
- World Health Organization (WHO), 2005b. *The Bangkok Charter for Health Promotion in Globalized World*. Bangkok: WHO. Available from: http://www.who.int/healthpromotion/conferences/6gchp/bangkok_charter/en/ [Accessed 16 November 2015].
- World Health Organization (WHO), 2007. WHO Country Cooperation Strategy 2006 - 2011 Nepal. Country Office for Nepal. Kathmandu: WHO. Available from: http://apps.searo.who.int/PDS_DOCS/B0408.pdf [Accessed 25 January 2017].
- World Health Organization (WHO), 2009a. *Milestones in Health Promotion:* Statements from Global Conferences. Geneva: WHO. Available from: http://www.who.int/healthpromotion/Milestones_Health_Promotion_05022010.p

df [Accessed 16 Nov 2015].

World Health Organization (WHO), 2009b. WHO/UNICEF Joint Statement: home visits for the newborn child: a strategy to improve survival. Geneva: World Health Organization, USAID and Save the Children. Geneva: WHO. Available from:

www.who.int/child_adolescent_health/documents/who_fch_cah_09_02/en/ [Accessed 3 July 2015].

 World Health Organization (WHO), 2011. International statistical classification of diseases and related health problems - 10th revision, edition 2010. Geneva: WHO. Available from: http://www.who.int/classifications/icd/ICD10Volume2_en_2010.pdf [Accessed 16 November 2015].

World Health Organization (WHO), 2012a. The WHO application of ICD-10 to deaths during pregnancy, childbirth and puerperium: ICD-MM. Geneva: WHO. Available from: http://apps.who.int/iris/bitstream/10665/70929/1/9789241548458_eng.pdf [Accessed 13 December 2016].

- World Health Organization (WHO), 2012b. Daily iron and folic acid supplementation in pregnant women: guidelines. Geneva: WHO. Available from: http://apps.who.int/iris/bitstream/10665/77770/1/9789241501996_eng.pdf?ua= 1 [Accessed 16 November 2015].
- World Health Organization (WHO), 2013. World Health Organization, Global Health Observatory Data Repository (GHO). Mortality: Neonatal deaths and Mortality Rate by Region, 1990-2012. Geneva: WHO. Available from: http://www.who.int/maternal_child_adolescent/epidemiology/newborn/en/index. html [Accessed 13 August 2015].
- World Health Organization (WHO), 2014a. WHO recommendations on postnatal care of the mother and newborn. Geneva: WHO. Available from http://apps.who.int/iris/bitstream/10665/97603/1/9789241506649_eng.pdf [Accessed 15 December 2015].

World Health Organization (WHO) and UNICEF, 2014b. Trends in maternal mortality: 1990 to 2013: estimates by WHO, UNICEF, UNFPA, The World Bank and the United Nations Population Division: executive summary. Geneva: WHO. Available from: http://apps.who.int/iris/bitstream/10665/194254/1/9789241565141_eng.pdf [Accessed 13 December 2016].

World Health Organization (WHO), 2014c. *Maternal mortality fact sheet.* Geneva: WHO. Available from: http://apps.who.int/iris/bitstream/10665/112318/1/WHO_RHR_14.06_eng.pdf? ua=1 [Accessed 24 January 2017].

World Health Organization (WHO), 2014d. The prevention and elimination of disrespect and abuse during facility-based childbirth: WHO statement. Geneva: WHO. Available from: http://www.who.int/reproductivehealth/topics/maternal_perinatal/statementchildbirth-data/en/ [Accessed 13 December 2016].

- World Health Organization (WHO), 2015a. Maternal mortality in 1990, WHO, UNICEF, UNFPA, World Bank Group and United Nations Population Division Maternal Mortality Estimation Inter-agency Group, Nepal. Available from: http://www.who.int/gho/maternal_health/countries/npl.pdf [Accessed 24 January 2017].
- World Health Organization (WHO), 2015b. WHO recommendations on health promotion interventions for maternal and newborn health 2015. Geneva: WHO. Available from: http://apps.who.int//iris/bitstream/10665/172427/1/9789241508742_report_eng. pdf?ua=1 [Accessed 16 November 2015].
- World Health Organization (WHO) and Unicef, 2010. UNFPA/World Bank: Trends in maternal mortality: 1990 to 2008. Geneva: WHO. http://whqlibdoc.who.int/publications/2010/9789241500265_eng.pdf [Accessed 16 November 2015].
- World Health Organization (WHO), Aga Khan University, and Partnership for Maternal Newborn and Child Health, 2011. Essential Interventions, Commodities and Guidelines for Reproductive, Maternal, Newborn and Child Health. Geneva: PMNCH. Available from: http://www.who.int/pmnch/knowledge/publications/201112_essential_interventi ons/en/ [Accessed 13 December 2014].
- Wright, L., 1996. A review of the Drinkwise campaign, 1989-94. London: Drinkwise.

Yanqiu, G., Ronsmans, C., and Lin, A., 2009. Time trends and regional differences in maternal mortality in China from 2000 to 2005. *Bulletin of the World Health Organization*, 87, 913–920. Available from: http://www.scielosp.org/scielo.php?script=sci_arttext&pid=S0042-96862009001200011&nrm=iso [Accessed 20 July 2015].

- Yargawa, J. and Leonardi-Bee, J., 2015. Male involvement and maternal health outcomes: systematic review and meta-analysis. *Journal of Epidemiology and Community Health*, pp.jech-2014.
- Zeitlin, J., Mohangoo, A.D., Delnord, M., Cuttini, M., Haidinger, G., Alexander, S., Pavlou, P., Velebil, P., Roos, J.L., Sakkeus, L. and Gissler, M., 2013. The second European Perinatal Health Report: documenting changes over 6 years in the health of mothers and babies in Europe. Journal of Epidemiology & Community Health 1; 67(12): 983-985. Available from: http://jech.bmj.com/content/67/12/983.long [Accessed 24 January 2017].